UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

	\cap	D	N/	1	n	-K
С,	\mathbf{U}	П	IV		u	-n

(Mark One)

Annual Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

For the fiscal year ended December 31, 2022

OR

Transition Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Commission File No. 1-5998



Marsh & McLennan Companies, Inc.

(Exact name of registrant as specified in its charter)

36-2668272

(I.R.S. Employer Identification No.)

Delaware

(State or other jurisdiction of incorporation or organization)

1166 Avenue of the Americas New York, New York 10036-2774

(Address of principal executive offices; Zip Code) (212) 345-5000

Registrant's telephone number, including area code Securities registered pursuant to Section 12(b) of the Act:

	Count	co registered paradant to occiton	12(b) of the 7 tot.	
Ti	tle of each class	Trading symbol(s)	Name of each exchange on wh	nich registered
Common Stoc	k, par value \$1.00 per share	MMC	New York Stock Exch	ange
			Chicago Stock Excha	ange
			London Stock Excha	inge
Indicate by check mark if the	registrant is a well-known seasoned issu	uer, as defined in Rule 405 of the S	ecurities Act. Yes ⊠ No □	
Indicate by check mark if the	registrant is not required to file reports p	ursuant to Section 13 or Section 15	5(d) of the Act. Yes □ No ⊠	
months (or for such shorter pe Indicate by check mark wheth	eriod that the registrant was required to t	file such reports), and (2) has been ically every Interactive Data File re	r 15(d) of the Securities Exchange Act of 1934 du subject to such filing requirements for the past 90 quired to be submitted pursuant to Rule 405 of Re (es ⊠ No □	days. Yes ⊠ No □
Indicate by check mark wheth		iler, an accelerated filer, a non-acce	elerated filer or a smaller reporting Company. See	the definitions of "large
Large Accelerated Filer	\boxtimes		Accelerated Filer	
Non-Accelerated Filer			Smaller Reporting Company	
			Emerging Growth Company	
	company, indicate by check mark if the red pursuant to Section 13(a) of the Excha	•	extended transition period for complying with any	new or revised financial
Indicate by check mark wheth	ner the registrant is a shell Company (as	defined in Rule 12b-2 of the Excha	ange Act). Yes □ No ⊠	
			assessment of the effectiveness of its internal cont firm that prepared or issued its audit report. Yes	
	regate market value of the registrant's co ock as reported on the New York Stock I	-	of the registrant was approximately \$77,420,628,3	314 computed by reference
As of February 9, 2023, there	were outstanding 494,571,451 shares of	of common stock, par value \$1.00 p	er share, of the registrant.	
Auditor Name: Deloitte	& Touche LLP Auditor Location:	New York, New York	Auditor Firm ID: 34	

DOCUMENTS INCORPORATED BY REFERENCE

Portions of Marsh & McLennan Companies, Inc.'s Notice of Annual Meeting and Proxy Statement for the 2023 Annual Meeting of Stockholders (the "2023 Proxy Statement") are incorporated by reference in Part III of this Form 10-K.

INFORMATION CONCERNING FORWARD-LOOKING STATEMENTS

This Annual Report on Form 10-K contains "forward-looking statements," as defined in the Private Securities Litigation Reform Act of 1995. These statements, which express management's current views concerning future events or results, use words like "anticipate," "assume," "believe," "continue," "estimate," "expect," "intend," "plan," "project" and similar terms, and future or conditional tense verbs like "could," "may," "might," "should," "will" and "would".

Forward-looking statements are subject to inherent risks and uncertainties that could cause actual results to differ materially from those expressed or implied in our forward-looking statements. Factors that could materially affect our future results include, among other things:

- the impact of geopolitical or macroeconomic conditions on us, our clients and the countries and industries in which we operate, including from conflicts such as the war in Ukraine, slower GDP growth or recession, capital markets volatility and inflation;
- the increasing prevalence of ransomware, supply chain and other forms of cyberattacks, and their potential to disrupt our operations and result in the disclosure of confidential client or company information:
- the impact from lawsuits or investigations arising from errors and omissions, breaches of fiduciary duty or other claims against us in our capacity as a broker or investment advisor, including claims related to our investment business' ability to execute timely trades;
- the financial and operational impact of complying with laws and regulations, including domestic and international sanctions regimes, anti-corruption laws such as the U.S. Foreign Corrupt Practices Act, U.K. Anti Bribery Act and cybersecurity and data privacy regulations;
- our ability to attract, retain and develop industry leading talent;
- our ability to compete effectively and adapt to competitive pressures in each of our businesses, including from disintermediation as well as technological change, digital disruption and other types of innovation;
- our ability to manage potential conflicts of interest, including where our services to a client conflict, or are perceived to conflict, with the interests of another client or our own interests:
- the impact of changes in tax laws, guidance and interpretations, or disagreements with tax authorities; and
- the regulatory, contractual and reputational risks that arise based on insurance placement activities and insurer revenue streams.

The factors identified above are not exhaustive. Further information concerning Marsh McLennan and its businesses, including information about factors that could materially affect our results of operations and financial condition, is contained in the Company's filings with the Securities and Exchange Commission, including the "Risk Factors" section in Part I, Item 1A of this report and the "Management's Discussion and Analysis of Financial Condition and Results of Operations" section in Part II, Item 7 of this report. Marsh McLennan and its subsidiaries operate in a dynamic business environment in which new risks emerge frequently. Accordingly, we caution readers not to place undue reliance on any forward-looking statements, which are based only on information currently available to us and speak only as of the dates on which they are made. The Company undertakes no obligation to update or revise any forward-looking statement to reflect events or circumstances arising after the date on which it is made.

TABLE OF CONTENTS

Information Concern	ning Forward-Looking Statements	<u> </u>
PART I		
Item 1 —	<u>Business</u>	<u>1</u>
Item 1A —	Risk Factors	<u>15</u>
Item 1B —	<u>Unresolved Staff Comments</u>	<u>34</u>
Item 2 —	<u>Properties</u>	<u>35</u>
Item 3 —	<u>Legal Proceedings</u>	<u>35</u>
Item 4 —	Mine Safety Disclosures	<u>36</u>
PART II		
Item 5 —	Market for the Company's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	<u>36</u>
Item 6 —	Selected Financial Data	<u>37</u>
Item 7 —	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>38</u>
Item 7A —	Quantitative and Qualitative Disclosures About Market Risk	<u>58</u>
Item 8 —	Financial Statements and Supplementary Data	<u>60</u>
Item 9 —	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	<u>117</u>
Item 9A —	Controls and Procedures	<u>117</u>
Item 9B —	Other Information	<u>119</u>
PART III		
Item 10 —	<u>Directors, Executive Officers and Corporate Governance</u>	<u>120</u>
Item 11 —	Executive Compensation	<u>120</u>
Item 12 —	<u>Security Ownership of Certain Beneficial Owners and Management and Related Stockholder</u> <u>Matters</u>	<u>120</u>
Item 13 —	Certain Relationships and Related Transactions, and Director Independence	<u>120</u>
Item 14 —	Principal Accountant Fees and Services	<u>120</u>
PART IV		
Item 15 —	Exhibits and Financial Statement Schedules	<u>121</u>
Item 16 —	Form 10-K Summary	<u>135</u>
Signatures		<u>136</u>

PART I

Item 1. Business.

References in this report to "we", "us" and "our" are to Marsh & McLennan Companies, Inc. and its consolidated subsidiaries (the "Company" or "Marsh McLennan"), unless the context otherwise requires.

GENERAL

Marsh McLennan is the world's leading professional services firm in the areas of risk, strategy and people. The Company's more than 85,000 colleagues advise clients in over 130 countries. With annual revenue of over \$20 billion, Marsh McLennan helps clients navigate an increasingly dynamic and complex environment through four market-leading businesses. Marsh provides data-driven risk advisory services and insurance solutions to commercial and consumer clients. Guy Carpenter develops advanced risk, reinsurance and capital strategies that help clients grow profitably and identify and capitalize on emerging opportunities. Mercer delivers advice and technology-driven solutions that help organizations redefine the future of work, shape retirement and investment outcomes, and advance health and well-being for a changing workforce. Oliver Wyman Group serves as a critical strategic, economic and brand advisor to private sector and governmental clients.

The Company conducts business through two segments:

- Risk and Insurance Services includes risk management activities (risk advice, risk transfer and risk control and mitigation solutions)
 as well as insurance and reinsurance broking and services. The Company conducts business in this segment through Marsh and
 Guy Carpenter.
- **Consulting** includes health, wealth and career solutions and products, and specialized management, strategic, economic and brand consulting services. The Company conducts business in this segment through Mercer and Oliver Wyman Group.

We describe our current segments in further detail below. We provide financial information about our segments in our consolidated financial statements included under Part II, Item 8 of this report.

OUR BUSINESSES

RISK AND INSURANCE SERVICES

The Risk and Insurance Services segment generated approximately 61% of the Company's total revenue in 2022 and employs approximately 48,600 colleagues worldwide. The Company conducts business in this segment through Marsh and Guy Carpenter.

MARSH

Marsh is the world's leading insurance broker and risk advisor, serving companies, institutions and individuals. From its founding in 1871 to the present day, Marsh has demonstrated a commitment to thought leadership, innovation and insurance expertise to meet its clients' needs. Marsh's pioneering contributions include introducing the practice of client representation through brokerage, the discipline of risk management, the globalization of risk management services and the development of service platforms that identify, quantify, mitigate and transfer risk.

Currently, approximately 45,200 Marsh colleagues provide risk management, insurance broking, insurance program management, risk consulting, analytical modeling and alternative risk financing services to a wide range of businesses, government entities, professional service organizations and individuals in more than 130 countries. Marsh generated approximately 51% of the Company's total revenue in 2022.

Insurance Broking and Risk Advisory

In its core insurance broking and risk advisory business, Marsh employs a team approach to identify, quantify and address clients' risk management and insurance needs. Marsh's product and service offerings include risk analysis, insurance program design and placement, insurance program support and administration, claims support and advocacy, alternative risk strategies and a wide array of risk analysis and risk management consulting services. Clients benefit from Marsh's advanced analytics, deep technical expertise, specialty and industry knowledge, collaborative global culture and the ability to

develop innovative solutions and products. The firm's resources also include nearly three dozen specialty and industry practices, including cyber, construction, renewable energy, healthcare, and financial and professional service practices, along with ESG products such as a D&O insurance initiative recognizing U.S. based clients with superior ESG frameworks, and an established employee health & benefits business.

Marsh provides services to clients of all sizes, including large multinational companies ("Risk Management"), high growth middle-market businesses ("Corporate"), small commercial enterprises and high net-worth private clients, and affinity group members ("Commercial & Consumer"). Marsh's segments are designed to build stronger value propositions and operating models to optimize solutions and services for clients depending on their needs.

Risk Management. Marsh has an extensive global footprint and market-leading advisory and placement services that benefit large domestic and international companies and institutions facing complex risk exposures. These clients are also supported by Marsh's robust analytics and a growing digital experience.

In addition, Marsh's largest global clients are serviced by Marsh Multinational, a dedicated team of colleagues from around the world focused on delivering service excellence and insurance solutions to clients wherever they are located. Marsh is digitizing the client experience through tools such as LINQ, Marsh's account and service application; Blue[i], a suite of analytics tools for clients; and Bluestream, a digital brokerage platform that enables clients to provide insurance to their customers or suppliers in a B2B2C distribution model. Marsh provides global expertise and an intimate knowledge of local markets, helping clients navigate local regulatory environments to address the worldwide risk issues that confront them.

• *Marsh Specialty* is an integrated and globally coordinated team of experts who provides clients in highly specialized industry and product areas with data driven insights, service, advice and access to global insurance markets. These specialists support clients who require advice and support across aviation & space, credit specialties, construction, energy & power, financial & professional services (FINPRO), marine & cargo, and private equity, mergers & acquisitions (PEMA).

Corporate. Middle market clients are served by Marsh's brokerage operations globally; this segment is also serviced by Marsh & McLennan Agency (MMA) in the United States.

 Marsh McLennan Agency (MMA) provides business insurance, employee health and benefits, retirement and wealth management, and private client insurance solutions to individuals and mid-market organizations. MMA advises on insurance program structure and market dynamics, along with industry expertise and transactional capability. Since its first acquisition in 2009, MMA has acquired more than 100 agencies.

Commercial & Consumer. Clients in this market segment typically face less complex risks and are served by Marsh's innovative product and placement offerings and growing capabilities in digitally enabled distribution and administration.

• Victor Insurance Managers (Victor) is one of the largest underwriting managers of professional liability and specialty insurance programs worldwide. In the United States, Victor Insurance Managers (US) and ICAT Managers deliver risk management and insurance solutions to insureds through a national third-party distribution network of licensed brokers. Through its Victor for Agents small business platform, Victor deploys cloud-based technology to enable independent insurance agents, on behalf of their small business clients, to obtain online quotes from multiple insurance providers and bind property and casualty and workers compensation insurance policies in real time. Victor Insurance Managers (Canada), a leading managing general agent in Canada, delivers professional liability and construction insurance and other P&C programs and administers group and retiree benefits programs and claims handling operations for individuals, organizations and businesses. Victor also has a business in the UK, the Netherlands, Italy and Germany. In addition, Victor manages Torrent Technologies, a service provider to Write Your Own (WYO) insurers and direct policy providers participating in the National Flood Insurance Program

(NFIP) in the United States. Torrent offers both NFIP and private and excess flood insurance products and services to WYO companies and agents.

- Marsh Affinity focuses on insurance programs sold to insureds or vendors through a corporate sponsor using an affinity distribution model.
- *High Net Worth (HNW)*. Individual high net worth clients and family offices are serviced by MMA in the United States and other Marsh personal lines businesses globally. These businesses provide a single-source solution for high net worth clients and are dedicated to sourcing protections across a broad spectrum of risk. Using a consultative approach, Marsh's HNW practices analyze exposures and customize programs to cover individual clients with complex asset portfolios.

Additional Services and Adjacent Businesses

In addition to insurance broking, Marsh provides certain other specialist advisory or placement services:

Marsh Advisory is a global practice comprising specialists who use data and analytics, including through Marsh's Blue[i] digital analytics platform. Marsh Advisory's four main service areas (Consulting, Claims, Analytics, and Captives) advise clients on existing and emerging risk exposures, protecting critical business activities and developing strategies to optimize total cost of risk.

Marsh Captive Solutions, a prominent part of the Marsh Advisory practice, helps organizations of all sizes retain risks through comprehensive and innovative captive solutions. This team is comprised of captive consultants, actuaries and captive management professionals which offer complete, end-to-end captive management services.

Bowring Marsh is an international placement broker. This unit's core strategy is to modernize risk transfer advice and solutions for clients. This is executed through a combination of data solutions, capacity creation vehicles, segmentation, placement platforms (on-shoring solutions within the network), and improved operational efficiency – all designed to yield a better client outcome and experience. The products Bowring Marsh places include property, casualty, terrorism, product recall, and special risks.

Mercer Marsh Benefits provides health benefits brokerage and consulting services to clients of all sizes in numerous countries across the globe, outside of the United States. As described below, Mercer and Marsh go to market together to provide strategic advice and services to help clients minimize risk, optimize benefits structure, drive efficiencies and maximize employee engagement.

Services for Insurers

Marsh's *Insurer Consulting Group (ICG)* provides services to insurance carriers. Through Marsh's patented electronic platform, MarketConnect, and sophisticated data analysis, ICG provides insurers with individualized preference setting and risk identification capabilities, as well as detailed performance data and metrics. Insurer consulting teams review performance metrics and preferences with insurers and provide customized consulting services to insurers designed to improve business planning and strategy implementation. ICG services are designed to improve the product offerings available to clients, assist insurers in identifying new opportunities and enhance insurers' operational efficiency. The scope and nature of the services vary by insurer and by geography.

GUY CARPENTER

Guy Carpenter, the Company's reinsurance intermediary and advisor, generated approximately 10% of the Company's total revenue in 2022. Currently, approximately 3,400 Guy Carpenter colleagues provide clients with a combination of specialized reinsurance broking expertise, strategic advisory services and analytics solutions. Guy Carpenter creates and executes reinsurance and risk management solutions for clients worldwide through risk assessment analytics, actuarial services, highly-specialized product knowledge and trading relationships with reinsurance markets. Client services also include contract and claims management, reinsurance accounting and fiduciary services.

Acting as a broker or intermediary on all classes of reinsurance, Guy Carpenter places two main types of property casualty and life / health reinsurance: treaty reinsurance, which involves the transfer of a portfolio of risks; and facultative reinsurance, which involves the transfer of part or all of the coverage provided by a single insurance policy.

Guy Carpenter provides reinsurance services in a broad range of centers of excellence, segments and specialties including: Automobile / Motor, Aviation, Captives, Crop/Agriculture, Cyber, Engineering / Construction, Financial Lines, InsurTech, Life / Accident / Health, Marine and Energy, Medical Professional, Personal Lines, Mortgage, Political Risk & Trade Credit, Primary & Excess Casualty, Managing General Agents and Program Manager Solutions, Property, Public Sector, Regional / Mutual, Retrocessional Reinsurance, Surety, Terror, and Workers Compensation / Employer Liability.

Guy Carpenter also offers clients alternatives to traditional reinsurance, including industry loss warranties and, through its licensed affiliates, capital markets alternatives such as transferring catastrophe risk through the issuance of insurance-linked securities. GC Securities, the Guy Carpenter division of MMC Securities LLC and MMC Securities (Europe) Limited, offer corporate finance solutions, including mergers & acquisitions advice and private debt and equity capital raising, and capital markets-based risk transfer solutions that complement Guy Carpenter's strong industry relationships, analytical capabilities and reinsurance expertise.

Guy Carpenter also provides its clients with reinsurance-related services, including actuarial, enterprise risk management, financial and regulatory consulting, portfolio analysis and advice on the efficient use of capital. Guy Carpenter's Global Strategic Advisory ("GSA") unit helps clients better understand and quantify the uncertainties inherent in their businesses. Working in close partnership with Guy Carpenter account executives, GSA specialists help support clients' critical decisions in numerous areas, including reinsurance utilization, catastrophe exposure portfolio management, new product and market development, rating agency, regulatory and account impacts, loss reserve risk, capital adequacy and return on capital.

Compensation for Services in Risk and Insurance Services

Marsh and Guy Carpenter are compensated for brokerage and consulting services through commissions and fees. Commission rates and fees vary in amount and can depend on a number of factors, including the type of insurance or reinsurance coverage provided, the particular insurer or reinsurer selected, and the capacity in which the broker acts and negotiates with clients. In addition to compensation from its clients, Marsh also receives other compensation, separate from retail fees and commissions, from insurance companies. This other compensation includes, among other things, payments for consulting and analytics services provided to insurers; compensation for administrative and other services (including fees for underwriting services and services provided to or on behalf of insurers relating to the administration and management of quota shares, panels and other facilities in which insurers participate); and contingent commissions, which are paid by insurers based on factors such as volume or profitability of Marsh's placements, primarily driven by MMA and parts of Marsh's international operations.

Marsh and Guy Carpenter receive interest income on certain funds (such as premiums and claims proceeds) held in a fiduciary capacity for others. For a more detailed discussion of revenue sources and factors affecting revenue in our Risk and Insurance Services segment, refer to Part II, Item 7 ("Management's Discussion and Analysis of Financial Condition and Results of Operations") of this report.

CONSULTING

The Company's Consulting segment generated approximately 39% of the Company's total revenue in 2022 and employs approximately 30,900 colleagues worldwide. The Company conducts business in this segment through Mercer and Oliver Wyman Group.

MERCER

Mercer is a leading provider in delivering advice and solutions that help organizations meet the health, wealth and career needs of a changing workforce. Mercer has approximately 24,200 colleagues based in 48 countries. Clients include a majority of the companies in the Fortune 1000 and FTSE 100, as well as medium- and small-market organizations, public sector entities and individual customers. Mercer generated approximately 26% of the Company's total revenue in 2022.

Mercer operates in the following areas:

Health. Mercer helps public and private sector employers design and manage employee health care and welfare programs; administer health benefits and flexible benefits programs, including benefits outsourcing; engage employees with their health benefits through a digital experience; and comply with local benefits-related regulations. Mercer provides a range of advice and solutions to clients, which, depending on the engagement, may include: total health and wellness management strategies; global health brokerage solutions; vendor performance and audit; life and disability management; and measurement of healthcare provider performance. These services are provided through traditional fee-based consulting as well as commission-based brokerage services in connection with the selection of insurance companies and healthcare providers. Mercer provides solutions for private active and retiree exchanges in the United States.

Mercer also provides consulting and actuarial services to U.S. state governments to support the purchase of healthcare through state Medicaid programs. Mercer offers clients tools to enhance employee engagement with their health benefits through its DarwinSM platform.

Outside of the United States, Mercer and Marsh go to market together for Health benefits brokerage and consulting under the Mercer Marsh BenefitsSM (MMB) brand, as described above.

Wealth. Through its Wealth business, Mercer assists clients worldwide in the design, governance and risk management of defined benefit, defined contribution, hybrid retirement plans and other pools of assets, and with investment of those assets.

Mercer provides actuarial consulting, investment consulting, investment management and related services to the sponsors and trustees of pension plans, master trusts, foundations, endowments, sovereign wealth funds, insurance companies and family offices. Mercer also provides investment consulting and investment management services to U.S. public sector clients, financial intermediaries and individuals. Mercer provides retirement plan outsourcing, including administration and delivery of defined benefit and defined contribution retirement benefits.

Mercer's investment consulting and investment management services (investment management services may also be referred to as "investment solutions," "delegated solutions," "fiduciary management" or "outsourced Chief Investment Officer (OCIO) services") cover a range of stages of the investment process, from investment research (through its Mercer-Insight service) strategy, asset allocation and implementation of investment strategies to ongoing portfolio management services. Mercer provides these services primarily to institutional and other sophisticated investors including retirement plans (e.g., defined benefit and defined contribution), master trusts, endowments and foundations, sovereign wealth funds, U.S. public sector clients, insurance companies and family offices, as well as wealth managers and other financial intermediaries, primarily through manager of manager strategies and funds sponsored and managed by Mercer. Mercer's clients invest in both traditional asset classes (e.g., equities, fixed income and cash equivalents) and alternative or private market strategies (e.g., private equity, private debt, real estate, other real assets and hedge funds). As of December 31, 2022, Mercer and its global affiliates had assets under management of approximately \$345 billion worldwide.

Mercer also provides services to individual retail clients, including financial planning, high net worth risk solutions and other discretionary investment services.

Career. Mercer advises organizations on the engagement, skill assessment, management and reward of employees; the design of executive remuneration programs; people strategies during business transformation; improvement of human resource (HR) effectiveness; and the implementation of digital and cloud-based Human Resource Information Systems. In addition, through proprietary survey data and decision support tools, Mercer provides clients with human capital information and analytical capabilities to improve strategic human capital decision making. Mercer's Career products include solutions relating to rewards, mobility, engagement, workforce analytics and assessments. Mercer helps clients plan and implement HR programs and other organizational changes designed to maximize employee engagement.

Mercer also provides advice relating to people and benefits-related issues to buyers and sellers in a variety of types of M&A transactions.

OLIVER WYMAN GROUP

With more than 6,700 professionals and offices in over 30 countries, Oliver Wyman Group delivers advisory services to clients through three operating units, each of which is a leader in its field: Oliver Wyman, Lippincott and NERA Economic Consulting. Oliver Wyman Group generated approximately 14% of the Company's total revenue in 2022.

Oliver Wyman is a global leader in management consulting and combines deep industry knowledge with specialized expertise in strategy, operations, risk management and organization transformation. The firm works with clients around the world to help optimize their business, improve their operations and risk profile, and accelerate their organizational performance to seize attractive opportunities. Industry groups include:

- · Automotive and Manufacturing Industries
- · Communications, Media & Technology
- · Energy and Natural Resources
- Financial Services (including corporate and institutional banking, public policy, and retail and business banking)
- · Insurance and Asset Management
- · Health and Life Sciences
- Public Sector
- Private Capital
- · Retail & Consumer Goods
- Transportation Services (including aviation; aerospace and defense; rail; express, postal and third party logistics; services, including
 travel and leisure, environmental and facility management, and business and tech services; and CAVOK, which provides technical
 consulting and market forecasting services)

Oliver Wyman overlays its industry knowledge with expertise in the following functional specializations:

- Actuarial. Oliver Wyman's Actuarial Practice uses mathematical and statistical modeling skills and qualitative assessment methodologies to assist clients in evaluating and addressing risk.
- Climate and Sustainability. Oliver Wyman assists clients in cutting through complex climate systems and solving for operational efficiencies. Oliver Wyman helps clients discover new business opportunities, create new pathways, and respond to climate risk, to make needed changes commercially compelling.
- Finance and Risk. Oliver Wyman provides leading financial institutions with custom solutions and insights covering all aspects of risk and finance functions, including credit risk, market risks, asset and liability management and liquidity risks, and non-financial risks, together with integrated risk management topics, such as aggregated risk analyses, business applications and culture and organization.

- Restructuring. Oliver Wyman offers a complete management solution and "one-stop-shop" approach to turning around companies, providing strategic, operational, and financial restructuring advice.
- *Digital*. Oliver Wyman partners with clients to address their digital challenges, blending the power of digital with deep industry expertise. By building strong capabilities and culture, Oliver Wyman accelerates and embeds digital transformation, working collaboratively with clients' leaders, employees, stakeholders, and customers to jointly define, design, and achieve lasting results.
- *Operations*. Oliver Wyman helps organizations leverage their operations for a competitive advantage using a comprehensive set of capabilities, including performance improvement, digital operations strategy, and risk management.
- People and Organizational Performance. Oliver Wyman's People and Organizational Performance capability brings together deep functional expertise and industry knowledge to enable the whole organization to work in service of its strategic vision and to address the most pressing organizational, people, and change issues.
- Payments. Oliver Wyman draws on years of industry-shaping work in the Financial Services and Retail industries, deep digital expertise, and renowned research partners in its Celent® business, to help clients from banks/issuers, to payments providers, to retailers to build growth strategies, form effective partnerships, optimize costs, and manage risk.
- Pricing, Sales, and Marketing. Oliver Wyman helps organizations drive top-line and margin growth through outstanding strategy and decision making on pricing, marketing optimization, and best practices on sales effectiveness.
- Customer First. Oliver Wyman helps bring together capabilities required to identify customer and business growth, conduct detailed business design, build and launch a business, and maintain a focus on realizing growth while de-risking delivery.
- Performance Transformation. Oliver Wyman helps clients to design, realize and sustain value growth via large-scale transformations.

Lippincott is a creative consultancy specializing in brand and innovation that shapes recognized brands and experiences for clients globally. Lippincott's designers have helped create some of the world's most recognized brands.

NERA Economic Consulting provides economic analysis and advice to public and private entities to achieve practical solutions to highly complex business and legal issues arising from competition, regulation, public policy, strategy, finance and litigation. NERA professionals operate worldwide assisting clients including corporations, governments, law firms, regulatory agencies, trade associations, and international agencies. NERA's specialized practice areas include: antitrust; securities; complex commercial litigation; energy; environmental economics; network industries; intellectual property; product liability and mass torts; and transfer pricing.

Compensation for Services in Consulting

Oliver Wyman Group is compensated for advice and services primarily through fees paid by clients. Mercer is compensated for advice and services through fees paid by clients, commissions and fees based on assets or members. In the majority of cases, Mercer's Health business is compensated through commissions for the placement of insurance contracts and supplemental compensation from insurers based on such factors as volume, growth of accounts, and total retention of accounts placed by Mercer. Mercer may receive commissions in other parts of its business, such as its Private Client Services business and certain financial advice businesses. Mercer's investments business and certain of Mercer's administration services are compensated typically through fees based on assets under administration or management or fee per member. For a majority of the Mercer-managed investment funds, revenue received from Mercer's investment management clients as subadvisor fees is reported in accordance with U.S. GAAP, on a gross basis rather than a net basis. For a more detailed discussion of revenue sources and factors affecting revenue in the Consulting segment, refer to Part II, Item 7 ("Management's Discussion and Analysis of Financial Condition and Results of Operations") of this report.

REGULATION

The Company's activities are subject to licensing requirements and extensive regulation under U.S. federal and state laws, as well as laws of other countries in which the Company's subsidiaries operate. Across most jurisdictions, we are also subject to various data privacy laws and regulations that apply to personal information. In addition, we are subject to various financial crime laws and regulations through our activities of associated persons, the products and services we provide and our business and client relationships. Such laws and regulations relate to, among other areas, sanctions and export control, anti-bribery, anti-corruption, anti-money-laundering and counter-terrorist financing. In certain circumstances, we are also required to maintain operating funds primarily related to regulatory requirements outside the U.S. See Part I, Item 1A ("Risk Factors") below for a discussion of how actions by regulatory authorities or changes in legislation and regulation in the jurisdictions in which we operate may have an adverse effect on our businesses.

Risk and Insurance Services. While laws and regulations vary from location to location, every state of the United States and most foreign jurisdictions require insurance market intermediaries and related service providers (such as insurance brokers, agents and consultants, reinsurance brokers and managing general agents) to hold an individual or company license from a government agency or self-regulatory organization. Some jurisdictions issue licenses only to individual residents or locally-owned business entities; in those instances, if the Company has no licensed subsidiary, it may maintain arrangements with residents or business entities licensed to act in such jurisdiction. Such arrangements are subject to an internal review and approval process. Licensing of reinsurance intermediaries is generally less rigorous compared to that of insurance brokers, and most jurisdictions require only corporate reinsurance intermediary licenses.

In the United Kingdom, our business is regulated by the Financial Conduct Authority ("FCA"). The FCA's responsibilities and powers include licensing of insurance and reinsurance intermediaries and related criteria such as professional competence, financial capacity and the requirement to hold professional indemnity insurance, the broking of premium finance to consumers, and competition powers that enable it to enforce prohibitions on anti-competitive behavior in relation to financial services.

Insurance authorities in the United States and certain other jurisdictions in which the Company's subsidiaries do business, including the FCA in the United Kingdom, also have enacted laws and regulations governing the investment of funds, such as premiums and claims proceeds, held in a fiduciary capacity for others. These laws and regulations typically provide for segregation of these fiduciary funds and limit the types of investments that may be made with them, and generally apply to both the insurance and reinsurance business.

Certain of the Company's Risk and Insurance Services activities are governed by other regulatory bodies, such as investment, securities and futures licensing authorities. In the United States, Marsh and Guy Carpenter use the services of MMC Securities LLC, a SEC registered broker-dealer and introducing broker in the United States. MMC Securities LLC is a member of the Financial Industry Regulatory Authority ("FINRA"), the National Futures Association and the Securities Investor Protection Corporation ("SIPC"), primarily in connection with capital markets and other investment banking-related services relating to insurance-linked and alternative risk financing transactions. Also in the United States, Marsh uses the services of MMA Securities LLC, a SEC registered broker-dealer, investment adviser and member of FINRA, SIPC and the Municipal Securities Rulemaking Board ("MSRB"), and MMA Asset Management LLC, a SEC registered investment adviser, primarily in connection with retirement, executive compensation and benefits consulting and advisory services to qualified and non-qualified benefits plans, companies and executives and personal wealth management. In the United Kingdom, Marsh and Guy Carpenter use the expertise of MMC Securities Limited, which is authorized and regulated by the FCA to provide advice on securities and investments, including mergers & acquisitions in the United Kingdom. In the European Union, Guy Carpenter uses MMC Securities (Ireland) Limited, which is authorized and regulated by the Central Bank of Ireland to place certain securities and investments in the European Union. MMC Securities LLC, MMC Securities Limited, MMC Securities (Ireland) Limited, MMA Securities LLC, and MMA Asset Management LLC are indirect, wholly-owned subsidiaries of Marsh & McLennan Companies, Inc.

Consulting. Mercer's retirement-related consulting and investment services are subject to pension law and financial regulation in many countries. Depending on the country, Mercer may rely on licensed colleagues or registered legal entities to engage in these services, or may utilize other Marsh McLennan entities or third parties. In addition, trustee services, investment services (including advice to persons, institutions and other entities on the investment of pension assets and assumption of discretionary investment management responsibilities) and retirement and employee benefit program administrative services provided by Mercer and its subsidiaries and affiliates may also be subject to investment and securities regulations in various jurisdictions, including (but not limited to) regulations imposed or enforced by the Securities and Exchange Commission (SEC) and the Department of Labor in the United States, the FCA in the United Kingdom, the Central Bank of Ireland and the Australian Prudential Regulation Authority and the Australian Securities and Investments Commission. In the United States, Mercer provides investment services through Mercer Investments LLC, (formerly Mercer Investment Management, Inc.), an SECregistered investment adviser, which consolidated the activities of each of Mercer's affiliated investment adviser entities in the United States (including Mercer Investment Consulting LLC and Pavilion Advisory Group) in 2019. Mercer Trust Company, a limited purpose New Hampshire chartered trust bank, may also provide services for certain clients of Mercer's investment management business in the United States. The benefits insurance consulting and brokerage services provided by Mercer and its subsidiaries and affiliates are subject to the same licensing requirements and regulatory oversight as the insurance market intermediaries described above regarding our Risk and Insurance Services businesses. Depending on the nature of the client and services performed. Mercer may also be subject to direct oversight by the Departments of Health and Human Services and other federal agencies in the United States. Mercer provides annuity buy-out support that is subject to regulations (for example, in the United States, state insurance licensing regulations and ERISA). Mercer uses the services of MMC Securities LLC to provide certain services, including executive benefit and compensation services and securities dealing services.

FATCA. Regulations promulgated by the U.S. Treasury Department pursuant to the Foreign Account Tax Compliance Act and related legislation (FATCA) require the Company to take various measures relating to non-U.S. funds, transactions and accounts. The regulations impose on Mercer and MMA certain client financial account obligations relating to non-U.S. financial institution and insurance clients.

COMPETITIVE CONDITIONS

The Company faces significant competition in all of its businesses from providers of similar products and services, including competition with regard to identifying and pursuing acquisition candidates. The Company also encounters strong competition throughout its businesses from both public corporations and private firms in attracting and retaining qualified employees. As the Company has clients across various geographies, industries and sizes, the competitive landscape is complex and varies across numerous markets. In addition to the discussion below, refer to "Risks Relating to the Company Generally — Competitive Risks," in Part I, Item 1A of this report.

Risk and Insurance Services. The Company's combined insurance and reinsurance services businesses are global in scope. Our insurance and reinsurance businesses compete principally on the sophistication, range, quality and cost of the services and products they offer to clients. The Company encounters strong competition from other insurance and reinsurance brokerage firms that operate on a global, regional, national or local scale in every geography in which it operates, from insurance and reinsurance companies that market, distribute and service their insurance and reinsurance products without the assistance of brokers and from other businesses, including commercial and investment banks, accounting firms, consultants and online platforms, that provide risk-related services and products or alternatives to traditional insurance brokerage services. In addition, third party capital providers have entered the insurance and reinsurance risk transfer market offering products and capital directly to the Company's clients. Their presence in the market increases the competitive pressures that the Company faces.

Certain insureds and groups of insureds have established programs of self-insurance as a supplement or alternative to purchasing traditional third-party insurance, thereby reducing in some cases their need for third-party insurance placements. Certain insureds also obtain coverage directly from insurance providers. There are also many other providers of managing general agency, affinity programs and private client services that compete with the Company's offerings.

Consulting. The Company's consulting businesses face strong competition from other privately and publicly held worldwide and national companies, as well as regional and local firms. These businesses generally compete on the basis of the range, quality and cost of the services and products they provide to clients. Competitors include independent consulting, broking and outsourcing firms, as well as consulting, broking and outsourcing operations affiliated with larger accounting, information systems, technology and financial services firms. Mercer's Health business faces additional competition from insurers and from non-traditional competitors seeking to enter or expand in the health benefits space (for example, payroll firms, large consumer businesses, and digitally oriented consultancies). Mercer's investments business faces competition from many sources, including investment consulting firms (many of which offer delegated services), investment management firms and other financial institutions. In some cases, clients have the option of handling the services provided by Mercer and Oliver Wyman Group internally, without assistance from outside advisors.

Segmentation of Activity by Type of Service and Geographic Area of Operation.

Financial information relating to the types of services provided by the Company and the geographic areas of its operations is incorporated herein by reference to Note 17, Segment Information, in the notes to the consolidated financial statements included under Part II, Item 8 of this report.

ENVIRONMENTAL, SOCIAL AND GOVERNANCE (ESG)

Since 2008, Marsh McLennan has had a framework for overseeing and managing the company's corporate responsibility initiatives at the Board and senior management levels. ESG is central to who we are and how we serve our clients and communities. We believe that Marsh McLennan is well positioned to help our clients tackle the challenges of climate resilience and we are committed to developing innovative solutions to help move the world towards a more sustainable future.

Our ESG Report provides more information about our ESG philosophy, goals and achievements. It also discloses against aspects of the Task Force on Climate-related Financial Disclosures, Sustainability Accounting Standards Board and Global Reporting Initiative standards and describes the six UN Sustainable Development Goals we have prioritized that most relate to our business. Our ESG Report, Pay Equity Statement, statement on Human Rights and related information is available on our website at marshmclennan.com/about/esg.html. These reports and our website are not deemed part of this report and are not incorporated by reference.

HUMAN CAPITAL

As a professional services firm, we believe the health of our business relies on the strength of our workforce. Our shared purpose is to make a difference in moments that matter, helping clients meet the challenges of our time. Measurement of our talent outcomes are, therefore, not just a human capital priority, but a business imperative.

For detailed information regarding our human capital management, we encourage investors to visit https://www.marshmclennan.com/about/esg.html for our consolidated ESG Report. The information on this website, and in the ESG report, does not constitute, and should not be viewed as, incorporation by reference of the information contained on, or available through, the website or the report and does not form part of this Form 10-K.

Our People. As of December 31, 2022, the Company and its consolidated subsidiaries employed more than 85,000 colleagues worldwide, including approximately 48,600 in Risk and Insurance Services and 30,900 in Consulting. Two-thirds of our global workforce are located in either North America or Europe. While these remain our largest work regions, we have also grown our presence in Asia-Pacific, Latin America and the Middle East over the last several years. Women comprise more than half of our global enterprise workforce, and approximately 32% of our senior leaders are women. In the United States, where we have the most complete data through workforce self-identification of race and ethnicity, approximately 1 in 4 U.S. colleagues and 17% of U.S. senior leaders identify as non-White.

Our Governance. The Chief People Officer is responsible for developing and executing our enterprise people strategy. This includes the attraction, recruitment, hiring, development and engagement of talent to deliver on our strategy and the design of colleague total rewards programs. The Chief People Officer and the Chief Inclusion & Diversity Officer are responsible for developing and integrating our inclusion and diversity approach into our strategy.

Our ESG Committee and Compensation Committee of the board of directors have oversight of these initiatives. The Compensation Committee has responsibility to review certain key human resource strategic activities, including those relating to diversity, training and recruitment. The Compensation Committee coordinates with the ESG Committee on diversity initiatives, and both committees receive reports at least annually on inclusion and diversity from the Company's Chief People Officer. The Chief Executive Officer and Chief People Officer regularly update our board of directors, the ESG Committee and the Compensation Committee on the Company's human capital trends and activities.

Inclusion & Diversity. Our Company's greatest strength is the collective talent of our people. We believe the more diverse our backgrounds and experiences, the more we can achieve together working side by side. We seek capable, creative and fair-minded people who can help us enable client success, find smarter ways to do things and live our Code of Conduct, The Greater Good. We believe that inclusion means more than acceptance—it means belonging. We have a Race Advisory Council to advise our Executive Committee and help amplify diverse perspectives in decision-making. Additionally, we have established cross-enterprise leadership development programs designed to expand professional development and opportunities for career progression, including our Black Leadership Program, Accelerated Leadership Program and our Racial Inclusion and Social Equity (RISE) MBA Fellowship program in partnership with the National Black MBA Association and Fisk University. We also offer regionally based and business-specific affinity programming to support leadership development for women, racially and ethnically underrepresented colleagues, and LGBTQIA+ identifying colleagues. Those programs include Marsh's global Diversity Sponsorship Program and women's Leadership Development Programs in South Africa and India.

Talent Development. We strive to create an environment where individuals and teams can perform to their highest potential and where career growth and mobility are encouraged and supported. We are committed to helping colleagues perform at their best by encouraging regular discussions about their goals, performance, career aspirations and development opportunities.

We also aim to build a learning culture and deliver a digital-first learning strategy, supplemented by formal programs for key groups. For example, we implemented a new learning platform to accelerate usage of digital learning as well as increasing participation in live learning. Our digital learning platform empowers colleagues to choose their own skill development, customize preferences, learn in local language, and get personalized development recommendations, and enables managers to have more meaningful development conversations. Parts of our business also sponsored Learning Days in 2022.

We also recognize the importance of our 16,800 people managers to our talent pipeline and have given them increased support and opportunities for promoting the growth of their teams. In 2022 we offered 440 live learning sessions in addition to more than 32,000 individual learning courses to help our colleagues grow and develop. Our People Manager Hub is a one-stop digital source for people managers globally. Through the Hub, people managers have access to suggested learning, webinars and resources to support development and provide guidance for leading with clarity and inclusion, including four new learning pathways on our digital learning platform and a series of People Manager Workshops on topics ranging from Building Inclusion in Your Team to Managing Change and Transitions, which were offered for the first time in 2022.

Colleague Engagement. Each year we ask our colleagues to share their views on working at Marsh McLennan through a company-wide engagement survey. Developed internally by our Global Talent Development team, the survey methodology has been consistent since 2011, with updates to specific questions as necessary. In 2022, we expanded the survey with questions on manager effectiveness, covering topics related to well-being, inclusion and diversity, feedback and colleague development. A third-party administers our survey in order to maintain confidentiality of responses. Collective survey outcomes allow us to monitor the evolution of our culture over time and identify opportunities to build on strengths and address challenges, all with the intention of furthering our productivity through an engaged workforce.

Health and Well-being. As a company, our success depends on the health and well-being of our colleagues—we want to support our colleagues with the resources, protection and peace of mind to live healthy lives. We offer comprehensive health insurance, including medical coverage and other core health benefits based on the market. We also prioritize our colleagues' mental wellness, including 24/7 access to

an Employee Assistance Program for confidential counselling on personal issues for 99% of our colleagues and their eligible family members, and critical incident support in countries where a disaster has occurred. In addition, we offer competitive time-off benefits, including a paid day off each year to volunteer. Supporting our colleagues as they navigate changing circumstances—health and economic challenges, new technologies and social inequities—has been our priority in 2022.

Total Rewards. We recognize how important it is to be financially secure through employment, so we offer competitive rewards to help build colleagues' personal wealth and improve their financial well-being. Base pay is just the start. Through our annual bonus program, we encourage performance that aligns with the Company's interests by providing eligible colleagues with discretionary awards. We also offer various incentives in certain circumstances, such as sales incentives and long-term incentives to people in roles that have a significant impact on our long-term performance and success. Our offerings also include retirement benefits, savings and stock investment plans in certain jurisdictions.

EXECUTIVE OFFICERS OF THE COMPANY

The executive officers and executive officer appointees of the Company are appointed annually by the Company's Board of Directors. The following individuals are the executive officers of the Company as of February 13, 2023:

Paul Beswick, age 48, is Senior Vice President and Global Chief Information Officer (CIO) of Marsh McLennan. In this role, he manages over 5,000 technologists supporting Marsh McLennan's global businesses. Prior to his appointment as Marsh McLennan CIO in January 2021, Mr. Beswick was a Partner and Global Head of Oliver Wyman Labs and the Digital Practice at Oliver Wyman. During more than two decades with Oliver Wyman, he worked in various sectors, including retail, transportation, telecom, and consumer goods. Before this, Mr. Beswick headed Oliver Wyman's North American Retail Practice. Mr. Beswick holds an MA (first class) in chemical engineering from Cambridge University.

Katherine J. Brennan, age 44, is Senior Vice President and General Counsel of Marsh McLennan. In this role, she leads Marsh McLennan's global legal, compliance and public affairs function, which supports the Company's four businesses, Marsh, Guy Carpenter, Mercer and Oliver Wyman. She also leads the Company's ESG efforts. Ms. Brennan has held several legal and compliance leadership roles at Marsh McLennan, serving most recently as General Counsel, Marsh LLC. She also served as Deputy General Counsel, Corporate Secretary and Chief Compliance Officer for Marsh McLennan from 2017 to 2021, and prior to that, as General Counsel of Guy Carpenter. Ms. Brennan currently serves on the Board of the Red Cross of Greater New York.

John Q. Doyle, age 59, is President and Chief Executive Officer of Marsh McLennan. Previously, from 2021 to 2022 he served as Group President and Chief Operating Officer, responsible for the strategy and operational objectives of Marsh McLennan's four global businesses. He joined the firm in 2016 as President of Marsh, then led Marsh as President and CEO from 2017 to 2021. An industry veteran with more than 35 years of management experience, Mr. Doyle began his career at AIG, where he held several executive positions. He is a member of the Board of the New York Police and Fire Widows' and Children's Benefit Fund, a Trustee of the Inner-City Scholarship Fund, a member of the Board of Overseers of the Maurice R. Greenberg School of Risk Management, Insurance and Actuarial Science at St. John's University and a former Director of the American Insurance Association. Mr. Doyle serves as the Chairman of the US Federal Advisory Committee on Insurance.

Martine Ferland, age 61, is President and Chief Executive Officer of Mercer. She also serves as Vice Chair of Marsh McLennan. Prior to assuming her current role in March 2019, she was Mercer's Group President, responsible for leading the firm's regions and Global Business Solutions. She joined Mercer in 2011 as Retirement Business Leader for EMEA, and has served as Europe and Pacific Region President and Co-President, Global Health. Ms. Ferland began her career as a pension actuary and consultant at Willis Towers Watson, where she spent 25 years and held various leadership positions in Montreal and New York. Ms. Ferland is a Fellow of the Society of Actuaries and of the Canadian Institute of Actuaries and a member of the Board of Trustees of the New York Academy of Medicine.

Carmen Fernandez, age 49, is Senior Vice President and Chief People Officer for Marsh McLennan. Prior to her appointment as Chief People Officer in January 2021, Ms. Fernandez held positions within Marsh McLennan for 15 years, most recently Deputy CHRO, CHRO of Guy Carpenter, and HR leadership roles at Mercer, including North America HR Leader, Global HR Leader for the Career business and Chief of Staff in the Office of the CEO. Before joining Marsh McLennan, Ms. Fernandez worked in investment banking at Bank of America and Goldman Sachs. She began her career as a consultant with PricewaterhouseCoopers.

John Jones, age 51, is Chief Marketing and Communications officer of Marsh McLennan. Previously, he served as Chief Marketing and Communications Officer of Marsh from 2018 to 2022. Mr. Jones joined Marsh in 2016 as senior vice president of Marsh's business planning, leading strategic planning and global growth initiatives. Prior to that, Mr. Jones was senior vice president of commercial marketing and strategy for AIG and has more than 25 years of marketing, communications and strategy experience.

Dean Klisura, age 59, is President and Chief Executive Officer of Guy Carpenter and serves as Vice Chair of Marsh McLennan. Prior to assuming this role in January 2022, he was President of Guy Carpenter, overseeing the North America, International, Specialty and Global Strategic Advisory business units. Prior to joining Guy Carpenter, Mr. Klisura was President of Marsh Global Placement and Advisory Services, leading property and casualty placement activities globally, as well as leading Bowring Marsh, the Insurer Consulting Group, and Marsh Advisory. He joined Marsh in 1993 and held several key global leadership roles including President of Global Specialties.

Mark McGivney, age 55, is Chief Financial Officer of Marsh McLennan. Prior to assuming this role in January 2016, Mr. McGivney held a number of senior financial management positions since joining the Company in 2007. Most recently he was Senior Vice President, Corporate Finance of Marsh McLennan, and was responsible for leading and directing the Company's Corporate Development, Treasury and Investor Relations functions from 2014 until 2016. Prior to that, he served as Chief Financial Officer of Marsh, and Chief Financial Officer and Chief Operating Officer of Mercer. His prior experience includes senior positions at The Hanover Insurance Group, including serving as Senior Vice President of Finance, Treasurer, and Chief Financial Officer of the Property & Casualty business, as well as positions with Merrill Lynch and PricewaterhouseCoopers.

Martin South, age 58, is President and Chief Executive Officer of Marsh, a position he assumed in January 2022, and oversees all of Marsh's businesses and operations globally. He also serves as Vice Chair of Marsh McLennan. With more than 30 years in the insurance industry, Mr. South joined Marsh for the first time in 1985 with Bowring Marsh, a Marsh McLennan broking unit. His industry experience includes senior leadership roles at Zurich Financial Services, where he was a member of the Group Management Board, responsible for all of Zurich's operations outside of North America and Europe, and CEO of Zurich's London operations. Since rejoining Marsh in 2007, Mr. South has served as CEO of Marsh's Asia-Pacific region, CEO of Marsh UK and Ireland, CEO of Marsh Europe and CEO of Marsh US and Canada.

Nicholas Studer, age 49, is Chief Executive Officer of Oliver Wyman Group, a role he assumed in July of 2021. He also serves as Vice Chair of Marsh McLennan. From 2017 to 2021, Mr. Studer was the Managing Partner of the Consumer, Industrial and Services Practice Group, before becoming Managing Partner of Oliver Wyman in 2021. He has held many senior positions at Oliver Wyman including Managing Partner of the Financial Services Practice Group, Head of the European Finance and Risk Practice and Global head of the Corporate and Institutional Banking practice. He has over 25 years of experience consulting in the UK, Continental Europe, and North America.

The Company is subject to the information reporting requirements of the Securities Exchange Act of 1934. In accordance with the Exchange Act, the Company files with, or furnishes to, the SEC its annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and proxy statement for its annual shareholders' meeting. The Company makes these reports and any amendments to these reports available free of charge through its website, www.marshmclennan.com, as soon as reasonably practicable after they are filed with or furnished to the SEC. The SEC also maintains a website at www.sec.gov that contains reports, proxy and information statements and other information regarding issuers, like the Company, that file electronically with the SEC.

The Company also posts on its website certain governance and other information for investors.

The Company encourages investors to visit these websites from time to time, as information is updated and new information is posted. Website references in this report are provided as a convenience and do not constitute, and should not be viewed as, incorporation by reference of the information contained on, or available through, the websites. Therefore, such information should not be considered part of this report.

Item 1A. Risk Factors.

You should consider the risks described below in conjunction with the other information presented in this report. These risks have the potential to materially adversely affect the Company's business, results of operations or financial condition.

SUMMARY RISK FACTORS

Some of the factors that could materially and adversely affect our business, financial condition, results of operations or prospects, include the following:

- Our results of operations and investments could be adversely affected by geopolitical or macroeconomic conditions;
- We could incur significant liability or our reputation could be damaged if our information systems are breached or we otherwise fail to protect client or Company data or information systems;
- The costs to comply with, or our failure to comply with, U.S. and foreign laws related to privacy, data security and data protection, such as the E.U. General Data Protection Regulation (GDPR) and the California Privacy Rights Act (CPRA), could adversely affect our financial condition, operating results and our reputation;
- Our business performance and growth plans could be negatively affected if we are not able to develop and implement improvements in technology or respond effectively to the threat of digital disruption and other technological change;
- We are subject to significant uninsured exposures arising from errors and omissions, breach of fiduciary duty and other claims;
- We cannot guarantee that we are or will be in compliance with all current and potentially applicable U.S. federal and state or foreign laws and regulations, and actions by regulatory authorities or changes in legislation and regulation in the jurisdictions in which we operate could have a material adverse effect on our business;
- Our business or reputation could be harmed by our reliance on third-party providers or introducers;
- · We may not be able to effectively identify and manage actual and apparent conflicts of interest;
- The loss of members of our senior management team or other key colleagues, or if we are unsuccessful in our efforts to attract, retain and develop talent, could have a material adverse effect on our business;
- Failure to maintain our corporate culture, particularly in a hybrid work environment, could damage our reputation;
- Increasing scrutiny and changing laws and expectations from regulators, investors, clients and our colleagues with respect to our
 environmental, social and governance (ESG) practices and disclosure may impose additional costs on us or expose us to new or
 additional risks;
- We face significant competitive pressures in each of our businesses, including from disintermediation, as our competitive landscape continues to evolve:
- We rely on a large number of vendors and other third parties to perform key functions of our business operations and to provide services to our clients. These vendors and third parties may act in ways that could harm our business;
- Our inability to successfully recover should we experience a disaster or other business continuity or data recovery problem could cause material financial loss, loss of human capital, regulatory actions, reputational harm or legal liability;
- We face risks when we acquire businesses:
- If we are unable to collect our receivables, our results of operations and cash flows could be adversely affected;
- · We may not be able to obtain sufficient financing on favorable terms;

- Our defined benefit pension plan obligations could cause the Company's financial position, earnings and cash flows to fluctuate;
- Our significant non-U.S. operations expose us to exchange rate fluctuations and various risks that could impact our business;
- Our quarterly revenues and profitability may fluctuate significantly:
- Credit rating downgrades would increase our financing costs and could subject us to operational risk;
- Our current debt level could adversely affect our financial flexibility;
- The current U.S. tax regime makes our results more difficult to predict;
- · We are exposed to multiple risks associated with the global nature of our operations;
- Results in our Risk and Insurance Services segment may be adversely affected by a general decline in economic activity;
- Volatility or declines in premiums and other market trends may significantly impede our ability to grow revenues and profitability;
- Adverse legal developments and future regulations concerning how intermediaries are compensated by insurers or clients, as well as
 allegations of anti-competitive behavior or conflicts of interest more broadly, could have a material adverse effect on Marsh's
 business, results of operations and financial condition;
- Mercer's Wealth business is subject to a number of risks, including risks related to public and private capital market fluctuations, third-party asset managers, operational and technology risks, conflicts of interest, ESG and greenwashing, asset performance and regulatory compliance, that, if realized, could result in significant damage to our business;
- Revenues for the services provided by our Consulting segment may decline for various reasons, including as a result of changes in
 economic conditions, the value of equity, debt and other asset classes, our clients' or an industry's financial condition or government
 regulation or an accelerated trend away from actively managed investments to passively managed investments;
- Factors affecting defined benefit pension plans and the services we provide relating to those plans could adversely affect Mercer;
- The profitability of our Consulting segment may decline if we are unable to achieve or maintain adequate utilization and pricing rates for our consultants.

RISKS RELATING TO THE COMPANY GENERALLY

Macroeconomic Risks

Our results of operations and investments could be adversely affected by geopolitical or macroeconomic conditions.

Geopolitical and macroeconomic conditions, including conflicts such as the war in Ukraine, slower GDP growth or recession, capital markets volatility and inflation affect our clients' businesses and the markets they serve. These conditions, including inflationary expense pressure with our clients, may reduce demand for our services or depress pricing for those services, which could have a material adverse effect on our results of operations.

For example, the war in Ukraine has continued to result in worldwide geopolitical and macroeconomic uncertainty and may negatively impact other regional and global economic markets (including Europe and the United States), companies in other countries (particularly those that have done business with Russia) and various sectors, industries and markets for securities and commodities globally, such as oil and natural gas, and may increase financial market volatility and adversely impact regional and global economic markets, industries and companies. Moreover, for nearly three years, the COVID-19 pandemic has impacted businesses, including our clients, third-party vendors and business partners, globally in every geography in which we operate. The ultimate extent of the impact of COVID-19, including the impact of hybrid working arrangements, on us will depend on future developments that we are unable to

predict. In addition, the continuing legal uncertainty, negotiations and potentially divergent laws and regulations as a result of Brexit may continue to lead to economic and legal uncertainty, causing increased economic volatility or disrupting the markets and clients we serve.

Changes in macroeconomic and geopolitical conditions could also shift demand to services for which we do not have a competitive advantage, and this could negatively affect the amount of business that we are able to obtain.

More generally, our investments, including our minority investments in other companies as well as our cash investments and those held in a fiduciary capacity, are subject to general credit, liquidity, counterparty, foreign exchange, market and interest rate risks. For example, in 2022, market conditions caused exchange rates to fluctuate significantly. These fluctuations in foreign exchange rates between the U.S. dollar and foreign currencies may adversely affect our results of operations.

These risks may be exacerbated by global macroeconomic conditions, market volatility and regulatory, financial and other difficulties affecting the companies in which we have invested or that may be faced by financial institution counterparties. During times of stress in the banking industry, counterparty risk can quickly escalate, potentially resulting in substantial trading and investment losses for corporate and other investors. In addition, we may incur investment losses as a result of unusual and unpredictable market developments, and we may continue to experience reduced investment earnings if the yields on investments deemed to be low risk remain at or near their current low levels. If the banking system or the fixed income, interest rate, credit or equity markets deteriorate, the value and liquidity of our investments could be adversely affected. Finally, the value of the Company's assets held in other jurisdictions, including cash holdings, may decline due to foreign exchange fluctuations.

Cybersecurity, Data Protection and Technology Risks

We could incur significant liability or our reputation could be damaged if our information systems are breached or we otherwise fail to protect client or Company data or information systems.

In operating our business and providing services and solutions to clients, we collect, use, store, transmit and otherwise process certain electronic information, including personal, confidential, proprietary and sensitive data such as information related to financial records, health care, mergers and acquisitions and personal data of our clients, colleagues and vendors. We rely on the efficient, uninterrupted and secure operation of complex information technology systems and networks to operate our business and securely process, transmit and store electronic information. In the normal course of business, we also share electronic information with our vendors and other third parties. This electronic information comprises sensitive and confidential data, including information related to financial records, health care, mergers and acquisitions and clients' personal data. Our information technology systems and safety control systems, and those of our numerous thirdparty providers, as well as the control systems of critical infrastructure they rely on, such as power grids, and undersea cables, are potentially vulnerable to unauthorized access, damage or interruption from a variety of external threats, including physical attack, cyberattacks, computer viruses and other malware, ransomware and other types of data and systems-related modes of attack. A disruption of physical infrastructure could impact our ability to conduct business and service clients. This may include deliberate or unintentional disruption of service to electrical systems, satellite communications, undersea or terrestrial cable systems, Internet services, or other systems our colleagues or third parties rely on us to conduct business in a multitude of jurisdictions across the globe. Disruptions may be the result of weather, natural disaster, war, terrorism, pandemic, or other natural or geopolitical event. Our systems are also subject to compromise from internal threats such as improper action by employees, vendors and other third parties with otherwise legitimate access to our systems. Moreover, we face the ongoing challenge of managing access controls in a complex environment. The latency of a compromise is often measured in months but could be years, and we may not be able to detect a compromise in a timely manner. We could experience significant financial and reputational harm if our information systems are breached, sensitive client or Company data are compromised, surreptitiously modified, rendered inaccessible for any period of time or maliciously made public, or if we fail to make adequate or timely disclosures to the public or law enforcement agencies following any such event, whether due to delayed discovery or a failure to follow existing protocols.

Cyberattacks are increasing in frequency and evolving in nature. We are at risk of attack by a variety of adversaries, including state-sponsored organizations, organized crime and hackers, through use of

increasingly sophisticated methods of attack, including the deployment of artificial intelligence to find and exploit vulnerabilities, "deep fakes", long-term, persistent attacks (referred to as advanced persistent threats) and the use of the IT supply chain to introduce malware through software updates or compromised suppliers accounts or hardware. In particular, we are at increased risk of a cyberattack during periods of heightened geopolitical conflict, such as the war in Ukraine, as diplomatic events and economic policies may trigger espionage or retaliatory cyber incidents.

The techniques used to obtain unauthorized access or sabotage systems include, among other things, computer viruses, malicious or destructive code, ransomware, social engineering attacks (including phishing and impersonation), hacking and denial-of-service attacks. Because these techniques change frequently and new techniques may not be identified until they are launched against a target, we may be unable to anticipate these techniques or implement adequate preventative measures, resulting in potential data loss, data unavailability, data corruption or other damage to information technology systems. In addition, an increased level of remote and hybrid work arrangements post COVID-19 has increased the risk of phishing and other cybersecurity attacks or unauthorized dissemination of personal, confidential, proprietary or sensitive data.

As the breadth and complexity of the technologies we use and the software and platforms we develop continue to grow, including as a result of the use of mobile devices, cloud services, "open source" software, social media tools and the increased reliance on devices connected to the Internet (known as the "Internet of Things"), the potential risk of security breaches and cyber-attacks also increases. Despite ongoing efforts to improve our ability to protect data from compromise, we may not be able to protect all of our data across our diverse systems. Our efforts to improve and protect data from compromise may also identify previously undiscovered instances of security breaches or other cyber incidents. Our policies, employee training (including phishing prevention training), procedures and technical safeguards may also be insufficient to prevent or detect improper access to confidential, personal or proprietary information. In addition, the competition for talent in the data privacy and cybersecurity space is intense, and we may also be unable to hire, develop or retain suitable talent capable of adequately detecting, mitigating or remediating these risks.

Should an attacker gain access to our network using compromised credentials of an authorized user, we are at risk that the attacker might successfully leverage that access to compromise additional systems and data. Certain measures that could increase the security of our systems, such as data encryption (including encryption of data at rest), heightened monitoring and logging, scanning for source code errors or deployment of multi-factor authentication, take significant time and resources to deploy broadly, and such measures may not be deployed in a timely manner or be effective against an attack. The inability to implement, maintain and upgrade adequate safeguards could have a material adverse effect on our business.

Our information systems must be continually updated, patched, and upgraded to protect against known vulnerabilities. The volume of new software vulnerabilities has increased markedly, as has the criticality of patches and other remedial measures. In addition to remediating newly identified vulnerabilities, previously identified vulnerabilities must also be continuously addressed. Accordingly, we are at risk that cyberattackers exploit these known vulnerabilities before they have been communicated by vendors or addressed. Due to the large number and age of the systems and platforms that we operate, the increased frequency at which vendors are issuing security patches to their products, the need to test patches and, in some cases coordinate with clients and vendors, before they can be deployed, we perpetually face the substantial risk that we cannot deploy patches in a timely manner. We are also dependent on third party vendors to keep their systems patched and secure in order to protect our data. Any failure related to these activities could have a material adverse effect on our business.

We have numerous vendors and other third parties who receive personal information from us in connection with the services we offer our clients and our employees. We also use hundreds of IT vendors and software providers to maintain and secure our global information systems infrastructure. In addition, we have migrated certain data, and may increasingly migrate data, to the cloud hosted by third-party providers. Some of these vendors and third parties also have direct access to our systems. We are at risk of a cyberattack involving a vendor or other third party, which could result in a breakdown of such third party's data protection processes or the cyberattackers gaining access to our infrastructure through a supply chain attack. Highly publicized data security breaches, such as the December 2020 large-scale

attack on SolarWinds that created security vulnerabilities for public and private organizations around the world may embolden malicious actors to target the IT supply chain and providers of business software. Other similar supply chain compromises could have a significant negative impact on our systems and operations.

We have a history of making acquisitions and investments, including the acquisition of JLT in 2019. The process of integrating the information systems of any businesses we acquire is complex and exposes us to additional risk. For instance, we may not adequately identify weaknesses and vulnerabilities in an acquired entity's information systems, either before or after the acquisition, which could affect the value we are able to derive from the acquisition, expose us to unexpected liabilities or make our own systems more vulnerable to a cyberattack. In addition, if we discover a historical compromise, security breach or other cyber incident related to the target's information systems following the close of the acquisition, we may be liable and exposed to significant costs and other unforeseen liabilities. We may also be unable to integrate the systems of the businesses we acquire into our environment in a timely manner, which could further increase these risks until such integration takes place.

We have experienced data incidents and cybersecurity breaches, such as malware incursions (including computer viruses and ransomware), vulnerabilities in the software on which we rely, users exceeding their data access authorization, employee misconduct and incidents resulting from human error, such as emails sent to the wrong recipient, loss of portable and other data storage devices or misconfiguration of software or hardware resulting in inadvertent exposure of personal, sensitive, confidential or proprietary information. In April 2021, an unauthorized actor leveraged a vulnerability in a third party's software and gained access to a limited set of data in our environment. Like many companies, we are also subject to social engineering attacks such as WhatsApp scams and regular phishing email campaigns directed at our employees that can result in malware infections, fraud and data loss. Although these incidents have resulted in data loss and other damages, to date, they have not had a material adverse effect on our business or operations. In the future, these types of incidents could result in personal, sensitive, confidential or proprietary information being lost or stolen, surreptitiously modified, rendered inaccessible for any period of time, or maliciously made public, including client, employee or Company data, which could have a material adverse effect on our business. In the event of a cyberattack, we might have to take our systems offline, which could interfere with services to our clients or other malware. We also may be unable to detect an incident, assess its severity or impact, or appropriately respond in a timely manner. In addition, our liability insurance, which includes cyber insurance, may not be sufficient in type or amount to cover us against claims related to security breaches, cyberattacks and other related data and system incidents.

The costs to comply with, or our failure to comply with, U.S. and foreign laws related to privacy, data security and data protection, such as the E.U. General Data Protection Regulation (GDPR) and the California Privacy Rights Act (CPRA), could adversely affect our financial condition, operating results and our reputation.

Improper collection, use disclosure, cross border transfer, and retention of confidential, personal, or proprietary data could result in regulatory scrutiny, legal and financial liability, or harm to our reputation. In operating our business and providing services and solutions to clients, we store and transfer sensitive employee and client data, including personal data, in and across multiple jurisdictions. We collect data from client and individuals located all over the world and leverage systems and teams to process it. As a result, we are subject to a variety of laws and regulations in the United States, Europe and around the world regarding privacy, data protection, data security and cyber security. These laws and regulations are continuously evolving and developing. Some of these laws and regulations are increasing the level of data handling restrictions, including rules on data localization, all of which could affect our operations and result in regulatory liability and high fines. In particular, high-profile data breaches at major companies continue to be disclosed regularly, which is leading to even greater regulatory scrutiny and fines at the highest levels they have ever been. These fines are not limited to data breaches and regulators are increasingly focusing on other data processing activities including those related to ad-tech and "data subject" rights.

The scope and interpretation of the laws that are or may be applicable to us are often uncertain and may be conflicting. For example, the GDPR, which became effective in May 2018, greatly increased the

European Commission's jurisdictional reach of its laws and added a broad array of requirements for handling personal data, such as the public disclosure of data breaches, privacy impact assessments, data portability and the appointment of data protection officers in some cases. In the U.S., CPRA was passed in late 2020 and has greatly expanded the requirements under the California Consumer Privacy Act (CCPA). Despite a proliferation of regulatory guidance papers, there remains uncertainty in key areas related to the GDPR and the CPRA, and that uncertainty could result in potential liability for our failure to meet our obligations under the GDPR and the CPRA. Given the breadth and depth of changes in data protection obligations, including classifying data and committing to a range of administrative, technical and physical controls to protect data and enable data transfers outside of the E.U., our compliance with laws such as the GDPR and the CPRA will continue to require time, resources and review of the technology and systems we use. Further, the European Union Court of Justice's "Schrems II" decision and Brexit continue to generate uncertainty with regard to the future of the flow of personal information between the U.S. and E.U and between the United Kingdom and the E.U., and that uncertainty may impair our ability to offer our existing and planned products and services or increase our cost of doing business.

Following the implementation of the GDPR, other jurisdictions have sought to amend, or propose legislation to amend, their existing data protection laws to align with the requirements of the GDPR with the aim of obtaining an adequate level of data protection to facilitate the transfer of personal data to most jurisdictions from the E.U. Accordingly, the challenges we face in the E.U. will likely also apply to other jurisdictions that adopt laws similar to the GDPR or regulatory frameworks of equivalent complexity. For example, Indonesia passed the Personal Data Protection Bill in 2022, Australia and Canada are seeking to make major amendments to their existing privacy laws and India is engaging in an ongoing effort to enact a new privacy law. Some of the laws enacted in recent years, including those in China and the Kingdom of Saudi Arabia, the laws include data localization elements that will require that certain personal data stay within their borders.

In the U.S., following the passage of the CCPA and CPRA, four other states (Colorado, Connecticut, Utah and Virginia) passed privacy laws and there remains continued legislative interest in passing laws in additional states, as well as a federal privacy law, though the prospects of such a law passing soon have diminished.

In addition to data protection laws, countries and states in the U.S. are enacting cybersecurity laws and regulations. For example, in 2017 the New York State Department of Financial Services (NYDFS) issued cybersecurity regulations which imposed an array of detailed security measures on covered entities. These requirements were phased in and the last of them came into effect on March 1, 2019. The NYDFS has now proposed an array of modifications to those rules which if passed would impose significant new requirements. A number of states have also adopted laws covering data collected by insurance licensees that include security and breach notification requirements. And at the federal level, the Securities and Exchange Commission is seeking to impose new cybersecurity requirements, including new reporting obligations, on publicly traded companies. All of these evolving compliance and operational requirements impose significant costs that are likely to increase over time, may divert resources from other initiatives and projects and could restrict the way services involving data are offered, all of which may adversely affect our results of operations.

Many statutory requirements, both in the United States and abroad, include obligations for companies to notify individuals of security breaches involving certain personal information, which could result from breaches experienced by us or our vendors. In addition to government regulation, privacy advocates and industry groups have and may in the future propose self-regulatory standards from time to time. These and other industry standards may legally or contractually apply to us, or we may elect to comply with such standards. We expect that there will continue to be new proposed laws and regulations concerning data privacy and security, and we cannot yet determine the impact such future laws, regulations and standards may have on our business. Data protection laws also include strict notification requirements for organizations related to confirmed or suspected breaches. With such a limited time available to validate indicators, there is an increased risk of reporting a false alarm or immaterial breach, which may lead to reputational damage despite there not being an actual data breach.

Furthermore, enforcement actions and investigations by regulatory authorities related to data security incidents and privacy violations, including a recent focus on website "cookies" compliance in some

countries, continue to increase. Privacy violations, including unauthorized use disclosure or transfer of sensitive or confidential client or Company data, whether through systems failure, employee negligence, fraud or misappropriation, by the Company, our vendors or other parties with whom we do business (if they fail to meet the standards we impose) could damage our reputation and subject us to significant litigation, monetary damages, regulatory enforcement actions, fines and criminal prosecution in one or more jurisdictions. Given the complexity of operationalizing the various privacy laws such as the GDPR and the CPRA, the maturity level of proposed compliance frameworks and the continued lack of certainty on how to implement their requirements, we and our clients are at risk of enforcement actions taken by E.U. and other data protection authorities or litigation from consumer advocacy groups acting on behalf of data subjects. We may not be able to respond quickly or effectively to regulatory, legislative and other developments, and these changes may in turn impair our ability to offer our existing or planned products and services and increase our cost of doing business.

Our business performance and growth plans could be negatively affected if we are not able to develop and implement improvements in technology or respond effectively to the threat of digital disruption and other technological change.

We depend in large part on our technology systems for conducting business, as well as for providing the data and analytics we use to manage our business. As a result, our business success is dependent on maintaining the effectiveness of existing technology systems and on continuing to develop and enhance technology systems that support our business processes and strategic initiatives in a cost and resource efficient manner, particularly as our business processes become more digital. We have a number of strategic initiatives involving investments in or partnerships with technology companies as part of our growth strategy, as well as investments in technology and infrastructure to support our own systems.

These investments may be costly and require significant capital expenditures, may not be profitable or may be less profitable than what we have experienced historically. In addition, investments in technology systems may not deliver the benefits or perform as expected, or may be replaced or become obsolete more quickly than expected, which could result in operational difficulties or additional costs. In some cases, we also depend on key vendors and partners to provide technology and other support for our strategic initiatives. If these vendors or partners fail to perform their obligations or otherwise cease to work with us, our ability to execute on our strategic initiatives could be adversely affected. If we do not keep up with technological changes or execute effectively on our strategic initiatives, our business and results of operations could be adversely impacted.

In addition, to remain competitive in many of our business areas, we must anticipate and respond effectively to the threat of digital disruption and other technological change. The threat comes from traditional players, such as insurers, through disintermediation as well as from new entrants, such as technology companies, "Insurtech" start-up companies and others. In the past few years, there has been a substantial increase in private equity investments into these Insurtech companies. These players are focused on using technology and innovation, including artificial intelligence (AI), digital platforms, data analytics, robotics and blockchain, to simplify and improve the client experience, increase efficiencies, alter business models and effect other potentially disruptive changes in the industries in which we operate.

Legal and Regulatory Risks

We are subject to significant uninsured exposures arising from errors and omissions, breach of fiduciary duty and other claims.

Our businesses provide numerous professional services, including the placement of insurance and the provision of consulting, investment advisory, investment management and actuarial services, to clients around the world. As a result, the Company and its subsidiaries are subject to a significant number of errors and omissions, breach of fiduciary duty, breach of contract and similar claims, which we refer to collectively as "E&O claims." In our Risk and Insurance Services segment, such claims include allegations of damages arising from our failure to assess clients' risks, advise clients, place coverage, or notify insurers of potential claims on behalf of clients in accordance with our obligations to them. For example, these claims could include allegations related to losses incurred by policyholders arising from the COVID-19 pandemic, or losses from cyberattacks associated with policies where cyber risk was not specifically included or excluded in policies, commonly referred to as "silent cyber." In our Consulting

segment, where we increasingly act in a fiduciary capacity through our investments business, such claims could include allegations of damages arising from the provision of consulting, investment management (including, for example, from trading or other operational errors), actuarial, pension administration and other services. We may also be exposed to claims related to assets or solutions offered by the Consulting segment in complement to its traditional consulting services. These Consulting segment services frequently involve complex calculations and services, including (i) making assumptions about, and preparing estimates concerning, contingent future events, (ii) drafting and interpreting complex documentation governing pension plans, (iii) calculating benefits within complex pension structures, (iv) providing individual financial planning advice including investment advice and advice relating to cashing out of defined benefit pension plans, (v) providing investment advice, including guidance on asset allocation and investment strategy, and (vi) managing client assets, including the selection of investment managers and implementation of the client's investment policy. We provide these services to a broad client base, including clients in the public sector for our investment services. Matters often relate to services provided by the Company dating back many years. Such claims may subject us to significant liability for monetary damages, including punitive and treble damages, negative publicity and reputational harm, and may divert personnel and management resources. We may be unable to effectively limit our potential liability in certain jurisdictions, including through insurance, or in connection with certain types of claims, particularly those concerning claims of a breach of fiduciary duty.

In establishing liabilities for E&O claims under U.S. generally accepted accounting principles ("U.S. GAAP"), the Company uses case level reviews by inside and outside counsel, actuarial analysis by Oliver Wyman, a subsidiary of the Company, and other methods to estimate potential losses. A liability is established when a loss is both probable and reasonably estimable. The liability is assessed quarterly and adjusted as developments warrant. In many cases, the Company has not recorded a liability, other than for legal fees to defend the claim, because we are unable, at the present time, to make a determination that a loss is both probable and reasonably estimable. Given the judgment involved in estimating and establishing liabilities in accordance with U.S. GAAP, as well as the unpredictability of E&O claims and the litigation that can flow from them, it is possible that an adverse outcome in a particular matter could have a material adverse effect on the Company's business, results of operations or financial condition.

We cannot guarantee that we are or will be in compliance with all current and potentially applicable U.S. federal and state or foreign laws and regulations, and actions by regulatory authorities or changes in legislation and regulation in the jurisdictions in which we operate could have a material adverse effect on our business.

Our activities are subject to extensive regulation under the laws of the United States and its various states, the United Kingdom, the European Union and its member states, Australia and the other jurisdictions in which we operate. We are also subject to trade sanctions laws relating to countries such as Afghanistan, Belarus, Cuba, Iran, North Korea, Russia, Syria, Ukraine (Russia-controlled territories) and Venezuela, and anti-corruption laws such as the U.S. Foreign Corrupt Practices Act and the U.K. Bribery Act. We are subject to numerous other laws on matters as diverse as internal control over financial reporting and disclosure controls and procedures, securities regulation, data privacy and protection, cybersecurity, taxation, anti-trust and competition, immigration, wage-and-hour standards and employment and labor relations.

The U.S. and foreign laws and regulations that apply to our operations are complex and may change rapidly, and our efforts to comply and keep up with them require significant resources. In some cases, these laws and regulations may decrease the need for our services, increase our costs, negatively impact our revenues or impose operational limitations on our business, including on the products and services we may offer or on the amount or type of compensation we may collect. In addition, the financial and operational impact of complying with laws and regulations has increased in the current environment of increased regulatory activity and enforcement. Changes with respect to the applicable laws and regulations may impose additional and unforeseen costs on us or pose new or previously immaterial risks to us. There can be no assurance that current and future government regulations will not adversely affect our business, and we cannot predict new regulatory priorities, the form, content or timing of regulatory actions, and their impact on our business and operations. In addition, geopolitical conflict, such as the war in Ukraine, has resulted in, and may continue to result in, new and rapidly evolving trade sanctions, which

may increase our costs, negatively impact our revenues or impose additional operational limitations on our businesses.

While we attempt to comply with applicable laws and regulations, there can be no assurance that we, our employees, our consultants and our contractors and other agents are in full compliance with such laws and regulations or interpretations at all times, or that we will be able to comply with any future laws or regulations. If we fail to comply or are accused of failing to comply with applicable laws and regulations, including those referred to above, or new and evolving regulations regarding cybersecurity, artificial intelligence or environmental, social and governance matters, we may become subject to investigations, criminal penalties, civil remedies or other consequences, including fines, injunctions, loss of an operating license or approval, increased scrutiny or oversight by regulatory authorities, the suspension of individual employees, limitations on engaging in a particular business or redress to clients or other parties, and we may become exposed to negative publicity or reputational damage. Moreover, our failure to comply with laws or regulations in one jurisdiction may result in increased regulatory scrutiny by other regulatory agencies in that jurisdiction or regulatory agencies in other jurisdictions. These inquiries consume significant management attention, and the cost of compliance and the consequences of failing to be in compliance could therefore have a material adverse effect on our business.

In addition, we may be responsible for the legal and regulatory liabilities of companies that we acquire. In particular, upon the consummation of the acquisition of JLT, the Company assumed the legal liabilities and became responsible for JLT's litigation and regulatory exposures as of April 1, 2019. Additional information regarding certain ongoing investigations and certain other legal and regulatory proceedings is set forth in Note 16, Claims, Lawsuits and Other Contingencies, in the notes to the consolidated financial statements included under Part II, Item 8 of this report.

In most jurisdictions, government regulatory authorities have the power to interpret and amend or repeal applicable laws and regulations, and have discretion to grant, renew and revoke the various licenses and approvals we need to conduct our activities. Such authorities may require the Company to incur substantial costs in order to comply with such laws and regulations. In some areas of our businesses, we act on the basis of our own or the industry's interpretations of applicable laws or regulations, which may conflict from state to state or country to country. In the event those interpretations eventually prove different from the interpretations of regulatory authorities, we may be penalized or precluded from carrying on our previous activities. Moreover, the laws and regulations to which we are subject may conflict among the various jurisdictions and countries in which we operate, which increases the likelihood of our businesses being non-compliant in one or more jurisdictions.

Our business or reputation could be harmed by our reliance on third-party providers or introducers.

We currently utilize the services of hundreds of third-party providers to meet the needs of our clients around the world.

There is a risk that our third-party providers or introducers engage in business practices that are prohibited by our internal policies or violate applicable laws and regulations, such as the U.S. Foreign Corrupt Practices Act and the U.K. Anti-Bribery Act.

We may not be able to effectively identify and manage actual and apparent conflicts of interest.

Given the significant volume of our engagements, potential conflicts of interest may arise across our businesses. There is a risk that we may not effectively identify and manage potential conflicts of interest, including but not limited to where our services to a client conflict, or are perceived to conflict, with the interests of another client or our own interests, where we receive revenue or benefits from third-parties with whom we conduct business (including but not limited to insurers, investment managers and vendors) and where our colleagues have personal interests.

Competitive Risks

The loss of members of our senior management team or other key colleagues, or if we are unsuccessful in our efforts to attract, retain and develop talent, could have a material adverse effect on our business.

We rely upon the contributions of our senior management team to establish and implement our business strategy and to manage the future growth of our business. We may be unable to retain them, particularly if we do not offer employment terms that are competitive with the rest of the labor market. The loss of any of the senior management team could limit our ability to successfully execute our business strategy or adversely affect our ability to retain existing and attract new clients. Moreover, we could be adversely affected if we fail to adequately plan for the succession of members of our senior management team or if our succession plans do not operate effectively.

Across all of our businesses, our colleagues are critical to developing and retaining client relationships as well as performing the services on which our revenues are earned. It is therefore important for us to attract, incentivize and retain significant revenue-producing employees and the key managerial and other professionals who support them. We face numerous challenges in this regard, including the intense competition for talent, which has accelerated through the pandemic. Such challenges include the general mobility of colleagues that has increased as a result of the COVID-19 pandemic as companies experiment with more flexible working models, market dislocation resulting from proposed and actual combinations in the industry, and fostering an inclusive and diverse workplace.

Losing colleagues who manage or support substantial client relationships or possess substantial experience or expertise could adversely affect our ability to secure and complete client engagements, which could adversely affect our results of operations. If a key employee were to join an existing competitor or form a competing company, some of our clients could choose to use the services of that competitor instead of our services. If a colleague joins us from a competitor and is subject to enforceable restrictive covenants, we may not be able to secure client engagements or maximize the colleague's potential. In addition, regulation or legislation impacting the workforce, such as the proposed U.S. Federal Trade Commission rule regarding noncompete clauses, may lead to increased uncertainty and competition for talent.

Failure to maintain our corporate culture, particularly in a hybrid work environment, could damage our reputation.

We strive to foster a culture in which our colleagues act with integrity and feel comfortable speaking up about potential misconduct. We are a people business, and our ability to attract and retain colleagues and clients is dependent upon our commitment to an inclusive and diverse workplace, trustworthiness, ethical business practices and other qualities. Our colleagues are the cornerstone of this culture, and acts of misconduct by any colleague, and particularly by senior management, could erode trust and confidence and damage our reputation among existing and potential clients and other stakeholders. Remote and hybrid work arrangements may also negatively impact our ability to maintain and promote our culture, as we believe being together is integral to promoting our culture.

Increasing scrutiny and changing laws and expectations from regulators, investors, clients and our colleagues with respect to our environmental, social and governance (ESG) practices and disclosure may impose additional costs on us or expose us to new or additional risks.

There is increased focus, including from governmental organizations, regulators (including the SEC in the United States), investors, colleagues and clients, on ESG issues such as environmental stewardship, climate change, greenhouse gas emissions, inclusion and diversity, human rights, racial justice, pay equity, workplace conduct, cybersecurity and data privacy. Negative public perception, adverse publicity or negative comments in social media could damage our reputation if we do not, or are not perceived to, adequately address these issues. Any harm to our reputation could impact colleague engagement and retention and the willingness of clients and our partners to do business with us.

Additionally, there has been increased regulatory focus on ESG and sustainability. For example, laws and regulations related to ESG issues continue to evolve, including in the U.S., the U.K., the EU and Australia, and these regulations may impose additional compliance or disclosure obligations on us. In particular, heightened demand for, and scrutiny of, ESG and sustainable-related products, funds, investment

strategies and advice has increased the risk that we could be perceived as, or accused of, making inaccurate or misleading statements, commonly referred to as "greenwashing" or that we have otherwise run afoul of regulation. Such perceptions or accusations could damage our reputation, result in litigation or regulatory enforcement actions, and adversely affect our business. Furthermore, perceptions of our efforts to achieve ESG goals or advance ESG and sustainable-related products, funds, investment strategies or advice may differ widely among stakeholders and could present risks to our reputation and business.

Moreover, as we continue to align with the recommendations of the Financial Stability Board's Task Force on Climate-related Financial Disclosures (TCFD), the Sustainability Accounting Standards Board (SASB), and our own ESG assessments and priorities, we have expanded our public disclosures in these areas, including providing additional metrics and goals. These disclosures, metrics and goals and any failure to accurately report or comply with federal, state or international ESG laws and regulations, or to achieve progress on our metrics and goals on a timely basis, or at all, may result in legal and regulatory proceedings against us and negatively impact our reputation.

Implementation of our ESG initiatives also depends in part on third-party performance or data that is outside the Company's control.

In addition, organizations that provide information to investors on corporate governance and related matters have developed ratings processes for evaluating companies on their approach to ESG matters, and unfavorable ratings of our company or our industries may lead to negative investor sentiment and the diversion of investment to other companies or industries, exclusion of our stock from ESG-oriented indices or investment funds or harm our relationships with regulators and the communities in which we operate.

We face significant competitive pressures in each of our businesses, including from disintermediation, as our competitive landscape continues to evolve.

As a global professional services firm, the Company faces competition in each of its businesses, and the competitive landscape continues to change and evolve. Our ability to compete successfully depends on a variety of factors, including the quality and expertise of our colleagues, our geographic reach, the sophistication and quality of our services, our pricing relative to competitors, our clients' ability to self-insure or use internal resources instead of consultants, and our ability to respond to changes in client demand and industry conditions. Some of our competitors may have greater financial resources, or may be better positioned to respond to technological and other changes in the industries we serve, and they may be able to compete more effectively. Additionally, the competition for talent has only accelerated following the COVID-19 pandemic and recent dislocation in the market resulting from proposed and actual combinations among our competitors.

Across our Risk and Insurance Services segment, we operate in a variety of markets and face different competitive landscapes. In addition to the challenges posed by capital market alternatives to traditional insurance and reinsurance, we compete against a wide range of other insurance and reinsurance brokerage and risk advisory firms that operate on a global, regional, national or local scale for both client business and employee talent. In recent years, private equity sponsors have invested tens of billions of dollars into the insurance brokerage sector, transforming existing players and creating new ones to compete with large brokers. We also compete with insurance companies that market and service their insurance products directly to consumers and reinsurance companies that market and service their products directly to insurance companies, in each case without the assistance of brokers or other market intermediaries, and with various other companies that provide risk-related services or alternatives to traditional brokerage services, including those that rely almost exclusively on technological solutions or platforms. This competition is intensified by an often "syndicated" or "distributed" approach to the purchase of insurance and reinsurance brokerage services, where a client engages multiple brokers to service different portions of the client's account. In addition, third party capital providers have entered the insurance and reinsurance risk transfer market offering products and capital directly to our clients that serve as substitutes for traditional insurance.

In our Consulting segment, we compete for business with numerous consulting firms and similar organizations, many of which also provide, or are affiliated with firms that provide, accounting, information systems, technology and financial services. Such competitors may be able to offer more comprehensive

products and services to potential clients, which may give them a competitive advantage. Some of our competitors also may be able to invest more significant capital in technology and digital solutions. In certain sub-segments, we compete in highly fragmented markets or with startups that may be able to offer solutions at a lower price or on more favorable conditions.

In addition, companies in the industries that we serve may seek to achieve economies of scale and other synergies by combining with or acquiring other companies. If two or more of our current clients merge, or consolidate or combine their operations, it may decrease the amount of work that we perform for these clients.

We rely on a large number of vendors and other third parties to perform key functions of our business operations and to provide services to our clients. These vendors and third parties may act in ways that could harm our business.

We rely on a large number of vendors and other third parties, and in some cases subcontractors, to provide services, data and information such as technology, information security, funds transfers, business process management, and administration and support functions that are critical to the operations of our business. These third parties include correspondents, agents and other brokers and intermediaries, insurance markets, data providers, plan trustees, payroll service providers, software and system vendors, health plan providers, investment managers, custodians, risk modeling providers, and providers of human resource functions, such as recruiters. Many of these providers are located outside the U.S., which exposes us to business disruptions and political risks inherent when conducting business outside of the U.S. As we do not control many of the actions of these third parties, we are subject to the risk that their decisions or operations may adversely impact us and replacing these service providers could create significant delay in services or operations and additional expense.

A failure by the third parties to (i) comply with service level agreements in a high quality and timely manner, particularly during periods of our peak demand for their services, (ii) maintain adequate internal controls that may impact our own financial reporting, or (iii) adequately maintain the confidentiality of any of our data or trade secrets or adequately protect or properly use other intellectual property to which they may have access, could result in economic and reputational harm to us. These third parties also face their own technology, operating, business and economic risks, and any significant failures by them, including the improper use or disclosure of our confidential client, employee, or Company information or failure to comply with applicable law, could cause harm to our reputation or otherwise expose us to liability. An interruption in or the cessation of service by any service provider as a result of systems failures, capacity constraints, non-compliance with legal, regulatory or contractual obligations, financial difficulties or for any other reason could disrupt our operations, impact our ability to offer certain products and services, and result in contractual or regulatory penalties, liability claims from clients or employees, damage to our reputation and harm to our business.

Business Resiliency Risks

Our inability to successfully recover should we experience a disaster or other business continuity or data recovery problem could cause material financial loss, loss of human capital, regulatory actions, reputational harm or legal liability.

If we experience a local or regional disaster or other business continuity event, such as an earthquake, hurricane, flood, terrorist attack, pandemic, war or other geopolitical tensions, protests or riots, security breach, cyberattack (including manipulating the control systems of critical infrastructure), power loss or telecommunications failure, our ability to operate will depend, in part, on the continued availability of our personnel, our office facilities and the proper functioning of our computer, telecommunication and other related systems and operations. In such an event, we could experience operational challenges that could have a material adverse effect on our business. The risk of business disruption is more pronounced in certain geographic areas, including major metropolitan centers, like New York or London, where we have significant operations and approximately 3,500 and 5,600 colleagues in those respective locations, and in certain countries and regions in which we operate that are subject to higher potential threat of terrorist attacks or military conflicts.

Our operations depend in particular upon our ability to protect our technology infrastructure against damage. If a business continuity event occurs, we could lose client or Company data or experience interruptions to our operations or delivery of services to our clients, which could have a material adverse

effect. Such risks have increased significantly due to the hybrid work environment following the COVID-19 pandemic. A cyberattack or other business continuity event affecting us or a key vendor or other third party could result in a significant and extended disruption in the functioning of our information technology systems or operations or our ability to recover data, requiring us to incur significant expense to address and remediate or otherwise resolve such issues. For example, hackers have increasingly targeted companies by attacking internet-connected industrial control and safety control systems. An extended outage could result in the loss of clients and a decline in our revenues. In the worst case, any manipulation of the control systems of critical infrastructure may even result in the loss of life.

We regularly assess and take steps to improve our existing business continuity, disaster recovery and data recovery plans and key management succession. However, a disaster or other continuity event on a significant scale or affecting certain of our key operating areas within or across regions, or our inability to successfully recover from such an event, could materially interrupt our business operations and result in material financial loss, loss of human capital, regulatory actions, reputational harm, damaged client relationships and legal liability. Our business disruption insurance may also not fully cover, in type or amount, the cost of a successful recovery in the event of such a disruption.

Acquisitions and Dispositions Risks

We face risks when we acquire businesses.

We have a history of making acquisitions and investments, including a total of 93 in the period from 2017 to 2022. We may not be able to successfully integrate the businesses that we acquire into our own business, or achieve any expected cost savings or synergies from the integration of such businesses. Subject to standard contractual protections, we may also be responsible for legacy liabilities of companies that we acquire.

In addition, if in the future the performance of our reporting units or an acquired business varies from our projections or assumptions, or estimates about future profitability of our reporting units or an acquired business change, the estimated fair value of our reporting units or an acquired business could change materially and could result in an impairment of goodwill and other acquisition-related intangible assets recorded on our balance sheet or in adjustments in contingent payment amounts. Given the significant size of the Company's goodwill and intangible assets, an impairment could have a material adverse effect on our results of operations in any given period.

We expect that acquisitions will continue to be a key part of our business strategy. Our success in this regard will depend on our ability to identify and compete for appropriate acquisition candidates and to finance and complete the transactions we decide to pursue on favorable terms with positive results.

When we dispose of businesses, we may continue to be subject to certain liabilities of that business after its disposition relating to the prior period of our ownership and may not be able to negotiate for limitations on those liabilities. We are also subject to the risk that the sales price is less than the amount reflected on our balance sheet.

Financial Risks

If we are unable to collect our receivables, our results of operations and cash flows could be adversely affected.

Our business depends on our ability to obtain payment from our clients of the amounts they owe us for the work we perform. As of December 31, 2022, our receivables for our commissions and fees were approximately \$5.3 billion, or approximately one-quarter of our total annual revenues, and portions of our receivables are increasingly concentrated in certain businesses and geographies.

Macroeconomic or geopolitical conditions, such as a slower economic growth or recession, the war in Ukraine, inflationary pressures or supply chain challenges, could result in financial difficulties for our clients, which could cause clients to delay payments to us, request modifications to their payment arrangements that could increase our receivables balance or default on their payment obligations to us.

We may not be able to obtain sufficient financing on favorable terms.

The maintenance and growth of our business, including our ability to finance acquisitions, the payment of dividends and our ability to make share repurchases rely on our access to capital, which depends in large part on cash flow generated by our business and the availability of equity and debt financing. Certain of

our businesses also rely on financings by the Company to fund the underwriting of their client's debt and equity capital raising transactions. There can be no assurance that our operations will generate sufficient positive cash flow to finance all of our capital needs or that we will be able to obtain equity or debt financing on favorable terms, particularly in an environment of rising interest rates. In addition, our ability to obtain financing will depend in part upon prevailing conditions in credit and capital markets, which are beyond our control.

Our defined benefit pension plan obligations could cause the Company's financial position, earnings and cash flows to fluctuate.

Our defined benefit pension obligations and the assets set aside to fund those obligations are sensitive to certain changes in the financial markets. Any such changes may result in increased pension expense or additional cash payments to fund these plans.

The Company has significant defined benefit pension obligations to its current and former employees, totaling approximately \$11.8 billion, and related plan assets of approximately \$13.0 billion, at December 31, 2022 on a U.S. GAAP basis. The Company's policy for funding its defined benefit pension plans is to contribute amounts at least sufficient to meet the funding requirements set forth by law. In the United States, contributions to these plans are based on ERISA guidelines. Outside the United States, contributions are generally based on statutory requirements and local funding practices, which may differ from measurements under U.S. GAAP. In the U.K., for example, the assumptions used to determine pension contributions are the result of legally-prescribed negotiations between the Company and the plan trustees. Currently, the use of these assumptions results in a lower funded status than determined under U.S. GAAP and may result in contributions irrespective of the U.S. GAAP funded status.

The financial calculations relating to our defined benefit pension plans are complex. Pension plan assets could decrease as the result of poor future asset performance. In addition, the estimated return on plan assets would likely be impacted by changes in the interest rate environment and other factors, including equity valuations, since these factors reflect the starting point used in the Company's projection models. For example, a reduction in interest rates may result in a reduction in the estimated return on plan assets. Also, pension plan liabilities, periodic pension expense and future funding amounts could increase as a result of a decline in the interest rates we use to discount our pension liabilities, longer lifespans than those reflected in our mortality assumptions, changes in investment markets that result in lower expected returns on assets, actual investment return that is less than the expected return on assets, adverse changes in laws or regulations and other variables.

While we have taken steps to mitigate the impact of pension volatility on our earnings and cash funding requirements, these strategies may not be successful. Accordingly, given the magnitude of our worldwide pension plans, variations in or reassessment of the preceding or other factors or potential miscalculations relating to our defined benefit pension plans could cause significant fluctuation from year to year in our earnings and cash flow, as well as our pension plan assets and liabilities, and may result in increased levels of contributions to our pension plans.

Our significant non-U.S. operations expose us to exchange rate fluctuations and various risks that could impact our business.

Approximately 51% of our total revenue reported in 2022 was from business outside of the United States. We are subject to exchange rate movement because we must translate the financial results of our foreign subsidiaries into U.S. dollars and also because some of our subsidiaries receive revenue other than in their functional currencies. Exchange rate movements may change over time, and they could have a material adverse impact on our financial results and cash flows reported in U.S. dollars. For additional discussion, see "Market Risk and Credit Risk-Foreign Currency Risk" in Part II, Item 7A ("Quantitative and Qualitative Disclosures about Market Risk") of this report.

Our quarterly revenues and profitability may fluctuate significantly.

Quarterly variations in revenues and operating results may occur due to several factors. These include:

- · the number of client engagements during a quarter;
- the possibility that clients may decide to delay or terminate a current or anticipated project as a result of factors unrelated to our work product or progress;

- fluctuations in capacity and utilization rates and clients' ability to terminate engagements without penalty;
- our net colleague hires and related compensation and benefits expense;
- potential limitations on the clients or industries we serve resulting from increased regulation or changing stakeholder expectations on ESG issues:
- the impact of changes in accounting standards or in our accounting estimates or assumptions;
- the impact on us or our clients of changes in legislation, regulation and legal guidance or interpretations in the jurisdictions in which we operate, in particular as a result of increased regulatory activity and enforcement;
- seasonality due to the impact of regulatory deadlines, policy renewals and other timing factors to which our clients are subject;
- · the success of our acquisitions or investments;
- macroeconomic factors such as changes in foreign exchange rates, interest rates and global public and private capital markets, particularly in the case of Mercer, where fees in its investments business and certain other business lines are derived from the value of assets under management, advisement or administration; and
- general economic conditions, including factors beyond our control affecting economic conditions such as global health crises or pandemics, severe weather, climate change, geopolitical unrest such as the war in Ukraine, protests and riots or other catastrophic events, since our results of operations are directly affected by the levels of business activity of our clients, which in turn are affected by the level of economic activity in the industries and markets that they serve.

A significant portion of our total operating expenses is relatively fixed in the short term. Therefore, a variation in the number of client assignments or in the timing of the initiation or the completion of client assignments can cause significant variations in quarterly operating results for these businesses.

Credit rating downgrades would increase our financing costs and could subject us to operational risk.

Currently, the Company's senior debt is rated A- by S&P and Fitch and Baa1 by Moody's. The Company carries a Stable outlook with both S&P and Fitch and Positive outlook with Moody's.

If we need to raise capital in the future (for example, in order to maintain adequate liquidity, fund maturing debt obligations or finance acquisitions or other initiatives), credit rating downgrades would increase our financing costs, and could limit our access to financing sources. A downgrade to a rating below investment-grade could result in greater operational risks through increased operating costs and increased competitive pressures.

Our current debt level could adversely affect our financial flexibility.

As of December 31, 2022, we had total consolidated debt outstanding of approximately \$11.5 billion.

The level of debt outstanding could adversely affect our financial flexibility by reducing our cash flows and our ability to use cash from operations for other purposes, including working capital, dividends to shareholders, share repurchases, acquisitions, capital expenditures and general corporate purposes. In addition, we are subject to risks that, at the time any of our outstanding debt matures, we will not be able to retire or refinance the debt on terms that are acceptable to us.

The current U.S. tax regime makes our results more difficult to predict.

Our effective tax rate may fluctuate in the future as a result of the current U.S. tax regime and the continuing issuance of interpretive guidance related to the operations of U.S.-based multinational corporations. These include significant provisions in U.S. income tax law that may have a meaningful impact on our income tax expense and require significant judgments and estimates in interpretation and calculations. Current tax legislation includes, among other provisions, limitations on the deductibility of net interest expense, a minimum tax on most non-U.S. income called Global Intangible Low-Taxed Income ("GILTI"), and the Base Erosion and Anti-Abuse Tax ("BEAT"). In addition, a recently enacted book minimum tax could increase the impact of these provisions on our income tax expense. Given the

significant complexity of the rules, and the potential for additional guidance from the U.S. Treasury, the Securities and Exchange Commission, the Financial Accounting Standards Board or other regulatory authorities, recognized impacts in future periods could be significantly different from our current estimates. Such uncertainty may also result in increased scrutiny from, or disagreements with, tax authorities. As a U.S.-domiciled company, any such increases would likely have a disproportionate impact on us compared to our foreign-based competitors.

Global Operations

We are exposed to multiple risks associated with the global nature of our operations.

We conduct business globally. In 2022, approximately 51% of the Company's total revenue was generated from operations outside the United States, and over one-half of our employees were located outside the United States. In addition, we conduct our operations through four separate businesses. Potential conflicts of interest may arise across our businesses given the significant volume of our engagements.

The geographic breadth of our activities also subjects us to significant legal, economic, operational, market, compliance and reputational risks. These include, among others, risks relating to:

- · economic and political conditions in the countries in which we operate;
- client concentration in certain high-growth countries in which we operate;
- the length of payment cycles and potential difficulties in collecting accounts receivable;
- unexpected increases in taxes or changes in U.S. or foreign tax laws, rulings, policies or related legal and regulatory interpretations, including upcoming changes to the U.K. statutory rate;
- the implementation of the Organization for Economic Cooperation and Development (OECD) international tax framework, including the Pillar 2 minimum tax regime and the Pillar 1 profit reallocation regime, which could substantially increase the Company's effective tax rate:
- international initiatives to require multinational enterprises, like ours, to calculate and report profitability on a country-by-country basis, which could increase scrutiny by, or cause disagreements with, foreign tax authorities;
- potential transfer pricing-related tax exposures that may result from the flow of funds among our subsidiaries and affiliates in the various jurisdictions in which we operate;
- unexpected reassessment by tax authorities of interpretations of existing rules which may require companies to defend previously
 accepted positions and may create both new and prior-year exposures;
- permanent establishments created due to colleagues traveling to and doing work in countries where the company has no presence, or living in such countries and working remotely post-pandemic, which are not properly compensated through transfer pricing:
- our ability to obtain dividends or repatriate funds from our non-U.S. subsidiaries, including as a result of the imposition of currency controls and other government restrictions on repatriation in the jurisdictions in which our subsidiaries operate, fluctuations in foreign exchange rates and the imposition of withholding and other taxes on such payments:
- geopolitical tensions, such as the war in Ukraine, in countries where we operate, international hostilities, international trade disputes, terrorist activities, natural disasters, pandemics, and infrastructure disruptions;
- local investment or other financial restrictions that foreign governments may impose;
- potential lawsuits, investigations, market studies, reviews or other activity by foreign regulatory or law enforcement authorities or legislatively appointed commissions, which may result in potential modifications to our businesses, related private litigation or increased scrutiny from U.S. or other regulators;
- potential costs and difficulties in complying with a wide variety of foreign laws and regulations (including tax systems) administered by foreign government agencies, some of which may conflict with U.S. or other sources of law;

- potential costs and difficulties in complying, or monitoring compliance, with foreign and U.S. laws and regulations that are applicable
 to our operations abroad, including trade sanctions laws relating to countries such as Afghanistan, Belarus, Cuba, Iran, North Korea,
 Russia, Syria, Ukraine (Russia-controlled territories) and Venezuela, anti-corruption laws such as the U.S. Foreign Corrupt Practices
 Act and the U.K. Bribery Act 2010;
- limitations or restrictions that foreign or U.S. governments and regulators may impose on the products or services we sell, the
 methods by which we sell our products and services and the manner in which and the amounts we are compensated;
- potential limitations or difficulties in protecting our intellectual property in various foreign jurisdictions;
- limitations that foreign governments may impose on the conversion of currency or the payment of dividends or other remittances to us from our non-U.S. subsidiaries;
- engaging and relying on third parties to perform services on behalf of the Company; and
- · potential difficulties in monitoring employees in geographically dispersed locations.

RISKS RELATING TO OUR RISK AND INSURANCE SERVICES SEGMENT

Our Risk and Insurance Services segment, conducted through Marsh and Guy Carpenter, represented 61% of the Company's total revenue in 2022. Our business in this segment is subject to particular risks.

Results in our Risk and Insurance Services segment may be adversely affected by a general decline in economic activity.

Demand for many types of insurance and reinsurance generally rises or falls as economic growth expands or slows. This dynamic affects the level of commissions and fees generated by Marsh and Guy Carpenter. To the extent our clients become adversely affected by declining business conditions, they may choose to limit their purchases of risk services and insurance and reinsurance coverage, as applicable, which would adversely impact our commission revenue and other revenue based on premiums placed and services provided by us. Also, the insurance they seek to obtain through us may be impacted by changes in their assets, property values, sales or number of employees, which may reduce our commission revenue, and they may decide not to purchase our risk advisory or other services, which would inhibit our ability to generate fee revenue. Moreover, insolvencies and combinations associated with an economic downturn, especially insolvencies and combinations in the insurance industry, could adversely affect our brokerage business through the loss of clients or by limiting our ability to place insurance and reinsurance business, as well as our revenues from insurers. Guy Carpenter is especially susceptible to this risk given the limited number of insurance company clients and reinsurers in the marketplace.

Volatility or declines in premiums and other market trends may significantly impede our ability to grow revenues and profitability.

A significant portion of our Risk and Insurance Services revenue consists of commissions paid to us out of the premiums that insurers and reinsurers charge our clients for coverage. We do not determine the insurance premiums on which our commissions are generally based. Our revenues and profitability are subject to change to the extent that premium rates fluctuate or trend in a particular direction. The potential for changes in premium rates is significant, due to the normal cycles of pricing in the commercial insurance and reinsurance markets.

As traditional insurance companies continue to rely on non-affiliated brokers or agents to generate premium, those insurance companies may seek to reduce their expenses by lowering their commission rates. The reduction of these commission rates, along with general volatility or declines in premiums, may significantly affect our revenue and profitability. Because we do not determine the timing or extent of premium pricing changes, it is difficult to accurately forecast our commission revenues, including whether they will significantly decline. As a result, we may have to adjust our plans for future acquisitions, capital expenditures, dividend payments, loan repayments and other expenditures to account for unexpected changes in revenues, and any decreases in premium rates may adversely affect the results of our operations.

In addition to movements in premium rates, our (and Mercer's Health business's) ability to generate premium-based commission revenue may be challenged by disintermediation and the growing availability of alternative methods for clients to meet their risk-protection needs. This trend includes a greater willingness on the part of corporations to self-insure, the expanded use of captive insurers, and the presence of capital markets-based solutions for traditional insurance and reinsurance needs. Further, the profitability of our Risk and Insurance Services segment depends in part on our ability to be compensated for the analytical services and other advice that we provide, including the consulting and analytics services that we provide to insurers. If we are unable to achieve and maintain adequate billing rates for all of our services, our margins and profitability could decline.

Adverse legal developments and future regulations concerning how intermediaries are compensated by insurers or clients, as well as allegations of anti-competitive behavior or conflicts of interest more broadly, could have a material adverse effect on our business, results of operations and financial condition.

The ways in which insurance intermediaries are compensated receive scrutiny from regulators in part because of the potential for anti-competitive behavior and conflicts of interest. The vast majority of the compensation that Marsh receives is in the form of retail fees and commissions that are paid by the client or paid from premium that is paid by the client. The amount of other compensation that we receive from insurance companies, separate from retail fees and commissions, has increased in the last several years, both on an underlying basis and through acquisition and represented approximately 6% of Marsh's revenue in 2022. This other compensation includes payment for (i) consulting and analytics services provided to insurers; (ii) administrative and other services provided to insurers (including underwriting services and services relating to the administration and management of quota shares, panels and other facilities); and (iii) contingent commissions, primarily at MMA and outside the U.S., paid by insurers based on factors such as volume or profitability. These other revenue streams present potential regulatory, litigation and reputational risks that may arise from alleged anti-competitive behavior or conflicts of interest, (including those arising from Guy Carpenter's role as intermediary and advisor for insurance companies), and future changes in the regulatory environment may impact our ability to collect such revenue. Adverse regulatory, legal or other developments could have a material adverse effect on our business and expose the Company to negative publicity and reputational harm.

RISKS RELATING TO OUR CONSULTING SEGMENT

Our Consulting segment, conducted through Mercer and Oliver Wyman Group, represented 39% of our total revenue in 2022. Our businesses in this segment are subject to particular risks.

Mercer's Wealth business is subject to a number of risks, including risks related to public and private capital market fluctuations, third-party asset managers, operational and technology risks, conflicts of interest, ESG and greenwashing, asset performance and regulatory compliance, that, if realized, could result in significant damage to our business.

Mercer's Investments business provides clients with digital tools, investment consulting and investment management services. Mercer's Investments business is subject to a number of risks, including risks related to litigation (both by clients and by plan participants, particularly as we increasingly act in a fiduciary capacity), liquidity and market volatility, third-parties, our operations and technology, trading errors, conflicts of interest, asset performance and regulatory compliance and scrutiny, which could arise in connection with these offerings. For example, Mercer's manager research or due diligence on an asset manager may fail to uncover material deficiencies or fraud that could result in investment losses to a client. There is a risk that Mercer will fail to properly or timely implement a client's investment policy or direction, which could cause an incorrect or untimely allocation of client assets among asset classes, asset managers, or strategies. Mercer may also be perceived as making inaccurate or misleading statements regarding the investment strategies of our offerings with respect to ESG or sustainability, commonly referred to as "greenwashing," or recommending certain asset managers to clients or offering delegated solutions to an investment consulting client, solely to enhance its own compensation or due to other perceived conflicts of interest. Asset classes may perform poorly, or asset managers may underperform their benchmarks, due to poor market performance, a downturn in the global markets, negligence or other reasons, resulting in poor returns or loss of client assets. Changes in the value of equity, debt, currency real estate, commodities, alternatives or other asset classes, in particular as a

result of a downturn in the global markets, could cause the value of our assets under management or advisement, and the fees earned by Mercer to decline. Mercer or its clients may be subject to claims or class action litigation relating to advice given or investment decisions made by plan sponsors and plan fiduciaries, particularly relating to 401(k) plans in the U.S. or pension schemes in the U.K. These risks, if realized, could result in significant liability and damage our business.

Revenues for the services provided by our Consulting segment may decline for various reasons, including as a result of changes in economic conditions, the value of equity, debt and other asset classes, our clients' or an industry's financial condition or government regulation or an accelerated trend away from actively managed investments to passively managed investments.

Global economic conditions, including slower GDP growth or recession, inflationary pressure and foreign exchange rate volatility, may negatively impact businesses and financial institutions. Many of our clients, including financial institutions, corporations, government entities and pension plans, have reduced expenses, including amounts spent on consulting services, and used internal resources instead of consultants during difficult economic periods. The evolving needs and financial circumstances of our clients may reduce demand for our consulting services and could adversely affect our revenues and profitability. If the economy or markets in which we operate experience weakness or deteriorate, our business, financial condition and results of operations could be materially and adversely affected. If our clients reduce their headcounts, they will have fewer employee lives covered under their health plans, which may reduce premiums and the commission or supplemental compensation Mercer may receive.

In addition, some of Mercer's Investments business generate fees based upon the value of the clients' assets under management, advisement or administration. Changes in the value of equity, debt, currency, real estate, commodities, alternatives or other asset classes could cause the value of assets under management, advisement or administration, and the fees received by Mercer, to decline. Such changes could also cause clients to withdraw funds from Mercer's Investments business in favor of other investment service providers. In either case, our business, financial condition and results of operations could be materially adversely affected. Mercer's Investments business also could be adversely affected by an accelerated shift away from actively managed investments to passively managed investments with associated lower fees. Further, revenue received by Mercer as investment manager to the majority of the Mercer-managed investment funds is reported in accordance with U.S. GAAP on a gross basis rather than a net basis, with sub-advisor fees reflected as an expense. Therefore, the reported revenue for these offerings does not fully reflect the amount of net revenue ultimately attributable to Mercer.

Demand for many of Mercer's benefits services is affected by government regulation and tax laws, rulings, policies and interpretations, which drive our clients' needs for benefits-related services. Significant changes in government regulations affecting the value, use or delivery of benefits and human resources programs, including changes in regulations relating to health and welfare plans, defined contribution plans or defined benefit plans, may adversely affect the demand for or profitability of Mercer's services.

Factors affecting defined benefit pension plans and the services we provide relating to those plans could adversely affect Mercer.

Mercer currently provides plan sponsors, plan trustees, multi-employer and public entity clients with actuarial, consulting and administration services relating to defined benefit pension plans. The nature of our work is complex. Many clients, particularly in the public sector, have sizeable pension deficits and are subject to impact from volatility in the global markets and interest rate fluctuations. A number of Mercer's clients have frozen or curtailed their defined benefit plans and have moved to defined contribution plans resulting in reduced revenue for Mercer's retirement business. These developments, fee compression pressures, and a continued or accelerated rate of decline in revenues for our defined benefit pension plans business could adversely affect Mercer's business and operating results. In addition, our actuarial services involve numerous assumptions and estimates regarding future and contingent events, including interest rates used to discount future liabilities, estimated rates of return for a plan's assets, healthcare cost trends, salary projections and participants' life expectancies. Mercer's consulting services involve the drafting and interpretation of trust deeds and other complex documentation governing pension plans. Mercer's administration services include calculating benefits within complicated pension plan structures. Mercer's investments services include investment advice and management relating to defined benefit pension plan assets intended to fund present and future benefit obligations. Clients dissatisfied with our

services have brought, and may bring, significant claims against us, particularly in the United States and the United Kingdom.

Additionally, a rapid rise in interest rates could result in higher defined benefit pension plan funding levels. In some markets, this could accelerate clients' desire to conduct a buyout or third-party risk transfer. Such a transaction could result in additional short-term revenue for Mercer to the extent we advise the client on the transaction, but a loss in longer term recurring revenue related to the plan.

The profitability of our Consulting segment may decline if we are unable to achieve or maintain adequate utilization and pricing rates for our consultants.

The profitability of our Consulting businesses depends in part on ensuring that our consultants maintain adequate utilization rates (i.e., the percentage of our consultants' working hours devoted to billable activities). Our utilization rates are affected by a number of factors, including:

- general economic conditions:
- our ability to transition consultants promptly from completed projects to new assignments, and to engage newly-hired consultants quickly in revenue-generating activities;
- our ability to continually secure new business engagements, particularly because a portion of our work is project-based rather than recurring in nature;
- our ability to forecast demand for our services and thereby maintain appropriate headcount in each of our geographies and workforces;
- our ability to retain key colleagues and consulting professionals;
- · unanticipated changes in the scope of client engagements;
- the potential for conflicts of interest that might require us to decline client engagements that we otherwise would have accepted;
- our need to devote time and resources to sales, training, professional development and other non-billable activities; and
- the potential disruptive impact of acquisitions and dispositions.

If the utilization rate for our consulting professionals declines, our revenues, profit margin and profitability could decline.

In addition, the profitability of our Consulting businesses depends in part on the prices we are able to charge for our services. The prices we charge are affected by a number of factors, including:

- general economic conditions;
- clients' perception of our ability to add value through our services;
- market demand for the services we provide;
- our ability to develop new services and the introduction of new services by competitors;
- · the pricing policies of our competitors; and
- the extent to which our clients develop in-house or other capabilities to perform the services that they might otherwise purchase from

If we are unable to achieve and maintain adequate billing rates for our services, our profit margin and profitability could decline.

Item 1B. Unresolved Staff Comments.

There are no unresolved comments to be reported pursuant to Item 1B.

Item 2. Properties.

The Company maintains its corporate headquarters in New York City. We also maintain other offices around the world, primarily in leased space. In certain circumstances we may have space that we sublet to third parties, depending upon our needs in particular locations.

The Company and certain of its subsidiaries own, directly and indirectly through special purpose subsidiaries, a 58% condominium interest covering approximately 900,000 square feet of office space in a 44 story condominium in New York City. This real estate serves as the Company's headquarters and is occupied primarily by the Company and its subsidiaries for general corporate use. The condominium interests are financed by a 30-year mortgage loan that is non-recourse to the Company unless the Company (i) is downgraded below B (stable outlook) by S&P or Fitch or B2 (stable outlook) by Moody's and such downgrade is continuing or (ii) an event of default under the mortgage loan has occurred. The mortgage is secured by a first priority assignment of leases and rents, including the leases which the Company and certain of its subsidiaries entered into with their affiliated special purpose subsidiaries which own the mortgaged condominium interests. The net rent due under those leases in effect services the mortgage debt.

Item 3. Legal Proceedings.

We and our subsidiaries are party to a variety of other legal, administrative, regulatory and government proceedings, claims and inquiries arising in the normal course of business.

Additional information regarding certain legal proceedings and related matters is set forth in Note 16, Claims, Lawsuits and Other Contingencies, in the notes to the consolidated financial statements appearing under Part II, Item 8 ("Financial Statements and Supplementary Data") of this report.

Item 4. Mine Safety Disclosures.

Not applicable.

Item 5. Market for the Company's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities.

The Company's common stock is listed on the New York, Chicago and London Stock Exchanges. The following table indicates the high and low prices (NYSE composite quotations) of the Company's common stock in 2022 and 2021 and each quarterly period thereof:

	20 Stock Pri	22 ce Range	2021 Stock Price Range		
	High	Low	High	Low	
First Quarter	\$173.34	\$142.80	\$122.09	\$106.95	
Second Quarter	\$183.14	\$143.33	\$141.41	\$121.31	
Third Quarter	\$174.23	\$146.82	\$162.26	\$137.85	
Fourth Quarter	\$176.75	\$148.14	\$175.12	\$151.37	
Full Year	\$183.14	\$142.80	\$175.12	\$106.95	

On March 23, 2022, the Board of Directors of the Company authorized an additional \$5 billion in share repurchases. This is in addition to the Company's existing share repurchase program, which had approximately \$1.3 billion of remaining authorization as of December 31, 2021. The Company repurchased approximately 12.2 million shares of its common stock for \$1.9 billion in 2022. As of December 31, 2022, the Company remained authorized to repurchase up to approximately \$4.3 billion in shares of its common stock. There is no time limit on the authorization.

The Company repurchased approximately 7.9 million shares of its common stock for \$1.2 billion in 2021.

The following information relates to the Company's repurchases of equity securities during any month within the fourth quarter of the fiscal year covered by this report:

Period	Total Number of Shares (or Units) Purchased	Average Price Paid per Share (or Unit)	Total Number of Shares (or Units) Purchased as Part of Publicly Announced Plans or Programs	(Maximum Number (or Approximate Dollar Value) of Shares (or Units) that May Yet Be Purchased Under the Plans or Programs
Oct 1-31, 2022	977,063	\$ 155.2316	977,063	\$	4,512,323,223
Nov 1-30, 2022	382,066	\$ 164.9085	382,066	\$	4,449,317,300
Dec 1-31, 2022	800,566	\$ 169.0335	800,566	\$	4,313,994,859
Total	2,159,695	\$ 162.0597	2,159,695	\$	4,313,994,859

As February 9, 2023, there were 4,210 stockholders of record.

Item 6. Selected Financial Data.

On November 19, 2020, the SEC adopted amendments to Regulation S-K (the "Amendments"), which modernize, simplify and enhance certain financial disclosure requirements. The Amendments are effective for fiscal years ending on or after August 9, 2021. The Company adopted the Amendments to Regulation S-K for the year-ended December 31, 2021 and elected to exclude Item 6. Selected Financial Data and the Selected Quarterly Data and Supplemental Information from this annual report on Form 10-K.

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations.

General

Marsh & McLennan Companies Inc., and its consolidated subsidiaries (the "Company") is a global professional services firm offering clients advice in the areas of risk, strategy and people. The Company's more than 85,000 colleagues advise clients in over 130 countries. With annual revenue of over \$20 billion, the Company helps clients navigate an increasingly dynamic and complex environment through four market-leading businesses. Marsh provides data-driven risk advisory services and insurance solutions to commercial and consumer clients. Guy Carpenter develops advanced risk, reinsurance and capital strategies that help clients grow profitably and identify and capitalize on emerging opportunities. Mercer delivers advice and technology-driven solutions that help organizations redefine the future of work, shape retirement and investment outcomes, and advance health and well-being for a changing workforce. Oliver Wyman Group serves as a critical strategic, economic and brand advisor to private sector and governmental clients.

The Company conducts business through two segments:

- Risk and Insurance Services includes risk management activities (risk advice, risk transfer and risk control and mitigation solutions) as well as insurance and reinsurance broking and services. The Company conducts business in this segment through Marsh and Guy Carpenter.
- **Consulting** includes health, wealth and career solutions and products, and specialized management, strategic, economic and brand consulting services. The Company conducts business in this segment through Mercer and Oliver Wyman Group.

The consolidated results of operations in the Management Discussion & Analysis ("MD&A") includes an overview of the Company's consolidated 2022 results compared to the 2021 results, and should be read in conjunction with the consolidated financial statements and notes. This section also includes a discussion of the key drivers impacting the Company's financial results of operations both on a consolidated basis and by reportable segments.

We describe the primary sources of revenue and categories of expense for each segment in the discussion of segment financial results. A reconciliation of segment operating income to total operating income is included in Note 17, Segment Information, in the notes to the consolidated financial statements included in Part II, Item 8, of this report.

For information and comparability of the Company's results of operations and liquidity and capital resources for fiscal year 2020, refer to "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" of our Form 10-K for the fiscal year ended December 31, 2021.

This MD&A contains forward-looking statements as defined in the Private Securities Litigation Reform Act of 1995. Refer to "Information Concerning Forward-Looking Statements" at the outset of this report.

Financial Highlights

- Consolidated revenue in 2022 was \$20.7 billion, an increase of 5% compared with 2021, or 9% on an underlying basis.
- Consolidated operating income decreased \$32 million, or 1% to \$4.3 billion in 2022, compared with 2021. Net income attributable to the Company was \$3.0 billion. Earnings per share decreased from \$6.13 to \$6.04, or 1% from the prior year.
- Risk and Insurance Services revenue in 2022 was \$12.6 billion, an increase of 5%, or 9% on an underlying basis. Operating income was \$3.1 billion in both 2022 and 2021.
- Consulting revenue in 2022 was \$8.1 billion, an increase of 5%, or 8% on an underlying basis. Operating income was \$1.6 billion and \$1.5 billion in 2022 and 2021, respectively.
- The Company's results of operations in 2022 were impacted by restructuring activities of \$427 million, primarily related to severance and lease exit charges for activities focused on workforce actions, technology rationalization and reductions in real estate.
- The Company completed 20 acquisitions in 2022, the largest being the acquisition of HMS Insurance Inc., a full service broker in the Risk and Insurance services segment, and the

Avascent Group Ltd, an aerospace and defense management consulting firm in the Consulting segment.

- In 2022, Mercer sold its U.S. affinity business that provided insurance marketing, brokerage and administration to association and affinity groups for cash proceeds of approximately \$140 million and a net gain of \$112 million.
- The Company issued senior notes of \$500 million due 2032 and \$500 million due 2052, and repaid senior notes of \$350 million due in March 2023.
- In 2022, the Company repurchased 12.2 million shares for \$1.9 billion.

For additional details, refer to the consolidated results of operations and liquidity and capital resources sections in this MD&A.

Acquisitions and dispositions impacting the Risk and Insurance Services and Consulting segments are discussed in Note 5, Acquisitions and Dispositions, in the notes to the consolidated financial statements.

Deconsolidation of Russia

On February 24, 2022, Russian forces launched a military invasion of Ukraine. In response, the United States (U.S.), the European Union (E.U.), United Kingdom (U.K.) and other governments have imposed significant economic sanctions on Russia, and Russia has responded with counter-sanctions.

The Company concluded that it did not meet the accounting criteria for control over its wholly-owned Russian businesses due to the evolving trade and economic sanctions, and recorded a loss of \$52 million on the deconsolidation of the Russian businesses and other related charges. Subsequently, the Company entered into a definitive agreement to exit its businesses in Russia and transfer ownership to local management, pending regulatory approvals. Refer to Note 5, Acquisitions and Dispositions, in the notes to the consolidated financial statements for additional information on the deconsolidation of the Russian businesses.

The war in Ukraine has continued to result in worldwide geopolitical and macroeconomic uncertainty. The Company continues to monitor the ongoing situation and its potential impact on our business, financial condition, results of operations and cash flows.

Business Update related to COVID-19

For nearly three years, the COVID-19 pandemic has impacted businesses globally including in every geography in which the Company operates. Our businesses have remained resilient throughout the pandemic and demand for our advice and services remains strong. The ultimate extent of the impact of COVID-19 to the Company will depend on future developments that it is unable to predict.

Factors that could adversely affect the Company's financial statements related to the financial and operational impact of COVID-19 are included in "Item 1A - Risk Factors" in Part I of this report.

Consolidated Results of Operations

For the Years Ended December 31,				
(In millions, except per share data)	2022	2021		2020
Revenue	\$ 20,720	\$ 19,820	\$	17,224
Expense				_
Compensation and benefits	12,071	11,425		10,129
Other operating expenses	4,369	4,083		4,029
Operating expenses	16,440	15,508		14,158
Operating income	\$ 4,280	\$ 4,312	\$	3,066
Income before income taxes	\$ 4,082	\$ 4,208	\$	2,793
Net income before non-controlling interests	\$ 3,087	\$ 3,174	\$	2,046
Net income attributable to the Company	\$ 3,050	\$ 3,143	\$	2,016
Net income per share attributable to the Company				
– Basic	\$ 6.11	\$ 6.20	\$	3.98
– Diluted	\$ 6.04	\$ 6.13	\$	3.94
Average number of shares outstanding				
– Basic	499	507		506
– Diluted	505	513	·	512
Shares outstanding at December 31,	495	504		508

Consolidated operating income decreased \$32 million, or 1% to \$4.3 billion in 2022, compared to the prior year, reflecting a 5% increase in revenue and a 6% increase in expenses. Revenue growth was driven by increases of 5% in both Risk and Insurance Services and Consulting, reflecting the continued strong demand for our advice and services and the expansion of the global economy. The increase in expenses is primarily due to increased headcount and higher incentive compensation, as well as severance and lease exit charges. Expenses also reflect higher travel and entertainment costs, partially offset by lower depreciation and amortization primarily in the Risk and Insurance Services segment in 2022 compared to the prior year. In 2022, net operating income was also impacted by foreign exchange movements across both segments due to the strengthening of the U.S. dollar.

Diluted earnings per share decreased from \$6.13 to \$6.04, or 1% from the prior year. The decrease is primarily the result of lower operating income, other net benefits credits and investment income, and higher interest expense in 2022, compared to the prior year. The decrease in net income attributable to the Company was offset by lower income taxes in 2022. Income taxes for 2021, included a net charge of \$110 million for the re-measurement of deferred tax assets and liabilities due to the enactment of a tax rate increase from 19% to 25% in the U.K, partially offset by no tax impact on the gain related to the consolidation of Marsh India.

In 2022 and 2021, the Company's results of operations and earnings per share were impacted by the following items:

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Restructuring, excluding JLT	\$ 312 \$	70 \$	89
Changes in contingent consideration	49	57	26
JLT integration and restructuring costs	115	93	251
JLT acquisition-related costs and other	28	81	54
JLT legacy legal charges	1	(69)	161
Legal claims	30	62	_
Disposal of businesses	(122)	(49)	(8)
Pre-acquisition related costs	21	_	_
Deconsolidation of Russian businesses and other related charges	52	_	_
Gain on consolidation of business	_	(267)	_
Other	_	_	5
Impact on income before taxes	\$ 486 \$	(22) \$	578

- Restructuring, excluding JLT: Primarily includes severance and lease exit charges for activities focused on workforce actions, rationalization of technology and functional resources, and reductions in real estate. Costs also reflect charges for Marsh's operational excellence program. These costs are discussed in more detail in Note 14, Restructuring Costs, in the notes to the consolidated financial statements.
- Changes in contingent consideration: Includes the change in fair value of contingent consideration related to acquisitions and dispositions measured each quarter.
- **JLT integration and restructuring costs:** Primarily reflects lease exit charges for a legacy JLT U.K. location in 2022. In 2021, costs incurred include severance, lease exit charges, technology costs, and consulting services related to the integration of JLT. The Company completed the integration of JLT in 2022.
- JLT acquisition-related costs and other: Retention costs and legal charges related to the acquisition of JLT.
- **JLT legacy legal charges:** Charges and recoveries related to legacy JLT legal matters. In 2021, the Company recorded a \$36 million reduction in the liability for a legacy JLT errors and omissions ("E&O") related to the suitability of advice provided to individuals for defined benefit pension transfers in the U.K., as well as \$33 million of recoveries under indemnities and insurance. The reduction in liability primarily reflects lower redress payments than previously estimated, partially offset by higher costs to review and calculate redress. Refer to Note 16, Claims, Lawsuits and Other Contingencies, in the notes to the consolidated financial statements.
- Legal claims: The Company recorded settlement and legal costs related to strategic recruiting.
- **Disposal of businesses:** Primarily reflects a net gain of \$112 million on sale of the Mercer U.S. affinity business in 2022, that provided insurance marketing, brokerage and administration to association and affinity groups. In 2021, the amount primarily reflects a gain on the sale of the U.K. commercial networks business that provided broking and back-office solutions for small independent brokers. These amounts are reflected as a component of revenue in the consolidated statements of income and excluded from the calculations of underlying revenue.
- **Pre-acquisition related costs:** Includes integration costs for the pending Westpac Banking Corporation superannuation fund transaction in Australia, which is expected to close in the first half of 2023. Refer to Note 5, Acquisitions and Dispositions, in the notes to the consolidated financial statements.

- **Deconsolidation of Russian businesses and other related charges:** Loss on deconsolidation of Russian businesses of \$39 million is included in revenue in the consolidated statements of income and excluded from the calculations of underlying revenue.
- Gain on consolidation of business: In 2021, the Company increased its ownership in Marsh India from 49% to 92%. Prior to the increase in ownership, the Company accounted for the investment under the equity method of accounting. In connection with the increased investment in Marsh India, the Company recorded a gain of \$267 million, related to the re-measurement of its previously held investment to fair value.

Consolidated Revenue and Expense

Revenue - Components of Change

The Company conducts business in 130 countries. As a result, foreign exchange rate movements may impact period-to-period comparisons of revenue. Similarly, certain other items such as the revenue impact of acquisitions and dispositions, including transfers among businesses, may impact period-to-period comparisons of revenue. Underlying revenue measures the change in revenue from one period to another by isolating these impacts.

The impact of foreign currency exchange fluctuations, acquisitions and dispositions, including transfers among businesses, on the Company's operating revenues by segment is as follows:

	r En embe	ded er 31,		% Change _	Comp	Components of Revenue Change**				
(In millions, except percentages)	202	2022 2021		% Change GAAP Revenue	Revenue excl. Marsh India Gain*	Currency Impact	Acquisitions/ Dispositions/ Other Impact	Underlying Revenue		
Risk and Insurance Services										
Marsh	\$ 10,50	5 \$	10,203	3 %	6 %	(3)%	1 %	8 %		
Guy Carpenter	2,02)	1,867	8 %		(2)%	1 %	9 %		
Subtotal	12,52	5	12,070	4 %	6 %	(3)%	1 %	8 %		
Fiduciary Interest Income	12)	15							
Total Risk and Insurance Services	12,64	5	12,085	5 %	7 %	(3)%	1 %	9 %		
Consulting										
Mercer	5,34	5	5,254	2 %		(5)%	1 %	6 %		
Oliver Wyman Group	2,79	1	2,535	10 %		(4)%	1 %	13 %		
Total Consulting	8,13)	7,789	5 %		(5)%	1 %	8 %		
Corporate Eliminations	(6	!)	(54)							
Total Revenue	\$ 20,72) \$	19,820	5 %	6 %	(4)%	1 %	9 %		

^{*} Percentage change excludes the gain from the consolidation of Marsh India of \$267 million from prior year's GAAP revenue.

The following table provides more detailed revenue information for certain of the components presented above:

	Year Ende December					% Change	Components of Revenue Change**				
(In millions, except percentages)	lions, except percentages) 20		2022 2021		% Change GAAP Revenue	Revenue excl. Marsh India Gain*	Currency Impact	Acquisitions/ Dispositions/ Other Impact	Underlying Revenue		
Marsh:											
EMEA	\$	2,879	\$	2,946	(2)%		(7)%	(3)%	8 %		
Asia Pacific		1,333		1,462	(9)%	12 %	(8)%	6 %	13 %		
Latin America		502		453	11 %		(1)%	_	11 %		
Total International		4,714		4,861	(3)%	3 %	(7)%	_	10 %		
U.S./Canada		5,791		5,342	8 %		_	1 %	7 %		
Total Marsh	\$	10,505	\$	10,203	3 %	6 %	(3)%	1 %	8 %		
Mercer:											
Wealth	\$	2,366	\$	2,509	(6)%		(6)%	_	_		
Health		2,017		1,855	9 %		(3)%	3 %	9 %		
Career		962		890	8 %		(6)%	_	14 %		
Total Mercer	\$	5,345	\$	5,254	2 %		(5)%	1 %	6 %		

^{*} Percentage change excludes the gain from the consolidation of Marsh India of \$267 million from prior year's GAAP revenue.

^{**} Components of revenue change may not add due to rounding.

^{**} Components of revenue change may not add due to rounding.

Consolidated Revenue

Consolidated revenue increased \$900 million, or 5%, to \$20.7 billion in 2022, compared to \$19.8 billion in 2021. Consolidated revenue increased 9% on an underlying basis and 1% from acquisitions, partially offset by a decrease of 4% from the impact of foreign currency translation. On an underlying basis, revenue increased 9% and 8% in 2022, in the Risk and Insurance Services and Consulting segments, respectively.

Underlying revenue growth in the Risk and Insurance Services and Consulting segments was driven by the continued strong demand for our advice and services, the expansion of the global economy, new business growth, and solid retention including continued benefits from pricing in the market place.

Consolidated Operating Expenses

Consolidated operating expenses increased \$932 million, or 6%, to \$16.4 billion in 2022, compared to \$15.5 billion in 2021, reflecting increases of 10% on an underlying basis and 1% from acquisitions, partially offset by a decrease of 4% from the impact of foreign currency translation. On an underlying basis, expenses increased 9% in 2022, in both the Risk and Insurance Services and Consulting segments.

The increase in underlying expenses is primarily due to increased headcount and higher incentive compensation, as well as severance and lease exit charges. Expenses also reflect higher travel and entertainment costs, partially offset by lower depreciation and amortization primarily in the Risk and Insurance Services segment in 2022 compared to the prior year.

In 2022, the Company incurred \$427 million of restructuring costs, primarily severance and lease exit charges, of which \$219 million related to the Company's activities focused on workforce actions, rationalization of technology and functional services, and reductions in real estate. These activities were initiated in the fourth quarter of 2022 and the Company expects related estimated savings in 2023 to be in the range of \$125 million to \$150 million. Based on current estimates, the Company anticipates these activities will continue throughout 2023 and into 2024. However, additional charges are unlikely to exceed costs incurred in 2022. The Company's plans are still being finalized, which may change the expected timing, estimates of expected costs and related savings, as the Company continues to refine its detailed plans for each business and location.

Expenses also reflect lease exit charges of \$89 million in the Risk and Insurance Services segment for a legacy JLT U.K. location. For additional details, refer to Note 14, Restructuring Costs, in the notes to the consolidated financial statements.

Risk and Insurance Services

In the Risk and Insurance Services segment, the Company's subsidiaries and other affiliated entities act as brokers, agents or consultants for insureds, insurance underwriters and other brokers in the areas of risk management, insurance broking, insurance program management, risk consulting, analytical modeling and alternative risk financing services, primarily under the name of Marsh, and engage in specialized reinsurance broking, strategic advisory services and analytics solutions, primarily under the name of Guy Carpenter.

Marsh and Guy Carpenter are compensated for brokerage and consulting services through commissions and fees. Commission rates and fees vary in amount and can depend on a number of factors, including the type of insurance or reinsurance coverage provided, the particular insurer or reinsurer selected, and the capacity in which the broker acts and negotiates with clients. Revenues can be affected by premium rate levels in the insurance/reinsurance markets, the amount of risk retained by insurance and reinsurance clients, and by the value of the risks that have been insured since commission-based compensation is frequently related to the premiums paid by insureds and reinsureds. In many cases, fee compensation may be negotiated in advance, based on the type of risk, coverage required and service provided by the Company and ultimately, the extent of the risk placed into the insurance market or retained by the client. The trends and comparisons of revenue from one period to the next can be affected by changes in premium rate levels, fluctuations in client risk retention and increases or decreases in the value of risks that have been insured, as well as new and lost business, and the volume of business from new and existing clients.

In addition to compensation from its clients, Marsh also receives other compensation, separate from retail fees and commissions, from insurance companies. This other compensation includes, among other things, payment for consulting and analytics services provided to insurers; compensation for administrative and other services (including fees for underwriting services and services provided to or on behalf of insurers relating to the administration and management of quota shares, panels and other facilities in which insurers participate); and contingent commissions, which are paid by insurers based on factors such as volume or profitability of Marsh's placements, primarily driven by Marsh McLennan Agency ("MMA") and parts of Marsh's international operations. Marsh and Guy Carpenter receive interest income on certain funds (such as premiums and claims proceeds) held in a fiduciary capacity for others. The investment of fiduciary funds is regulated by state and other insurance authorities. These regulations typically require segregation of fiduciary funds and limit the types of investments that may be made. Interest income from these investments varies depending on the amount of funds invested and applicable interest rates, both of which vary from time to time. For presentation purposes, fiduciary interest income is segregated from the other revenues of Marsh and Guy Carpenter and separately presented within the segment, as shown in the previous revenue by segments tables.

The results of operations for the Risk and Insurance Services segment are as follows:

(In millions, except percentages)	2022	2021			2020
Revenue	\$ 12,645	\$	12,085	\$	10,337
Compensation and benefits	6,938		6,506		5,690
Other operating expenses	2,618		2,499		2,301
Operating expenses	9,556		9,005		7,991
Operating income	\$ 3,089	\$	3,080	\$	2,346
Operating income margin	24.4%		25.5%		22.7%

Revenue

Revenue in the Risk and Insurance Services segment increased \$560 million, or 5%, to \$12.6 billion in 2022, compared with \$12.1 billion in 2021. Revenue increased by 7% excluding the Marsh India gain in 2021. This reflects increases of 9% on an underlying basis and 1% from acquisitions, offset by a decrease of 3% from the impact of foreign currency translation. Interest earned on fiduciary funds increased by \$105 million to \$120 million, compared to \$15 million in the prior year.

The increase in underlying revenue in the Risk and Insurance Services segment was primarily due to growth in new business from existing clients, investments in talent, and solid retention including continued benefits from pricing in the marketplace. The increase in interest earned on fiduciary funds in 2022 is a result of higher interest rates compared to the prior year.

In Marsh, revenue increased \$302 million, or 3%, to \$10.5 billion in 2022, compared to \$10.2 billion in 2021. Revenue increased by 6% excluding the Marsh India gain in 2021. This reflects increases of 8% on an underlying basis and 1% from acquisitions, offset by a decrease of 3% from the impact of foreign currency translation. On an underlying basis, U.S./Canada rose 7%. Total International operations produced underlying revenue growth of 10%, reflecting growth of 13% in Asia Pacific, 11% in Latin America and 8% in EMEA.

Results in 2022 also include a charge of \$27 million related to the loss on deconsolidation of the Company's Russian businesses. In 2021, revenue from acquisitions reflected a gain of \$267 million related to the re-measurement of the previously held equity method investment in Marsh India and a net gain of approximately \$50 million related to the disposition of the commercial networks business in the U.K.

At Guy Carpenter, revenue increased \$153 million, or 8%, to \$2.0 billion in 2022, compared with \$1.9 billion in 2021. This reflects increases of 9% on an underlying basis and 1% from acquisitions, offset by a decrease of 2% related to the impact of foreign currency translation.

The Risk and Insurance Services segment completed 16 acquisitions in 2022. Information regarding these acquisitions is included in Note 5, Acquisitions and Dispositions, in the notes to the consolidated financial statements.

Operating Expenses

Expenses in the Risk and Insurance Services segment increased \$551 million, or 6%, to \$9.6 billion in 2022, compared with \$9.0 billion in 2021. This reflects increases of 9% on an underlying basis and 1% from acquisitions, offset by a decrease of 4% from the impact of foreign currency translation.

The increase in underlying expenses in 2022 is primarily due to increased headcount and higher incentive compensation, as well as severance and lease exit charges. In 2022, the Company incurred \$254 million for restructuring costs in Risk and Insurance Services, of which \$104 million related to the Company's activities focused on workforce actions, rationalization of technology and functional services, and reductions in real estate.

Expenses also reflect higher lease related exit costs for a legacy JLT U.K. location of \$89 million, and higher travel and entertainment costs, partially offset by lower depreciation and amortization in 2022, compared to the prior year.

Consulting

The Company conducts business in its Consulting segment through Mercer and Oliver Wyman Group. Mercer delivers advice and technology-driven solutions that help organizations redefine the future of work, shape retirement and investment outcomes, and advance health and well-being for a changing workforce. Oliver Wyman serves as a critical strategic, economic and brand advisor to private sector and governmental clients.

The major component of revenue in the Consulting business is fees paid by clients for advice and services. Mercer, principally through its health line of business, also earns revenue in the form of commissions received from insurance companies for the placement of group (and occasionally individual) insurance contracts, primarily life, health and accident coverages. Revenue for Mercer's investment management business and certain of Mercer's defined benefit and contribution administration services consists principally of fees based on assets under management or administration. For a majority of the Mercer managed investment funds, revenue is recorded on a gross basis with subadvisor fees included in other operating expenses.

Revenue in the Consulting segment is affected by, among other things, global economic conditions, including changes in clients' particular industries and markets. Revenue is also affected by competition due to the introduction of new products and services, broad trends in employee demographics, including levels of employment and the effect of government policies and regulations. Revenues from investment management services and retirement trust and administrative services are significantly affected by the level of assets under management or administration, which is impacted by securities market performance.

The results of operations for the Consulting segment are as follows:

(In millions, except percentages)	2022	2021	2020		
Revenue	\$ 8,139	\$ 7,789	\$	6,976	
Compensation and benefits	4,626	4,435		3,995	
Other operating expenses	1,960	1,850		1,987	
Operating expenses	6,586	6,285		5,982	
Operating income	\$ 1,553	\$ 1,504	\$	994	
Operating income margin	19.1%	19.3%		14.3%	

Revenue

Consulting revenue increased \$350 million, or 5%, to \$8.1 billion in 2022, compared with \$7.8 billion in 2021. This reflects increases of 8% on an underlying basis and 1% from acquisitions, partially offset by a decrease of 5% from the impact of foreign currency translation.

Mercer's revenue increased \$91 million, or 2%, to \$5.3 billion in 2022, compared to the prior year. This reflects increases of 6% on an underlying basis and 1% from dispositions, offset by a decrease of 5% from the impact of foreign currency translation. On an underlying basis, revenue for Career and Health increased 14%, and 9%, respectively, while underlying revenue for Wealth remained flat.

The increase in underlying revenue at Mercer in 2022 was primarily due to continued strong demand for our advice and services and the expansion of the global economy. The increase in Career products and services was due to continued demand for solutions linked to new ways of working, skill gaps, workforce transformation and diversity & inclusion issues. Health continued to benefit from growth in new business, higher retention, increased enrolled lives from a strong labor market, and medical inflation. Underlying revenue in Wealth grew in retirement solutions in 2022, offset by a decrease in investment management fees from the decline in assets under management due to market volatility on investments.

Results in 2022 also include a net gain of \$112 million from the sale of the Mercer U.S. affinity business.

Oliver Wyman Group's revenue increased \$259 million, or 10%, to \$2.8 billion in 2022, compared with \$2.5 billion in 2021, reflecting increases of 13% on an underlying basis and 1% from acquisitions, offset by a decrease of 4% from the impact of foreign currency translation.

The increase in underlying revenue at Oliver Wyman in 2022 was primarily due to increased demand for project-based services across all industries. Results in 2022, also include a charge of \$12 million related to the loss on deconsolidation of the Company's Russian businesses.

The Consulting segment completed 4 acquisitions in 2022. Information regarding these acquisitions is included in Note 5, Acquisitions and Dispositions, in the notes to the consolidated financial statements.

Operating Expenses

Consulting expenses increased \$301 million, or 5%, to \$6.6 billion in 2022, compared to \$6.3 billion in 2021. This reflects an increase of 9% on an underlying basis, partially offset by a decrease of 5% from the impact of foreign currency translation.

The increase in underlying expenses in the Consulting segment in 2022 is primarily due to increased headcount and higher incentive compensation, as well as severance and lease exit charges. In 2022, the Company incurred \$77 million for restructuring costs in Consulting, of which \$53 million related to the Company's activities focused on workforce actions, rationalization of technology and functional services, and reductions in real estate.

Expenses also reflect higher travel and entertainment costs compared to the prior year. Results in 2021 also included a \$69 million reduction in the liability for a legacy JLT E&O, including recoveries under indemnities and insurance, recorded in other expenses.

Corporate and Other

Corporate expenses increased \$90 million, or 33% to \$362 million in 2022, compared with \$272 million in 2021. This reflects an increase of 34% on an underlying basis, offset by a decrease of 1% from the impact of foreign currency translation. The increase in underlying expenses is primarily related to \$62 million in lease exit charges for reductions in real estate.

Interest

Interest expense was \$469 million in 2022, compared to \$444 million in 2021. Interest expense increased \$25 million due to higher short term borrowings in 2022 at higher interest rates compared to the prior year.

Investment Income (Loss)

The caption "Investment income (loss)" in the consolidated statements of income comprises realized and unrealized gains and losses from investments. It includes, when applicable, other than temporary declines in the value of securities, mark-to-market increases or decreases in equity investments with readily determinable fair values and equity method gains or losses on its investments in private equity funds. The Company's investments may include direct investments in insurance, consulting or other strategically linked companies and investments in private equity funds.

The Company recorded net investment income of \$21 million in 2022, compared to \$61 million in 2021. Net investment income in 2022 is driven primarily by lower mark-to-market gains from the Company's private equity investments compared to the prior year.

Income and Other Taxes

The Company's consolidated effective tax rate for 2022 and 2021 was 24.4% and 24.6%, respectively.

The tax rates in both years reflect the impact of discrete tax matters such as excess tax benefits related to share-based compensation, enacted tax legislation, changes in uncertain tax positions, deferred tax adjustments, non-taxable adjustments related to contingent consideration for acquisitions, and valuation allowances for certain tax credits. The rates in both years also reflect tax benefits from planning that postponed the utilization of current year losses in the U.K. to a future year when the tax rate will be 25%.

The excess tax benefit related to share-based payments is the most significant discrete item in 2022, reducing the effective tax rate by 0.7%. The reduction in 2021 was also 0.7%.

The effective tax rate for 2021 reflects a net charge of \$100 million related to the re-measuring of the Company's U.K. deferred tax assets and liabilities upon enactment of legislation on June 10, 2021, increasing the U.K. corporate income tax rate from 19% to 25%, effective April 1, 2023. The re-measurement of the Company's U.K. deferred tax assets and liabilities was the most significant discrete item in the prior year, increasing the Company's effective tax rate by 2.6% in 2021.

The effective tax rate in 2021 also decreased 1.5% due to the Company not recording a tax on the gain from the fair value re-measurement of the Company's previously-held equity method investment in Marsh India when the Company increased its ownership interest from 49% to 92%. The Company does not intend to dispose the business and has indefinitely reinvested this gain.

The effective tax rate may vary significantly from period to period. The effective tax rate is sensitive to the geographic mix of earnings and the cost to repatriate the Company's earnings, which may result in higher or lower effective tax rates. Therefore, a shift in the mix of profits among jurisdictions, or changes in the Company's repatriation strategy to access offshore cash, can affect the effective tax rate. In 2022, pretax income in the U.K., Barbados, Canada, Ireland, Bermuda, Australia, Japan, India, and Germany accounted for approximately 65% of the Company's total non-U.S. pre-tax income, with effective rates in those countries of 17.8%, 1.2%, 26.8%, 11.9%, 3.7%, 29.3%, 32.8%, 24.9%, and 32.6%, respectively.

In addition, losses in certain jurisdictions cannot be offset by earnings from other operations and may require valuation allowances that affect the rate in a particular period, depending on estimates of the value of associated deferred tax assets which can be realized. A valuation allowance was recorded to reduce deferred tax assets to the amount that the Company believes is more likely than not to be realized. Details are provided in Note 7, Income Taxes, in the notes to the consolidated financial statements. The effective tax rate is also sensitive to changes in unrecognized tax benefits, including the impact of settled tax audits and expired statutes of limitations.

Changes in tax laws, rulings, policies or related legal and regulatory interpretations occur frequently and may also have significant favorable or adverse impacts on our effective tax rate.

On August 16, 2022, the Inflation Reduction Act of 2022 ("IRA") was enacted into law. The Company evaluated the provisions of the new legislation, the most significant of which are the corporate alternative minimum tax and the share repurchase tax. The IRA is effective as of January 1, 2023, and will not have a significant impact on the Company's financial results of operations.

As a U.S.-domiciled parent holding company, the Company is the issuer of essentially all of the Company's external indebtedness, and incurs the related interest expense in the U.S. The Company's interest expense deductions are not currently limited. Further, most senior executive and oversight functions are conducted in the U.S. and the associated costs are incurred primarily in the U.S. Some of these expenses may not be deductible in the U.S., which may impact the effective tax rate.

Changes to the U.S. tax law in recent years have allowed the Company to repatriate foreign earnings without incurring additional U.S. federal income tax costs as foreign income is generally already taxed in the U.S. However, permanent reinvestment continues to be a component of the Company's global capital strategy. The Company continues to evaluate its global investment and repatriation strategy in light of our capital requirements and potential costs of repatriation, which are generally limited to local country withholding taxes.

The Coronavirus Aid, Relief and Economic Security Act (the "CARES Act") was signed into law on March 27, 2020. The CARES Act provided over \$2 trillion in economic relief to individuals, governmental agencies and companies, to deal with the public health and economic impacts of COVID-19. Pursuant to the CARES Act, the Company deferred payroll taxes due from March 27, 2020 through December 31, 2020, and paid 50% in 2021 and 2022, respectively.

Liquidity and Capital Resources

The Company is organized as a legal entity separate and distinct from its operating subsidiaries. As the Company does not have significant operations of its own, the Company is dependent upon dividends and other payments from its operating subsidiaries to pay principal and interest on its outstanding debt obligations, pay dividends to stockholders, repurchase its shares and pay corporate expenses. The Company can also provide financial support to its operating subsidiaries for acquisitions, investments and certain parts of their business that require liquidity, such as the capital markets business of Guy Carpenter. Other sources of liquidity include borrowing facilities in financing cash flows.

The Company derives a significant portion of its revenue and operating profit from operating subsidiaries located outside of the U.S. Funds from those operating subsidiaries are regularly repatriated to the U.S. out of annual earnings. At December 31, 2022, the Company had approximately \$921 million of cash and cash equivalents in its foreign operations, which includes \$325 million of operating funds required to be maintained for regulatory requirements or as collateral under certain captive insurance arrangements. The Company expects to continue its practice of repatriating available funds from its non-U.S. operating subsidiaries out of current annual earnings. Where appropriate, a portion of the current year earnings will continue to be permanently reinvested.

In 2022, the Company recorded foreign currency translation adjustments which decreased net equity by \$1 billion. Continued strengthening of the U.S. dollar against foreign currencies would further decrease the translated U.S. dollar value of the Company's net investments in its non-U.S. subsidiaries, as well as the translated U.S. dollar value of cash repatriations from those subsidiaries.

Cash and cash equivalents on our consolidated balance sheets includes funds available for general corporate purposes. Funds held on behalf of clients in a fiduciary capacity are segregated and shown separately in the consolidated balance sheets as an offset to fiduciary liabilities. Fiduciary funds cannot be used for general corporate purposes, and should not be considered as a source of liquidity for the Company.

Operating Cash Flows

The Company provided \$3.5 billion of cash from operations in both 2022 and 2021. These amounts reflect the net income of the Company during those periods, excluding gains or losses from investments, adjusted for non-cash charges and changes in working capital which relate primarily to the timing of payments of accrued liabilities, including incentive compensation, or receipts of receivables and pension contributions. The Company used cash of \$193 million related to its restructuring activities in 2022, compared to \$178 million in 2021.

Pension Related Items

Contributions

In 2022, the Company contributed \$30 million to its U.S. defined benefit pension plans and \$139 million to its non-U.S. defined benefit pension plans. In 2021, the Company contributed \$35 million to its U.S. defined benefit pension plans and \$95 million to its non-U.S. defined benefit pension plans.

In the U.S., contributions to the tax-qualified defined benefit plans are based on Employee Retirement Income Security Act ("ERISA") guidelines and the Company generally expects to maintain a funded status of 80% or more of the liability determined in accordance with the ERISA guidelines. In 2022, the Company made contributions of \$30 million to its non-qualified plans and expects to contribute approximately \$31 million in 2023. The Company was not required to and made no contributions to its U.S. qualified plans in 2022, and is not required to make any contributions in 2023.

Outside the U.S., the Company has a large number of non-U.S. defined benefit pension plans, the largest of which are in the U.K., which comprise approximately 79% of non-U.S. plan assets at December 31, 2022. Contribution rates for non-U.S. plans are generally based on local funding practices and statutory requirements, which may differ significantly from measurements in accordance with U.S. GAAP.

The Company contributed \$113 million to its U.K. plans (including the JLT section) in 2022. The Company's contributions to its U.K. plans (including the JLT section) for 2023 are expected to be approximately \$41 million.

In the U.K., the assumptions used to determine pension contributions are the result of legally-prescribed negotiations between the Company and the plans' trustee that typically occur every 3 years in conjunction with the actuarial valuation of the plans. Currently, this results in a lower funded status compared to U.S. GAAP and may result in contributions irrespective of the U.S. GAAP funded status.

In 2021, the JLT Pension Scheme was merged into the MMC U.K. Pension Fund with a new segregated JLT section created. The Company made deficit contributions of \$103 million to the JLT section in 2022, and is expected to make contributions totaling approximately \$39 million in 2023.

For the MMC U.K. Pension Fund, excluding the JLT section, an agreement was reached with the trustee in the fourth quarter of 2022, based on the surplus funding position at December 31, 2021. In accordance with the agreement, no deficit funding is required until 2026. The funding level will be re-assessed during 2025 as part of the December 31, 2024 actuarial valuation to determine if contributions are required in 2026. As part of a long-term strategy which depends on having greater influence over asset allocation and overall investment decisions, in December 2022, the Company renewed its agreement to support annual deficit contributions by the U.K. operating companies under certain circumstances, up to £450 million (or \$541 million) over a 7-year period.

The Company expects to contribute approximately \$76 million to its non-U.S. defined benefit plans in 2023, comprising approximately \$41 million to the U.K. plans and \$35 million to plans outside of the U.K.

Changes in Funded Status and Expense

The year-over-year change in the funded status of the Company's pension plans is impacted by the difference between actual and assumed results, particularly with regard to return on assets, and changes in the discount rate, as well as the amount of Company contributions, if any. Unrecognized actuarial losses as of December 31, 2022 were approximately \$1.4 billion and \$2.6 billion for the U.S. plans and non-U.S. plans, respectively, compared with losses of \$1.8 billion and \$2.9 billion as of December 31, 2021. The decreases in both the U.S. and non-U.S. plans were primarily due to an increase in the discount rate used to measure plan liabilities and the impact of foreign exchange, partially offset by a decrease in asset values. In the past several years, the amount of unamortized losses has been significantly impacted, both positively and negatively, by actual asset performance and changes in discount rates. The discount rate used to measure plan liabilities in 2022 increased in the U.S. and U.K., the Company's largest plans, following an increase in 2021 and a decrease in 2020. An increase in the discount rate decreases the measured plan benefit obligation, resulting in actuarial gains, while a decrease in the discount rate increases the measured plan obligation, resulting in actuarial losses. In 2022, the Company's defined benefit pension plan assets had losses of 18.3% and 29.2% in the U.S. and U.K., respectively, as compared to gains of 13.2% and 1.9% in the U.S. and U.K., respectively, in 2021.

Overall, based on the measurement at December 31, 2022, net benefit credits related to the Company's defined benefit plans are expected to decrease in 2023 by approximately \$12 million compared to 2022, reflecting a decrease in the U.S. plans of \$40 million, offset by an increase in non-U.S. plans of approximately \$28 million.

The Company's accounting policies for its defined benefit pension plans, including the selection of and sensitivity to assumptions, are discussed in Management's Discussion of Critical Accounting Policies. For additional information regarding the Company's retirement plans, refer to Note 1, Summary of Significant Accounting Policies, and Note 8, Retirement Benefits, in the notes to the consolidated financial statements

Financing Cash Flows

Net cash used for financing activities was \$1.0 billion in 2022, compared with \$1.3 billion used by financing activities in 2021.

Credit Facilities

The Company has a multi-currency unsecured \$2.8 billion five-year revolving credit facility (the "Credit Facility"), entered into in April 2021. The interest rate on the Credit Facility is based on LIBOR plus a fixed margin which varies with the Company's credit ratings. The Credit Facility expires in April 2026 and requires the Company to maintain certain coverage and leverage ratios which are tested quarterly. The Credit Facility includes provisions for determining a LIBOR successor rate in the event LIBOR reference

rates are no longer available or in certain other circumstances which are determined to make using an alternative rate desirable. As of December 31, 2022 and 2021, the Company had no borrowings under this facility.

In May 2022, the Company secured a \$250 million uncommitted revolving credit facility. The facility expires in May 2023, and has similar coverage and leverage ratios as the Credit Facility. The Company had no borrowings outstanding under this facility at December 31, 2022.

The Company also maintains other credit facilities, guarantees and letters of credit with various banks aggregating \$514 million at December 31, 2022, and \$508 million at December 31, 2021. There were no outstanding borrowings under these facilities as of December 31, 2022 or as of December 31, 2021.

Deht

In October 2022, the Company issued \$500 million of 5.75% senior notes due 2032 and \$500 million of 6.25% senior notes due 2052. The Company used the net proceeds from these issuances for general corporate purposes, and repaid \$350 million of 3.30% senior notes in November 2022, with an original maturity date of March 2023.

In October 2022, the Company increased its short-term commercial paper financing program to \$2.8 billion from \$2 billion. The Company had no commercial paper outstanding at December 31, 2022.

In December 2021, the Company issued \$400 million of 2.375% senior notes due 2031 and \$350 million of 2.90% senior notes due 2051. The Company used the net proceeds from these issuances for general corporate purposes, and repaid \$500 million of 2.75% senior notes with an original maturity date of January 2022, in December 2021.

On April 15, 2021, the Company repaid \$500 million of senior notes maturing in July 2021.

The Company's senior debt is currently rated A- by Standard & Poor's ("S&P"), Baa1 by Moody's and A- by Fitch. The Company's short-term debt is currently rated A-2 by S&P, P-2 by Moody's and F2 by Fitch. The Company carries a Positive outlook with Moody's and a Stable outlook with S&P and Fitch.

Share Repurchases

On March 23, 2022, the Board of Directors of the Company authorized an additional \$5 billion in share repurchases. This is in addition to the Company's existing share repurchase program, which had approximately \$1.3 billion of remaining authorization as of December 31, 2021.

In 2022, the Company repurchased 12.2 million shares of its common stock for \$1.9 billion. As of December 31, 2022, the Company remained authorized to repurchase up to approximately \$4.3 billion in shares of its common stock. There is no time limit on this authorization. In 2021, the Company repurchased 7.9 million shares of its common stock for \$1.2 billion.

Dividends

The Company paid dividends on its common stock shares of \$1.1 billion (\$2.25 per share) in 2022 and \$1.0 billion (\$2.00 per share) in 2021.

Contingent Payments Related To Acquisitions

The classification of contingent consideration in the consolidated statements of cash flows is dependent upon whether the receipt or payment was part of the initial liability established on the acquisition date (financing) or an adjustment to the acquisition date liability (operating).

The following amounts are included in the consolidated statements of cash flows as operating and financing activities:

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Operating:			
Contingent consideration payments for prior year acquisitions	\$ (38)	\$ (49)	\$ (48)
Receipt of contingent consideration for dispositions	_	19	_
Acquisition/disposition related net charges for adjustments	49	57	26
Adjustments and payments related to contingent consideration	\$ 11	\$ 27	\$ (22)
Financing:			
Contingent consideration for prior year acquisitions	\$ (32)	\$ (28)	\$ (54)
Deferred consideration related to prior year acquisitions	(126)	(89)	(68)
Payments of deferred and contingent consideration for acquisitions	\$ (158)	\$ (117)	\$ (122)
Receipt of contingent consideration for dispositions	\$ 3	\$ 71	\$ _

For acquisitions completed in 2022, and in prior years, remaining estimated future contingent payments of \$377 million and deferred consideration payments of \$142 million, are recorded in accounts payable and accrued liabilities or other liabilities in the consolidated balance sheets at December 31, 2022.

Derivatives - Net Investment Hedge

The Company has investments in various subsidiaries with Euro functional currencies. As a result, the Company is exposed to the risk of fluctuations between the Euro and U.S. dollar exchange rates. As part of its risk management program, the Company issued €1.1 billion senior notes, and designated the debt instruments as a net investment hedge of its Euro denominated subsidiaries. The hedge is reassessed each quarter to confirm that the designated equity balance at the beginning of each period continues to equal or exceed 80% of the outstanding balance of the Euro debt instrument and that all the critical terms of the hedging instrument and the hedged net investment continue to match. If the hedge is highly effective, the change in the debt balance related to foreign exchange fluctuations is recorded in accumulated other comprehensive loss in the consolidated balance sheets.

The U.S. dollar value of the Euro notes decreased by \$82 million in 2022 related to the change in foreign exchange rates. The Company concluded that the hedge was highly effective and recorded a gain as a decrease to accumulated other comprehensive loss for the year ended December 31, 2022.

Fiduciary Liabilities

Since cash and cash equivalents held in a fiduciary capacity are not available for corporate use, they are shown in the consolidated balance sheets as an offset to fiduciary liabilities. Financing cash flows reflect an increase of \$1.7 billion and \$1.2 billion in 2022 and 2021, respectively, related to fiduciary liabilities.

Investing Cash Flows

Net cash used for investing activities amounted to \$850 million in 2022, compared with \$1.2 billion used for investing activities in 2021.

The Company paid \$572 million and \$859 million, net of cash, cash equivalents and cash and cash equivalents held in a fiduciary capacity acquired, for acquisitions it made in 2022 and 2021, respectively, including the Company's increased ownership interest in Marsh India from 49% to 92% in December 2021.

In 2022, the Company sold certain business, primarily Mercer's U.S. affinity business, for cash proceeds of approximately \$155 million, partially offset by \$36 million primarily related to cash and cash equivalents held in a fiduciary capacity in the disposed businesses. In 2021, the Company sold certain of its businesses, primarily in the U.S. and U.K., for cash proceeds of approximately \$84 million.

In the third quarter of 2022, the Company sold the remaining investment in the common stock of Alexander Forbes ("AF"), for cash proceeds of approximately \$62 million.

The Company's additions to fixed assets and capitalized software, which amounted to \$470 million in 2022 and \$406 million in 2021, primarily related to computer equipment purchases, the refurbishing and modernizing of office facilities, and software development costs.

The Company has commitments for potential future investments of approximately \$160 million in private equity funds that invest primarily in financial services companies, including a \$100 million commitment to invest in a private equity fund entered into on April 1, 2022.

Commitments and Obligations

The following sets forth the Company's future contractual obligations by type as of December 31, 2022:

	Payment due by Period									
(In millions)		Total	Withi	n 1 Year		1-3 Years		4-5 Years		After 5 Years
Current portion of long-term debt	\$	268	\$	268	\$	_	\$	_	\$	_
Long-term debt		11,304		_		2,138		1,224		7,942
Interest on long-term debt		6,021		458		794		692		4,077
Net operating leases		2,228		362		615		500		751
Service agreements		509		256		194		59		_
Other long-term obligations (a)		600		253		274		68		5
Total	\$	20,930	\$	1,597	\$	4,015	\$	2,543	\$	12,775

⁽a) Primarily reflects future payments of deferred and contingent purchase consideration.

The table does not include the liability for unrecognized tax benefits of \$97 million as the Company is unable to reasonably predict the timing of settlement of these liabilities, other than approximately \$32 million that may become payable within one year. The table also does not include the remaining transitional tax payments related to the Tax Cuts and Jobs Act ("TCJA") of \$62 million, which will be paid in installments beginning in 2023 through 2026.

Management's Discussion of Critical Accounting Policies and Estimates

Management makes estimates and judgments that affect reported amounts of assets, liabilities, revenue and expenses, and disclosure of contingent assets and liabilities. Management considers the following policies to be critical to understanding the Company's financial statements because their application places the most significant demands on management's judgment, and requires management to make estimates about the effect of matters that are inherently uncertain. Actual results may differ from those estimates.

Revenue Recognition

In the Risk and Insurance Services segment, judgments related to the amount of variable revenue consideration to ultimately be received on placement of quota share reinsurance treaties and contingent commission from insurers, which was previously recognized when the contingency was resolved, now requires significant judgments and estimates.

Management also makes significant judgments and estimates to measure the progress toward completing performance obligations and realization rates for consideration related to contracts as well as potential performance-based fees in the Consulting segment.

The Company capitalizes the incremental costs to obtain contracts primarily related to commissions or sales bonus payments. These deferred costs are amortized over the expected life of the underlying customer relationships. The Company also capitalizes certain preplacement costs that are considered fulfillment costs that are amortized at a point in time when the associated revenue is recognized.

Refer to Note 2, Revenue, in the notes to the consolidated financial statements for additional information.

Legal and Other Loss Contingencies

The Company and its subsidiaries are subject to numerous claims, lawsuits and proceedings including claims for E&O. The Company records a liability when a loss is both probable and reasonably estimable which requires significant management judgment. The Company utilizes case level reviews by inside and outside counsel, an internal actuarial analysis by Oliver Wyman, a subsidiary of the Company, and other methods to estimate potential losses. The liability is reviewed quarterly and adjusted based on claims developments. In many cases, the Company has not recorded a liability, other than for legal fees to defend the claim, because the Company is unable, at present time, to make a determination that a loss is both probable and reasonably estimable. Given the unpredictability of E&O claims and of litigation that could arise from such claims, it is possible that an adverse outcome in a particular matter could have a material adverse effect on the Company's businesses, results of operations, financial condition or cash flows in a given quarterly or annual period.

In addition, to the extent that insurance coverage is available, significant management judgment is required to determine the amount of recoveries that are probable of collection under the Company's various insurance programs.

Retirement Benefits

The Company maintains qualified and non-qualified defined benefit pension and defined contribution plans for its eligible U.S. employees and a variety of defined benefit and defined contribution plans for its eligible non-U.S. employees. The Company's policy for funding its tax-qualified defined benefit retirement plans is to contribute amounts at least sufficient to meet the funding requirements set forth in U.S. and applicable foreign laws.

The Company recognizes the funded status of its over-funded defined benefit pension and retiree medical plans as a net benefit plan asset and its unfunded and underfunded plans as a net benefit plan liability. The gains or losses and prior service costs or credits that have not been recognized as components of net benefit (credit) costs are recorded as a component of Accumulated Other Comprehensive Income ("AOCI"), net of tax, in the Company's consolidated balance sheets. The gains and losses that exceed specified corridors, 10% of the greater of the projected benefit obligation or the market-related value of plan assets, are amortized prospectively out of AOCI over a period that approximates the remaining life expectancy of participants in plans where substantially all participants are inactive or the average remaining service period of active participants for plans with active participants. The vast majority of

unrecognized losses relate to inactive plans and are amortized over the remaining life expectancy of the participants.

The determination of net periodic benefit (credit) cost is based on a number of assumptions, including an expected long-term rate of return on plan assets, the discount rate, mortality and assumed rate of salary increase. The assumptions used in the calculation of net periodic pension costs and pension liabilities are disclosed in Note 8, Retirement Benefits, in the notes to the consolidated financial statements.

The long-term rate of return on plan assets assumption is determined for each plan based on the facts and circumstances that exist as of the measurement date, and the specific portfolio mix of each plan's assets. The Company utilizes a model developed by Mercer, a subsidiary of the Company, to assist in the determination of this assumption. The model takes into account several factors, including: target portfolio allocation, investment, administrative and trading expenses incurred directly by the plan trust, historical portfolio performance, relevant forward-looking economic analysis, and expected returns, variances and correlations for different asset classes. These measures are used to determine probabilities using standard statistical techniques to calculate a range of expected returns on the portfolio.

The target asset allocation for the U.S. plans is 50% equities and equity alternatives and 50% fixed income. At the end of 2022, the actual allocation for the U.S. plans was 61% equities and equity alternatives and 39% fixed income. The target asset allocation for the U.K. plans, which comprise approximately 79% of non-U.S. plan assets, is 14% equities and equity alternatives and 86% fixed income. At the end of 2022, the actual allocation for the U.K. plans was 16% equities and equity alternatives and 84% fixed income.

The discount rate selected for each U.S. plan is based on a model bond portfolio with coupons and redemptions that closely match the expected liability cash flows from the plan. Discount rates for non-U.S. plans are based on appropriate bond indices adjusted for duration. In the U.K., the plan duration is reflected using the Mercer yield curve.

The following table shows the weighted average assumed rate of return and the discount rate at the December 31, 2022 measurement date used to measure pension expense in 2023 for the total Company, the U.S. and the Rest of World ("ROW").

	Total Company	U.S.	ROW
Assumed rate of return on plan assets	5.31 %	6.49 %	4.74 %
Discount rate	5.16 %	5.53 %	4.89 %

Holding all other assumptions constant, a 0.5 percentage point change in the rate of return on plan assets and discount rate assumptions would affect net periodic pension cost for the U.S. and U.K. plans, which together comprise approximately 85% of total pension plan liabilities, as follows:

	0.5 Per Point Ir				
(In millions)	U.S.	U.K.		U.S.	U.K.
Assumed rate of return on plan assets	\$ (24)	\$ (45)	\$	24 \$	45
Discount Rate	\$ (1)	\$ 6	\$	1 \$	(8)

The impact of discount rate changes relates to the increase or decrease in actuarial gains or losses being amortized through net periodic pension cost, as well as the increase or decrease in interest expense, with all other facts and assumptions held constant. It does not contemplate nor include potential future impacts a change in the interest rate environment and discount rates might cause, such as the impact on the market value of the plans' assets. In addition, the assumed return on plan assets would likely be impacted by changes in the interest rate environment and other factors, including equity valuations, since these factors reflect the starting point used in the Company's projection models. For example, a reduction in interest rates may result in a reduction in the assumed return on plan assets. Changing the discount rate and leaving the other assumptions constant, may also not be representative of the impact on expense, because the long-term rates of inflation and salary increases are often correlated with the discount rate. Changes in these assumptions will not necessarily have a linear impact on the net periodic pension cost.

The Company contributes to certain health care and life insurance benefits provided to its retired employees. The cost of these post-retirement benefits for employees in the U.S. is accrued during the period up to the date employees are eligible to retire but is funded by the Company as incurred. The key assumptions and sensitivity to changes in the assumed health care cost trend rate are discussed in Note 8, Retirement Benefits, in the notes to the consolidated financial statements.

Income Taxes

Significant judgment is required in determining the annual effective tax rate and in evaluating uncertain tax positions. The Company reports a liability for unrecognized tax benefits resulting from uncertain tax positions taken or expected to be taken in a tax return. The evaluation of a tax position is a two-step process:

- First, the Company determines whether it is more-likely-than-not a tax position will be sustained upon tax examination, including resolution of any related appeals or litigation, based on only the technical merits of the position. If a tax position does not meet the more-likely-than-not recognition threshold, the benefit of that position is not recognized in the financial statements.
- The second step is measurement. A tax position that meets the more-likely-than-not recognition threshold is measured to determine the amount of benefit to recognize in the financial statements. The tax position is measured as the largest amount of benefit that is greater than 50% likely to be realized upon ultimate resolution with a taxing authority. Uncertain tax positions are evaluated based upon the facts and circumstances that exist at each reporting period and involve significant management judgment. Subsequent changes in judgment based upon new information may lead to changes in recognition, de-recognition, and measurement. Adjustments may result, for example, upon resolution of an issue with the taxing authorities, or expiration of a statute of limitations barring an assessment for an issue.

The Company recognizes interest and penalties, if any, related to unrecognized tax benefits in income tax expense. The Company's accounting policy follows the portfolio approach that leaves stranded income tax effects in accumulated other comprehensive income.

Certain items are included in the Company's tax returns at different times than the items are reflected in the financial statements. As a result, the annual tax expense reflected in the consolidated statements of income is different than that reported in the tax returns. Some of these differences are permanent, such as non-deductible expenses, and some differences are temporary and reverse over time, such as depreciation expense. Temporary differences create deferred tax assets and liabilities, which are measured at existing tax rates. Deferred tax liabilities generally represent tax expense recognized in the financial statements for which payment has been deferred, or expense for which a deduction has been taken already in the tax return but the expense has not yet been recognized in the financial statements. Deferred tax assets generally represent items that can be used as a tax deduction or credit in tax returns in future years for which a benefit has already been recorded in the financial statements. The Company evaluates all significant available positive and negative evidence, including the existence of losses in recent years and its forecast of future taxable income by jurisdiction, in assessing the need for a valuation allowance against deferred tax assets. The Company also considers tax planning strategies that would result in realization of deferred tax assets, and the presence of taxable income in prior period tax filings in jurisdictions that allow for the carry back of tax attributes pursuant to the applicable tax law. The underlying assumptions the Company uses in forecasting future taxable income require significant judgment and take into account the Company's recent performance. The ultimate realization of deferred tax assets is dependent on the generation of future taxable income during the periods in which temporary differences or carry-forwards are deductible or creditable. Valuation allowances are established for deferred tax assets when it is estimated that it is more-likely-than-not that future taxable income will be insufficient to fully use a deduction or credit in that jurisdiction.

Fair Value Determinations

Goodwill Impairment Testing – The Company is required to assess goodwill and any indefinite-lived intangible assets for impairment annually, or more frequently if circumstances indicate impairment may have occurred. The Company performs the annual impairment assessment for each of its reporting units during the third quarter of each year. A company can assess qualitative factors to determine whether it is necessary to perform a quantitative goodwill impairment test. Alternatively, the Company may elect to proceed directly to the quantitative goodwill impairment test. In 2022, the Company elected to perform a qualitative impairment assessment. As part of its assessment, the Company considered numerous factors, including:

- that the fair value of each reporting unit exceeds its carrying value by a substantial margin based on its most recent quantitative assessment in 2019:
- whether significant acquisitions or dispositions occurred which might alter the fair value of its reporting units;
- · macroeconomic conditions and their potential impact on reporting unit fair values;
- actual performance compared with budget and prior projections used in its estimation of reporting unit fair values;
- · industry and market conditions; and
- the year-over-year change in the Company's share price.

The Company completed its qualitative assessment in the third quarter of 2022 and concluded that a quantitative goodwill impairment test was not required in 2022 and that goodwill was not impaired.

Purchase Price Allocation

Assets acquired and liabilities assumed, including contingent consideration, as part of a business acquisition are generally recorded at their fair value at the date of acquisition. The excess of purchase price over the fair value of assets acquired and liabilities assumed is recorded as goodwill. Determining fair value of identifiable assets, particularly intangibles, and liabilities acquired also requires management to make estimates, which are based on all available information and in some cases assumptions with respect to the timing and amount of future revenues and expenses associated with an asset. These estimates directly impact the amount of identified intangible assets recognized and the related amortization expense in future periods.

New Accounting Pronouncements

Note 1, Summary of Significant Accounting Policies, in the notes to the consolidated financial statements contains a summary of the Company's significant accounting policies, including a discussion of recently issued accounting pronouncements and their impact or potential future impact on the Company's financial results, if determinable, under the sub-heading "New Accounting Pronouncements."

Item 7A. Quantitative and Qualitative Disclosures About Market Risk.

Market Risk and Credit Risk

Certain of the Company's revenues, expenses, assets and liabilities are exposed to the impact of interest rate changes and fluctuations in foreign currency exchange rates and equity markets.

Interest Rate Risk and Credit Risk

Interest income generated from the Company's cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity will vary with the general level of interest rates.

The Company had the following investments subject to variable interest rates:

For the Years Ended December 31,		
(In millions)	2022	2021
Cash and cash equivalents	\$ 1,442 \$	1,752
Cash and cash equivalents held in a fiduciary capacity	\$ 10,660 \$	9,622

Based on the above balances at December 31, 2022, if short-term interest rates increased or decreased by 10%, or 25 basis points for the year 2023, annual interest income, including interest earned on cash and cash equivalents held in a fiduciary capacity, would increase or decrease by approximately \$30 million. At December 31, 2021, a change in short-term interest rates of 10%, or 1 basis point, would have increased or decreased interest income by approximately \$1 million. The change in interest rate risk at December 31, 2022 is due to higher short-term interest rates compared to the prior year.

In addition to interest rate risk, our cash investments and fiduciary cash investments are subject to potential loss of value due to counterparty credit risk. To minimize this risk, the Company and its subsidiaries invest pursuant to a Board approved investment policy. The policy mandates the preservation of principal and liquidity and requires broad diversification with counter-party limits assigned based primarily on credit rating and type of investment. The Company carefully monitors its cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity, and will further restrict the portfolio as appropriate to market conditions. The majority of cash, cash equivalents and cash and cash equivalents held in a fiduciary capacity are invested in short-term bank deposits and liquid money market funds.

Foreign Currency Risk

The translated values of revenue and expense from the Company's international operations are subject to fluctuations due to changes in currency exchange rates. The non-U.S. based revenue that is exposed to foreign exchange fluctuations is approximately 51% of total revenue. We periodically use forward contracts and options to limit foreign currency exchange rate exposure on net income and cash flows for specific, clearly defined transactions arising in the ordinary course of business. Although the Company has significant revenue generated in foreign locations which is subject to foreign exchange rate fluctuations, in most cases both the foreign currency revenue and expenses are in the functional currency of the foreign location. As such, under normal circumstances, the U.S. dollar translation of both the revenue and expense, as well as the potentially offsetting movements of various currencies against the U.S. dollar, generally tend to mitigate the impact on net operating income of foreign currency risk.

However, there have been periods where the impact was not mitigated due to external market factors, and external macroeconomic events may result in greater foreign exchange rate fluctuations in the future. If foreign exchange rates of major currencies (Euro, Sterling, Australian dollar and Canadian dollar) moved 10% in the same direction against the U.S. dollar compared with the foreign exchange rates in 2022, the Company estimates net operating income would increase or decrease by approximately \$74 million. The corresponding increase or decrease in net operating income in 2021 was estimated at \$57 million. The Company has exposure to approximately 80 foreign currencies overall.

In Continental Europe, the largest amount of revenue from renewals for the Risk and Insurance Services segment occurs in the first quarter.

Equity Price Risk

The Company holds investments in both public and private companies as well as private equity funds, including investments of approximately \$17 million that are valued using readily determinable fair values and approximately \$42 million of investments without readily determinable fair values. The Company also has investments of approximately \$215 million that are accounted for using the equity method. The Company's investments are subject to risk of decline in market value, which, if determined to be other than temporary, could result in realized impairment losses. The Company periodically reviews the carrying value of such investments to determine if any valuation adjustments are appropriate under the applicable accounting pronouncements.

Other

A number of lawsuits and regulatory proceedings are pending. Refer to Note 16, Claims, Lawsuits and Other Contingencies, in the notes to the consolidated financial statements included in this report.

Item 8. Financial Statements and Supplementary Data.

MARSH & McLENNAN COMPANIES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF INCOME

For the Years Ended December 31,				
(In millions, except per share data)		2022	2021	2020
Revenue	\$	20,720	\$ 19,820	\$ 17,224
Expense:				
Compensation and benefits		12,071	11,425	10,129
Other operating expenses		4,369	4,083	4,029
Operating expenses		16,440	15,508	14,158
Operating income		4,280	4,312	3,066
Other net benefits credits		235	277	257
Interest income		15	2	7
Interest expense		(469)	(444)	(515)
Investment income (loss)		21	61	(22)
Income before income taxes		4,082	4,208	2,793
Income tax expense		995	1,034	747
Net income before non-controlling interests		3,087	3,174	2,046
Less: Net income attributable to non-controlling interests		37	31	30
Net income attributable to the Company	\$	3,050	\$ 3,143	\$ 2,016
Net income per share attributable to the Company				
– Basic	\$	6.11	\$ 6.20	\$ 3.98
– Diluted	\$	6.04	\$ 6.13	\$ 3.94
Average number of shares outstanding				
– Basic		499	507	506
– Diluted		505	513	512
Shares outstanding at December 31,	-	495	504	508

MARSH & McLENNAN COMPANIES, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME

For the Years Ended December 31, (In millions)	2022	2021	-	2020
Net income before non-controlling interests	\$ 3,087	\$ 3,174	\$	2,046
Other comprehensive (loss) income, before tax:				
Foreign currency translation adjustments	(1,198)	(389)		559
Gain (loss) related to pension and post-retirement plans	641	1,229		(784)
Other comprehensive (loss) income, before tax	(557)	840		(225)
Income tax expense (credit) on other comprehensive loss	182	305		(170)
Other comprehensive (loss) income, net of tax	(739)	535		(55)
Comprehensive income	2,348	3,709		1,991
Less: Comprehensive income attributable to non-controlling interests	37	31		30
Comprehensive income attributable to the Company	\$ 2,311	\$ 3,678	\$	1,961

MARSH & McLENNAN COMPANIES, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS

December 31,				
(In millions, except share data)		2022		2021
ASSETS				
Current assets:	_		_	
Cash and cash equivalents	\$	1,442	\$	1,752
Receivables				
Commissions and fees		5,293		5,093
Advanced premiums and claims		103		136
Other		616		523
		6,012		5,752
Less-allowance for credit losses		(160)		(166)
Net receivables		5,852		5,586
Other current assets		1,005		926
Total current assets		8,299		8,264
Goodwill		16,251		16,317
Other intangible assets		2,537		2,810
Fixed assets, net		871		847
Pension related assets		2,127		2,270
Right of use assets		1,562		1,868
Deferred tax assets		358		551
Other assets		1,449		1,461
	\$	33,454	\$	34,388
LIABILITIES AND EQUITY				
Current liabilities:				
Short-term debt	\$	268	\$	17
Accounts payable and accrued liabilities		3,278		3,165
Accrued compensation and employee benefits		3,095		2,942
Current lease liabilities		310		332
Accrued income taxes		221		198
Total current liabilities		7,172		6,654
Fiduciary liabilities		10,660		9,622
Less - cash and cash equivalents held in a fiduciary capacity		(10,660)		(9,622)
		_		
Long-term debt		11,227		10,933
Pension, postretirement and postemployment benefits		921		1,632
Long-term lease liabilities		1,667		1,880
Liability for errors and omissions		355		355
Other liabilities		1,363		1,712
Commitments and contingencies				_
Equity:				
Preferred stock, \$1 par value, authorized 6,000,000 shares, none issued		_		_
Common stock, \$1 par value, authorized 1,600,000,000 shares,				
issued 560,641,640 shares at December 31, 2022 and 2021		561		561
Additional paid-in capital		1,179		1,112
Retained earnings		20,301		18,389
Accumulated other comprehensive loss		(5,314)		(4,575)
Non-controlling interests		229		213
		16,956		15,700
Less – treasury shares, at cost, 65,855,914 shares at December 31, 2022		,,,,		_3,.03
and 57,105,619 shares at December 31, 2021		(6,207)		(4,478)
Total equity		10,749		11,222
	\$	33,454	\$	34,388

MARSH & McLENNAN COMPANIES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CASH FLOWS

For the Years Ended December 31,				
(In millions)		2022	2021	2020
Operating cash flows:				
Net income before non-controlling interests	\$	3,087	\$ 3,174	\$ 2,046
Adjustments to reconcile net income to cash provided by operations:				
Depreciation and amortization of fixed assets and capitalized software		381	382	390
Amortization of intangible assets		338	365	351
Non-cash lease expense		404	327	355
Adjustments and payments related to contingent consideration assets and liabilities		11	27	(22)
Deconsolidation of Russian businesses		39	_	_
Gain on consolidation of entity		(2)	(267)	_
Net (gain) loss on investments		(21)	(61)	22
Net (gain) loss on disposition of assets		(127)	(33)	24
Share-based compensation expense		367	348	290
Changes in assets and liabilities:				
Net receivables		(492)	(570)	412
Other assets		(122)	(354)	255
Accrued compensation and employee benefits		171	574	171
Provision for taxes, net of payments and refunds		(54)	(33)	63
Contributions to pension and other benefit plans in excess of current year credit		(385)	(372)	(356)
Other liabilities		193	358	(268)
Operating lease liabilities		(323)	(349)	(351)
Net cash provided by operations		3,465	 3,516	3,382
Financing cash flows:		5,.55	0,010	0,002
-		(4.050)	(1.150)	
Purchase of treasury shares		(1,950)	(1,159)	1 000
Borrowings from term-loan and credit facilities				1,000
Proceeds from issuance of debt		984	743	737
Repayments of debt		(365)	(1,016)	(2,515)
Net purchase of non-controlling interests		(7)		(3)
Shares withheld for taxes on vested units – treasury shares		(198)	(101)	(132)
Issuance of common stock from treasury shares		126	161	132
Payments of deferred and contingent consideration for acquisitions		(158)	(117)	(122)
Receipts of contingent consideration for dispositions		3	71	_
Distributions of non-controlling interests		(27)	(36)	(34)
Dividends paid		(1,138)	(1,026)	(943)
Change in fiduciary liabilities		1,684	1,183	955
Net cash used for financing activities		(1,046)	(1,297)	 (925)
Investing cash flows:				
Capital expenditures		(470)	(406)	(348)
Purchases of long term investments		(22)	(28)	(23)
Sales of long term investments		86	41	130
Dispositions		119	84	98
Acquisitions, net of cash and cash held in a fiduciary capacity acquired		(572)	(859)	(647
Other, net		9	4	(3)
Net cash used for investing activities		(850)	 (1,164)	 (793)
Effect of exchange rate changes on cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity		(841)	(355)	511
Increase in cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity		728	700	2,175
Cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity at beginning of year		11,374	10,674	8,499
Cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity at end of year	\$	12,102	\$ 11,374	\$ 10,674
Reconciliation of cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity to the Consolidated Bala	ance Sheets		 	
For the Years Ended December 31,		2022	2021	2020
(In millions)				
Cash and cash equivalents	\$	1,442	\$ 1,752	\$ 2,089
Cash and cash equivalents held in a fiduciary capacity		10,660	 9,622	 8,585
Total cash, cash equivalents, and cash and cash equivalents held in a fiduciary capacity	\$	12,102	\$ 11,374	\$ 10,674

${\it MARSH~\&~McLENNAN~COMPANIES,~INC.~AND~SUBSIDIARIES}\\ {\it CONSOLIDATED~STATEMENTS~OF~EQUITY}$

For the Years Ended December 31,				
(In millions, except per share data)		2022	2021	2020
COMMON STOCK				
Balance, beginning and end of year	\$	561	\$ 561	\$ 561
ADDITIONAL PAID-IN CAPITAL				
Balance, beginning of year	\$	1,112	\$ 943	\$ 862
Change in accrued stock compensation costs		(2)	124	75
Issuance of shares under stock compensation plans and employee stock purchase plans		80	45	7
Other		(11)	_	(1)
Balance, end of year	\$	1,179	\$ 1,112	\$ 943
RETAINED EARNINGS				
Balance, beginning of year	\$	18,389	\$ 16,272	\$ 15,199
Net income attributable to the Company		3,050	3,143	2,016
Dividend equivalents declared and paid - (per share amounts: \$2.25 in 2022, \$2.00 in 2021, an \$1.84 in 2020)	d	(13)	(12)	(11)
Dividends declared and paid – (per share amounts: \$2.25 in 2022, \$2.00 in 2021, and \$1.84 in 2020)		(1,125)	(1,014)	(932)
Balance, end of year	\$	20,301	\$ 18,389	\$ 16,272
ACCUMULATED OTHER COMPREHENSIVE INCOME (LOSS)				
Balance, beginning of year	\$	(4,575)	\$ (5,110)	\$ (5,055)
Other comprehensive (loss) income, net of tax		(739)	535	(55)
Balance, end of year	\$	(5,314)	\$ (4,575)	\$ (5,110)
TREASURY SHARES				
Balance, beginning of year	\$	(4,478)	\$ (3,562)	\$ (3,774)
Issuance of shares under stock compensation plans and employee stock purchase plans		221	243	212
Purchase of treasury shares		(1,950)	(1,159)	_
Balance, end of year	\$	(6,207)	\$ (4,478)	\$ (3,562)
NON-CONTROLLING INTERESTS				
Balance, beginning of year	\$	213	\$ 156	\$ 150
Net income attributable to non-controlling interests		37	31	30
Distributions and other changes		(28)	(38)	(21)
Net non-controlling interests acquired		7	64	(3)
Balance, end of year	\$	229	\$ 213	\$ 156
TOTAL EQUITY	\$	10,749	\$ 11,222	\$ 9,260
-			 	

MARSH & McLENNAN COMPANIES, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. Summary of Significant Accounting Policies

Nature of Operations: Marsh & McLennan Companies, Inc., and its consolidated subsidiaries (the "Company"), a global professional services firm, is organized based on the different services that it offers. Under this structure, the Company's two business segments are Risk and Insurance Services and Consulting.

The Risk and Insurance Services segment ("RIS") includes risk management activities (risk advice, risk transfer and risk control and mitigation solutions) as well as insurance and reinsurance broking and services for businesses, public entities, insurance companies, associations, professional services organizations, and private clients. The Company conducts business in this segment through Marsh and Guy Carpenter. Marsh provides data-driven risk advisory services and insurance solutions to commercial and consumer clients. Guy Carpenter develops advanced risk, reinsurance and capital strategies that help clients grow profitably and identify and capitalize on emerging opportunities.

The Consulting segment includes health, wealth and career solutions and products, and specialized management, strategic, economic and brand consulting services. The Company conducts business in this segment through Mercer and Oliver Wyman Group. Mercer delivers advice and technology-driven solutions that help organizations redefine the future of work, shape retirement and investment outcomes, and advance health and well-being for a changing workforce. Oliver Wyman Group serves as a critical strategic, economic and brand advisor to private sector and governmental clients.

Deconsolidation of Russia

On February 24, 2022, Russian forces launched a military invasion of Ukraine. In response, the United States (U.S.), the European Union (E.U.), United Kingdom (U.K.) and other governments have imposed significant economic sanctions on Russia, and Russia has responded with counter-sanctions

The Company concluded that it did not meet the accounting criteria for control over its wholly-owned Russian businesses due to the evolving trade and economic sanctions, and recorded a loss of \$52 million on the deconsolidation of the Russian businesses and other related charges. Subsequently, the Company entered into a definitive agreement to exit its businesses in Russia and transfer ownership to local management, pending regulatory approvals.

The Company continues to monitor the ongoing situation and its potential impact on our business, financial condition, results of operations and cash flows.

Business Update related to COVID-19

For nearly three years, the COVID-19 pandemic has impacted businesses globally including in every geography in which the Company operates. Our businesses have remained resilient throughout the pandemic and demand for our advice and services remains strong. The ultimate extent of the impact of COVID-19 to the Company will depend on future developments that it is unable to predict.

Principles of Consolidation: The accompanying consolidated financial statements include all wholly-owned and majority-owned subsidiaries. All significant inter-company transactions and balances have been eliminated.

Revenue: The Company provides detailed discussion regarding its revenue policies in Note 2, Revenue.

Cash and Cash Equivalents: Cash and cash equivalents primarily consist of certificates of deposit and time deposits, with original maturities of three months or less, and money market funds. The estimated fair value of the Company's cash and cash equivalents approximates their carrying value. The Company is required to maintain operating funds primarily related to regulatory requirements outside of the U.S. or as collateral under captive insurance arrangements. At December 31, 2022, the Company maintained \$348 million compared to \$303 million at December 31, 2021 related to these regulatory requirements.

Fixed Assets: Fixed assets are stated at cost less accumulated depreciation and amortization. Expenditures for improvements are capitalized. Upon sale or retirement of an asset, the cost and related accumulated depreciation and amortization are removed from the accounts and any gain or loss is reflected in income. Expenditures for maintenance and repairs are charged to operations as incurred.

Depreciation of buildings, building improvements, furniture, and equipment is provided on a straight-line basis over the estimated useful lives of these assets. Furniture and equipment are depreciated over periods ranging from 3 to 10 years. Leasehold improvements are amortized on a straight-line basis over the periods covered by the applicable leases or the estimated useful life of the improvement, whichever is less. Buildings are depreciated over periods ranging from 30 to 40 years. The Company periodically reviews long-lived assets for impairment whenever events or changes indicate that the carrying value of assets may not be recoverable.

The components of fixed assets are as follows:

December 31,		
(In millions)	2022	2021
Furniture and equipment	\$ 772	\$ 811
Land and buildings	372	385
Leasehold and building improvements	1,258	1,240
	2,402	2,436
Less-accumulated depreciation and amortization	(1,531)	(1,589)
Fixed assets, net	\$ 871	\$ 847

Investments: The caption "Investment income (loss)" in the consolidated statements of income comprises realized and unrealized gains and losses from investments recognized in earnings. It includes, when applicable, other than temporary declines in the value of securities, mark-to-market increases or decreases in equity investments with readily determinable fair values and equity method gains or losses on the Company's investments in private equity funds.

The Company holds investments in private equity funds. Investments in private equity funds are accounted for in accordance with the equity method of accounting using a consistently applied three-month lag period adjusted for any known significant changes from the lag period to the reporting date of the Company. The underlying private equity funds follow investment company accounting, where investments within the fund are carried at fair value. Investment gains or losses for its proportionate share of the change in fair value of the funds are recorded in earnings. Investments accounted for in accordance with the equity method of accounting are included in "other assets" in the consolidated balance sheets.

In 2022, the Company recorded net investment income of \$21 million, compared to \$61 million in 2021, and a net investment loss of \$22 million in 2020. Net Investment income in 2022 is driven primarily by lower mark-to-market gains from the Company's private equity investments compared to the prior year. The net investment loss in 2020 is primarily due to the loss on the sale of shares of Alexander Forbes ("AF").

Goodwill and Other Intangible Assets: Goodwill represents acquisition costs in excess of the fair value of net assets acquired. Goodwill is assessed at least annually for impairment. The Company performs an annual impairment test for each of its reporting units during the third quarter of each year. A company can assess qualitative factors to determine whether it is necessary to perform a goodwill impairment test. Alternatively, a company may elect to proceed directly to the quantitative goodwill impairment test. When a quantitative test is performed, fair values of the reporting units are estimated using either a market approach or a discounted cash flow model. Carrying values for the reporting units are based on balances at the prior quarter-end and include directly identified assets and liabilities as well as an allocation of those assets and liabilities not recorded at the reporting unit level. As discussed in Note 6, Goodwill and Other Intangibles, the Company elected to perform a qualitative impairment assessment in 2022.

Other intangible assets, which primarily consist of acquired customer lists that are not deemed to have an indefinite life, are amortized over their estimated lives, typically ranging from 10 to 15 years, and assessed for impairment upon the occurrence of certain triggering events in accordance with applicable accounting literature. The Company had no indefinite lived identified intangible assets at December 31, 2022 and 2021.

Retirement Benefits: The Company maintains qualified and non-qualified defined benefit pension plans for its U.S. and non-U.S. eligible employees. The Company's policy for funding its tax qualified defined benefit retirement plans is to contribute amounts at least sufficient to meet the funding requirements set forth by U.S. law and the laws of the non-U.S. jurisdictions in which the Company offers defined benefit plans. The net benefit cost of the Company's defined benefit plans is measured on an actuarial basis using various methods and assumptions.

The Company uses actuaries from Mercer, a subsidiary of the Company, to perform valuations of its pension plans. The long-term rate of return on plan assets assumption is determined for each plan based on the facts and circumstances that exist as of the measurement date, and the specific portfolio mix of each plan's assets. The Company utilizes a model developed by the Mercer actuaries to assist in the determination of this assumption. The model takes into account several factors, including: actual and target portfolio allocation; investment, administrative and trading expenses incurred directly by the plan trust; historical portfolio performance; relevant forward-looking economic analysis; and expected returns, variances and correlations for different asset classes. These measures are used to determine probabilities using standard statistical techniques to calculate a range of expected returns on the portfolio. Generally, the Company does not adjust the rate of return assumption from year to year if, at the measurement date, it is within the range between the 25th and 75th percentile of the expected long-term annual returns. Historical long-term average asset returns of the most significant plans are also reviewed to determine whether they are consistent and reasonable compared with the rate selected. The expected return on plan assets is determined by applying the assumed long-term rate of return to the market-related value of plan assets. This market-related value recognizes investment gains or losses over a five-year period from the year in which they occur. Investment gains or losses for this purpose are the difference between the expected return calculated using the market-related value of assets and the actual return based on the market value of assets. Since the market-related value of assets recognizes gains or losses over a five-year period, the future market-related value of the assets will be impacted as previously deferred gains or losses are reflected. The Company reviews its actuarial assumptions on an annual basis and modifies these assumptions based on current rates and trends.

The funded status of the Company's pension plans is recorded in the consolidated balance sheets and provides for a delayed recognition of actuarial gains or losses arising from changes in the projected benefit obligation due to changes in the assumed discount rates, differences between the actual and expected value of plan assets and other assumption changes. The unrecognized pension plan actuarial gains or losses and prior service costs not yet recognized in net periodic pension cost are recognized in Accumulated Other Comprehensive Income ("AOCI"), net of tax. These gains and losses are amortized prospectively out of AOCI over a period that approximates the remaining life expectancy of participants in plans where substantially all participants are inactive, or the average remaining service period of active participants for plans with active participants. The vast majority of unrecognized losses relate to inactive plans and are amortized over the remaining life expectancy of the participants.

The discount rate selected for each U.S. plan is based on a model bond portfolio with coupons and redemptions that closely match the expected liability cash flows from the plan. Discount rates for non-U.S. plans are based on appropriate bond indices adjusted for duration. In the U.K., the plan duration is reflected using the Mercer yield curve.

Defined Benefit Pension Plans in the U.K. and certain other countries allow participants an option for the payment of a lump sum distribution from plan assets before retirement in full satisfaction of the retirement benefits due to the participant as well as any survivor's benefit. The Company's policy is to treat these lump sum payments as a partial settlement of the plan liability if they exceed the total of interest plus service costs.

Refer to Note 8, Retirement Benefits, for additional information.

Lease: A lease is defined as a party obtaining the right to use an asset legally owned by another party. The Company determines if an arrangement is a lease at inception. Right-of-use ("ROU") assets and lease liabilities are recorded at the lease commencement date. Lease liabilities are recognized at the present value of the contractual fixed lease payments. The Company uses discount rates to determine the present value of future lease payments. The Company primarily uses its incremental borrowing rate adjusted to reflect a secured rate, based on the information available for leases, including the lease term

and interest rate environment in the country in which the lease exists. The lease terms used to calculate the ROU asset and lease liability may include options to extend or terminate when it is reasonably certain that the Company will exercise that option. ROU assets are recognized equal to lease liabilities, adjusted for prepaid lease payments, initial direct costs and lease incentives. Operating lease expense is recognized on a straight-line basis over the lease term, while variable lease payments are expensed as incurred.

Leases are negotiated with third-parties and, in some instances, contain renewal, expansion and termination options. The Company also subleases certain office facilities to third-parties when the Company no longer utilizes the space. In addition to the base rental costs, the Company's lease agreements generally provide for rent escalations resulting from increased assessments for real estate taxes and other charges. A portion of the Company's real estate lease portfolio contains base rents subject to annual changes in the Consumer Price Index ("CPI") as well as charges for operating expenses which are reimbursable to the landlord based on actual usage. Changes to the CPI and payments for such reimbursable operating expenses are considered variable and are recognized as variable lease costs in the period in which the obligation for those payments was incurred. Approximately 99% of the Company's lease obligations are for the use of office space. All of the Company's material leases are operating leases.

As a practical expedient, the Company has elected an accounting policy not to separate non-lease components from lease components and instead account as a single lease component. The Company has also elected not to recognize ROU assets and lease liabilities for leases that, at the commencement date, are for 12 months or less. Refer to Note 12, Leases for additional information.

Capitalized Software Costs: The Company capitalizes certain costs to develop, purchase or modify software for the internal use of the Company. These costs are amortized on a straight-line basis over periods ranging from 3 to 10 years. Costs incurred during the preliminary project stage and post implementation stage are expensed as incurred. Costs incurred during the application development stage are capitalized. Costs related to updates and enhancements are only capitalized if they will result in additional functionality. Capitalized computer software costs of \$492 million and \$475 million, net of accumulated amortization of \$1.8 billion and \$1.7 billion as of December 31, 2022 and 2021, respectively, are included in other assets in the consolidated balance sheets.

Legal and Other Loss Contingencies: The Company and its subsidiaries are subject to a significant number of claims, lawsuits and proceedings including claims for errors and omissions ("E&O"). The Company records a liability when a loss is both probable and reasonably estimable which requires significant management judgment. The Company utilizes case level reviews by inside and outside counsel, an internal actuarial analysis by Oliver Wyman, a subsidiary of the Company, and other methods to estimate potential losses, including estimated legal costs. The liability is reviewed quarterly and adjusted as developments warrant. In many cases, the Company has not recorded a liability, other than for legal fees to defend the claim, because the Company is unable, at present time, to make a determination that a loss is both probable and reasonably estimable. Given the unpredictability of E&O claims and of litigation that could arise from such claims, it is possible that an adverse outcome in a particular matter could have a material adverse effect on the Company's businesses, results of operations, financial condition or cash flows in a given quarterly or annual period.

As of December 31, 2022, the Company's liability for errors and omissions was \$419 million, compared to \$434 million at December 31, 2021, of which \$64 million and \$79 million, respectively, were current liabilities and included in accounts payable and accrued liabilities in the consolidated balance sheets. In addition, to the extent that insurance coverage is available, significant management judgment is required to determine the amount of recoveries that are probable of collection under the Company's various insurance programs.

The legal and other contingent liabilities described above are not discounted.

Income Taxes: The Company's effective tax rate reflects its income, statutory tax rates and tax planning in the various jurisdictions in which it operates. Significant judgment is required in determining the annual tax provision and in evaluating uncertain tax positions and the ability to realize deferred tax assets.

The Company reports a liability for unrecognized tax benefits resulting from uncertain tax positions taken or expected to be taken in a tax return. The evaluation of a tax position is a two-step process. The first

step involves recognition. The Company determines whether it is more likely than not that a tax position will be sustained upon tax examination, including resolution of any related appeals or litigation, based on only the technical merits of the position. The technical merits of a tax position derive from both statutory and judicial authority (legislation and statutes, legislative intent, regulations, rulings, and case law) and their applicability to the facts and circumstances of the tax position. If a tax position does not meet the more-likely-than-not recognition threshold, the benefit of that position is not recognized in the financial statements. The second step is measurement. A tax position that meets the more-likely-than-not recognition threshold is measured to determine the amount of benefit to recognize in the financial statements. The tax position is measured as the largest amount of benefit that is greater than 50% likely to be realized upon ultimate resolution with a taxing authority. Uncertain tax positions are evaluated based on the facts and circumstances that exist at each reporting period. Subsequent changes in judgment based on new information may lead to changes in recognition, de-recognition, and measurement. Adjustments may result, for example, upon resolution of an issue with the taxing authorities, or expiration of a statute of limitations barring an assessment for an issue. The Company recognizes interest and penalties, if any, related to unrecognized tax benefits in income tax expense.

Tax law may require items be included in the Company's tax returns at different times than the items are reflected in the financial statements. As a result, the annual tax expense reflected in the consolidated statements of income is different than that reported in the income tax returns. Some of these differences are permanent, such as expenses that are not deductible in the returns, and some differences are temporary and reverse over time, such as depreciation expense. Temporary differences create deferred tax assets and liabilities. Deferred tax assets generally represent items that can be used as a tax deduction or credit in tax returns in future years for which benefit has already been recorded in the financial statements. Valuation allowances are established for deferred tax assets when it is estimated that future taxable income will be insufficient to use a deduction or credit in that jurisdiction. Deferred tax liabilities generally represent tax expense recognized in the financial statements for which payment has been deferred, or expense for which a deduction has been taken already in the tax return but the expense has not yet been recognized in the financial statements.

Restructuring Costs: Charges associated with restructuring activities are recognized in accordance with applicable accounting guidance which includes accounting for disposal or exit activities, guidance related to impairment of ROU assets related to real estate leases, as well as other costs resulting from accelerated depreciation or amortization of leasehold improvements and other property and equipment.

Severance and related costs are recognized based on amounts due under established severance plans or estimates of one-time benefits that will be provided. Typically, severance benefits are recognized when the impacted colleagues are notified of their expected termination and such termination is expected to occur within the legally required notification period. These costs are included in compensation and benefits in the consolidated statements of income.

Costs for real estate rationalization are recognized based on the type of cost and the expected future use of the facility. For locations where the Company does not expect to sub-lease the property, the amortization of any ROU asset is accelerated from the decision date to the cease use date. For locations where the Company expects to sub-lease the properties subsequent to its vacating the property, the ROU asset is reviewed for potential impairment at the earlier of the cease use date or the date a sub-lease is signed. To determine the amount of impairment, the fair value of the ROU asset is determined based on the present value of the estimated net cash flows related to the property. Contractual costs outside of the ROU asset are recognized based on the net present value of expected future cash outflows for which the Company will not receive any benefit. Such amounts are reliant on estimates of future sub-lease income to be received and future contractual costs to be incurred.

These costs are included in other operating expenses in the consolidated statements of income.

Other costs related to restructuring, such as moving, legal or consulting costs, are recognized as incurred. These costs are included in other operating expenses in the consolidated statements of income.

Derivative Instruments: All derivatives, whether designated in hedging relationships or not, are recorded on the consolidated balance sheets at fair value. If the derivative is designated as a fair value hedge, the changes in the fair value of the derivative and of the hedged item attributable to the hedged risk are recognized in earnings. The fair value of the derivative is recorded in the consolidated balance sheets in

other receivables or accounts payable and accrued liabilities. If the derivative is designated as a cash flow hedge, the effective portions of changes in the fair value of the derivative are recorded in other comprehensive income and are recognized in the consolidated statements of income when the hedged item affects earnings. Changes in the fair value attributable to the ineffective portion of cash flow hedges are recognized in earnings. If a derivative is not designated as an accounting hedge, such as forward contracts periodically used by the Company to limit foreign currency exchange rate exposure on net income, the change in fair value is recorded in earnings.

Per Share Data: Basic net income per share attributable to the Company is calculated by dividing the after-tax income attributable to the Company by the weighted average number of outstanding shares of the Company's common stock.

Diluted net income per share attributable to the Company is calculated by dividing the after-tax income attributable to the Company by the weighted average number of outstanding shares of the Company's common stock, which have been adjusted for the dilutive effect of potentially issuable common shares.

Basic and Diluted EPS Calculation			
(In millions, except per share data)	2022	2021	2020
Net income before non-controlling interests	\$ 3,087	\$ 3,174	\$ 2,046
Less: Net income attributable to non-controlling interests	37	31	30
Net income attributable to the Company	\$ 3,050	\$ 3,143	\$ 2,016
Basic weighted average common shares outstanding	499	507	506
Dilutive effect of potentially issuable common shares	6	6	6
Diluted weighted average common shares outstanding	505	513	512
Average stock price used to calculate common stock equivalents	\$ 160.39	\$ 141.57	\$ 109.12

Fiduciary Assets and Liabilities: The Company, in its capacity as an insurance broker or agent, generally collects premiums from insureds and after deducting its commissions, remits the premiums to the respective insurance underwriters. The Company also collects claims or refunds from underwriters on behalf of insureds. Unremitted insurance premiums and claims proceeds are held by the Company in a fiduciary capacity. The Company's fiduciary assets primarily include bank or short-term time deposits and liquid money market funds, classified as cash and cash equivalents. Risk and Insurance Services revenue includes interest on fiduciary funds of \$120 million, \$15 million and \$46 million in 2022, 2021 and 2020, respectively. Since cash and cash equivalents held in a fiduciary capacity are not available for corporate use, they are shown in the consolidated balance sheets as an offset to fiduciary liabilities.

Net uncollected premiums and claims and the related payables were \$13.0 billion at December 31, 2022 and 2021. The Company is not a principal to the contracts under which the right to receive premiums or the right to receive reimbursement of insured losses arises. Accordingly, net uncollected premiums and claims and the related payables are not assets and liabilities of the Company and are not included in the accompanying consolidated balance sheets.

In certain instances, the Company advances premiums, refunds or claims to insurance underwriters or insureds prior to collection. These advances are made from corporate funds and are reflected in the accompanying consolidated balance sheets as receivables.

The Company, through its Mercer subsidiary, manages assets in trusts or funds for which Mercer's management or trustee fee is not considered a variable interest, since the fees are commensurate with the level of effort required to provide those services. Mercer is not the primary beneficiary of these trusts or funds. Mercer's maximum exposure to loss of its interests is, therefore, limited to collection of its fees.

Estimates: The preparation of the consolidated financial statements requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosures of assets and liabilities at the date of the financial statements, and the reported amounts of revenue and expense during the reporting period.

On an ongoing basis, the Company evaluates its estimates, judgments and methodologies. The estimates are based on historical experience and on various other assumptions that the Company believes are reasonable.

Such matters include:

- estimates of revenue;
- impairment assessments and charges;
- · recoverability of long-lived assets;
- · liabilities for errors and omissions;
- deferred tax assets, uncertain tax positions and income tax expense;
- share-based and incentive compensation expense;
- the allowance for current expected credit losses on receivables:
- useful lives assigned to long-lived assets, and depreciation and amortization; and
- fair value estimates of contingent consideration receivable or payable related to acquisitions or dispositions.

The Company believes these estimates are reasonable based on information currently available at the time they are made. The Company also considered potential impact of macroeconomic factors including inflation, volatility in interest rates, the war in Ukraine and COVID-19 to its customer base in various industries and geographies. Insurance exposures subject to variable factors are subject to mid-term and end of term adjustments, as well as policy audits, which may reduce premiums and corresponding commissions. Estimates were updated based on internal and industry specific economic data. Actual results may differ from these estimates.

New Accounting Pronouncement Adopted Effective January 1, 2022:

In October 2021, the Financial Accounting Standards Board ("FASB") issued new guidance for measuring contract assets and contract liabilities acquired in a business combination. In accordance with the new guidance, contract assets and contract liabilities should be measured in accordance with the guidance for revenue from contracts with customers as opposed to the guidance for business combinations. The guidance must be applied on a prospective basis, and is effective for fiscal years beginning after December 15, 2022, including interim periods therein. Early adoption is permitted. The Company elected to adopt this new standard effective January 1, 2022. Adoption of this guidance did not have a material impact on the Company's financial position or results of operations.

New Accounting Pronouncements Adopted Effective January 1, 2021:

In January 2020, the FASB issued guidance that addresses accounting for the transition into and out of the equity method and measuring certain purchased options and forward contracts to acquire investments. The standard takes effect for public business entities for fiscal years, and interim periods within those fiscal years, beginning after December 15, 2020. The adoption of this standard did not have a material impact on the Company's financial position or its results of operations.

In December 2019, the FASB issued guidance related to the accounting for income taxes. The standard removes specific exceptions in the current rules and eliminates the need for an organization to analyze whether the following apply in a given period: (a) exception to the incremental approach for intraperiod tax allocation; (b) exceptions to accounting for basis differences when there are ownership changes in foreign investments and (c) exception in interim period income tax accounting for year-to-date losses that exceed anticipated losses. The standard also is designed to improve financial statement preparers' application of income tax-related guidance and simplify GAAP for (a) franchise taxes that are partially based on income; (b) transactions with a government that result in a step-up in the tax basis of goodwill; (c) separate financial statements of legal entities that are not subject to tax and (d) enacted changes in tax laws in interim periods. The standard takes effect for public business entities for fiscal years, and interim periods within those fiscal years, beginning after December 15, 2020. The adoption of this standard did not have a material impact on the Company's financial position or its results of operations.

2. Revenue

The core principle of the revenue recognition guidance is that an entity should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those goods or services. To achieve that principle, the entity applies the following steps: identify the contract(s) with the customer, identify the performance obligations in the contract(s), determine the transaction price, allocate the transaction price to the performance obligations in the contract and recognize revenue when (or as) the entity satisfies a performance obligation. In accordance with the accounting guidance, a performance obligation is satisfied either at a "point in time" or "over time", depending on the nature of the product or service provided, and the specific terms of the contract with customers.

Other revenue included in the consolidated statements of income that is not from contracts with customers is less than 2% of total revenue and is not presented as a separate line item.

Risk and Insurance Services

Risk and Insurance Services revenue reflects compensation for brokerage and consulting services through commissions and fees. Commission rates and fees vary in amount and can depend on a number of factors, including the type of insurance or reinsurance coverage provided, the particular insurer or reinsurer selected, and the capacity in which the broker acts and negotiates with clients. For the majority of the insurance and reinsurance brokerage arrangements, advice and services provided which culminate in the placement of an effective policy are considered a single performance obligation. Arrangements with clients may include the placement of a single policy, multiple policies or a combination of policy placements and other services. Consideration related to such "bundled arrangements" is allocated to the individual performance obligations based on their relative fair value. Revenue for policy placement is generally recognized on the policy effective date, at which point control over the services provided by the Company has transferred to the client and the client has accepted the services. In many cases, fee compensation may be negotiated in advance, based on the type of risk, coverage required and service provided by the Company and ultimately, the extent of the risk placed into the insurance market or retained by the client. The trends and comparisons of revenue from one period to the next can be affected by changes in premium rate levels, fluctuations in client risk retention and increases or decreases in the value of risks that have been insured, as well as new and lost business, and the volume of business from new and existing clients. For such arrangements, revenue is recognized using output measures, which correspond to the progress toward completing the performance obligation. Fees for non-risk transfer services provided to clients are recognized over time in the period the services are provided, using a proportional performance model, primarily based on input measures. These measures of progress provide a faithful depiction of the progress towards completion of the performance obligation.

Revenue related to reinsurance brokerage for excess of loss ("XOL") treaties is estimated based on contractually specified minimum or deposit premiums, and adjusted as additional evidence of the ultimate amount of brokerage is received. Revenue for quota share treaties is estimated based on indications of estimated premium income provided by the ceding insurer. The estimated brokerage revenue recognized for quota share treaties is constrained to an amount that is probable to not have a significant negative adjustment. The estimated revenue and the constraint are evaluated as additional evidence of the ultimate amount of underlying risks to be covered and are received over the 12 to 18 months following the effective date of the placement.

In addition to compensation from its clients, the Company also receives other compensation, separate from retail fees and commissions, from insurance companies. This other compensation includes, among other things, payments for consulting and analytics services provided to insurers; compensation for administrative and other services (including fees for underwriting services and services provided to or on behalf of insurers relating to the administration and management of quota shares, panels and other facilities in which insurers participate); and contingent commissions, which are paid by insurers based on factors such as volume or profitability of Marsh's placements primarily driven by the Marsh McLennan Agency ("MMA") and parts of Marsh's international operations. Revenue for contingent commissions from insurers is estimated based on historical evidence of the achievement of the respective contingent metrics and recorded as the underlying policies that contribute to the achievement of the metric are placed. Due to the uncertainty of the amount of contingent consideration that will be received, the estimated revenue is

constrained to an amount that is probable to not have a significant negative adjustment. Contingent consideration is generally received in the first quarter of the subsequent year.

A significant portion of the Company's Risk and Insurance Services revenue is commission revenue for brokerage arrangements recognized at a point in time on the effective date of the underlying policy. Commission revenue is estimated using historical information about the risks to be covered over the policy period, some of which are dependent on variable factors such as number of employees covered, covered payroll, airline passenger miles flown, shipped tonnage of marine cargo and others. Marsh and Guy Carpenter also receive interest income on certain funds (such as premiums and claims proceeds) held in a fiduciary capacity for others.

Insurance brokerage commissions are generally invoiced on the policy effective date. Fee based arrangements generally include a percentage of the total fee due upon signing the arrangement, with additional fixed installments payable over the remainder of the year. Payment terms range from receipt of invoice up to 30 days from invoice date.

Reinsurance brokerage revenue is recognized on the effective date of the treaty. Payment terms depend on the type of reinsurance. For XOL treaties, brokerage revenue is typically collected in four installments during an annual treaty period based on a contractually specified minimum or deposit premium. For proportional or quota share treaties, brokerage is billed as underlying insured risks attach to the reinsurance treaty, generally over 12 to 18 months.

Consulting

The major component of revenue in the Consulting business is fees paid by clients for advice and services. Mercer, principally through its health line of business, also earns revenue in the form of commissions received from insurance companies for the placement of group (and occasionally individual) insurance contracts, primarily health, life and accident coverages. Revenue for Mercer's investment management business and certain of Mercer's defined benefit and contribution administration services consists principally of fees based on assets under delegated management or administration.

Consulting projects in Mercer's wealth and career businesses, and consulting projects in Oliver Wyman Group, typically consist of a single performance obligation, which is recognized over time as control is transferred continuously to customers. Therefore, revenue is typically recognized over time using an input measure of time expended to date relative to total estimated time to be incurred at project completion. Incurred hours represent services rendered and thereby faithfully depicts the transfer of control to the customer.

On a limited number of engagements, performance fees may also be earned for achieving certain prescribed performance criteria. Revenue for achievement is estimated and constrained to an amount that is probable to not have a significant negative adjustment.

A significant majority of fee revenues in the Consulting segment is recognized over time.

For consulting projects, Mercer generally invoices monthly in arrears with payment due within 30 days of the invoice date. Fees for delegated management services are either deducted from the net asset value of the fund or invoiced to the client on a monthly or quarterly basis in arrears. Oliver Wyman Group typically bills its clients 30 to 60 days in arrears with payment due upon receipt of the invoice.

Health brokerage and consulting services are components of both Marsh, which includes MMA, and Mercer, with approximately 59% of such revenues reported in Mercer. Health contracts typically involve a series of distinct services that are treated as a single performance obligation. Revenue for these services is recognized over time based on the amount of remuneration the Company expects to be entitled in exchange for these services. Payments for health brokerage and consulting services are typically paid monthly in arrears from carriers based on insured lives under the contract.

The following table disaggregates various components of the Company's revenue:

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Marsh:			_
EMEA (a)	\$ 2,879	\$ 2,946	\$ 2,575
Asia Pacific ^(b)	1,333	1,462	1,059
Latin America	502	453	424
Total International	4,714	4,861	4,058
U.S./Canada	5,791	5,342	4,537
Total Marsh	10,505	10,203	8,595
Guy Carpenter	2,020	1,867	1,696
Subtotal	12,525	12,070	10,291
Fiduciary interest income	120	15	46
Total Risk and Insurance Services	\$ 12,645	\$ 12,085	\$ 10,337
Mercer:			
Wealth	\$ 2,366	\$ 2,509	\$ 2,348
Health (c)	2,017	1,855	1,793
Career	962	890	787
Total Mercer	5,345	5,254	4,928
Oliver Wyman Group ^(a)	2,794	2,535	2,048
Total Consulting	\$ 8,139	\$ 7,789	\$ 6,976

⁽a) Revenue in 2022 includes the loss on deconsolidation of the Company's Russian businesses at Marsh and Oliver Wyman of \$27 million and \$12 million, respectively. Revenue in 2021 includes a net gain on the disposition of businesses of approximately \$50 million.

The following table provides contract assets and contract liabilities information from contracts with customers:

December 31, (In millions)	2022	2021	2020
Contract assets	\$ 335 \$	290 \$	236
Contract liabilities	\$ 837 \$	776 \$	676

The Company records accounts receivable when the right to consideration is unconditional, subject only to the passage of time. Contract assets primarily relate to quota share reinsurance brokerage and contingent insurer revenue. The Company does not have the right to bill and collect revenue for quota share brokerage until the underlying policies written by the ceding insurer attach to the treaty. Estimated revenue related to achievement of volume or loss ratio metrics cannot be billed or collected until all related policy placements are completed and the contingency is resolved.

Contract assets are included in other current assets in the Company's consolidated balance sheets. Contract liabilities primarily relate to the advance consideration received from customers. Contract liabilities are included in current liabilities in the Company's consolidated balance sheets.

⁽b) Revenue in 2021 includes gain on consolidation of Marsh India of \$267 million.

⁽c) Revenue in 2022 includes a net gain from the sale of the Mercer U.S. affinity business of \$112 million.

Details of the change in Contract Assets and Contract Liabilities for 2022 and 2021 are as follows:

For the Years Ended December 31,		
(In millions)	2022	2021
Contract Assets		
Balance beginning of the year	\$ 290 \$	236
Additions	661	547
Transfers to accounts receivable (a)	(614)	(493)
Effect of foreign exchange rate changes	(2)	_
Balance at the end of the year	\$ 335 \$	290
Contract Liabilities		
Balance beginning of the year	\$ 776 \$	676
Cash received for performance obligations not yet fulfilled	726	642
Revenue recognized	(640)	(539)
Effect of foreign exchange rate changes	(25)	(3)
Balance at the end of the year	\$ 837 \$	776

⁽a) Amounts transferred to accounts receivable as the rights to bill and collect became unconditional.

The amount of revenue recognized in 2022, 2021 and 2020 from performance obligations satisfied in previous periods, mainly due to variable consideration from contracts with insurers, quota share business and consulting contracts previously considered constrained was \$83 million, \$84 million, and \$97 million respectively.

The Company applies the practical expedient and does not disclose the value of unsatisfied performance obligations for (1) contracts with original contract terms of one year or less and (2) contracts where the Company has the right to invoice for services performed. The revenue expected to be recognized in future periods during the non-cancellable term of existing contracts greater than one year that is related to performance obligations that are unsatisfied or partially satisfied at the end of the reporting period is approximately \$231 million, primarily related to Mercer. The Company expects revenue in 2023, 2024, 2025, 2026 and 2027 and beyond of \$99 million, \$63 million, \$42 million, \$18 million and \$9 million, respectively, related to these performance obligations.

Costs to Obtain and Fulfill a Contract

The Company capitalizes the incremental costs to obtain contracts primarily related to commissions or sales bonus payments in both segments. These deferred costs are amortized over the expected life of the underlying customer relationships.

In Risk and Insurance Services, the Company capitalizes certain pre-placement costs that are considered fulfillment costs that meet the following criteria: these costs (1) relate directly to a contract, (2) enhance resources used to satisfy the Company's performance obligation and (3) are expected to be recovered through revenue generated by the contract. These costs are amortized at a point in time when the associated revenue is recognized.

In Consulting, the Company incurs implementation costs necessary to facilitate the delivery of the contracted services. These costs are capitalized and amortized over the initial contract term plus expected renewal periods.

At December 31, 2022, the Company's capitalized assets related to deferred implementation costs, costs to obtain and costs to fulfill were \$19 million, \$328 million and \$320 million, respectively. At December 31, 2021, the Company's capitalized assets related to deferred implementation costs, costs to obtain and costs to fulfill were \$24 million, \$290 million and \$316 million, respectively. Costs to obtain and deferred implementation costs are primarily included in other assets and costs to fulfill are primarily included in other current assets in the Company's consolidated balance sheets. The Company recorded compensation and benefits expense of \$1.6 billion, \$1.5 billion and \$1.3 billion for the years ended December 31, 2022, 2021 and 2020, respectively, related to the amortization of these capitalized assets.

A significant portion of deferred costs to fulfill in Risk and Insurance Services is amortized within 3 to 6 months. Therefore, the deferral of the cost and its amortization often occur in the same annual period.

The Company has elected to use the practical expedient and recognizes the incremental costs of obtaining contracts as an expense when incurred if the amortization period of the assets is one year or less.

3. Supplemental Disclosures

The following table provides additional information concerning acquisitions, interest and income taxes paid:

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Assets acquired, excluding cash, and cash and cash equivalents held in a fiduciary capacity	\$ 734	\$ 1,697	\$ 929
Acquisition-related deposit	24		_
Fiduciary liabilities assumed	(6)	(18)	(21)
Liabilities assumed	(49)	(213)	(78)
Non-controlling interests assumed	(5)	(64)	_
Fair value of previously-held equity method investment	(6)	(390)	_
Contingent/deferred purchase consideration	(120)	(153)	(183)
Net cash outflow for acquisitions	\$ 572	\$ 859	\$ 647
(In millions)	2022	2021	2020
Interest paid	\$ 431	\$ 441	\$ 481
Income taxes paid, net of refunds	\$ 1,049	\$ 1,069	\$ 673

The classification of contingent consideration in the consolidated statement of cash flows is dependent upon whether the receipt or payment was part of the initial liability established on the acquisition date (financing) or an adjustment to the acquisition date liability (operating).

The following amounts are included in the consolidated statements of cash flows as operating and financing activities:

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Operating:			
Contingent consideration payments for prior year acquisitions	\$ (38)	\$ (49)	\$ (48)
Receipt of contingent consideration for dispositions	_	19	_
Acquisition/disposition related net charges for adjustments	49	57	26
Adjustments and payments related to contingent consideration	\$ 11	\$ 27	\$ (22)
Financing:			
Contingent consideration for prior year acquisitions	\$ (32)	\$ (28)	\$ (54)
Deferred consideration related to prior year acquisitions	(126)	(89)	(68)
Payments of deferred and contingent consideration for acquisitions	\$ (158)	\$ (117)	\$ (122)
Receipt of contingent consideration for dispositions	\$ 3	\$ 71	\$ _

The Company had non-cash issuances of common stock under its share-based payment plan of \$372 million, \$228 million and \$219 million in 2022, 2021 and 2020, respectively. The Company recorded share-based compensation expense related to restricted stock units, performance stock units and stock options of \$367 million, \$348 million and \$290 million in 2022, 2021 and 2020, respectively.

Statement of Cash Flows Reclassifications

In 2022, the Company refined the statements of cash flows presentation to combine and reclassify certain line items in the operating cash flows section and separately present purchases and sales of long-term investments in the investing cash flows section. The prior year presentation was conformed to the current presentation with no impact on either operating or investing cash flows.

Allowance for Credit Losses on Accounts Receivable

The Company's policy for providing an allowance for credit losses on its accounts receivable is a combination of factors, including historical write-offs, aging of balances, and other qualitative and quantitative analyses.

An analysis of the allowance for credit losses is provided below:

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Balance at beginning of year	\$ 166	\$ 142	\$ 140
Provision charged to operations	17	46	47
Accounts written-off, net of recoveries	(17)	(16)	(30)
Effect of exchange rate changes and other	(6)	(6)	(15)
Balance at end of year	\$ 160	\$ 166	\$ 142

4. Accumulated Other Comprehensive Income (Loss)

The changes, net of tax, in the balances of each component of AOCI for the years ended December 31, 2022 and 2021, including amounts reclassified out of AOCI, are as follows:

(In millions)	Pension and Post- Retirement Plans Losses			oreign Currency Translation Adjustments	Total
Balance as of January 1, 2022	\$	(3,202)	\$	(1,373)	\$ (4,575)
Other comprehensive income (loss) before reclassifications		371		(1,220)	(849)
Amounts reclassified from accumulated other comprehensive loss		110		_	110
Net current period other comprehensive income (loss)		481		(1,220)	(739)
Balance as of December 31, 2022	\$	(2,721)	\$	(2,593)	\$ (5,314)

(In millions)	-	ension and Post- Retirement Plans Losses	F	oreign Currency Translation Adjustments	Total
Balance as of January 1, 2021	\$	(4,126)	\$	(984)	\$ (5,110)
Other comprehensive gain (loss) before reclassifications		765		(389)	376
Amounts reclassified from accumulated other comprehensive loss		159		_	159
Net current period other comprehensive income (loss)	<u> </u>	924		(389)	535
Balance as of December 31, 2021	\$	(3,202)	\$	(1,373)	\$ (4,575)

The components of other comprehensive (loss) income for the years ended December 31, 2022, 2021 and 2020 are as follows:

For the Year Ended December 31,	2022			
(In millions)	Pre-Tax	Tax	Net of Tax	
Foreign currency translation adjustments	\$ (1,198) \$	22	2 \$	(1,220)
Pension/post-retirement plans:				
Amortization of (gains) losses included in net benefit (credit) cost:				
Prior service credits (a)	(2)	_	-	(2)
Net actuarial losses (a)	150	38	3	112
Effect of settlement (a)	2	_	-	2
Subtotal	150	38	3	112
Net gains arising during period	203	52	L	152
Foreign currency translation adjustments	285	73	L	214
Other adjustments	3	_	-	3
Pension/post-retirement plans gains	641	160)	481
Other comprehensive (loss) income	\$ (557) \$	182	2 \$	(739)

⁽a) Included in other net benefit credits in the consolidated statements of income. Income tax expense on net actuarial losses is included in income tax expense.

For the Year Ended December 31,	2021			
(In millions)	Pre-Tax	Tax (Credit)	Net of Tax	
Foreign currency translation adjustments	\$ (389) \$	— \$	(389)	
Pension/post-retirement plans:				
Amortization of (gains) losses included in net benefit (credit) cost:				
Prior service credits (a)	(2)	_	(2)	
Net actuarial losses (a)	208	52	156	
Effect of curtailment ^(a)	2	1	1	
Effect of settlement (a)	5	1	4	
Subtotal	213	54	159	
Net losses arising during period	1,003	249	754	
Foreign currency translation adjustments	19	4	15	
Other adjustments	(6)	(2)	(4)	
Pension/post-retirement plans gains	1,229	305	924	
Other comprehensive income	\$ 840 \$	305 \$	535	

⁽a) Included in other net benefit credits in the consolidated statements of income. Income tax expense on net actuarial losses is included in income tax expense.

For the Year Ended December 31,	2020			
(In millions)	Pre-Tax Tax (Credit)		Net of Tax	
Foreign currency translation adjustments	\$ 559 \$	— \$	559	
Pension/post-retirement plans:				
Amortization of (gains) losses included in net benefit (credit) cost:				
Prior service credits (a)	(2)	(1)	(1)	
Net actuarial losses ^(a)	161	37	124	
Effect of settlement (a)	3	1	2	
Subtotal	162	37	125	
Net losses arising during period	(772)	(177)	(595)	
Foreign currency translation adjustments	(163)	(28)	(135)	
Other adjustments	(11)	(2)	(9)	
Pension/post-retirement plans losses	(784)	(170)	(614)	
Other comprehensive loss	\$ (225) \$	(170) \$	(55)	

⁽a) Included in other net benefit credits in the consolidated statements of income. Income tax expense on net actuarial losses is included in income tax expense.

The components of accumulated other comprehensive loss are as follows:

(In millions)	December 31, 2022	December 31, 2021
Foreign currency translation adjustments (net of deferred tax liability of \$8 in 2022 and deferred tax asset of \$13 in 2021, respectively)	\$ (2,593)	\$ (1,373)
Net charges related to pension/post-retirement plans (net of deferred tax asset of \$1,340 and \$1,501 in 2022 and 2021, respectively)	(2,721)	(3,202)
	\$ (5,314)	\$ (4,575)

5. Acquisitions and Dispositions

The Company's acquisitions have been accounted for as business combinations. Net assets and results of operations are included in the Company's consolidated financial statements commencing at the respective purchase closing dates. In connection with acquisitions, the Company records the estimated values of the net tangible assets and the identifiable intangible assets purchased, which typically consist of customer relationships, developed technology, trademarks and non-compete agreements. The valuation of purchased intangible assets involves significant estimates and assumptions. The Company estimates the fair value of purchased intangible assets, primarily using the income approach, by determining the present value of future cash flows over the remaining economic life of the respective assets. The significant estimates and assumptions used in this approach include the determination of the discount rate, economic life, future revenue growth rates, expected account attrition rates and earnings margins. Refinement and completion of final valuation of net assets acquired could affect the carrying value of tangible assets, goodwill and identifiable intangible assets.

The Risk and Insurance Services segment completed 16 acquisitions in 2022:

- January MMA acquired Heil & Kay Insurance Agency Inc., an Illinois-based full-service broker providing business insurance, employee health benefits services and personal lines insurance.
- April Marsh acquired the business of Regional Treaty Services Corporation, a Rhode Island-based managing general underwriter, which manages reinsurance facilities for small to midsize U.S.-based insurers primarily writing personal lines, small agriculture, and main street commercial business.
- June MMA acquired Clark Insurance, a Maine-based full-service broker providing business insurance, employee health and benefits and private client services to businesses and individuals across the region.
- July MMA acquired CS Insurance Strategies, Inc., an Illinois-based full-service broker providing employee health and benefits, business insurance, and risk management consulting services to organizations of all sizes across the U.S. and Suchanek Partners LLC, an Ohio-based employee benefits insurance broker.
- August Marsh acquired Best Insurance Co. Ltd, a Japan-based insurance broker that provides affinity type schemes, general and personal lines insurance.
- September MMA acquired Steinberg & Associates, Inc., a South Carolina-based insurance broker that primarily offers employee health benefit services to group clients and Leykell, Inc., a Texas-based full-service broker that provides specialty insurance focused on trade credit.
- October MMA acquired Galbraith Group, a Texas-based employee health and benefits insurance broker.
- November MMA acquired Focus Insurance and Financial Services, a Texas-based personal insurance broker and Bradley
 Insurance Agency, a commercial insurance broker in Knoxville, Tennessee, with expertise serving the hospitality and construction
 industries. Marsh increased its ownership interest in Beassur SARL, a Morocco-based multi-line insurance broker, from 35% to 70%.
- December MMA acquired McDonald-Zaring Insurance, Inc., a Washington-based full-service broker focused on agri-business, wineries, crops and contractors, Chartwell Insurance Brokers, Inc., a Massachusetts-based full-service broker that specializes in commercial Property & Casualty insurance in the technology, financial services and non-profit space, and HMS Insurance Associates, Inc., a Maryland-based full-service broker providing commercial, surety, employee benefits, and personal lines insurance. Marsh acquired BHM Consultores S.A., d/b/a Grupo Mesos, a leading auto affinity insurance broker specialist in Chile that has extensive distribution partnerships with car dealerships, original equipment manufacturers and auto finance companies.

The Consulting segment completed 4 acquisitions in 2022:

- February Oliver Wyman acquired Azure Consulting, an Australia-based management consulting firm with expertise in strategy development, organizational design and operations in the industrials, energy and natural resources sectors.
- March Mercer acquired GeFi Assurances, a France-based brokerage and consulting firm specializing in collective corporate social protection.
- September Oliver Wyman acquired Booz Allen Hamilton's strategy consulting business serving the Middle East and North Africa.
- November Oliver Wyman acquired the Avascent Group Ltd, an aerospace and defense management consulting firm focused on the corporate and private equity sectors based in the U.S., U.K., Canada and France.

Total purchase consideration for acquisitions made during 2022 was \$705 million, which consisted of cash paid of \$579 million, deferred purchase and estimated contingent consideration of \$120 million, and the fair value of a previously-held equity method investment of \$6 million. Contingent consideration arrangements are based primarily on earnings before interest, tax, depreciation and amortization ("EBITDA") or revenue targets over a period of 2 to 4 years. In 2022, the Company also paid \$126 million of deferred purchase consideration and \$70 million of contingent consideration related to acquisitions made in prior years. Estimated fair values of assets acquired and liabilities assumed are subject to adjustment until purchase accounting is finalized.

The following table presents the preliminary allocation of purchase consideration to the assets acquired and liabilities assumed in 2022, based on the estimated fair values for the acquisitions as of their respective acquisition dates.

Acquisitions for the Year Ended December 31, 2022		
(In millions) Cash	\$	579
Estimated fair value of deferred/contingent consideration	Ψ	120
Fair value of previously-held equity method investment		6
Total consideration	<u> </u>	705
Allocation of purchase price:	·	
Cash and cash equivalents	\$	25
Cash and cash equivalents held in a fiduciary capacity		6
Net receivables		32
Other current assets		3
Goodwill		460
Other intangible assets		228
Fixed assets, net		3
Right of use assets		7
Other assets		1
Total assets acquired		765
Current liabilities		35
Fiduciary liabilities		6
Long-term lease liabilities		7
Other liabilities		7
Total liabilities assumed		55
Non-controlling interests		5
Net assets acquired	\$	705

Note: Amounts in the table above primarily reflect the acquisition of HMS Insurance Associates, Inc.

The purchase price allocation for assets acquired and liabilities assumed is based on estimates that are preliminary in nature and subject to adjustments, which could be material. Any necessary adjustments must be finalized during the measurement period, which for a particular asset, liability, or non-controlling interest ends once the acquirer determines that either (1) the necessary information has been obtained or (2) the information is not available. However, the measurement period for all items is limited to one year from the acquisition date.

Items subject to change include:

- amounts of intangible assets, fixed assets, capitalized software assets and right-of-use assets, subject to finalization of valuation efforts;
- amounts for contingencies, pending the finalization of the Company's assessment of the portfolio of contingencies;
- amounts for deferred tax assets and liabilities pending the finalization of valuations of the assets acquired, liabilities assumed and associated goodwill discussed below; and
- amounts for income tax assets, receivables and liabilities, pending the filing of the acquired companies' pre-acquisition income tax returns and receipt of information from taxing authorities which may change certain estimates and assumptions used.

The estimation of fair value requires numerous judgments, assumptions and estimates about future events and uncertainties, which could materially impact these values, and the related amortization, where applicable, in the Company's results of operations.

The following table provides information about other intangible assets acquired in 2022:

(In millions)	Amount	Weighted Average Amortization Period
Customer relationships	\$ 219	12.7 years
Other	9	2.3 years
Total other intangible assets	\$ 228	

The consolidated statements of income include the results of operations of acquired companies since their respective acquisition dates. The consolidated statement of income for 2022 includes approximately \$58 million of revenue and an operating loss of \$5 million related to acquisitions made in 2022. The consolidated statement of income for 2021 includes approximately \$114 million of revenue and \$3 million of operating income related to acquisitions made in 2021, and the consolidated statement of income for 2020 includes approximately \$169 million of revenue and \$11 million of operating income related to acquisitions made in 2020.

Pending transactions with Westpac Banking Corporation

In May 2022, the Company entered into an agreement with Westpac Banking Corporation ("Westpac") to acquire one of its financial advisory businesses, Advance Asset Management. Westpac will also transfer BT Financial Group's personal and corporate pension funds to the Mercer Super Trust managed by Mercer Australia (collectively, the "Transaction"). The Transaction is expected to be completed in the first half of 2023, subject to regulatory approvals. In 2022, the Company incurred approximately \$21 million of integration expenses, primarily for technology, consulting, legal and people related costs.

Dispositions

In April 2022, Mercer sold its U.S. affinity business that provided insurance marketing, brokerage and administration to association and affinity groups for cash proceeds of approximately \$140 million and recorded a net gain of \$112 million which is included in revenue in the consolidated statements of income.

In addition, in 2022, the Company sold certain businesses in the U.K., the Czech Republic, Brazil and Belgium for cash proceeds of approximately \$15 million and recorded a net gain of \$15 million.

The cash proceeds received were partially offset by \$36 million primarily related to cash and cash equivalents held in a fiduciary capacity in the disposed businesses.

Deconsolidation of Russia

In the first quarter of 2022, the Company concluded that it does not meet the accounting criteria for control over its wholly-owned Russian subsidiaries due to the evolving trade and economic sanctions against Russia and the related Russian counter sanctions. These sanctions included restrictions on payments to and from Russian companies and reduced currency access through official exchange markets that have significantly impacted the Company's ability to effectively manage and operate its Russian businesses.

As a result, the Company deconsolidated its Russian businesses effective as of the end of the first quarter, and recorded a loss of \$39 million included in revenue in the consolidated statements of income. The loss consisted of the reclassification of cumulative translation losses from accumulated other comprehensive income and a charge for the write-off of the Russia businesses' net assets.

In June 2022, the Company entered into a definitive agreement to exit its businesses in Russia and transfer ownership to local management, pending regulatory approvals.

Prior year acquisitions

The Risk and Insurance Services segment completed 8 acquisitions in 2021:

- April MMA acquired PayneWest Insurance, Inc., a Montana-based full-service broker providing business insurance, surety, employee benefits and personal insurance services to companies and individuals, and The Pryor Group, LLC, a Texas-based full-service broker providing business insurance with a specialty in quick service restaurants and the personal lines of franchise owners.
- September MMA acquired Vaaler Insurance, Inc., a North Dakota-based insurance broker providing business insurance, employee health and benefits, and personal lines solutions, with specialized expertise in the construction, education, and healthcare industries.
- November MMA acquired Pelnik Insurance, a North Carolina-based full-service broker providing business insurance, employee health and benefits, and private client services to midsize businesses and individuals throughout the Mid-Atlantic, Southwest Truck Insurance Agency, Inc., a Texas-based broker providing business insurance for the trucking industry, serving clients in the U.S., and Mexico and InSource Insurance Group LLC, a Texas-based full-service broker providing business insurance, employee health and benefits, private client and surety services to the oil and gas, construction, manufacturing, and transportation industries.
- December Marsh acquired Services Assurance Monétique (SAM), a France-based affinity insurance broker specializing in bank and retail insurance markets and increased its ownership interest in Marsh India Insurance Broker Private Limited ("Marsh India") from 49% to 92%.

The Consulting segment completed 1 acquisition in 2021:

November – Oliver Wyman Group acquired Huron Consulting Group's life sciences strategy consulting practice in the U.S. and the
U.K., which assists clients in addressing their most important commercial strategy, marketing, pricing, market access and research
and development challenges.

Total purchase consideration for acquisitions made in 2021 was approximately \$1.4 billion, which consisted of cash paid of \$888 million and deferred purchase and estimated contingent consideration of \$153 million and the fair value of a previously held equity method investment in Marsh India of \$390 million. Contingent consideration arrangements are based primarily on EBITDA or revenue targets over periods of 2 to 4 years. The fair value of the contingent consideration was based on projected revenue and earnings of the acquired entities. In 2021, the Company also paid \$89 million of deferred purchase consideration and \$77 million of contingent consideration related to acquisitions made in prior years. Estimated fair values of assets acquired and liabilities assumed are subject to adjustment when purchase accounting is finalized.

In December 2021, in connection with its increased investment in Marsh India, the Company recorded a gain of \$267 million related to the remeasurement and consolidation of its previously held equity method investment to fair value. The fair value of the pre-existing equity method investment was calculated using an average of applying an income approach based on discounted future cash flows and market approach.

Prior year dispositions

In 2021, the Company sold certain businesses, primarily in the U.S. and the U.K., for cash proceeds of approximately \$84 million and recognized a net gain of approximately \$50 million, primarily related to the commercial networks business in the U.K. that provided broking and back-office solutions for small independent brokers.

Pro-Forma Information

The following unaudited pro-forma financial data gives effect to the acquisitions made by the Company in 2022, 2021 and 2020. In accordance with accounting guidance related to pro-forma disclosures, the information presented for current year acquisitions is as if they occurred on January 1, 2021 and reflects acquisitions made in 2021 as if they occurred on January 1, 2020. The 2020 information includes 2020 acquisitions as if they occurred on January 1, 2019. The un-audited pro-forma information includes the effects of amortization of acquired intangibles in all years. The unaudited pro-forma financial data is presented for illustrative purposes only and is not necessarily indicative of the operating results that would have been achieved if such acquisitions had occurred on the dates indicated, nor is it necessarily indicative of future consolidated results.

For the Years Ended December 31,			
(In millions, except per share data)	2022	2021	2020
Revenue	\$ 20,880	\$ 20,220	\$ 17,586
Net income attributable to the Company	\$ 3,078	\$ 3,177	\$ 2,042
Basic net income per share attributable to the Company	\$ 6.16	\$ 6.27	\$ 4.03
Diluted net income per share attributable to the Company	\$ 6.10	\$ 6.20	\$ 3.99

6. Goodwill and Other Intangibles

The Company is required to assess goodwill and any indefinite-lived intangible assets for impairment annually, or more frequently if circumstances indicate impairment may have occurred. The Company performs the annual impairment assessment for each of its reporting units during the third quarter of each year. In accordance with applicable accounting guidance, a company can assess qualitative factors to determine whether it is necessary to perform a quantitative goodwill impairment test. Alternatively, the Company may elect to proceed directly to the quantitative goodwill impairment test. In the third quarter of 2022, the Company completed a qualitative impairment assessment and concluded that goodwill was not impaired. As part of its assessment, the Company considered numerous factors, including:

- that the fair value of each reporting unit exceeds its carrying value by a substantial margin based on its most recent quantitative assessment in 2019:
- whether significant acquisitions or dispositions occurred which might alter the fair value of its reporting units;
- · macroeconomic conditions and their potential impact on reporting unit fair values;
- actual performance compared with budget and prior projections used in its estimation of reporting unit fair values;
- · industry and market conditions; and
- the year-over-year change in the Company's share price.

Other intangible assets that are not deemed to have an indefinite life are amortized over their estimated lives and assessed for impairment upon the occurrence of certain triggering events in accordance with applicable accounting literature. Based on its assessment, the Company concluded that other intangible assets were not impaired. The Company had no indefinite lived identified intangible assets at December 31, 2022 and 2021.

Changes in the carrying amount of goodwill are as follows:

(In millions)	2022	2021
Balance as of January 1	\$ 16,317	\$ 15,517
Goodwill acquired	460	1,045
Other adjustments (a)	(526)	(245)
Balance at December 31,	\$ 16,251	\$ 16,317

⁽a) Primarily reflects the impact of foreign exchange and dispositions.

The goodwill arising from acquisitions in 2022 and 2021 consists largely of the synergies and economies of scale expected from combining the operations of the Company and the acquired entities and the trained assembled workforce acquired.

The goodwill acquired in 2022 included approximately \$348 million and \$64 million in the Risk and Insurance Services and Consulting segments, respectively, which is deductible for tax purposes. The goodwill acquired in 2021 included approximately \$96 million, primarily related to the Risk and Insurance Service segment, which is deductible for tax purposes.

Goodwill allocable to the Company's reportable segments as of December 31, 2022, is \$12.5 billion for Risk and Insurance Services and \$3.8 billion for Consulting.

The gross cost and accumulated amortization of other intangible assets at December 31, 2022 and 2021 are as follows:

(In millions)		2022			2021	
	Gross Cost	Accumulated Amortization	Net Carrying Amount	Gross Cost	Accumulated Amortization	Net Carrying Amount
Customer relationships	\$ 3,993	\$ 1,508	\$ 2,485	\$ 4,066	\$ 1,334	\$ 2,732
Other ^(a)	360	308	52	365	287	78
Other intangible assets	\$ 4,353	\$ 1,816	\$ 2,537	\$ 4,431	\$ 1,621	\$ 2,810

⁽a) Primarily non-compete agreements, trade names and developed technology.

Aggregate amortization expense was \$338 million, \$365 million, and \$351 million for the years ended December 31, 2022, 2021 and 2020, respectively. The estimated future aggregate amortization expense is as follows:

For the Years Ended December 31, (In millions)	
2023	\$ 327
2024	307
2025	274
2026	253
2027	249
Subsequent years	1,127
Total future amortization	\$ 2,537

7. Income Taxes

For financial reporting purposes, income before income taxes includes the following components:

(In millions)		2022	2021		2020
Income before income taxes:					
U.S.	\$	1,468	\$ 1,590	\$	1,075
Other		2,614	2,618		1,718
	\$	4,082	\$ 4,208	\$	2,793
The expense (benefit) for income taxes is comprised of:					
Current –					
U.S. Federal	\$	262	\$ 251	\$	172
Other national governments		653	714		456
U.S. state and local		123	132		79
		1,038	1,097		707
Deferred –					
U.S. Federal		38	(40)		40
Other national governments		(91)	(12)		(14
U.S. state and local		10	(11)		14
		(43)	(63)		40
Total income taxes	\$	995	\$ 1,034	\$	747
The significant components of deferred income tax assets and December 31,	liabilities and their balance	e sheet class		llows:	
(In millions)			2022		2021
Deferred tax assets:					
Accrued expenses not currently deductible (a)		\$	670 \$		647
Differences related to non-U.S. operations ^(b)			275		293
Accrued U.S. retirement benefits			172		293
Net operating losses ^(c)			285		183
Income currently recognized for tax			34		29
Other			25		32
		\$	1,461 \$		1,477
Deferred tax liabilities:					
Deferred tax liabilities: Differences related to non-U.S. operations		\$	543 \$		624

Capitalized expenses currently recognized for tax

Depreciation and amortization

Other

Accrued retirement & postretirement benefits - non-U.S. operations

\$

\$

506

403

98

32

1,663

510

408

107

1,625

57

 $^{^{(}a)}$ Net of valuation allowances of \$5 million in 2022 and \$2 million in 2021.

 $^{^{(}b)}$ Net of valuation allowances of \$160 million in 2022 and \$144 million in 2021.

⁽c) Net of valuation allowances of \$69 million in 2022 and \$88 million in 2021.

December 31,		
(In millions)	2022	2021
Balance sheet classifications:		
Deferred tax assets	\$ 358 \$	551
Other liabilities	\$ 522 \$	737

The amount of cumulative undistributed earnings that are indefinitely reinvested in non-U.S. subsidiaries is approximately \$830 million as of December 31, 2022. While no additional U.S. federal income tax would be required if such earnings were repatriated, additional state and withholding taxes would apply. The amount of these additional taxes is estimated to be approximately \$80 million.

Future U.S. federal tax costs related to basis differences in non-U.S. subsidiaries would primarily be realized through the U.S. Global Intangible Low-Taxed Income ("GILTI") minimum tax regime. The Company elected to recognize GILTI tax costs as a period cost and has not provided deferred tax liabilities on these basis differences.

A reconciliation from the U.S. federal statutory income tax rate to the Company's effective income tax rate is shown below:

For the Years Ended December 31,	2022	2021	2020
U.S. Federal statutory rate	21.0 %	21.0 %	21.0 %
U.S. state and local income taxes—net of U.S. Federal income tax benefit	2.7	2.3	2.5
Differences related to non-U.S. operations	0.7	0.1	2.3
U.K. statutory rate change	_	2.6	_
Gain on consolidation of business	_	(1.5)	_
Equity compensation	(0.7)	(0.7)	(1.4)
Uncertain tax positions	0.1	0.1	1.1
Other	0.6	0.7	1.2
Effective tax rate	24.4 %	24.6 %	26.7 %

The rates in all periods reflect the effects of tax planning and the ongoing impact of regulatory and other guidance as it became available. The tax rates in all periods include a valuation allowance for certain tax credits, the impact of uncertain tax positions, and certain tax planning benefits. The tax rate in 2021 also included the effect of a statutory rate change in the U.K. and the tax effect of a gain from the fair value remeasurement of the Company's previously held equity method investment in Marsh India upon the Company increasing its ownership interest from 49% to 92%. The Company does not intend to dispose the business and has indefinitely reinvested this gain.

A valuation allowance was recorded to adjust deferred tax assets to the amount that the Company believes is more likely than not to be realized. Valuation allowances had a net decrease of \$1 million in 2022 and increases of \$36 million and \$72 million in 2021, and 2020, respectively. Adjustments of the beginning of the year balances of valuation allowances decreased income tax expense by \$5 million in 2022. Adjustments of the beginning of the year valuation allowances increased tax expense by \$2 million in 2021 and decreased income tax expense by \$14 million in 2020. Approximately 14% of the Company's net operating loss carryforwards expire from 2023 through 2038, and others are unlimited. The potential tax benefit from net operating loss carryforwards at the end of 2022 comprised of federal, state and local, and non-U.S. tax benefits of \$5 million, \$13 million, and \$342 million, respectively, before reduction for valuation allowances.

Following is a reconciliation of the Company's total gross unrecognized tax benefits for the years ended December 31, 2022, 2021 and 2020:

(In millions)	2022	2021	2020
Balance at January 1,	\$ 94 \$	98 \$	86
Additions, based on tax positions related to current year	1	2	9
Additions for tax positions of prior years	15	11	25
Reductions for tax positions of prior years	(2)	(1)	(9)
Settlements	(2)	(1)	(4)
Lapses in statutes of limitations	(9)	(15)	(9)
Balance at December 31,	\$ 97 \$	94 \$	98

Of the total unrecognized tax benefits at December 31, 2022, 2021, and 2020, \$94 million, \$87 million and \$90 million, respectively, represent the amount that, if recognized, would favorably affect the effective tax rate in any future periods. The total gross amount of accrued interest and penalties at December 31, 2022, 2021 and 2020, before any applicable federal benefit, was \$48 million, \$45 million, and \$40 million, respectively.

The Company is routinely examined by the jurisdictions in which it has significant operations. In the U.S. federal jurisdiction, the Company participates in the Internal Revenue Service's ("IRS") Compliance Assurance Process ("CAP"), which is structured to be, in effect, a real-time audit. In 2021, the IRS concluded its examination of the Company's 2017, 2018, and 2019 tax returns. The Company was accepted into the Bridge phase of the CAP program for tax years 2020 and 2021, and generally will not be audited by the IRS for those years. In 2022, the IRS began its pre-filing examination of the Company's 2022 tax year.

New York is a significant tax jurisdiction for the Company. New York State and New York City have continuing examinations underway in 2022 and 2021 for various entities covering the years 2010 through 2018. The New York State audits for 2010-2012 were finalized in 2022.

We conduct business through multiple legal entities in significant jurisdictions outside the U.S. Separate audits for individual entities within a jurisdiction may open or close within a particular year.

The status of audits for significant jurisdictions outside the U.S. are summarized in the table below:

Tax Audit (Years)			
Jurisdiction:	Initiated in 2022	Ongoing	Concluded
France			2017, 2018 during 2022
Germany		2013-2016	2017-2018 during 2022
Hong Kong			2019 during 2022
Italy		2015-2016	2017 during 2022
Singapore	2018-2021	2017-2021	2018-2020 during 2022
United Kingdom	2020	2016-2020	

The Company has established liabilities for uncertain tax positions in relation to potential assessments in the jurisdictions in which it operates. The Company believes the resolution of tax matters will not have a material effect on the consolidated financial position of the Company, although a resolution of tax matters could have a material impact on the Company's net income or cash flows and on its effective tax rate in a particular future period. It is reasonably possible that the total amount of unrecognized tax benefits could decrease up to approximately \$42 million within the next 12 months due to settlement of audits and expiration of statutes of limitations.

8. Retirement Benefits

The Company maintains qualified and non-qualified defined benefit pension plans for its U.S. and non-U.S. eligible employees.

Combined U.S. and Non-U.S. Plans

The weighted average actuarial assumptions utilized for the U.S. and significant non-U.S. defined benefit plans and post-retirement benefit plans are as follows:

	Pension Benefits		Post-retirem Benefits	ent
	2022	2021	2022	2021
Weighted average assumptions:				
Discount rate (for expense)	2.28 %	1.92 %	2.36 %	2.42 %
Expected return on plan assets	4.57 %	4.73 %	_	_
Rate of compensation increase (for expense)*	3.34 %	1.85 %	_	_
Discount rate (for benefit obligation)	5.16 %	2.28 %	4.92 %	2.36 %
Rate of compensation increase (for benefit obligation)*	3.16 %	2.16 %	_	_

^{*}There are no rate of compensation increase assumptions for the U.S. defined benefit plans since future benefit accruals were discontinued for those plans after December 31, 2016 and earned benefits are not subject to final salary level adjustments.

The target asset allocation for the U.S. plans is 50% equities and equity alternatives and 50% fixed income. At December 31, 2022, the actual allocation for the U.S. plans was 61% equities and equity alternatives and 39% fixed income. The target asset allocation for the U.K. plans, which comprise approximately 79% of non-U.S. plan assets, is 14% equities and equity alternatives and 86% fixed income. At December 31, 2022, the actual allocation for the U.K. plans was 16% equities and equity alternatives and 84% fixed income. The assets of the Company's defined benefit plans are diversified and are managed in accordance with applicable laws and with the goal of maximizing the plans' real return within acceptable risk parameters. The Company uses threshold-based portfolio re-balancing to ensure the actual portfolio remains consistent with target asset allocation ranges.

The net benefit (credit) or cost of the Company's defined benefit and other post-retirement plans is measured on an actuarial basis using various methods and assumptions. The components of the net benefit (credit) or cost for the years 2022, 2021 and 2020 are as follows:

Combined U.S. and significant non-U.S. Plans			Pe	nsion			Р	ost-r	etireme	nt	
For the Years Ended December 31,	Benefits							Be	enefits		
(In millions)		2022		2021		2020	2022		2021		2020
Service cost	\$	28	\$	38	\$	36	\$ _	\$	1	\$	_
Interest cost		389		341		421	3		2		3
Expected return on plan assets		(778)		(832)		(844)	_		_		_
Amortization of prior service		1		_		_	(2)		(2)		(2)
Recognized actuarial loss (gain)		149		206		161	1		1		_
Net periodic benefit (credit) cost		(211)		(247)		(226)	2		2		1
Curtailment loss		_		2			_		_		_
Plan termination		_		_		1	_		_		_
Settlement loss		2		5		3	_		_		_
Net benefit (credit) cost	\$	(209)	\$	(240)	\$	(222)	\$ 2	\$	2	\$	1

The following table provides the amounts reported in the consolidated statements of income:

Combined U.S. and significant non-U.S. Plans For the Years Ended December 31,		ension enefits			retirement enefits	
(In millions)	2022	2021	2020	2022	2021	2020
Compensation and benefits expense	\$ 28 \$	38 \$	36 \$	— \$	1 \$	_
Other net benefit (credit) cost	(237)	(278)	(258)	2	1	1
Net benefit (credit) cost	\$ (209) \$	(240) \$	(222) \$	2 \$	2 \$	1

Plan Assets

For the U.S. plans, investment allocation decisions are made by a fiduciary committee composed of senior executives appointed by the Company's Chief Executive Officer. For the non-U.S. plans, investment allocation decisions are made by local fiduciaries, in consultation with the Company for the larger plans. Plan assets are invested in a manner consistent with the fiduciary standards set forth in all relevant laws relating to pensions and trusts in each country. Primary investment objectives are (1) to achieve an investment return that, in combination with current and future contributions, will provide sufficient funds to pay benefits as they become due, and (2) to minimize the risk of large losses. The investment allocations are designed to meet these objectives by broadly diversifying plan assets among numerous asset classes with differing expected returns, volatilities, and correlations.

The major categories of plan assets include equity securities, equity alternative investments, and fixed income securities. For the U.S. plan, the category ranges are 46%-54% for both equities and equity alternatives, and for fixed income. For the U.K. plans, the category ranges are 11%-17% for equities and equity alternatives, and 83%-89% for fixed income. Asset allocation is frequently monitored and re-balancing actions are taken as appropriate.

Plan investments are exposed to stock market, interest rate, and credit risk. Concentrations of these risks are generally limited due to diversification by investment style within each asset class, diversification by investment manager, diversification by industry sectors and issuers, and the dispersion of investments across many geographic areas.

U.S. Plans
The following tables provide information concerning the Company's U.S. defined benefit pension and post-retirement benefit plans:

	U.S. P Ben	on	U.S. Post- Ben	retire efits	ment
(In millions)	2022	2021	2022		2021
Change in benefit obligation:					
Benefit obligation at beginning of year	\$ 6,594	\$ 6,914	\$ 28	\$	31
Interest cost	193	184	1		1
Employee contributions	_	_	3		4
Plan amendments	_	1	_		_
Actuarial gain	(1,625)	(227)	(3)		(1)
Benefits paid	(286)	(278)	(7)		(7)
Benefit obligation, December 31	\$ 4,876	\$ 6,594	\$ 22	\$	28
Change in plan assets:					
Fair value of plan assets at beginning of year	\$ 5,537	\$ 5,100	\$ 2	\$	2
Actual return on plan assets	(1,005)	680	_		_
Employer contributions	30	35	4		3
Employee contributions	_	_	3		4
Benefits paid	(286)	(278)	(7)		(7)
Fair value of plan assets, December 31	\$ 4,276	\$ 5,537	\$ 2	\$	2
Net funded status, December 31	\$ (600)	\$ (1,057)	\$ (20)	\$	(26)
Amounts recognized in the consolidated balance sheets:					
Current liabilities	\$ (31)	\$ (31)	\$ (1)	\$	(1)
Non-current liabilities	(569)	(1,026)	(19)		(25)
Net liability recognized, December 31	\$ (600)	\$ (1,057)	\$ (20)	\$	(26)
Amounts recognized in other comprehensive income (loss):					
Prior service cost	\$ (1)	\$ (1)	\$ _	\$	_
Net actuarial (loss) gain	(1,419)	(1,777)	8		3
Total recognized accumulated other comprehensive (loss) income, December 31	\$ (1,420)	\$ (1,778)	\$ 8	\$	3
Cumulative employer contributions in excess of (less than) net benefit (credit) cost	820	721	(28)		(29)
Net amount recognized in consolidated balance sheet	\$ (600)	\$ (1,057)	\$ (20)	\$	(26)
Accumulated benefit obligation at December 31	\$ 4,876	\$ 6,594	\$ _	\$	

	U.S. F Ber	ensionefits	on	U.S. Post- Ber	retire efits	ment
(In millions)	2022		2021	2022		2021
Reconciliation of net actuarial (loss) gain recognized in accumulated other comprehensive income (loss):						
Beginning balance	\$ (1,777)	\$	(2,446)	\$ 3	\$	3
Recognized as component of net benefit (credit) cost	74		90	_		(1)
Changes in plan assets and benefit obligations recognized in other comprehensive income (loss):						
Other	_		(1)	2		_
Liability experience	1,625		227	3		1
Asset experience	(1,341)		353	_		_
Total gain recognized as change in plan assets and benefit obligations	284		579	5		1
Net actuarial (loss) gain, December 31	\$ (1,419)	\$	(1,777)	\$ 8	\$	3

For the Years Ended December 31,	U₊S. Pension Benefits						U.S.	S. Post-retirement Benefits					
(In millions)		2022		2021	2020		2022	2021	2020				
Total recognized in net benefit (credit) cost and other comprehensive (income) loss	\$	(427)	\$	(722)	\$ 272	\$	(4) \$	s — \$	2				

The weighted average actuarial assumptions utilized in determining expense during the year and benefit obligation at the end of the year for the U.S. defined benefit and other U.S. post-retirement plans are as follows:

	U.S. Per Benef		U.S. Post-retirem	ent Benefits
	2022	2021	2022	2021
Weighted average assumptions:				
Discount rate (for expense)	3.00 %	2.73 %	2.56 %	2.18 %
Expected return on plan assets	6.88 %	7.03 %	_	_
Discount rate (for benefit obligation)	5.53 %	3.00 %	5.31 %	2.56 %

The accumulated benefit obligation and aggregate fair value of plan assets for U.S. pension plans with accumulated benefit obligations in excess of plan assets were \$4.9 billion and \$4.3 billion, respectively, as of December 31, 2022 and \$6.6 billion and \$5.5 billion, respectively, as of December 31, 2021. The decrease in the benefit obligation in 2022 compared to 2021 reflects the increase in discount rates used to measure plan liabilities.

The projected benefit obligation and fair value of plan assets for U.S. pension plans with projected benefit obligations in excess of plan assets was \$4.9 billion and \$4.3 billion, respectively, as of December 31, 2022 and \$6.6 billion and \$5.5 billion, respectively, as of December 31, 2021. The decrease in the benefit obligation in 2022 compared to 2021 reflects the increase in discount rates used to measure plan liabilities.

As of December 31, 2022, the U.S. qualified plan holds 2 million shares of the Company's common stock which were contributed to the qualified plan by the Company in 2005. This represented approximately 7.8% of that plan's assets as of December 31, 2022.

The components of the net benefit (credit) cost for the U.S. defined benefit and other post-retirement benefit plans are as follows:

U.S. Plans only For the Years Ended December 31,		Pension Benefits				ſ	 retirement enefits	
(In millions)	2022 2021 2020					2022	2021	2020
Interest cost	\$ 193	\$ 184	\$	213	\$	1	\$ 1	\$ 1
Expected return on plan assets	(336)	(327)		(345)		_	_	_
Recognized actuarial loss (gain)	74	90		72		_	(1)	_
Net benefit (credit) cost	\$ (69)	\$ (53)	\$	(60)	\$	1	\$ _ :	\$ 1

The assumed health care cost trend rate for Medicare eligibles and non-Medicare eligibles is approximately 6.4% in 2022, gradually declining to 4% in 2046. Assumed health care cost trend rates have a small effect on the amounts reported for the U.S. health care plans because the Company caps its share of health care trend at 5%.

Estimated Future Contributions

The Company expects to contribute approximately \$31 million to its non-qualified U.S. plans in 2023. The Company's policy for funding its tax-qualified defined benefit retirement plans is to contribute amounts at least sufficient to meet the funding requirements set forth in the U.S. and applicable foreign law. The Company was not required to and made no contributions to its U.S. qualified plans in 2022, and is not required to make any contributions in 2023.

Non-U.S. Plans
The following tables provide information concerning the Company's non-U.S. defined benefit pension and post-retirement benefit plans:

		Non-U.S Ber	. Pe efits		Non Post-retiren	
(In millions)		2022		2021	2022	2021
Change in benefit obligation:						
Benefit obligation at beginning of year	\$	12,057	\$	12,998	\$ 68	\$ 73
Service cost		28		38	_	1
Interest cost		196		157	2	1
Employee contributions		3		2	_	_
Plan combination		2		_	_	_
Actuarial (gain) loss		(3,953)		(617)	(14)	(4)
Plan amendments		_		7	_	_
Effect of settlement		(22)		(16)	_	_
Effect of curtailment		_		(2)	_	_
Benefits paid		(342)		(395)	(2)	(3)
Foreign currency changes		(1,083)		(115)	(6)	_
Benefit obligation, December 31	\$	6,886	\$	12,057	\$ 48	\$ 68
Change in plan assets:						
Fair value of plan assets at beginning of year	\$	13,855	\$	14,028	\$ _	\$ _
Plan combination		1		_	_	_
Actual return on plan assets		(3,609)		306	_	_
Effect of settlement		(22)		(16)	_	_
Company contributions		139		95	2	3
Employee contributions		3		2	_	_
Benefits paid		(342)		(395)	(2)	(3)
Foreign currency changes		(1,261)		(165)	_	_
Fair value of plan assets, December 31	\$	8,764	\$	13,855	\$ _	\$ _
Net funded status, December 31	\$	1,878	\$	1,798	\$ (48)	\$ (68)
Amounts recognized in the consolidated balance sheets:						
Non-current assets	\$	2,127	\$	2,269	\$ _	\$ _
Current liabilities		(6)		(6)	(3)	(3)
Non-current liabilities		(243)		(465)	(45)	(65)
Net asset (liability) recognized, December 31	\$	1,878	\$	1,798	\$ (48)	\$ (68)
Amounts recognized in other comprehensive loss:						
Prior service (credit) cost	\$	(16)	\$	(18)	\$ 5	\$ 7
Net actuarial loss		(2,610)		(2,904)	6	(10)
Total recognized accumulated other comprehensive (loss) income December 31	e, \$	(2,626)	\$	(2,922)	\$ 11	\$ (3)
Cumulative employer contributions in excess of (less than) net benefit (credit) cost		4,504		4,720	(59)	(65)
Net asset (liability) recognized in consolidated balance sheets, December 31	\$	1,878	\$	1,798	\$ (48)	\$ (68)
Accumulated benefit obligation, December 31	\$	6,776	\$	11,830	\$ _	\$ _

			Non-U.S Ben				Po	Non st-retiren			;
(In millions)			2022		2021			2022			2021
Reconciliation of prior service (cost) credit recognized in accumula other comprehensive income (loss):	ated										
Beginning balance	\$	3	(18)	\$	(13)	\$		7	\$		9
Recognized as component of net benefit (credit) cost:											
Amortization of prior service credit			_		_			(2)			(2)
Effect of curtailment			_		2			_			_
Total recognized as component of net benefit (credit) cost			_		2			(2)			(2)
Changes in plan assets and benefit obligations recognized in othe comprehensive income:	er										
Plan amendments			_		(7)			_			_
Exchange rate adjustments			2		_			_			_
Prior service (cost) credit, December 31	\$	3	(16)	\$	(18)	\$		5	\$		7
			Non-U.S. Ben				Po	Non- est-retirem		-	
(In millions)			2022		2021			2022			2021
Reconciliation of net actuarial (loss) gain recognized in accumulat other comprehensive (loss) income:	ted										
Beginning balance	\$;	(2,904)	\$	(3,467)	\$		(10)	\$		(16)
Recognized as component of net benefit (credit) cost:					-						
Amortization of net loss			75		116			1			2
Effect of settlement			2		5			_			_
Total recognized as component of net benefit (credit) cost			77		121			1			2
Changes in plan assets and benefit obligations recognized in othe comprehensive income (loss):	er										
Liability experience			3,953		617			14			4
Asset experience			(4,051)		(199)			_			_
Effect of curtailment			_		2			_			_
Total amount recognized as change in plan assets and benefit obligations			(98)		420			14			4
Exchange rate adjustments			315		22			1			_
Net actuarial (loss) gain, December 31	\$	i	(2,610)	\$	(2,904)	\$		6	\$		(10)
	Nic	on II G	S. Pensio	n			Non	U.S. Post	t roti	romont	
For the Years Ended December 31,	INC		nefits	11			INOII	Bene		CHICHI	
	2022		2021		2020	2	022		021		2020
Total recognized in net benefit (credit) cost and other								_			
	(436)	\$	(745)	\$	261 \$		(13)	\$	(2)	\$	13

The weighted average actuarial assumptions utilized in determining expense during the year and benefit obligation at the end of the year for the non-U.S. defined benefit plans are as follows:

	Non-U.S. Per Benefits		Non-U.S. Post-retirement B	enefits
	2022	2021	2022	2021
Weighted average assumptions:				
Discount rate (for expense)	1.89 %	1.49 %	2.28 %	1.96 %
Expected return on plan assets	3.64 %	3.89 %	_	_
Rate of compensation increase (for expense)	3.34 %	2.84 %	_	_
Discount rate (for benefit obligation)	4.89 %	1.89 %	4.73 %	2.28 %
Rate of compensation increase (for benefit obligation)	3.16 %	3.34 %	_	_

The accumulated benefit obligation and fair value of plan assets for the non-U.S. pension plans with accumulated benefit obligations in excess of plan assets were \$935 million and \$718 million, respectively, as of December 31, 2022 and \$1.6 billion and \$1.2 billion, respectively, as of December 31, 2021.

The projected benefit obligation and fair value of plan assets for non-U.S. pension plans with projected benefit obligations in excess of plan assets was \$1.0 billion and \$723 million, respectively, as of December 31, 2022 and \$1.7 billion and \$1.2 billion, respectively, as of December 31, 2021. The decrease in the benefit obligation in 2022 compared to 2021 reflects an actuarial gain primarily due to the increase in discount rates used to measure plan liabilities.

Components of Net Benefit (Credit) or Cost

The components of the net benefit (credit) or cost for the non-U.S. defined benefit and other post-retirement benefit plans and the curtailment, settlement and termination expenses are as follows:

For the Years Ended December 31,	Non-l	Post-retiremen enefits	t			
(In millions)	2022	2021	2020	2022	2021	2020
Service cost	\$ 28 \$	38 \$	36 \$	— \$	1 \$	_
Interest cost	196	157	208	2	1	2
Expected return on plan assets	(442)	(505)	(499)	_	_	_
Amortization of prior service credit	1	_	_	(2)	(2)	(2)
Recognized actuarial loss	75	116	89	1	2	_
Net periodic benefit (credit) cost	(142)	(194)	(166)	1	2	
Settlement loss	2	5	3	_	_	_
Curtailment loss	_	2	_	_	_	_
Special termination benefits	_	_	1	_	_	_
Net benefit (credit) cost	\$ (140) \$	(187) \$	(162) \$	1 \$	2 \$	

The assumed health care cost trend rate was approximately 4.91% in 2022, gradually declining to 4.45% in 2040. Assumed health care cost trend rates can have a significant effect on the amounts reported for the non-U.S. health care plans.

Estimated Future Contributions

The Company expects to contribute approximately \$76 million to its non-U.S. pension plans in 2023. Funding requirements for non-U.S. plans vary by country. Contribution rates are generally based on local funding practices and requirements, which may differ significantly from measurements under U.S. GAAP. Funding amounts may be influenced by future asset performance, the level of discount rates and other variables impacting the assets and/or liabilities of the plan. Discretionary contributions may also be affected by alternative uses of the Company's cash flows, including dividends, investments and share repurchases.

In the U.K., the assumptions used to determine pension contributions are the result of legally prescribed negotiations between the Company and the plans' trustee that typically occurs every three years in conjunction with the actuarial valuation of the plans. Currently, this results in a lower funded status than under U.S. GAAP and may result in contributions irrespective of the U.S. GAAP funded status.

In 2021, the JLT Pension Scheme was merged into the MMC U.K. Pension Fund with a new segregated JLT section created. The Company made deficit contributions of \$103 million to the JLT section in 2022 and is expected to make contributions totaling approximately \$39 million in 2023.

For the MMC U.K. Pension Fund, excluding the JLT section, an agreement was reached with the trustee in the fourth quarter of 2022 based on the surplus funding position at December 31, 2021. In accordance with the agreement no deficit funding is required until 2026. The funding level will be re-assessed during 2025, as part of the December 31, 2024 actuarial valuation, to determine if contributions are required in 2026. In November 2022, as part of a long-term strategy to have greater influence over asset allocation and overall investment decisions, the Company renewed its agreement to support annual deficit contributions by the U.K. operating companies under certain circumstances, up to £450 million (or \$541 million) over a 7 year period.

Estimated Future Benefit Payments

The estimated future benefit payments for the Company's pension and post-retirement benefit plans are as follows:

For the Years Ended December 31,	Per Ber	Post-retirement Benefits					
(In millions)	U.S.		U.S.		Non-U.S.		
2023	\$ 312	\$	337	\$	3	\$	3
2024	\$ 324	\$	344	\$	3	\$	3
2025	\$ 337	\$	352	\$	3	\$	3
2026	\$ 343	\$	360	\$	2	\$	3
2027	\$ 347	\$	371	\$	2	\$	3
2028-2032	\$ 1,760	\$	2,054	\$	8	\$	15

Defined Benefit Plans Fair Value Disclosures

The U.S. and non-U.S. plan investments are classified into Level 1, which refers to investments valued using quoted prices from active markets for identical assets; Level 2, which refers to investments not traded on an active market but for which observable market inputs are readily available; Level 3, which refers to investments valued based on significant unobservable inputs; and investments valued using net asset value ("NAV") as a practical expedient. Assets and liabilities are classified in their entirety based on the lowest level of input that is significant to the fair value measurement. Refer to Note 10, Fair Value Measurements, for further description of fair value hierarchy leveling.

The following table sets forth, by level within the fair value hierarchy, a summary of the U.S. and non-U.S. plans' investments measured at fair value on a recurring basis at December 31, 2022 and 2021:

			F	air Value Measur	eme	ents at December	31, 2	2022	
Assets (In millions)	Ac	oted Prices in tive Markets or Identical Assets (Level 1)		Significant Other Observable Inputs (Level 2)	Other Significant Unobservable Unputs Inputs			NAV	Total
Common/collective trusts	\$	43	\$	_	\$	_	\$	3,995	\$ 4,038
Corporate obligations		_		2,402		_		_	2,402
Corporate stocks		527		36		1		_	564
Private equity/partnerships		_		_		_		1,433	1,433
Government securities		15		4,662		_		_	4,677
Real estate		_		_		_		261	261
Short-term investment funds		609		3		_		_	612
Company common stock		331		_		_		_	331
Other investments		10		11		308		_	329
Total investments	\$	1,535	\$	7,114	\$	309	\$	5,689	\$ 14,647
Net derivative liabilities		_		(1,605)		_		_	(1,605)
Net investments	\$	1,535	\$	5,509	\$	309	\$	5,689	\$ 13,042

			F	air Value Measu	reme	ents at December 3	31, 2	021		
Assets (In millions)	Ä	oted Prices in ctive Markets for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)			Significant Unobservable Inputs (Level 3)		NAV	Total	
Common/collective trusts	\$	476	\$	_	\$	_	\$	5,221	\$	5,697
Corporate obligations		_		4,209		_		_		4,209
Corporate stocks		2,368		44		1		_		2,413
Private equity/partnerships		_		_		_		1,531		1,531
Government securities		15		7,364		_		_		7,379
Real estate		_		_		_		356		356
Short-term investment funds		681		_		_		_		681
Company common stock		348		_		_		_		348
Other investments		11		6		662		_		679
Total investments	\$	3,899	\$	11,623	\$	663	\$	7,108	\$	23,293
Net derivative liabilities		_		(3,899)		_		_		(3,899)
Net investments	\$	3.899	\$	7.724	\$	663	\$	7.108	\$	19.394

The tables below set forth a summary of changes in the fair value of the plans' Level 3 assets for the years ended December 31, 2022 and December 31, 2021:

Assets (In millions)	air Value, anuary 1, 2022	Purchases		Sal	es	Unrealized Gain/ (Loss)	Realized Gain/ (Loss)	Exchange Rate Impact	Transfers in/(out) and Other	Dec	Fair Value, cember 31, 2022
Other investments	\$ 662	\$	18	\$	(19)	\$ (302)	\$ _	\$ (51)	\$ _	\$	308
Corporate stocks	1		_		_	_	_	_	_		1
Total assets	\$ 663	\$	18	\$	(19)	\$ (302)	\$ _	\$ (51)	\$ _	\$	309

Assets (In millions)	air Value, anuary 1, 2021	Purchases		Sale	es	Unrealized Gain/ (Loss)	Realized Gain/ (Loss)	Exchange Rate Impact	Transfers in/(out) and Other	D	Fair Value, ecember 31, 2021
Other investments	\$ 773	\$	19	\$	(15)	\$ (78)	\$ 1	\$ (36)	\$ (2)	\$	662
Corporate stocks	1		1		_	(1)	_	_	_		1
Total assets	\$ 774	\$	20	\$	(15)	\$ (79)	\$ 1	\$ (36)	\$ (2)	\$	663

The following is a description of the valuation methodologies used for assets measured at fair value:

Company common stock: Valued at the closing price reported on the New York Stock Exchange.

Common stocks, preferred stocks, convertible equity securities, rights/warrants and real estate investment trusts (included in Corporate stocks): Valued at the closing price reported on the primary exchange.

Corporate bonds (included in Corporate obligations): The fair value of corporate bonds is estimated using recently executed transactions, market price quotations (where observable) and bond spreads. The spread data used are for the same maturity as the bond. If the spread data does not reference the issuer, then data that references a comparable issuer are used. When observable price quotations are not available, fair value is determined based on cash flow models.

Commercial mortgage-backed and asset-backed securities (included in Corporate obligations): Fair value is determined using discounted cash flow models. Observable inputs are based on trade and quote activity of bonds with similar features including issuer vintage, purpose of underlying loan (first or second lien), prepayment speeds and credit ratings. The discount rate is the combination of the appropriate rate from the benchmark yield curve and the discount margin based on quoted prices.

Common/Collective trusts: Trust assets include mutual funds that are valued based on readily determinable market values and other assets valued at the net asset value of units of a bank collective trust. The net asset value as provided by the trustee, is used as a practical expedient to estimate fair value. The net asset value is based on the fair value of the underlying investments held by the fund less its liabilities. This practical expedient is not used when it is determined to be probable that the fund will sell the investment for an amount different than the reported net asset value.

U.S. government bonds (included in Government securities): The fair value of U.S. government bonds is estimated by pricing models that utilize observable market data including quotes, spreads and data points for yield curves.

U.S. agency securities (included in Government securities): U.S. agency securities are comprised of two main categories consisting of agency issued debt and mortgage pass-throughs. Agency issued debt securities are valued by benchmarking market-derived prices to quoted market prices and trade data for identical or comparable securities. Mortgage pass-throughs include certain "To-be-announced" (TBA) securities and mortgage pass-through pools. TBA securities are generally valued using quoted market prices or are benchmarked thereto. Fair value of mortgage pass-through pools are model driven with respect to spreads of the comparable TBA security.

Private equity and real estate partnerships: Investments in private equity and real estate partnerships are valued based on the fair value reported by the manager of the corresponding partnership and reported on

a one quarter lag. The managers provide unaudited quarterly financial statements and audited annual financial statements which set forth the value of the fund. The valuations obtained from the managers are based on various analyses on the underlying holdings in each partnership, including financial valuation models and projections, comparable valuations from the public markets, and precedent private market transactions. Investments are valued in the accompanying financial statements based on the Plan's beneficial interest in the underlying net assets of the partnership as determined by the partnership agreement.

Insurance group annuity contracts: The fair values for these investments are based on the current market value of the aggregate accumulated contributions plus interest earned.

Net derivative liabilities: Includes interest rate swaps, inflation swaps, total return swaps, repurchase agreements and equity based derivatives, primarily related to the U.K. plans. These derivatives are structured to hedge interest rate, inflation and equity exposure in the U.K. plans. Fair values for interest rate, inflation and equity based derivatives are calculated using a discounted cash flow pricing model. These models use observable market data such as contractual fixed rate, spot equity price or index value and dividend data.

Short-term investment funds: Primarily high-grade money market instruments valued at a readily determinable price.

Registered investment companies: Valued at the closing price reported on the primary exchange.

Defined Contribution Plans

The Company maintains certain defined contribution plans for its employees, including the Marsh & McLennan Companies 401(k) Savings & Investment Plan ("MMC 401(k) Plan") and the Marsh & McLennan Agency Savings and Investment Plan (collectively, the "401(k) Plans"), that are qualified under U.S. tax laws. For the 401(k) Plans, eligible employees may contribute a percentage of their base salary, subject to certain limitations, and the Company matches a fixed portion of the employees' contributions. In addition, the Company also amended the MMC 401(k) Plan for most of its U.S. employees to add an automatic Company contribution equal to 4% of eligible base pay beginning on January 1, 2017. The 401(k) Plans contain an Employee Stock Ownership Plan feature under U.S. tax law. Approximately \$677 million of the 401(k) Plans' assets at December 31, 2022 and \$742 million at December 31, 2021 were invested in the Company's common stock. If a participant does not choose an investment direction for their future contributions, they are automatically invested in a BlackRock LifePath Portfolio that most closely matches the participant's expected retirement year. The cost of these defined contribution plans was \$161 million in 2022, \$150 million in 2021 and \$145 million in 2020. In addition, the Company has significant defined contribution plans in the U.K. Effective August 1, 2014, a newly formed defined contribution plan replaced the existing defined contribution and defined benefit plans with regard to future service. In addition, the Company assumed responsibility for the defined contribution section in 2021. The cost of the JLT U.K. plan defined contribution section transferred to the MMC U.K. Pension Fund defined contribution section in 2021. The cost of the U.K. defined contribution plan was \$140 million, \$141 million and \$121 million in 2022, 2021 and 2020, respectively.

9. Stock Benefit Plans

The Company maintains multiple stock-based payment arrangements under which employees may be awarded restricted stock units, stock options and other forms of stock-based benefits.

Marsh & McLennan Companies, Inc. Incentive and Stock Award Plans

On May 21, 2020, the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan (the "2020 Plan") was approved by the Company's stockholders. The 2020 Plan replaced the Company's previous equity incentive plan, the 2011 Incentive and Stock Award Plan.

The types of awards permitted under the 2020 Plan include stock options, restricted stock units payable in Company common stock or cash, and other stock-based awards. Performance-based restricted stock units are referred to as performance stock units. The 2020 Plan contains a provision which, in the event of a change in control of the Company, may accelerate the vesting of awards. This provision requires both a change in control of the Company and a subsequent specified termination of employment for vesting to be accelerated. There are 20 million shares approved for issuance under the 2020 plan. The total number

of shares issued in connection with full-value awards may not exceed 12.5 million shares. Full-value awards include awards such as restricted stock units and performance stock units but exclude stock options.

The Company's current practice is to grant non-qualified stock options, restricted stock units ("RSUs") and/or performance stock units ("PSUs") on an annual basis to certain employees as part of their annual total compensation. Senior executives are granted options and PSU awards. In addition, a small group of other employees are granted options, PSU and RSU awards and a larger group of other employees are granted RSU awards. RSU awards are also granted to new hires or as retention awards for certain employees.

Stock Options: The Company currently grants non-qualified stock options under the 2020 Plan. The Compensation Committee determines when the options vest and may be exercised and under what terms the options are forfeited. Options are generally granted with an exercise price equal to the market value of the Company's common stock on the date of grant. These option awards generally vest 25% per year and have a contractual term of 10 years.

The estimated fair value of options granted is calculated using the Black-Scholes option pricing valuation model. This model considers several factors and assumptions. The expected dividend yield is based on expected dividends for the expected life of the stock options.

The assumptions used in the Black-Scholes option pricing valuation model for options granted by the Company in 2022, 2021 and 2020 are as follows:

	2022	2021	2020
Risk-free interest rate	1.88 %	0.79 %	1.44 %
Expected life (in years)	5.8	6.0	6.0
Expected volatility	22.58 %	23.45 %	20.33 %
Expected dividend yield	1.41 %	1.58 %	1.53 %

A summary of the status of the Company's stock option awards as of December 31, 2022 and changes during the year then ended are presented below:

	Shares	Av	Weighted erage Exercise Price	Weighted Average Remaining Contractual Term	Aggregate Intrinsic Value (\$000)
Balance at January 1, 2022	7,217,125	\$	90.17		
Granted	1,026,615	\$	151.37		
Exercised	(648,016)	\$	77.37		
Forfeited	(62,679)	\$	141.68		
Balance as of December 31, 2022	7,533,045	\$	99.18	5.9 years	\$ 494,391
Options vested or expected to vest at December 31, 2022	7,458,604	\$	98.96	6.1 years	\$ 491,121
Options exercisable at December 31, 2022	4,634,039	\$	82.46	4.7 years	\$ 381,622

In the above table, forfeited options are unvested options whose requisite service period has not been met. Expired options are vested options that were not exercised. The weighted-average grant-date fair value of the Company's option awards granted in 2022, 2021 and 2020 was \$31.38, \$22.25 and \$21.09, respectively. The total intrinsic value of options exercised during the same periods was \$56 million, \$138 million and \$159 million, respectively.

As of December 31, 2022, there was \$22 million of unrecognized compensation cost related to the Company's option awards. The weighted-average period over which that cost is expected to be recognized is approximately 1.43 years. Cash received from the exercise of stock options in 2022, 2021 and 2020 was \$50 million, \$103 million and \$72 million, respectively.

The Company's policy is to issue treasury shares upon option exercises or share unit conversion. The Company intends to issue treasury shares as long as an adequate number of those shares is available.

Restricted Stock Units and Performance Stock Units: The Company currently grants RSU and PSU awards under the 2020 Plan. The Compensation Committee determines the restrictions on such units, when the restrictions lapse, when the units vest and are paid, and under what terms the units are forfeited. The cost of these awards is amortized over the vesting period, which is generally three years. Dividend equivalents are not paid out unless and until such time that the award vests and shares are distributed.

The payout for PSU awards granted prior to 2020 is based on the achievement of the Company's performance measures, based on adjusted EPS growth as modified for executive compensation purposes and measured on a three-year annualized growth basis, and paid out generally at the end of February after the three-year performance period is completed. The Company accounts for these awards as performance condition restricted stock units. The performance condition is not considered in the determination of grant date fair value of such awards. Compensation cost is recognized over the performance period based on management's estimate of the number of units expected to vest and shares to be paid and is adjusted to reflect the actual number of shares paid out at the end of the three-year performance period.

The payout for PSU awards granted beginning in 2020 is based on the achievement of the Company's adjusted EPS growth as well as a relative total stockholder return ("TSR") modifier versus the S&P 500 constituents. The TSR modifier is a market condition with the grant-date fair value determined using a Monte Carlo simulation model. The Monte Carlo model considers several factors and assumptions including the risk-free interest rate, historical volatility of and correlations between the stock prices of the Company and the S&P 500 constituents, and the Company's relative TSR versus S&P 500 constituents for the brief portion of the three-year performance period prior to the grant date.

The number of shares earned at the end of the three-year vesting period will vary, based on actual Company financial performance, and for 2020, 2021 and 2022 PSU awards, relative TSR, from 0% to 200% of the number of performance share units granted.

The assumptions used in the Monte Carlo simulation model for PSU awards granted with the TSR modifier by the Company in 2022 include:

	2022
Risk-Free Interest Rate	1.75 %
Dividend Yield	1.4 %
Volatility	25.0 %
Initial TSR	(12.8)%

A summary of the status of the Company's RSU and PSU awards as of December 31, 2022 and changes during the period then ended are presented below:

	Restricted Stoc	k Units	Performance St	ock Units
		Weighted Average Grant Date Fair Value	Shares	Weighted Average Grant Date Fair Value
Non-vested balance at January 1, 2022	5,548,462 \$	110.86	662,190 \$	114.35
Granted	2,048,576 \$	152.34	212,875\$	151.00
Vested	(3,173,780)\$	104.56	(208,730)\$	91.12
Forfeited	(240,563) \$	127.32	(27,357) \$	132.57
Non-vested balance at December 31, 2022	4,182,695 \$	135.01	638,978\$	133.36

The weighted-average grant-date fair value of the Company's RSU awards granted in 2021 and 2020 was \$120.19 and \$118.20, respectively. The weighted-average grant-date fair value of the Company's PSU awards granted in 2021 and 2020 was \$122.77 and \$127.71, respectively. The total fair value of the

shares distributed in 2022, 2021 and 2020 in connection with the Company's non-option equity awards was \$560 million, \$278 million and \$290 million, respectively.

The payout of shares in 2022 with respect to the PSU awards granted in 2019 was 195% of target based on performance for the three-year performance period. In aggregate, 407,203 shares became distributable in respect to PSUs vested in 2022.

As of December 31, 2022, there was \$351 million of unrecognized compensation cost related to the Company's RSU and PSU awards. The weighted-average period over which that cost is expected to be recognized is approximately one year.

Marsh & McLennan Companies Stock Purchase Plans

In May 1999, the Company's stockholders approved an employee stock purchase plan (the "1999 Plan") to replace the 1994 Employee Stock Purchase Plan (the "1994 Plan"), which terminated on September 30, 1999 following its fifth annual offering. In accordance with the current terms of the 1999 Plan, shares are purchased four times during the plan year at a price that is 95% of the average market price on each quarterly purchase date. In accordance with the 1999 Plan, after including the available remaining unused shares in the 1994 Plan and reducing the shares available by 10,000,000 consistent with the Company's Board of Directors' action in March 2007 and the addition of 4,750,000 shares due to a shareholder action in May 2018, no more than 40,350,000 shares of the Company's common stock may be sold. Employees purchased 322,999 shares in 2022 and at December 31, 2022, 4,193,759 shares were available for issuance for the 1999 Plan.

In accordance with the 1995 Company Stock Purchase Plan for International Employees (the "International Plan"), after reflecting the additional 5,000,000 shares of common stock for issuance approved by the Company's Board of Directors in July 2002, the addition of 4,000,000 shares due to a shareholder action in May 2007 and reducing the shares available by 1,000,000 consistent with the Company's Board of Directors' action in March 2018, no more than 11,000,000 shares of the Company's common stock may be sold. Employees purchased 113,321 shares in 2022 and there were 920,811 shares available for issuance at December 31, 2022 for the International Plan. The plans are considered non-compensatory.

10. Fair Value Measurements

Fair Value Hierarchy

The Company has categorized its assets and liabilities that are valued at fair value on a recurring basis into a three-level fair value hierarchy as defined by the FASB. The fair value hierarchy gives the highest priority to quoted prices in active markets for identical assets and liabilities (Level 1) and lowest priority to unobservable inputs (Level 3). In some cases, the inputs used to measure fair value might fall into different levels of the fair value hierarchy. In such cases, the level in the fair value hierarchy, for disclosure purposes, is determined based on the lowest level input that is significant to the fair value measurement. Assets and liabilities recorded in the consolidated balance sheets at fair value are categorized based on the inputs in the valuation techniques as follows:

Level 1. Assets and liabilities whose values are based on unadjusted quoted prices for identical assets or liabilities in an active market (examples include active exchange-traded equity securities and exchange-traded money market mutual funds).

Assets and liabilities measured using Level 1 inputs include exchange-traded equity securities, exchange-traded mutual funds and money market funds.

Level 2. Assets and liabilities whose values are based on the following:

- a) quoted prices for similar assets or liabilities in active markets;
- b) quoted prices for identical or similar assets or liabilities in non-active markets (examples include corporate and municipal bonds, which trade infrequently);
- c) pricing models whose inputs are observable for substantially the full term of the asset or liability (examples include most over-the-counter derivatives, including interest rate and currency swaps); and

d) pricing models whose inputs are derived principally from or corroborated by observable market data through correlation or other means for substantially the full asset or liability (for example, certain mortgage loans).

Assets and liabilities using Level 2 inputs are related to an equity security.

Level 3. Assets and liabilities whose values are based on prices, or valuation techniques that require inputs that are both unobservable and significant to the overall fair value measurement. These inputs reflect management's own assumptions about the assumptions a market participant would use in pricing the asset or liability.

Assets and liabilities measured using Level 3 inputs relate to assets and liabilities for contingent purchase consideration.

Valuation Techniques

Equity Securities, Money Market Funds and Mutual Funds - Level 1

Investments for which market quotations are readily available are valued at the sale price on their principal exchange or, for certain markets, official closing bid price. Money market funds are valued at a readily determinable price.

Contingent Purchase Consideration Assets and Liability - Level 3

Purchase consideration for some acquisitions and dispositions made by the Company include contingent consideration arrangements. Contingent consideration arrangements are based primarily on EBITDA or revenue targets over a period of 2 to 4 years. The fair value of contingent purchase consideration asset and liability is estimated as the present value of future cash flows to be paid, based on projections of revenue and earnings and related targets of the acquired and disposed entities.

The following fair value hierarchy table presents information about the Company's assets and liabilities measured at fair value on a recurring basis as of December 31, 2022 and 2021:

(In millions)	Identica (Lev		Observa (Lev		Unobserva (Lev		Total			
	12/31/22	12/31/21	12/31/22	12/31/21	12/31/22	12/31/21	12/31/22		12/31/21	
Assets:										
Financial instruments owned:										
Exchange traded equity securities (a)	\$ 6	\$ 61	\$ _	\$ _	\$ _	\$ _	\$ 6	\$	61	
Mutual funds ^(a)	162	192	_	_	_	_	162		192	
Money market funds ^(b)	146	425	_	_	_	_	146		425	
Other equity investment (a)	_	_	13	8	_	_	13		8	
Contingent purchase consideration asset (c)	_	_	_	_	3	5	3		5	
Total assets measured at fair value	\$ 314	\$ 678	\$ 13	\$ 8	\$ 3	\$ 5	\$ 330	\$	691	
Fiduciary Assets:										
U.S. treasury bills (d)	\$ _	\$ 55	\$ _	\$ _	\$ _	\$ _	\$ _	\$	55	
Money market funds	201	527	_	_	_	_	201		527	
Total fiduciary assets measured at fair value	\$ 201	\$ 582	\$ _	\$ 	\$ _	\$ 	\$ 201	\$	582	
Liabilities:										
Contingent purchase consideration liability ^(e)	\$ _	\$ _	\$ _	\$ _	\$ 377	\$ 352	\$ 377	\$	352	
Total liabilities measured at fair value	\$ _	\$ _	\$ _	\$ _	\$ 377	\$ 352	\$ 377	\$	352	

⁽a) Included in other assets in the consolidated balance sheets.

The Level 3 assets in the table reflect contingent purchase consideration from the sale of businesses. The change in the contingent purchase consideration assets from December 31, 2021 is driven by cash receipts.

⁽b) Included in cash and cash equivalents in the consolidated balance sheets.

⁽c) Included in other receivables in the consolidated balance sheets.

⁽d) Maturity dates of three months or less.

⁽e) Included in accounts payable and accrued liabilities and other liabilities in the consolidated balance sheets.

In 2022 and 2021, there were no assets or liabilities that were transferred between levels.

The following table sets forth a summary of the changes in fair value of the Company's Level 3 liabilities for the years ended December 31, 2021 and December 31, 2021.

(In millions)	2022	2021
Balance at January 1,	\$ 352	\$ 243
Net additions	46	107
Payments	(70)	(77)
Revaluation impact	49	81
Other ^(a)	_	(2)
Balance at December 31,	\$ 377	\$ 352

⁽a) Primarily reflects the impact of foreign exchange.

Long-Term Investments

The Company holds investments in public and private companies as well as certain private equity investments that are accounted for using the equity method of accounting. The carrying value of these investments was \$215 million and \$207 million at December 31, 2022 and 2021, respectively.

Investments in Public and Private Companies

The Company has investments in private insurance and consulting companies with a carrying value of \$56 million and \$58 million at December 31, 2022 and 2021, respectively. These investments are accounted for using the equity method of accounting, the results of which are included in revenue in the consolidated statements of income and the carrying value of which is included in other assets in the consolidated balance sheets. The Company records its share of income or loss on its equity method investments, some of which are on a one quarter lag basis. In December 2021, the Company increased its ownership in Marsh India from 49% to 92%. Prior to the increase in ownership, the Company accounted for the investment under the equity method of accounting.

Private Equity Investments

The Company's investments in private equity funds were \$159 million and \$149 million at December 31, 2022 and 2021, respectively. The carrying values of these private equity investments approximates fair value. The underlying private equity funds follow investment company accounting, where investments within the fund are carried at fair value. The Company records in earnings its proportionate share of the change in fair value of the funds on the investment income line in the consolidated statements of income. These investments are included in other assets in the consolidated balance sheets. The Company recorded net investment gains of \$18 million, \$56 million and \$3 million from these investments in 2022, 2021 and 2020, respectively.

The Company has commitments for potential future investments of approximately \$160 million in private equity funds that invest primarily in financial services companies.

Other Investments

At December 31, 2022 and 2021, the Company held certain equity investments with readily determinable market values of \$17 million and \$75 million, respectively. In 2022, 2021, and 2020, the Company recorded mark-to-market investment gains on these investments of \$11 million and \$5 million, and losses of \$27 million, respectively.

The Company also held investments without readily determinable market values of \$42 million and \$36 million at December 31, 2022 and 2021, respectively.

In 2022, the Company sold certain of these investments for cash proceeds of approximately \$62 million, including its remaining investment in the common stock of AF, and recorded a net loss of \$4 million.

11. Derivatives

Net Investment Hedge

The Company has investments in various subsidiaries with Euro functional currencies. As a result, the Company is exposed to the risk of fluctuations between the Euro and U.S. dollar exchange rates. The Company designated its €1.1 billion senior note debt instruments ("Euro notes") as a net investment hedge (the "hedge") of its Euro denominated subsidiaries. The hedge effectiveness is re-assessed each quarter to confirm that the designated equity balance at the beginning of each period continues to equal or exceed 80% of the outstanding balance of the Euro debt instrument and that all the critical terms of the hedging instrument and the hedged net investment continue to match. If the Company concludes that the hedge is highly effective, the change in the debt balance related to foreign exchange fluctuations is recorded in accumulated other comprehensive loss in the consolidated balance sheets.

The U.S. dollar value of the Euro notes decreased by \$82 million in 2022 related to the change in foreign exchange rates. The Company concluded that the hedge was highly effective and recorded a gain as a decrease to accumulated other comprehensive loss for the year ended December 31, 2022.

12. Leases

The Company leases office facilities under non-cancelable operating leases with terms generally ranging between 10 and 25 years. The Company utilizes these leased office facilities for use by its employees in countries in which the Company conducts its business. The Company's leases have no restrictions on the payment of dividends, the acquisition of debt or entering into additional lease obligations. The leases also do not contain any significant purchase options. Operating leases are recognized on the consolidated balance sheets as ROU assets and operating lease liabilities based on the present value of the remaining future minimum payments over the lease term at the commencement date of the lease.

The Company determined that \$118 million and \$16 million of its ROU assets were impaired and recorded a charge to the consolidated statements of income in 2022 and 2021, respectively.

The following table provides additional information about the Company's property leases:

For the Years Ended December 31,		
(In millions, except weighted average data)	2022	2021
Lease Cost:		
Operating lease cost (a)	\$ 343 \$	374
Short-term lease cost	4	4
Variable lease cost	133	144
Sublease income	(17)	(20)
Net lease cost	\$ 463 \$	502
Other information:		
Operating cash outflows from operating leases	\$ 380 \$	412
Right of use assets obtained in exchange for new operating lease liabilities	\$ 196 \$	348
Weighted-average remaining lease term – real estate	8.37 years	8.87 years
Weighted-average discount rate – real estate leases	2.90 %	2.72 %

⁽a) Excludes ROU asset impairment charges.

Future minimum lease payments for the Company's operating leases as of December 31, 2022 are as follows:

(In millions)	Real Es	state Leases
2023	\$	362
2024		324
2025		291
2026		268
2027		232
Subsequent years		751
Total future lease payments		2,228
Less: imputed interest		(251)
Total	\$	1,977
Current lease liabilities	\$	310
Long-term lease liabilities		1,667
Total lease liabilities	\$	1,977

Note: The above table excludes obligations for leases with original terms of 12 months or less which have not been recognized as a ROU asset or liability in the consolidated balance sheets.

As of December 31, 2022, the Company had additional operating real estate leases that had not yet commenced of \$33 million. These operating leases will commence over the next 12 months.

13. Debt

The Company's outstanding debt is as follows:

December 31,			
(In millions)	20	22	2021
Short-term:			
Current portion of long-term debt	\$ 26	8 \$	17
	26	8	17
Long-term:			
Senior notes – 3.30% due 2023	-	_	349
Senior notes – 4.05% due 2023	25	0	249
Senior notes – 3.50% due 2024	59	9	599
Senior notes – 3.875% due 2024	99	8	997
Senior notes – 3.50% due 2025	49	9	498
Senior notes – 1.349% due 2026	58	7	629
Senior notes – 3.75% due 2026	59	8	598
Senior notes – 4.375% due 2029	1,49	9	1,499
Senior notes – 1.979% due 2030	57	6	614
Senior notes – 2.25% due 2030	73	9	739
Senior notes – 2.375% due 2031	39	7	397
Senior notes – 5.750% due 2032	49	3	_
Senior notes – 5.875% due 2033	29	8	298
Senior notes – 4.75% due 2039	49	5	495
Senior notes – 4.35% due 2047	49	3	493
Senior notes – 4.20% due 2048	59	3	593
Senior notes – 4.90% due 2049	1,23	8	1,238
Senior notes – 2.90% due 2051	34	6	346
Senior notes – 6.250% due 2052	49	2	_
Mortgage – 5.70% due 2035	30	1	316
Other		4	3
	11,49	5	10,950
Less current portion	26	8	17
	\$ 11,22	7 \$	10,933

The senior notes in the table above are registered by the Company with the Securities and Exchange Commission and are not guaranteed.

In October 2022, the Company increased its short-term commercial paper financing program to \$2.8 billion from \$2.0 billion. The Company had previously increased its short-term commercial financial program on April 9, 2021, to \$2.0 billion from \$1.5 billion. The Company had no commercial paper outstanding at December 31, 2022 and 2021.

Credit Facilities

The Company has a multi-currency unsecured \$2.8 billion five-year revolving credit facility (the "Credit Facility") entered into on April 1, 2021. The interest rate on the Credit Facility is based on LIBOR plus a fixed margin which varies with the Company's credit ratings. The Credit Facility expires in April 2026 and requires the Company to maintain certain coverage and leverage ratios which are tested quarterly. The Credit Facility includes provisions for determining a LIBOR successor rate in the event LIBOR reference rates are no longer available or in certain other circumstances which are determined to make using an alternative rate desirable. As of December 31, 2022 and 2021, the Company had no borrowings under this facility.

In connection with the Credit Facility, the Company terminated its previous multi-currency unsecured \$1.8 billion five-year and its unsecured \$1 billion 364-day revolving credit facilities.

In May 2022, the Company secured a \$250 million uncommitted revolving credit facility. The facility expires in May 2023, and has similar coverage and leverage ratios as the Credit Facility. The Company had no outstanding borrowings under this facility at December 31, 2022.

The Company also maintains other credit facilities, guarantees and letters of credit with various banks aggregating \$514 million at December 31, 2022 and \$508 million at December 31, 2021. There were no outstanding borrowings under these facilities at December 31, 2022 and December 31, 2021.

Senior Notes

In October 2022, the Company issued \$500 million of 5.75% senior notes due 2032 and \$500 million of 6.25% senior notes due 2052. The Company used the net proceeds from these issuances for general corporate purposes, and repaid \$350 million of 3.30% senior notes in November 2022, with an original maturity date of March 2023.

In December 2021, the Company issued \$400 million of 2.375% senior notes due 2031 and \$350 million of 2.90% senior notes due 2051. The Company used the net proceeds from these issuances for general corporate purposes and repaid \$500 million of 2.75% senior notes with an original maturity date of January 2022 in December 2021.

On April 15, 2021, the Company repaid \$500 million of senior notes maturing in July 2021.

Scheduled repayments of long-term debt in 2023 and in the four succeeding years are \$268 million, \$1.6 billion, \$518 million, \$1.2 billion and \$21 million, respectively.

Fair Value of Short-term and Long-term Debt

The estimated fair value of the Company's short-term and long-term debt is provided below. Certain estimates and judgments were required to develop the fair value amounts. The fair value amounts shown below are not necessarily indicative of the amounts that the Company would realize upon disposition, nor do they indicate the Company's intent or need to dispose of the financial instrument.

	December 31, 2022		December 31, 2021			
(In millions)	Carrying Amount		Fair Value	Carrying Amount		Fair Value
Short-term debt	\$ 268	\$	265	\$ 17	\$	17
Long-term debt	\$ 11,227	\$	10,544	\$ 10,933	\$	12,466

The fair value of the Company's short-term debt consists primarily of term debt maturing within the next year and its fair value approximates its carrying value. The estimated fair value of a primary portion of the Company's long-term debt is based on discounted future cash flows using current interest rates available for debt with similar terms and remaining maturities. Short- and long-term debt would be classified as Level 2 in the fair value hierarchy.

14. Restructuring Costs

In the fourth quarter of 2022, the Company initiated activities focused on workforce actions, rationalization of technology and functional services, and reductions in real estate. The Company incurred \$219 million related to these activities, primarily severance and lease exit charges. Based on current estimates, the Company anticipates that these activities will continue throughout 2023 and into 2024. However, additional charges are unlikely to exceed costs incurred in 2022. The Company's plans are still being finalized, which may change the expected timing, estimates of expected costs and related savings, as the Company continues to refine its detailed plans for each business and location.

Restructuring activities also include charges related to improving the Company's global information technology and HR functions, JLT integration costs, and improving efficiencies and client services related to the Marsh operational excellence program. Expenses also reflect additional lease related exit charges of \$89 million in the Risk and Insurance Services segment for a legacy JLT U.K. location.

In 2022, the Company incurred costs related to these initiatives of \$427 million, reflecting \$254 million in RIS, \$77 million in Consulting, and \$96 million and in Corporate.

In 2021, the Company incurred restructuring costs of \$163 million, reflecting \$84 million in RIS, \$48 million in Consulting, and \$31 million in Corporate.

Details of the restructuring activity from January 1, 2021 through December 31, 2022, are as follows:

(In millions)	Sev	erance	F	Real Estate Related Costs ^(a)	Information Technology	(Consulting and Other Outside Services	Total
Liability at 1/1/21	\$	52	\$	51	\$ 2	\$	1	\$ 106
2021 charges		38		31	23		71	163
Cash payments		(55)		(26)	(25)		(72)	(178)
Non-cash charges		_		(22)	_		_	(22)
Liability at 12/31/21	\$	35	\$	34	\$ _	\$	_	\$ 69
2022 charges		111		195	15		106	427
Cash payments		(58)		(25)	(6)		(104)	(193)
Non-cash charges		_		(148)	(9)		_	(157)
Liability at 12/31/22	\$	88	\$	56	\$ _	\$	2	\$ 146

⁽a) Includes ROU and fixed asset impairments and other related costs.

The expenses associated with these initiatives are included in compensation and benefits and other operating expenses in the consolidated statements of income. The liabilities associated with these initiatives are classified on the consolidated balance sheets as accounts payable and accrued liabilities, other liabilities or accrued compensation and employee benefits, depending on the nature of the items.

15. Common Stock

On March 23, 2022, the Board of Directors of the Company authorized an additional \$5 billion in share repurchases. This is in addition to the Company's existing share repurchase program, which had approximately \$1.3 billion of remaining authorization as of December 31, 2021.

In 2022, the Company repurchased 12.2 million shares of its common stock for \$1.9 billion. The Company remains authorized to repurchase up to approximately \$4.3 billion in shares of its common stock. There is no time limit on the authorization. In 2021, the Company repurchased 7.9 million shares of its common stock for \$1.2 billion.

The Company issued approximately 3.5 million and 3.8 million shares related to stock compensation and employee stock purchase plans during the years ended December 31, 2022 and 2021, respectively.

16. Claims, Lawsuits and Other Contingencies

Acquisition of Jardine Lloyd Thompson Group plc

On April 1, 2019, the Company completed its previously announced acquisition of all of the outstanding shares of JLT. Upon the consummation of the acquisition of JLT, the Company assumed the legal liabilities and became responsible for JLT's litigation and regulatory exposures as of April 1, 2019.

Nature of Contingencies

The Company and its subsidiaries are subject to a significant number of claims, lawsuits and proceedings in the course of our business. Such claims and lawsuits consist principally of alleged errors and omissions in connection with the performance of professional services, including the placement of insurance, the provision of actuarial services for corporate and public sector clients, the provision of investment advice and investment management services to pension plans, the provision of advice relating to pension buy-out transactions and the provision of consulting services relating to the drafting and interpretation of trust deeds and other documentation governing pension plans. These claims often seek damages, including punitive and treble damages, in amounts that could be significant. In establishing liabilities for errors and omissions claims in accordance with FASB guidance on Contingencies - Loss Contingencies, the Company utilizes case level reviews by inside and outside counsel, and internal actuarial analysis by Oliver Wyman, a subsidiary of the Company, and other methods to estimate potential losses. A liability is established when a loss is both probable and reasonably estimable. The liability is reviewed quarterly and

adjusted as developments warrant. In many cases, the Company has not recorded a liability, other than for legal fees to defend the claim, because we are unable, at the present time, to make a determination that a loss is both probable and reasonably estimable. To the extent that expected losses exceed our deductible in any policy year, the Company also records an asset for the amount that we expect to recover under any available third-party insurance programs. The Company has varying levels of third-party insurance coverage, with policy limits and coverage terms varying significantly by policy year.

Our activities are regulated under the laws of the U.S. and its various states, the U.K., the E.U. and its member states, and the many other jurisdictions in which the Company operates. The Company also receives subpoenas in the ordinary course of business, and from time, to time requests for information in connection with government investigations.

Current Matters

Risk and Insurance Services Segment

- In January 2019, the Company received a notice that the Administrative Council for Economic Defense anti-trust agency in Brazil
 had commenced an administrative proceeding against a number of insurance brokers, including both Marsh and JLT, and insurers "to
 investigate an alleged sharing of sensitive commercial and competitive confidential information" in the aviation insurance and
 reinsurance sector.
- In 2017, JLT identified payments to a third-party introducer that had been directed to unapproved bank accounts. These payments related to reinsurance placements made on behalf of an Ecuadorian state-owned insurer between 2014 and 2017. In early 2018, JLT voluntarily reported this matter to law enforcement authorities. In February and March 2020, money laundering charges were filed in the United States against a former employee of JLT, the principals of the third-party introducer and a former official of the state-owned insurer. These individuals, including the former JLT employee, have since pleaded guilty to criminal charges. In March 2022, the U.S. Department of Justice (DOJ) issued a declination letter declining to pursue any charges against any JLT entity and seeking disgorgement of \$29 million in alleged gross profits on this account. As previously disclosed, the Company recorded a charge for this amount in the fourth quarter of 2021. In addition, in March 2022, the Colombian Superintendecia de Sociedades (SS) concluded its investigation of this matter and notified JLT of its intention to seek \$2 million in civil penalties which was recorded in the first quarter of 2022. The SS issued its final resolution in May 2022. In June 2022, JLT reached an agreement to settle the investigation by the U.K. Financial Conduct Authority (FCA) for £7.9 million (or \$11 million) in civil penalties which concluded the FCA's investigation into this matter, and the Company recorded a charge for this amount in the second quarter of 2022.
- From 2014, Marsh Ltd. was engaged by Greensill Capital (UK) Limited as its insurance broker. Marsh Ltd. placed a number of trade credit insurance policies for Greensill. On March 1, 2021, Greensill filed an action against certain of its trade credit insurers in Australia seeking a mandatory injunction compelling these insurers to renew coverage under expiring policies. Later that day, the Australian court denied Greensill's application. Since then, a number of Greensill entities have filed for, or been subject to, insolvency proceedings, and several litigations and investigations have been commenced in the U.K., Australia, Germany, Switzerland and the U.S.

Consulting Segment

• In 2014, the FCA conducted an industry-wide review of the suitability of financial advice provided to individuals by a number of companies, including JLT, relating to enhanced transfer value ("ETV") defined benefit pension transfers. In January 2015, the FCA notified JLT that it was commissioning a Skilled Person review of ETV pension transfer advice given by JLT and a business acquired by JLT in 2012. Following the Skilled Person review, which took place between 2015 and 2018, JLT engaged a compliance consulting firm to conduct an analysis of approximately 14,000 individual files to assess the suitability of the advice provided and, where appropriate, the amount of redress to be paid. In February 2019, prior to the completion of its acquisition by the Company, JLT recorded a gross liability of £59 million (or \$77 million). This preliminary estimate by JLT reflected projected redress amounts based on the limited number of files examined as part of the Skilled Person's review and report. Thereafter, the FCA expanded

the scope of the review. As of December 31, 2020, the updated redress liability, including the projected costs of completing the review, increased to £155 million (or \$210 million) resulting from the expansion in the scope of the review, and the significant progress made in completing the individual suitability reviews. Payments of redress and expenses in 2021 and 2022 reduced the recorded liability to £5 million (or \$6 million) as of December 31, 2022. The suitability review and calculation redress for affected costumers is now complete, and redress payments have been made. In July 2022, the FCA notified JLT that it had closed its review of this matter. This gross liability has been, and we anticipate will continue to be, partially offset by a contractual indemnity obligation and insurance recoveries from third-party E&O insurers.

At this time, we are unable to predict the likely timing, outcome or ultimate impact of the foregoing matters. Adverse determinations in one or more of these matters could have a material impact on the Company's consolidated results of operations, financial condition or cash flows in a future period.

Other Contingencies-Guarantees

In connection with its acquisition of U.K.-based Sedgwick Group in 1998, the Company acquired several insurance underwriting businesses that were already in run-off, including River Thames Insurance Company Limited ("River Thames"), which the Company sold in 2001. Sedgwick guaranteed payment of claims on certain policies underwritten through the Institute of London Underwriters (the "ILU") by River Thames. The policies covered by this guarantee are partly reinsured by a related party of River Thames. Payment of claims under the reinsurance agreement is collateralized by funds withheld by River Thames from the reinsurer. To the extent River Thames or the reinsurer is unable to meet its obligations under those policies, a claimant may seek to recover from the Company under the guarantee.

From 1980 to 1983, the Company owned indirectly the English & American Insurance Company ("E&A"), which was a member of the ILU. The ILU required the Company to guarantee a portion of E&A's obligations. After E&A became insolvent in 1993, the ILU agreed to discharge the guarantee in exchange for the Company's agreement to post an evergreen letter of credit that is available to pay claims by policyholders on certain E&A policies issued through the ILU and incepting between July 3, 1980 and October 6, 1983. Certain claims have been paid under the letter of credit and the Company anticipates that additional claimants may seek to recover against the letter of credit.

* * * *

The pending proceedings described above and other matters not explicitly described in this Note 16 on Claims, Lawsuits and Other Contingencies may expose the Company or its subsidiaries to liability for significant monetary damages, fines, penalties or other forms of relief. Where a loss is both probable and reasonably estimable, the Company establishes liabilities in accordance with FASB guidance on Contingencies - Loss Contingencies. Except as described above, the Company is not able at this time to provide a reasonable estimate of the range of possible loss attributable to these matters or the impact they may have on the Company's consolidated results of operations, financial position or cash flows. This is primarily because these matters are still developing and involve complex issues subject to inherent uncertainty. Adverse determinations in one or more of these matters could have a material impact on the Company's consolidated results of operations, financial condition or cash flows in a future period.

17. Segment Information

The Company is organized based on the types of services provided. Under this structure, the Company's segments are:

- Risk and Insurance Services. comprising insurance services (Marsh) and reinsurance services (Guy Carpenter); and
- Consulting, comprising Mercer and Oliver Wyman Group.

The accounting policies of the segments are the same as those used for the consolidated financial statements described in Note 1, Summary of Significant Accounting Policies. Segment performance is evaluated based on segment operating income, which includes directly related expenses, and charges or credits related to restructuring costs but not the Company's corporate-level expenses. Revenues are attributed to geographic areas on the basis of where the services are performed.

Selected information about the Company's segments and geographic areas of operation are as follows:

				Operating	T-+-1		Depreciation	0
For the Years Ended December 31, (In million	s)	Revenue		Income (Loss)	Total Assets		and Amortization	Capital Expenditures
2022 –	,			,				
Risk and Insurance Services	\$	12,645	(a)	\$ 3,089	\$ 22,776	(c)	\$ 469	\$ 283
Consulting		8,139	(b)	1,553	10,032	(d)	158	109
Total Segments		20,784		4,642	32,808		627	392
Corporate/Eliminations		(64)		(362)	646	(e)	92	78
Total Consolidated	\$	20,720		\$ 4,280	\$ 33,454		\$ 719	\$ 470
2021 –								
Risk and Insurance Services	\$	12,085	(a)	\$ 3,080	\$ 21,996	(c)	\$ 505	\$ 214
Consulting		7,789	(b)	1,504	10,346	(d)	171	109
Total Segments		19,874		4,584	32,342		676	323
Corporate/Eliminations		(54)		(272)	2,046	(e)	71	83
Total Consolidated	\$	19,820		\$ 4,312	\$ 34,388		\$ 747	\$ 406
2020 –								
Risk and Insurance Services	\$	10,337	(a)	\$ 2,346	\$ 20,612	(c)	\$ 500	\$ 170
Consulting		6,976	(b)	994	9,571	(d)	174	107
Total Segments		17,313		3,340	30,183		674	277
Corporate/Eliminations		(89)		(274)	2,866	(e)	67	71
Total Consolidated	\$	17,224		\$ 3,066	\$ 33,049		\$ 741	\$ 348

⁽a) Includes inter-segment revenue of \$5 million in each of 2022, 2021 and 2020, interest income on fiduciary funds of \$120 million, \$15 million and \$46 million in 2022, 2021 and 2020, respectively, and equity method income of \$12 million, \$31 million and \$27 million in 2022, 2021 and 2020, respectively. Revenue in 2022 also includes the loss on deconsolidation of the Russian businesses of \$27 million. Revenue in 2021 includes the gain on the consolidation of Marsh India of \$267 million and a net gain on disposition of business of approximately \$50 million.

⁽b) Includes inter-segment revenue of \$59 million, \$49 million and \$84 million in 2022, 2021 and 2020, respectively, and equity method income of \$1 million in 2022 and \$5 million in 2020, respectively. Revenue in 2022, also includes a net gain on the sale of the Mercer U.S. affinity business of \$112 million, partially offset by the loss on deconsolidation of the Russian businesses of \$12 million.

⁽c) Includes equity method investments of \$50 million, \$53 million and \$165 million at December 31, 2022, 2021 and 2020, respectively.

⁽d) Includes equity method investments of \$6 million at December 31, 2022 and \$5 million at December 31, 2021 and 2020.

⁽e) Corporate assets primarily include insurance recoverables, pension related assets, the owned portion of the Company headquarters building and intercompany eliminations.

Details of operating segment revenue are as follows:

Asia Pacific

Total

Corporate/Eliminations

Other

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Risk and Insurance Services			
Marsh	\$ 10,585 \$	10,214 \$	8,628
Guy Carpenter	2,060	1,871	1,709
Total Risk and Insurance Services	12,645	12,085	10,337
Consulting			
Mercer	5,345	5,254	4,928
Oliver Wyman Group	2,794	2,535	2,048
Total Consulting	8,139	7,789	6,976
Total Segments	20,784	19,874	17,313
Corporate/Eliminations	(64)	(54)	(89)
Total	\$ 20,720 \$	19,820 \$	17,224
Information by geographic area is as follows:			
For the Years Ended December 31,			
(In millions)	2022	2021	2020
Revenue			
United States	\$ 10,215 (a) \$	9,343 \$	8,168
United Kingdom	3,114	3,130 ^(c)	2,818
Continental Europe	3,223 (b)	3,219	2,881

2,537

1,695

20,784

20,720

(64)

\$

2,093

1,353

(89)

17,313

17,224

2,617

1,565

19,874

19,820

(54)

⁽d) Revenue in 2021 includes gain on the consolidation of Marsh India of \$267 million.

For the Years Ended December 31,			
(In millions)	2022	2021	2020
Fixed Assets, Net			
United States	\$ 473	\$ 484	\$ 492
United Kingdom	166	116	115
Continental Europe	68	68	74
Asia Pacific	81	96	105
Other	83	83	70
Total	\$ 871	\$ 847	\$ 856

 $^{^{(}a)}$ Revenue in 2022 includes a net gain from the sale of the Mercer U.S. affinity business of \$112 million.

⁽b) Revenue in 2022 includes the loss on deconsolidation of the Company's Russian businesses at Marsh and Oliver Wyman of \$27 million and \$12 million, respectively.

⁽c) Revenue in 2021 includes a net gain on the disposition of businesses of approximately \$50 million.

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and the Board of Directors of Marsh & McLennan Companies, Inc.

Opinion on the Financial Statements

We have audited the accompanying consolidated balance sheets of Marsh & McLennan Companies, Inc. and subsidiaries (the "Company") as of December 31, 2022 and 2021, the related consolidated statements of income, comprehensive income, cash flows, and equity for each of the three years in the period ended December 31, 2022, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2022 and 2021, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2022, in conformity with accounting principles generally accepted in the United States of America.

We have also audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the Company's internal control over financial reporting as of December 31, 2022, based on criteria established in *Internal Control — Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission and our report dated February 13, 2023 expressed an unqualified opinion on the Company's internal control over financial reporting.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

Critical Audit Matter

The critical audit matter communicated below is a matter arising from the current-period audit of the financial statements that was communicated or required to be communicated to the audit committee and that (1) relates to accounts or disclosures that are material to the financial statements and (2) involved our especially challenging, subjective, or complex judgments. The communication of critical audit matters does not alter in any way our opinion on the financial statements, taken as a whole, and we are not, by communicating the critical audit matter below, providing a separate opinion on the critical audit matter or on the accounts or disclosures to which it relates.

Liability for Errors and Omissions — Refer to Notes 1 and 16 to the financial statements

Critical Audit Matter Description

The Company is subject to a significant number of claims, lawsuits and proceedings in the ordinary course of business. Such claims and lawsuits consist principally of alleged errors and omissions ("E&O") in connection with the performance of professional services. These claims may seek damages, including punitive and treble damages, in amounts that could be significant. The Company uses case level reviews performed by inside and outside counsel, internal actuarial analysis and other methods to estimate potential losses resulting from reported and unreported claims.

Given that the determination of the liability for E&O requires management to make significant estimates and assumptions in projecting ultimate settlement values of reported and unreported claims, performing audit procedures to evaluate the reasonableness of such estimates and assumptions required a high degree of auditor judgment, including the need to involve our actuarial specialists.

How the Critical Audit Matter Was Addressed in the Audit

Our audit procedures related to the determination of the liability for E&O included the following, among others:

- We tested the effectiveness of internal controls related to the determination of the liability for E&O, including controls over the projection of ultimate settlement values of reported and unreported claims determined through internal actuarial analyses, management's review of the appropriateness of the assumptions used and calculation of case loss estimates, and management's independent review of case level estimates provided by inside and outside counsel, as applicable.
- For selected E&O matters, we evaluated the reasonableness of management's case loss estimates and, as applicable, made inquiries of the Company's inside and outside counsel regarding the status of these matters and likelihood of settlement.
- We compared total incurred losses and current case estimates as of the balance sheet date to amounts reported in prior periods to evaluate trends and developments in reported cases.
- With the assistance of our actuarial specialists, we evaluated the reasonableness of the assumptions and methodologies involved in the development of the liability for E&O by:
 - Testing the underlying data that served as the basis for the actuarial analysis, including historical claims and case loss estimates, to evaluate whether the inputs to the actuarial estimate were reasonable.
 - Comparing management's prior-year assumptions of expected development and ultimate loss to actual amounts incurred during
 the current year to identify potential bias in the determination of the liability for E&O.
 - Developing a range of independent estimates and comparing those to the liability for E&O recorded by the Company.

/s/ Deloitte & Touche LLP New York, New York February 13, 2023

We have served as the Company's auditor since 1989.

Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure.

None.

Item 9A. Controls and Procedures.

<u>Disclosure Controls and Procedures</u>. Based on their evaluation, as of the end of the period covered by this annual report on Form 10-K, the Company's chief executive officer and chief financial officer have concluded that the Company's disclosure controls and procedures (as defined in Rules 13a-15(e) or 15d-15(e) under the Securities Exchange Act of 1934) are effective.

Internal Control over Financial Reporting.

(a) Management's Annual Report on Internal Control Over Financial Reporting

MANAGEMENT'S ANNUAL REPORT ON INTERNAL CONTROL OVER FINANCIAL REPORTING

The management of Marsh & McLennan Companies, Inc. is responsible for establishing and maintaining adequate internal control over financial reporting for the Company. The Company's internal control over financial reporting is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles.

The Company's internal control over financial reporting includes those policies and procedures relating to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the Company; the recording of all necessary transactions to permit the preparation of the Company's consolidated financial statements in accordance with generally accepted accounting principles; the proper authorization of receipts and expenditures in accordance with authorizations of the Company's management and directors; and the prevention or timely detection of the unauthorized acquisition, use or disposition of assets that could have a material effect on the Company's consolidated financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

Management evaluated the effectiveness of the Company's internal control over financial reporting as of December 31, 2022 under the supervision and with the participation of the Company's principal executive and principal financial officers. In making this evaluation, management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission (COSO) in Internal Control—Integrated Framework issued in 2013. Based on its evaluation, management determined that the Company maintained effective internal control over financial reporting as of December 31, 2022.

Deloitte & Touche LLP, the Independent Registered Public Accounting Firm that audited and reported on the Company's consolidated financial statements included in this annual report on Form 10-K, also issued an audit report on the effectiveness of the Company's internal control over financial reporting as of December 31, 2022.

(b) Audit Report of the Registered Public Accounting Firm.

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and the Board of Directors of Marsh & McLennan Companies, Inc.

Opinion on Internal Control over Financial Reporting

We have audited the internal control over financial reporting of Marsh & McLennan Companies, Inc. and subsidiaries (the "Company") as of December 31, 2022, based on criteria established in *Internal Control — Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). In our opinion, the Company maintained, in all material respects, effective internal control over financial reporting as of December 31, 2022, based on criteria established in *Internal Control — Integrated Framework (2013)* issued by COSO.

We have also audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the consolidated financial statements as of and for the year ended December 31, 2022, of the Company and our report dated February 13, 2023, expressed an unqualified opinion on those financial statements.

Basis for Opinion

The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting, included in the accompanying Management's Annual Report on Internal Control Over Financial Reporting. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control based on the assessed risk, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

Definition and Limitations of Internal Control over Financial Reporting

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ Deloitte & Touche LLP New York, New York February 13, 2023

(c) Changes in Internal Control Over Financial Reporting

There were no changes in the Company's internal control over financial reporting identified in connection with the evaluation required by Rules 13a-15(d) or 15d-15(d) under the Securities Exchange Act of 1934 that occurred during the quarter ended December 31, 2021 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

Item 9B. Other Information.

None.

PART III

Item 10. Directors, Executive Officers and Corporate Governance.

Information as to the directors and nominees for the board of directors of the Company is incorporated herein by reference to the material set forth under the heading "Item 1: Election of Directors" in the 2023 Proxy Statement.

The executive officers and executive officer appointees of the Company are Paul Beswick, Katherine J. Brennan, John Q. Doyle, Martine Ferland, Carmen Fernandez, Dean Klisura, Mark C. McGivney, Martin South and Nick Studer. Information with respect to these individuals is provided in Part I, Item 1 above under the heading "Executive Officers of the Company".

The information set forth in the 2023 Proxy Statement in the sections "Corporate Governance—Codes of Conduct", "Board of Directors and Committees—Committees—Audit Committee" and "Additional Information—Transactions with Management and Others" is incorporated herein by reference.

Item 11. Executive Compensation.

The information set forth in the sections "Additional Information—Director Compensation" and "Executive Compensation—Compensation of Executive Officers" in the 2023 Proxy Statement is incorporated herein by reference.

Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters.

The information set forth in the sections "Additional Information—Stock Ownership of Directors, Management and Certain Beneficial Owners" and "Additional Information—Equity Compensation Plan Information" in the 2023 Proxy Statement is incorporated herein by reference.

Item 13. Certain Relationships and Related Transactions, and Director Independence.

The information set forth in the sections "Corporate Governance—Director Independence", "Corporate Governance—Review of Related-Person Transactions" and "Additional Information—Transactions with Management and Others" in the 2023 Proxy Statement is incorporated herein by reference.

Item 14. Principal Accountant Fees and Services.

The information set forth under the heading "Item 3: Ratification of Selection of Independent Registered Public Accounting Firm—Fees of Independent Registered Public Accounting Firm" in the 2023 Proxy Statement is incorporated herein by reference.

PART IV

Item 15. Exhibits and Financial Statement Schedules. †

The following documents are filed as a part of this report:

(1) Consolidated Financial Statements:

Consolidated Statements of Income for each of the three years in the period ended December 31, 2022

Consolidated Statements of Comprehensive Income for each of the three years in the period ended December 31, 2022

Consolidated Balance Sheets as of December 31, 2022 and 2021

Consolidated Statements of Cash Flows for each of the three years in the period ended December 31, 2022

Consolidated Statements of Shareholders Equity for each of the three years in the period ended December 31, 2022

Notes to Consolidated Financial Statements

Report of Independent Registered Public Accounting Firm

- (2) All required Financial Statement Schedules are included in the Consolidated Financial Statements or the Notes to Consolidated Financial Statements.
- (3) The following exhibits are filed as a part of this report:
- (2.1) Stock Purchase Agreement, dated as of June 6, 2010, by and between Marsh & McLennan Companies, Inc. and

 Altegrity, Inc. (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2010)
- (2.2) Rule 2.7 Announcement, dated as of September 18, 2018 (incorporated by reference to the Company's Current Report on Form 8-K dated September 18, 2018)

rmitted by Item 601(b)(4)(iii)(A) of Regulation S-K, the Company has not filed with this Form 10-K certain instruments defining the rights of holders of long-term debt of the Company and its subsidiaries because the total amount of securities authorized under any of such instruments does not exceed 10% of the total assets of the Company and its subsidiaries on a consolidated basis. The Company agrees to furnish a copy of any such agreement to the Commission upon request.

- (2.3) Co-operation Agreement, dated as of September 18, 2018, by and among Marsh & McLennan Companies, Inc., MMC Treasury

 Holdings (UK) Limited and Jardine Lloyd Thompson Group plc. (incorporated by reference to the Company's Current Report

 on Form 8-K dated September 18, 2018)
- (3.1) Restated Certificate of Incorporation of Marsh & McLennan Companies, Inc. (incorporated by reference to the Company's Current Report on Form 8-K dated July 17, 2008)
- (3.2) Amended and Restated By-Laws of Marsh & McLennan Companies, Inc. (incorporated by reference to the Company's Current Report on Form 8-K dated January 12, 2017)
- (4.1) Indenture dated as of June 14, 1999 between Marsh & McLennan Companies, Inc. and State Street Bank and Trust

 Company, as trustee (incorporated by reference to the Company's Registration Statement on Form S-3, Registration

 No. 333-108566)
- (4.2) Third Supplemental Indenture dated as of July 30, 2003 between Marsh & McLennan Companies, Inc. and U.S. Bank National

 Association (as successor to State Street Bank and Trust Company), as trustee (incorporated by reference to the Company's

 Quarterly Report on Form 10-Q for the quarter ended June 30, 2003)
- (4.3) <u>Indenture dated as of March 19, 2002 between Marsh & McLennan Companies, Inc. and State Street Bank and Trust Company, as trustee (incorporated by reference to the Company's Registration Statement on Form S-4, Registration No. 333-87510)</u>
- (4.4) Indenture, dated as of July 15, 2011, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee

 (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011)
- (4.5) First Supplemental Indenture, dated as of July 15, 2011, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011)
- (4.6) Form of Third Supplemental Indenture between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Current Report on Form 8-K dated September 24, 2013)
- (4.7) Form of Fourth Supplemental Indenture between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Current Report on Form 8-K dated May 27, 2014)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

- (4.8) Form of Fifth Supplemental Indenture between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee

 (incorporated by reference to the Company's Current Report on Form 8-K dated September 10, 2014)
- (4.9) Sixth Supplemental Indenture, dated as of March 6, 2015, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2015)
- (4.10) Seventh Supplemental Indenture, dated as of September 14, 2015, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Current Report on Form 8-K filed on September 14, 2015)
- (4.11) <u>Eighth Supplemental Indenture, dated as of March 14, 2016, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Quarterly Report on Form 10-Q filed on May 2, 2016)</u>
- (4.12) Ninth Supplemental Indenture, dated as of January 12, 2017, between Marsh & McLennan Companies, Inc. and The Bank of

 New York Mellon, as trustee (incorporated by reference to the Company's Annual Report on Form 10-K filed on February 24,

 2017)
- (4.13) <u>Tenth Supplemental Indenture, dated as of March 1, 2018, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Current Report on Form 8-K filed on March 1, 2018)</u>
- (4.14) <u>Eleventh Supplemental Indenture, dated January 15, 2019, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Current Report on Form 8-K filed on January 15, 2019)</u>
- (4.15) Twelfth Supplemental Indenture, dated March 21, 2019, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to the Company's Current Report on Form 8-K filed on March 21, 2019)
- (4.16) Thirteenth Supplemental Indenture, dated May 7, 2020, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to Company's Current Report on Form 8-K dated May 7, 2020)
- (4.17) <u>Fourteenth Supplemental Indenture, dated December 8, 2021, between Marsh & McLennan Companies, Inc. and The Bank of New York Mellon, as trustee (incorporated by reference to Company's Current Report on Form 8-K dated December 9, 2021)</u>

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

(4.18)Fifteenth Supplemental Indenture, dated October 31, 2022, between Marsh & McLennan Companies, Inc. and the Bank of New York Mellon, as trustee (incorporated by reference to the Company's Current Report on Form 8-K dated October 31, 2022) (4.19)Description of Marsh & McLennan Companies, Inc.'s Securities Registered Pursuant to Section 12 of the Securities Exchange Act of 1934 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2019) (10.1)*Marsh & McLennan Companies, Inc. U.S. Employee 1996 Cash Bonus Award Voluntary Deferral Plan (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 1996) (10.2)*Marsh & McLennan Companies, Inc. U.S. Employee 1997 Cash Bonus Award Voluntary Deferral Plan (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 1997) (10.3)*Marsh & McLennan Companies, Inc. U.S. Employee 1998 Cash Bonus Award Voluntary Deferral Plan (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 1998) (10.4)*Marsh & McLennan Companies, Inc. 2000 Senior Executive Incentive and Stock Award Plan (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 1999) (10.5)*Amendments to Marsh & McLennan Companies, Inc. 2000 Senior Executive Incentive and Stock Award Plan and the Marsh & McLennan Companies, Inc. 2000 Employee Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2005) *Form of Awards under the Marsh & McLennan Companies, Inc. 2000 Senior Executive Incentive and Stock Award (10.6)Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended September 30, 2004) (10.7)*Additional Forms of Awards under the Marsh & McLennan Companies, Inc. 2000 Senior Executive Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March

31, 2005)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

- (10.8)*Marsh & McLennan Companies, Inc. 2000 Employee Incentive and Stock Award Plan (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2001) (10.9)*Form of Awards under the Marsh & McLennan Companies, Inc. 2000 Employee Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2004) (10.10)*Additional Forms of Awards under the Marsh & McLennan Companies, Inc. 2000 Employee Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2005) (10.11)*Form of Long-term Incentive Award under the Marsh & McLennan Companies, Inc. 2000 Senior Executive Incentive and Stock Award Plan and the Marsh & McLennan Companies, Inc. 2000 Employee Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2006) (10.12)*Form of 2012 Long-term Incentive Award under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2012) (10.13)*Form of 2013 Long-term Incentive Award under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended March 31, 2013)
- *Form of 2014 Long-term Incentive Award under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock

 Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31,

 2014)
- (10.15) *Form of 2015 Long-term Incentive Award under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock

 Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31,

 2015)
- (10.16) *Form of 2016 Long-term Incentive Award under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock

 Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31,

 2016)
- (10.17) *Form of Deferred Stock Unit Award, with grant dates from March 1, 2018 through February 1, 2019, under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2018)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

- (10.18) *Form of Deferred Stock Unit Award, with grant dates from March 1, 2019 through February 1, 2020, under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2019).
- *Form of Deferred Stock Unit Award, with grant dates from May 1, 2019 through February 1, 2020, under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan Form A (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended June 30, 2019)
- *Form of Deferred Stock Unit Award, with grant dates from May 1, 2019 through February 1, 2020, under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan Form B (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended June 30, 2019)
- *Form of Deferred Stock Unit Award, with grant dates from March 1, 2020 through February 1, 2021, under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2020)
- (10.22) *Form of Deferred Stock Unit Award, with grant dates from March 1, 2021 through February 1, 2022, under the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan Cliff Vesting (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2021)
- *Form of Deferred Stock Unit Award, with grant dates from March 1, 2021 through February 1, 2022, under the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan Ratable Vesting (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2021)
- *Form of Restricted Stock Unit Award, dated as of February 21, 2018 under the Marsh & McLennan Companies, Inc.

 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended March 31, 2018)
- *Form of Restricted Stock Unit Award, dated as of February 19, 2019, under the Marsh & McLennan Companies, Inc.

 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2019)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

- (10.26) *Form of Restricted Stock Unit Award, dated as of May 1, 2019, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan Form A (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2019)
- *Form of Restricted Stock Unit Award, dated as of May 1, 2019, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan Form B (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2019)
- (10.28) *Form of Restricted Stock Unit Award, dated as of February 22, 2021, under the Marsh & McLennan Companies, Inc.

 2020 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2021)
- (10.29) *Form of Restricted Stock Unit Award, dated as of May 1, 2019, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan Form C (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2019)
- (10.30) *Form of Restricted Stock Unit Award, dated as of February 19, 2020, under the Marsh & McLennan Companies, Inc.

 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2020)
- *Form of Performance Stock Unit Award, dated as of February 21, 2018, under the Marsh & McLennan Companies, Inc.

 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2018)
- (10.32) *Form of Performance Stock Unit Award, dated as of February 19, 2019, under the Marsh & McLennan Companies, Inc.

 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended March 31, 2019)
- (10.33) *Form of Performance Stock Unit Award, dated as of May 1, 2019, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2019)
- (10.34) *Form of Performance Stock Unit Award, dated as of February 19, 2020, under the Marsh & McLennan Companies, Inc.

 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2020)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

- *Form of Performance Stock Unit Award, dated as of February 22, 2021, under the Marsh & McLennan Companies, Inc.

 2020 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2021)
- (10.36) *Form of Stock Option Award, dated as of February 22, 2017, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2017)
- (10.37) *Form of Stock Option Award, dated as of February 21, 2018, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2018)
- (10.38) *Form of Stock Option Award, dated as of February 19, 2019, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2019)
- (10.39) *Form of Stock Option Award, dated as of May 1, 2019, under the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2019).
- (10.40) *Form of Stock Option Award, dated as of February 19, 2020, under the Marsh & McLennan Companies, Inc. 2011

 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2020)
- (10.41) *Form of Stock Option Award, dated as of February, 22, 2021, under the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2021)
- (10.42) *Form of Deferred Stock Unit Award, with grant dates from March 1, 2022 through February 1, 2023, under the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan Cliff Vesting (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022)
- *Form of Deferred Stock Unit Award, with grant dates from March 1, 2022 through February 1, 2023, under the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan Ratable Vesting (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022)
- *Form of Restricted Stock Unit Award, dated as of February 23, 2022, under the Marsh & McLennan Companies, Inc.

 2020 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

- *Form of Performance Stock Unit Award, dated as of February 23, 2022, under the Marsh & McLennan Companies, Inc.

 2020 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022).
- (10.46) *Form of Stock Option Award, dated as of February 23, 2022, under the Marsh & McLennan Companies, Inc. 2020

 Incentive and Stock Award Plan (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022)
- (10.47) *Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Registration Statement on Form S-8 dated August 5, 2011, Registration No. 333-176084)
- (10.48) *Amendment to the Marsh & McLennan Companies, Inc. 2011 Incentive and Stock Award Plan (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2018)
- (10.49) *Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan (incorporated by reference from Exhibit C to the Company's Definitive Proxy Statement on Schedule 14A filed on April 3, 2020)
- (10.50) 2023 Amendment to the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan effective January
 12, 2023
- *Amendments to Certain Marsh & McLennan Companies Equity-Based Awards Due to U.S. Tax Law Changes Affecting
 Equity-Based Awards granted under the Marsh & McLennan Companies, Inc. 2000 Senior Executive Incentive and Stock
 Award Plan and the Marsh & McLennan Companies, Inc. 2000 Employee Incentive and Stock Award Plan, effective

 January 1, 2009 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended

 December 31, 2008)
- (10.52) <u>*Section 409A Amendment Document, effective as of January 1, 2009 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2008)</u>
- (10.53) *Section 409A Amendment Regarding Payments Conditioned Upon Employment-Related Action to Any and All Plans or
 Arrangements Entered into by the Marsh & McLennan Companies, Inc., or any of its Direct or Indirect Subsidiaries, that
 Provide for the Payment of Section 409A Nonqualified Deferred Compensation, effective December 21, 2012 (incorporated
 by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2012)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

- (10.54) *Marsh & McLennan Companies Supplemental Savings & Investment Plan (formerly the Marsh & McLennan

 Companies Stock Investment Supplemental Plan) Restatement, effective January 1, 2012 (incorporated by reference to the

 Company's Annual Report on Form 10-K for the year ended December 31, 2012)
- (10.55) *First Amendment to the Marsh & McLennan Companies Supplemental Savings & Investment Plan Restatement
 effective January 1, 2012 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended
 December 31, 2016)
- (10.56) *Second Amendment to the Marsh & McLennan Companies Supplemental Savings & Investment Plan Restatement effective January 1, 2012 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2017)
- *Third Amendment to the Marsh & McLennan Companies Supplemental Savings & Investment Plan Restatement

 effective January 1, 2012 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended

 December 31, 2018)
- *Fourth Amendment to the Marsh & McLennan Companies Supplemental Savings & Investment Plan Restatement effective January 1, 2012 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2020)
- (10.59) <u>Marsh & McLennan Companies Supplemental Savings & Investment Plan (formerly the Marsh & McLennan Companies</u>

 <u>Stock Investment Supplemental Plan) Restatement effective January 1, 2022)</u>
- (10.60) *Marsh & McLennan Companies Benefit Equalization Plan and Marsh & McLennan Companies Supplemental

 Retirement Plan as Restated, effective January 1, 2012 (incorporated by reference to the Company's Annual Report on Form

 10-K for the year ended December 31, 2012)
- *First Amendment to the Marsh & McLennan Companies Benefit Equalization Plan and Marsh & McLennan Companies

 Supplemental Retirement Plan as Restated effective January 1, 2012 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2016
- *Second Amendment to the Marsh & McLennan Companies Benefit Equalization Plan and Marsh & McLennan

 Companies Supplemental Retirement Plan as Restated effective January 1, 2012 (incorporated by reference to the

 Company's Annual Report on Form 10-K for the year ended December 31, 2016)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

(10.63)	*Marsh & McLennan Companies, Inc. Senior Executive Severance Pay Plan (incorporated by reference to the
	Company's Quarterly Report on Form 10-Q for the Quarter ended March 31, 2008)
(10.64)	*Amendment to the Marsh & McLennan Companies, Inc. Senior Executive Severance Pay Plan, effective December 3
	2009 (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2009)
(10.65)	*Marsh & McLennan Companies, Inc. Senior Management Incentive Compensation Plan (incorporated by reference to
	the Company's Annual Report on Form 10-K for the year ended December 31, 1994)
(10.66)	*Marsh & McLennan Companies, Inc. Directors' Stock Compensation Plan - May 31, 2009 Restatement (incorporated
	by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2009)
(10.67)	*Marsh & McLennan Companies International Retirement Plan As Amended and Restated Effective January 1, 2009
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2014)
(10.68)	*Description of compensation arrangements for independent directors of Marsh & McLennan Companies, Inc. effective
	June 1, 2021 (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30,
	<u>2021)</u>
(10.69)	*Letter Agreement, effective as of March 20, 2013, between Marsh & McLennan Companies, Inc. and Daniel S. Glase
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2013)
(10.70)	*Non-Competition and Non-Solicitation Agreement, effective as of September 18, 2013, between Marsh & McLennan
	Companies, Inc. and Daniel S. Glaser (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the
	quarter ended September 30, 2013)
(10.71)	*Letter Agreement, effective as of May 14, 2014, between Marsh & McLennan Companies, Inc. and Daniel S. Glaser
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2014)
(10.72)	*Letter Agreement, effective as of February 22, 2016, between Marsh & McLennan Companies, Inc. and Daniel S.
	Glaser (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2016)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

(10.73)*Letter Agreement, effective as of February 22, 2017, between Marsh & McLennan Companies, Inc. and Daniel S. Glaser (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended September 30, 2017) (10.74)*Letter Agreement, dated as of September 18, 2019, between Marsh & McLennan Companies, Inc. and Daniel S. Glaser (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2019) (10.75)*Letter Agreement Amendment, dated September 23, 2022, between Marsh & McLennan Companies, Inc. and Daniel S. Glaser (incorporated by reference to the Company's Current Report on Form 8-K dated September 23, 2022) (10.76)*Letter Agreement, effective as of January 1, 2016, between Marsh & McLennan Companies, Inc. and Mark C. McGivney (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2015) (10.77)*Non-Competition and Non-Solicitation Agreement, effective as of January 1, 2016, between Marsh & McLennan Companies, Inc. and Mark C. McGivney (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2015) (10.78)*Letter Agreement, effective as of January 17, 2018, between Marsh & McLennan Companies, Inc. and Mark C. McGivney (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2017) (10.79)*Letter Agreement, effective as of January 16, 2019, between Marsh & McLennan Companies, Inc. and Mark C. McGivney (incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2018)

(10.81) *Letter Agreement, effective as of July 5, 2017, between Marsh & McLennan Companies, Inc. and John Q. Doyle (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended March 31, 2018)

*Letter Agreement, effective as of September 22, 2022, between Marsh & McLennan Companies, Inc. and Mark C.

McGivney (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended September 30,

(10.80)

2022)

- (10.82) *Non-Competition and Non-Solicitation Agreement, dated as of February 25, 2016, between Marsh & McLennan

 Companies, Inc. and John Q. Doyle (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended March 31, 2018)
- (10.83) *Letter Agreement, effective as of January 15, 2020, between Marsh & McLennan Companies, Inc. and John Q. Doyle (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the guarter ended March 31, 2020)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

(10.84)	*Letter Agreement, effective as of January 1, 2022 between Marsh & McLennan Companies, Inc. and John Q. Doyle
	(incorporated by reference to the Company's Annual Report on Form 10-K for the year ended December 31, 2021)
(10.85)	*Letter Amendment, dated November 10, 2022, between Marsh & McLennan Companies, Inc. and John Q. Doyle
	(incorporated by reference to the Company's Current Report on Form 8-K/A dated September 26, 2022)
(10.86)	*Letter Agreement, effective as of March 1, 2019, between Marsh & McLennan Companies, Inc. and Martine Ferland
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2020)
(10.87)	*Non-Competition and Non-Solicitation Agreement, effective as of March 1, 2016, between Marsh & McLennan
	Companies, Inc. and Martine Ferland (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the
	quarter ended March 31, 2020)
(10.88)	*Letter Agreement, effective as of January 20, 2021, between Marsh & McLennan Companies, Inc. and Martine Ferland
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2021)
(10.89)	*Letter Agreement, effective as of April 1, 2022, between Marsh & McLennan Companies, Inc. and Martine Ferland
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022)
(10.90)	*Letter Agreement, effective as of February 19, 2019, between Marsh & McLennan Companies, Inc. and Peter C. Hearn
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2019)
(10.91)	*Non-Competition and Non-Solicitation Agreement, effective as of June 1, 2016, between Marsh & McLennan
	Companies, Inc. and Peter C. Hearn (incorporated by reference to the Company's Quarterly Report on Form 10-Q for the
	quarter ended March 31, 2019)
(10.92)	*Letter Agreement, effective as of January 1, 2022, between Marsh & McLennan Companies, Inc. and Peter C. Hearn
	(incorporated by reference to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022)
(10.93)	Calculation Agency Agreement, dated as of January 15, 2019, between Marsh & McLennan Companies, Inc. and The
	Bank of New York Mellon, as calculation agent (incorporated by reference to the Company's Current Report on Form 8-K
	filed on January 15, 2019)

^{*}Management contract or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 15(b) of Form 10-K.

(10.94)	Paying Agency Agreement, dated as of March 21, 2019, between Marsh & McLennan Companies, Inc. and The Bank of
	New York Mellon, London Branch, as paying agent (incorporated by reference to the Company's Current Report on Form 8-K
	filed on March 21, 2019)
(10.95)	Shareholder Undertaking, dated as of September 18, 2018 (incorporated by reference to the Company's Current Report
	on Form 8-K dated September 18, 2018)
(10.96)	Form of Director Undertaking, dated as of September 18, 2018 (incorporated by reference to the Company's Current
	Report on Form 8-K dated September 18, 2018)
(10.97)	Bridge Loan Agreement, dated as of September 18, 2018 by and between Marsh & McLennan Companies, Inc., the
	lenders party thereto and Goldman Sachs Bank USA, as administrative agent (incorporated by reference to the Company's
	Current Report on Form 8-K dated September 18, 2018)
(10.98)	Calculation Agency Agreement, dated as of January 15, 2019, between Marsh & McLennan Companies, Inc. and The
	Bank of New York Mellon, as calculation agent (incorporated by reference to the Company's Current Report on Form 8-K
	filed on January 15, 2019)
(10.99)	Amended and Restated 5 Year Credit Agreement, dated as of April 2, 2021, among Marsh & McLennan Companies,
	Inc., the designated subsidiaries party thereto as borrowers, Citibank, N.A., as administrative agent, and the lenders from
	time to time party thereto (incorporated by reference to the Company's Current Report on Form 8-K filed on April 2, 2021)
(14.1)	Code of Ethics for Chief Executive and Senior Financial Officers (incorporated by reference to the Company's Annual
	Report on Form 10-K for the year ended December 31, 2002)
(21.1)	List of Subsidiaries of Marsh & McLennan Companies, Inc.
(23.1)	Consent of Independent Registered Public Accounting Firm
(24.1)	Power of Attorney (included on signature page)
(31.1)	Rule 13a-14(a)/15d-14(a) Certification of Chief Executive Officer
(31.2)	Rule 13a-14(a)/15d-14(a) Certification of Chief Financial Officer
(32.1)	Section 1350 Certifications
101.INS	XBRL Instance Document
101.SCH	XBRL Taxonomy Extension Schema
101.CAL	XBRL Taxonomy Extension Calculation Linkbase
101.DEF	XBRL Taxonomy Extension Definition Linkbase
101.LAB	XBRL Taxonomy Extension Label Linkbase
101.PRE	XBRL Taxonomy Extension Presentation Linkbase

Cover Page Interactive Data File (formatted as inline XBRL and contained in Exhibit 101)

104.

Item 16. Form 10-K Summary.

None.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

MARSH & McLENNAN COMPANIES, INC.

Dated: February 13, 2023 By /S/ JOHN Q. DOYLE

John Q. Doyle President and Chief Executive Officer

Each person whose signature appears below hereby constitutes and appoints Courtenay Birchler and Connor Kuratek, and each of them singly, such person's lawful attorneys-in-fact and agents, with full power to them and each of them to sign for such person, in the capacity indicated below, any and all amendments to this Annual Report on Form 10-K filed with the Securities and Exchange Commission.

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities indicated this 13th day of February, 2023.

Name	Title	Date
/s/ JOHN Q. DOYLE John Q. Doyle	Director, President & Chief Executive Officer	February 13, 2023
/s/ MARK C. McGIVNEY Mark C. McGivney	Chief Financial Officer	February 13, 2023
/s/ STACY M. MILLS Stacy M. Mills	Vice President & Controller (Chief Accounting Officer)	February 13, 2023
/s/ Anthony K. Anderson Anthony K. Anderson	Director	February 13, 2023
/s/ Hafize Gaye Erkan Hafize Gaye Erkan	Director	February 13, 2023
/s/ Oscar Fanjul Oscar Fanjul	Director	February 13, 2023
/s/ H. EDWARD HANWAY H. Edward Hanway	Director	February 13, 2023
/s/ DEBORAH C. HOPKINS Deborah C. Hopkins	Director	February 13, 2023
/s/ Tamara Ingram Tamara Ingram	Director	February 13, 2023
/s/ JANE H. LUTE Jane H. Lute	Director	February 13, 2023
/s/ Steven A. Mills Steven A. Mills	Director	February 13, 2023
/s/ Bruce P. Nolop Bruce P. Nolop	Director	February 13, 2023
/s/ Morton O. Schapiro Morton O. Schapiro	Director	February 13, 2023
/s/ LLOYD M. YATES Lloyd M. Yates	Director	February 13, 2023
/S/ R. DAVID YOST R. David Yost	Director	February 13, 2023

2023 AMENDMENT TO THE MARSH & MCLENNAN COMPANIES, INC. 2020 INCENTIVE AND STOCK AWARD PLAN

Effective January 12, 2023, the last sentence of Section 10(e) of the Marsh & McLennan Companies, Inc. 2020 Incentive and Stock Award Plan shall be deleted in its entirety.

MARSH & MCLENNAN COMPANIES

SUPPLEMENTAL SAVINGS & INVESTMENT PLAN

(formerly the Marsh & McLennan Companies Stock Investment Supplemental Plan)

Restatement effective January 1, 2022

{00835594-7}

PREAMBLE

Effective July 1, 1992, Marsh & McLennan Companies, Inc. (the "Company") adopted the Marsh & McLennan Companies Stock Investment Supplemental Plan (the "Plan"). The Plan provides benefits to certain employees of the Company and its participating subsidiaries whose benefits and contributions under the Marsh & McLennan Companies 401(k) Savings & Investment Plan (formerly the Marsh & McLennan Companies Stock Investment Plan) (the "Basic Plan") are limited by certain provisions of the Internal Revenue Code of 1986, as amended (the "Code").

It is intended that benefits paid under the Plan shall be paid under an arrangement that is, for purposes of Employee Retirement Income Security Act of 1974, unfunded and maintained primarily for the purpose of providing deferred compensation for a select group of management or highly compensated employees. It is further intended that the Plan complies with the requirements of Section 409A of the Code, including the transition rules issued by the Internal Revenue Service in Notice 2005-1, 2005-1 C.B. 274 and subsequent guidance, to the extent applicable.

The Plan has been amended and restated several times since its establishment, with the most recent restatement effective January 1, 2012 (the "January 2012 Restatement"). The Plan has since been amended. To incorporate the amendments previously made to the January 2012 Restatement and for administrative convenience, the Company hereby restates the Plan in its entirety, effective January 1, 2022.

{00835594-7}

ARTICLE 1 DEFINITIONS

The following terms when used in this Plan have the designated meanings unless a different meaning is clearly required by the context.

- 1.1 Board of Directors, Catch-Up Contributions, Code, Company, Contribution Authorization, Eligible

 Employee, Employee, Investment Fund, MMC Fixed Company Contributions, MMC Stock, Participating Company, Matching

 Contributions, Non-Covered Company, Plan Year, Pre-Tax Contributions, Roth Contributions, Spouse, Stock Fund, and Year of

 Service have the meanings given them in the Basic Plan as from time to time in effect.
- 1.2 Account means an account established on behalf of a Participant by the Company pursuant to Section 4.1(a). Solely for the purpose of ensuring the Company's compliance with Section 162(m) of the Code and the Treasury Regulations thereunder, the Company shall maintain, as applicable, a record of the portion of each Participant's Benefit that is determined by the Administrative Committee not to be subject to the amendments made to Section 162(m) by Section 13601 of Public Law 115-97 (commonly referred to as the Tax Cuts and Jobs Act).
- 1.3 <u>Administrative Committee</u> means the individual or entity appointed from time to time by the Company to address certain administrative matters under the Plan.
- 1.4 <u>Annual Enrollment</u> means the process adopted by the Plan Administrator for an Eligible Employee to enroll in the Plan with respect to a Plan Year, provided that each Annual Enrollment period as determined by the Plan Administrator

{00835594-7} -ii-

shall end no later than the end of the Plan Year immediately preceding the Plan Year to which an Enrollment Authorization relates.

- 1.5 <u>Basic Plan</u> means the Marsh & McLennan Companies 401(k) Savings & Investment Plan (formerly the Marsh & McLennan Companies Stock Investment Plan).
- 1.6 <u>Beneficiary</u> means the person or persons designated pursuant to Article 6 to receive a benefit in the event of a Participant's death before his or her Benefit has been paid in full.
 - 1.7 <u>Benefit</u> means the benefit amount described in Section 5.1.
 - 1.8 <u>Change in Control</u> has the meaning set forth in Section 5.5.
- 1.9 <u>Committee</u> means either the Administrative Committee or the Investment Committee. Any reference to "Committee" shall be considered a reference to either one or both of the Administrative Committee and Investment Committee as the context may require.
- 1.10 <u>Compensation</u> means a Participant's base salary from a Participating Company before any deductions for taxes and shall include any reductions in Compensation under the Plan, Pre-Tax Contributions, Roth Contributions, and any salary reduction amounts under any plan described in Section 125 of the Code or any "qualified transportation fringe benefit plan" described in Section 132(f) of the Code.
- 1.11 <u>Compensation Limit</u> means, with respect to any Plan Year, the limit established for such Plan Year pursuant to Section 401(a)(17) of the Code.
- 1.12 <u>Deferral Limit</u> means, with respect to any Plan Year and only with respect to Pre-Tax Contributions and/or Roth Contributions made under the Basic

Plan, the limit on elective deferrals for such calendar year provided by the Internal Revenue Service pursuant to Section 402(g) of the Code, including the Catch-Up Contribution limitation amount set forth in Section 414(v) of the Code with respect to an eligible individual; provided, however, that the Catch-Up Contribution limitation amount shall be added to a Participant's Deferral Limit for a Plan Year only if he or she made Catch-Up Contributions under the Basic Plan for such Plan Year.

- 1.13 <u>Disabled or Disability</u> means a Participant who suffers from any medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, as evidenced by the Participant receiving income replacement benefits for such impairment under the Company's short term and long term benefit programs for a continuous period of 12 months.
- 1.14 <u>Distribution Election</u> means a written election of a Participant, provided on a form and in a manner specified by the Plan Administrator (which may include electronic means) from time to time, directing the time and form of payment of his or her Post-2004 Benefit or, in the absence of such an affirmative election, the Plan's default election under Section 5.3(d) which default election shall be deemed to be the Participant's Distribution Election. A Participant's distribution election with respect to the time and form of payment of his or her Pre-2005 Benefit shall be made in accordance with Section 5.2.
- 1.15 <u>Enrollment Authorization</u> means a written agreement of an Eligible Employee who has been selected to participate in the Plan, provided on a form and in a manner specified by the Plan Administrator (which may include electronic

{00835594-7} **-iv-**

means) from time to time, that authorizes the reduction of such Eligible Employee's Compensation that would otherwise be payable to him or her by the Participating Company employing such individual for services performed after such agreement goes into effect in accordance with Section 3.1 and, in the case of an Eligible Employee's initial enrollment under Section 2.2, his or her initial Distribution Election.

- 1.16 <u>ERISA</u> means the Employee Retirement Income Security Act of 1974, as amended.
- 1.17 <u>Fair Market Value</u> of a share of MMC Stock on any date means the closing price per share reported on the New York Stock Exchange for such date or, if no trading occurs on such date, for the last preceding day on which trading occurred.
- 1.18 <u>Fund</u> means any of the notional investment alternatives, other than Notional Shares, made available from time to time as a notional investment vehicle for Participants' Accounts under the Plan, including those, if any, based on the Investment Funds (other than the Stock Fund) made available from time to time under the Basic Plan.
- 1.19 <u>Investment Committee</u> means the individual or entity appointed from time to time by the Company to address investment-related matters under the Plan.
- 1.20 <u>Investment Direction</u> means the instructions a Participant provides to the Plan Administrator, on a form and in a manner specified by the Plan Administrator (which may include electronic means) from time to time, with respect to the (i) notional investment of future Compensation reduction credits under Section 4.2 credited to his or her Account, (ii) notional investment of future Matching Credits under Section 4.3(a) credited to his or her Account, (iii) notional investment of future MMC

Fixed Company Credits under Section 4.3A credited to his or her Account, or (iv) reallocation of the notional balance, or a portion thereof, credited to his or her Account. In the absence of affirmative instructions provided by a Participant with respect to clauses (i), (ii), (iii) or (iv) of this Section 1.20, the Plan's default investment direction provided under Section 4.10 shall be deemed the Participant's Investment Direction.

- 1.21 <u>Notional Investment</u> means a bookkeeping entry made to a Participant's Account pursuant to Article 4 that records a Participant's notional investment in shares or units, as the case may be, of a Fund.
- 1.22 <u>Notional Investment Value</u> means, as of the close of any business day, the notional value based on the net asset value of an investment vehicle upon which a Fund is based.
- 1.23 <u>Notional Share</u> means a bookkeeping entry made to a Participant's Account pursuant to Article 4 in respect to his or her notional investment in a share or fractional share of MMC Stock.
 - 1.24 Participant means an individual who has an Account that has not been terminated pursuant to Section 4.1.
- 1.25 <u>Participating Company</u> means the Company and any subsidiary or affiliate thereof whose Eligible Employees are eligible to participate in the Basic Plan; provided, however, that for the period commencing on January 1, 2017 and ending on August 31, 2020, Dovetail Insurance Corporation shall not be a Participating Company with respect to the MMC Fixed Company Credits portion of the Plan (as described in Section 4.3A).

- 1.26 <u>Payment Date</u> means the date determined pursuant to Section 5.3(b) for the commencement of the payment of a Participant's Post-2004 Benefit. The date of payment of a Participant's Pre-2005 Benefit shall be determined pursuant to Section 5.2.
- 1.27 <u>Plan</u> means this Marsh & McLennan Companies Supplemental Savings & Investment Plan (formerly the Marsh & McLennan Companies Stock Investment Supplemental Plan) as in effect from time to time.
- 1.28 <u>Plan Administrator</u> means the individual or entity appointed from time to time by the Company to administer the Plan.
 - 1.29 <u>Post-2004 Benefit</u> means the Benefit reduced by the Pre-2005 Benefit.
- 1.30 <u>Pre-2005 Benefit</u> means the portion of the Benefit attributable to amounts credited, adjusted for notional earnings and losses, to a Participant's Pre-2005 Sub-Account.
- 1.31 <u>Pre-2005 Sub-Account</u> means a sub-account established as part of an Account of a Participant by the Company pursuant to Section 4.1(b).
- 1.32 <u>Retirement</u> means (i) with respect to a Participant's Post-2004 Benefit, a Participant's Termination of Employment that occurs (A) on or after a Participant's fifty-fifth (55th) birthday provided he or she has been credited with at least five (5) Years of Service or (B) on or after his or her sixty-fifth (65th) birthday, and (ii) with respect to a Participant's Pre-2005 Benefit, a Participant's separation from service, as determined by the Plan Administrator under the rules and administrative practices of the Plan that were in effect prior to January 1, 2005 and in accordance with the

employment practices of the Company at such time, that occurs (A) on or after a Participant's fifty-fifth (55th) birthday provided he or she has been credited with at least five (5) Years of Service or (B) on or after his or her sixty-fifth (65th) birthday.

- 1.33 Specified Employee means a Participant who is an Eligible Employee and has met the requirements of Section 416(i)(1)(A)(i), (ii), or (iii) of the Code (applied in accordance with Treasury regulations and disregarding Section 416(i)(5) of the Code) at any time during a twelve (12) month period ending on December 31 (the "Identification Date"). A Participant who meets any of those requirements at any time during a twelve (12) month period ending on an Identification Date shall be treated as a Specified Employee with respect to a Termination of Employment that occurs during the 12-month period beginning on the first day of the fourth month following the Identification Date.
- 1.34 Termination of Employment or Terminates Employment means termination of all active employment with the Company, and all related entities aggregated with the Company under Section 414(b) or Section 414(c) of the Code. For purposes of this Section, a Participant's "active employment" is considered to have terminated when the number of hours of service per week performed by the Participant for a Participating Company or Non-Covered Company in a week are twenty percent (20%) or less of the average weekly hours worked by the Participant during the previous thirty-six (36) month period. Notwithstanding the foregoing to the contrary, a Participant, who is not performing services for a Participating Company or Non-Covered Company because he or she is on a bona fide leave of absence, Terminates Employment under the Plan only after such leave of absence exceeds six (6) months or

{00835594-7} -Viii-

such longer period of time as provided under an applicable statute or by contract. This Section 1.34 shall be administered in accordance with Treas. Reg. §1.409A-1(h)(1).

{00835594-7} **-i**X**-**

ARTICLE 2 ELIGIBILITY AND ENROLLMENT

2.1 <u>Eligibility</u>.

- (a) Employee Deferrals and Matching Credits. An Eligible Employee who actively participates in the Basic Plan, including a former Eligible Employee who is rehired by a Participating Company or an Eligible Employee who has been transferred from a Non-Covered Company to a Participating Company, whose Contribution Authorization under the Basic Plan is not suspended pursuant to Section 3.5 of the Basic Plan and whose opportunity to cause contributions to be made pursuant to Section 3.1 of the Basic Plan could reasonably be expected to be limited in any Plan Year by the operation of the Compensation Limit, may be selected by the Plan Administrator to enroll in the Plan in order to (i) reduce and defer a portion of his or her Compensation during such Plan Year pursuant to Section 3.1 of this Plan and have such deferred amount credited to his or her Account pursuant to Section 4.1 and (ii) have Matching Credits credited to his or her Account pursuant to Section 4.3 of this Plan. Notwithstanding the foregoing, no Matching Credits shall be credited with respect to periods prior to the Eligible Employee's attainment of one (1) Year of Service.
- (b) MMC Fixed Company Credits. An Eligible Employee (i) whose annual Compensation exceeds the Compensation Limit, and (ii) whose opportunity to receive MMC Fixed Company Contributions pursuant to Section 3.9 of the Basic Plan could reasonably be expected to be limited in any Plan Year by the operation of the Compensation Limit (regardless of whether such Eligible Employee has completed an Enrollment Authorization with respect to employee deferrals under this Plan), may be selected by the Plan Administrator to participate in the Plan and have MMC Fixed

{00835594-7} -X-

Company Credits credited to his or her Account pursuant to Section 4.3A of this Plan. Notwithstanding the foregoing, (i) no MMC Fixed Company Credits shall be credited with respect to periods prior to the Eligible Employee's attainment of one (1) Year of Service and (ii) for the period commencing on January 1, 2017 and ending on August 31, 2020, any employee of Dovetail Insurance Corporation shall not be an Eligible Employee with respect to the MMC Fixed Company Credits portion of the Plan.

2.2 <u>Initial Enrollment</u>.

- (a) New Hire. In the case of any individual who is newly hired by a Participating Company and who satisfies the eligibility requirements of Section 2.1(a) and/or (b), such individual may, subject to the provisions of this Article 2, complete and submit to the Plan Administrator his or her Enrollment Authorization and Distribution Election (as applicable) during the first Annual Enrollment period ending on or after his or her date of hire.
- (b) Rehires and Transferees. A former Participant who has Terminated Employment and is rehired by the Company or Participating Company or an individual transferred to the Company or Participating Company by a Non-Covered Company who satisfies the eligibility requirements of Section 2.1(a) and/or (b) may re-enroll or enroll in the Plan no earlier than during the first Annual Enrollment period ending after the date such former Participant or individual satisfies the eligibility requirements of Section 2.1(a) and/or (b). If such rehired or transferred individual has attained one (1) Year of Service as of the Plan Year in which the rehire or transfer occurs, notwithstanding the attainment of one (1) Year of Service, any MMC Fixed Company Credits with respect to such rehired or transferred individual shall commence or

recommence no earlier than the Plan Year following the Plan Year in which the rehire or transfer occurs.

(c) Acquisitions. In the case of any individual who is employed by an Acquired Company on its date of acquisition by a Participating Company and satisfies the eligibility requirements of Section 2.1(a) and/or (b) of the Plan, such individual may, subject to the following sentences of this Section 2.2(c), complete and submit to the Plan Administrator his or her Enrollment Authorization and Distribution Election during the first Annual Enrollment period ending on or after the closing date of the Acquired Company's acquisition by a Participating Company. If the closing date of an Acquired Company's acquisition is scheduled to occur on or about, but no later than, the last day of a Plan Year, and if the Participating Company so requests, the Plan Administrator may in its discretion provide an individual not otherwise eligible to participate in the Plan under the provisions of this Section 2.2(c) with his or her initial Enrollment Authorization and Distribution Election during the Annual Enrollment period occurring during the Plan Year in which the closing date of Acquired Company's acquisition occurs; provided, however, that the acceptance of such Enrollment Authorization and Distribution Election by the Plan Administrator shall be contingent upon such acquisition closing on a date that is no later than the last day of such Plan Year. If an individual described in this Section 2.2(c) has attained one (1) Year of Service as of the Plan Year in which the closing date of the Acquired Company's acquisition occurs, notwithstanding the attainment of one (1) Year of Service, any MMC Fixed Company Credits with respect to such individual shall commence no earlier than the Plan Year following the Plan Year in which such acquisition occurs.

- (d) Employees Newly Eligible for MMC Fixed Company Credits. In the case of any individual who (i) has not elected to enroll in the Plan under paragraphs (a), (b) or (c) of this Section 2.2 and (ii) is newly eligible to participate in the MMC Fixed Company Credits portion of the Plan pursuant to Section 2.1(b), such individual may complete and submit to the Plan Administrator his or her Distribution Election during the Annual Enrollment period preceding the Plan Year in which the first MMC Fixed Company Credit is scheduled to be credited to the Plan on his or her behalf.
- (e) No Distribution Election. If an Eligible Employee selected under Section 2.1 to participate in the Plan completes and submits to the Plan Administrator an Enrollment Authorization, but fails to timely complete and submit his or her Distribution Election to the Plan Administrator in accordance with paragraphs (a), (b) or (c) of this Section 2.2, such Eligible Employee shall be deemed to have made the default election described in Section 5.3(d) but shall be eligible to make a subsequent Distribution Election in accordance with the provisions of Section 5.3(e).
- 2.3 Annual Election of Employee Deferrals. Except as provided in Section 2.2 (and whether or not an Eligible Employee completed and submitted to the Plan Administrator his or her Enrollment Authorization during the first Annual Enrollment period applicable to such Eligible Employee under Section 2.2 (a), (b) or (c)), each Eligible Employee who is eligible to participate in the Plan for a Plan Year shall complete and submit to the Plan Administrator a Contribution Authorization under the Basic Plan and an Enrollment Authorization related to such Plan Year during the Annual Enrollment period. An Eligible Employee who fails to timely submit a completed Contribution Authorization or Enrollment Authorization shall not participate

{00835594-7} -Xiii-

in the employee deferral and Matching Credit portion of the Plan for such Plan Year. An Enrollment Authorization under this Section 2.3 shall become effective on the first day of the Plan Year to which it relates and irrevocable no later than the last day of such Annual Enrollment period.

- 2.4 <u>Timing of Elections</u>. Notwithstanding anything to the contrary in Sections 2.2 or 2.3, the Plan Administrator shall have the discretion to:
- (a) accept an Eligible Employee's Enrollment Authorization for a Plan Year after the end of the Annual Enrollment period for such Plan Year, but no later than the last business day of the Plan Year that includes such Annual Enrollment period; and
- (b) accept an Eligible Employee's initial Distribution Election after the end of the Annual Enrollment period in which the initial Distribution Election is otherwise permitted to be made pursuant to Section 2.2, but no later than the last business day of the Plan Year that includes such Annual Enrollment period.

{00835594-7} **-XiV-**

ARTICLE 3

ELIGIBLE EMPLOYEE DEFERRALS

- 3.1 <u>Deferral Election</u>. Subject to Section 3.2, an Eligible Employee who is eligible pursuant to Section 2.1 to participate in the Plan for a Plan Year may direct the Participating Company that employs him or her to reduce his or her Compensation for the calendar year that includes such Plan Year from 1% to 30% of the Eligible Employee's Compensation and to pay such amount (as adjusted pursuant to this Plan) to such Eligible Employee or his or her Beneficiary in the future as deferred compensation under this Plan. Any designation by an Eligible Employee pursuant to this Section 3.1 shall be contained in his or her annual Enrollment Authorization.
- 3.2 <u>Timing of Compensation Reduction</u>. Reductions in Compensation effected by payroll deduction shall commence in, or immediately following, the payroll period in which the Eligible Employee's Compensation first exceeds the Compensation Limit or his or her Pre-Tax Contributions and/or Roth Contributions exceed the Deferral Limit.
- 3.3 Suspension of Enrollment Authorization. Notwithstanding any other provision contained herein, a Participant's Enrollment Authorization for a Plan Year shall be suspended during such Plan Year for the period during which such Participant is (i) on an unpaid leave of absence from the Participating Company (but only during the period of time that precedes a Participant's Termination of Employment if one is deemed to occur), or (ii) eligible for, and in receipt of, benefit payments under the Participating Company's long-term disability plan. Such suspension shall cease to apply and the Participant's Enrollment Authorization shall be reinstated if and when such Participant returns to active employment with a Participating Company during such

Plan Year. There shall be no retroactive employee deferrals or Matching Credits for the period of the leave of absence (including long-term disability).

3.4 Contingent Enrollment Authorization. If a Participant's period of (i) unpaid leave of absence from a Participating Company, or (ii) eligibility for, and receipt of, benefit payments under such Participating Company's long-term disability plan occurs during a Plan Year, and is expected to continue beyond the end of that Plan Year, such Participant may make deferrals to the Plan in a future Plan Year to the extent the Participant completes an Enrollment Authorization under Section 2.3 for such future Plan Year; provided, however, that the Participant's Enrollment Authorization shall become effective and irrevocable in the first payroll period following such Participant's return to active employment with a Participating Company during such future Plan Year.

{00835594-7} -XVİ-

ARTICLE 4 ACCOUNTS

4.1 Accounts.

- (a) Establishment of Accounts. The Company shall establish an Account for each Participant which shall be credited with Notional Shares and/or Notional Investments in accordance with his or her Investment Direction based upon such Participant's Compensation reductions made pursuant to his or her timely completed Enrollment Authorization, matching amounts attributable thereto, and any MMC Fixed Company Credits pursuant to Section 4.3A. Each Account shall also be credited with reinvested notional dividends and notional investment gains, if any. A Participant's Account shall be debited for Benefits paid to or in respect of such Participant, including tax withholding pursuant to Section 5.7, and notional investment losses, if any. An Account shall be terminated when there are no longer any Notional Shares or Notional Investments credited to it.
- (b) <u>Pre-2005 Sub-Accounts</u>. The Company shall establish a Pre-2005 Sub-Account for each Participant who participated in the Plan before January 1, 2005 and shall credit each Pre-2005 Sub-Account with the December 31, 2004 balance of each such Participant's Account, respectively. The Pre-2005 Sub-Account shall be credited/debited with post-2004 notional earnings/losses, and debited with Pre-2005 Benefits paid to or in respect of such Participant, including tax withholding pursuant to Section 5.7. Effective for Plan Years beginning on or after January 1, 2005, the Pre-2005 Sub-Account shall not be credited with any amount described in Section 4.2, Section 4.3 or Section 4.4.

{00835594-7} -XVII-

4.2 <u>Credits for Compensation Reductions</u>. The reduction in a Participant's Compensation for any payroll period made pursuant to his or her Enrollment Authorization shall be credited to the Participant's Account in accordance with procedures and at such times established from time to time by the Plan Administrator and notionally invested in Notional Investments (including Notional Shares) in accordance with the Participant's Investment Direction.

4.3 <u>Credits for Company Match</u>.

- (a) Matching Credits. Provided that a Participant has completed at least one (1) Year of Service and reductions of his or her Compensation have been credited in accordance with Section 4.2 to his or her Account, then such Participant shall be credited with additional amounts based on the formula applicable to Participating Company Matching Contributions under Section 3.7.1 of the Basic Plan ("Matching Credits") and notionally invested in Funds and/or Notional Shares in accordance with the Participant's Investment Direction; provided, however, that no portion of a Participant's Compensation (and any reductions related thereto) shall be taken into account during any payroll period for both Participating Company Matching Contributions under the Basic Plan and Matching Credits under this Plan.
- (b) <u>Performance-Based Credits</u>. Effective for the Plan Years that commence January 1, 2006, January 1, 2007 and January 1, 2008, at the time and at the same rate that the Company allocates Participating Company Matching Contributions with respect to a Plan Year in accordance with Section 3.7.2 of the Basic Plan ("Performance-Based Credits"), the Company shall credit Performance-Based Credits to the Account of a Participant in the same manner as Matching Credits are credited to his

{00835594-7} -XVIII-

or her Account provided that the Participant is actively employed by a Participating Company on the last business day of the Plan Year, or incurred a disability (as defined in Section 1.22 of the Basic Plan) during the Plan Year, or terminated employment (as determined in accordance with Section 1.70 of the Basic Plan) during the Plan Year by reason of Retirement or death.

- (c) The sum for any Plan Year of the Matching Credits and Performance-Based Credits to a Participant's Account pursuant to paragraphs (a) and (b) of this Section 4.3 of this Plan and the Participating Company Matching Contributions allocated to him or her pursuant to Section 3.7 of the Basic Plan shall not exceed in the aggregate the maximum rate specified from time to time under Section 3.7 of the Basic Plan. The amount credited pursuant to Section 4.3 of this Plan shall be adjusted to the extent necessary to comply with the limitation set forth in the preceding sentence.
- 4.3A <u>MMC Fixed Company Credits</u>. The following provisions shall apply with respect to Plan Years beginning on or after January 1, 2017:
- (a) Provided that an Eligible Employee has completed at least one (1) Year of Service and is employed by a Participating Company (other than Dovetail Insurance Corporation for the period commencing on January 1, 2017 and ending on August 31, 2020), and subject to the provisions of Section 2.2, beginning with the first payroll period of a Plan Year in which the Eligible Employee's Compensation first exceeds the Compensation Limit, such Participant shall be credited with fixed amounts under this Plan based on the formula applicable to MMC Fixed Company Contributions under Section 3.9 of the Basic Plan ("MMC Fixed Company Credits").

{00835594-7} -XiX-

- (b) For purposes of determining the amount of the MMC Fixed Company Credits to be made on behalf of an Eligible Employee pursuant to Section 4.3A(a), Compensation shall be limited to amounts earned both (i) after the Eligible Employee has attained one (1) Year of Service and (ii) while the employee was an Eligible Employee with respect to the MMC Fixed Company Credits portion of the Plan. No MMC Fixed Company Credits shall be credited with respect to Compensation for which an MMC Fixed Company Contribution is provided under the Basic Plan.
- (c) Notwithstanding any other provision contained herein, a Participant's MMC Fixed Company Credits for a Plan Year shall be suspended during such Plan Year for the period during which such Participant is (i) on an unpaid leave of absence from the Participating Company (but only during the period of time that precedes a Participant's Termination of Employment if one is deemed to occur), or (ii) eligible for, and in receipt of, benefit payments under the Participating Company's long-term disability plan. The MMC Fixed Company Credits shall be reinstated if and when such Participant returns to active employment with a Participating Company, to the extent that the Participant otherwise remains eligible for such credits upon resumption of employment. There shall be no retroactive MMC Fixed Company Credits for the period of the leave of absence (including long-term disability).
- 4.4 <u>Post-Termination Credits</u>. If an Account is credited under Section 4.2, Section 4.3 or Section 4.3A after the Participant incurred a Termination of Employment, such credit or credits shall be notionally invested in accordance with the Investment Direction last in effect immediately prior to the Participant's Termination of Employment.

- 4.5 <u>Nonforfeitable Account</u>. A Participant's right to the balance credited to his or her Account shall be fully vested and nonforfeitable at all times.
 - 4.6 <u>Credits for Dividend Reinvestment, Other Distributions and Adjustments.</u>
- (a) Whenever a cash dividend is declared on MMC Stock, each Account shall receive a notional credit in the same dollar amount as the cash dividend for each Notional Share credited to each such Account on the record date for dividend payment. The notional dollar amount in each Account shall be converted into additional Notional Shares based on the Fair Market Value of MMC Stock on the dividend payment date. In the event of a stock dividend or distribution, stock split, recapitalization or the like, each Account shall be credited on the record date of such event with a number of Notional Shares equal to the number of shares of MMC Stock payable in respect of a share of MMC Stock for each Notional Share credited to each such Account on such record date.
- (b) Whenever a cash dividend, income distribution or other amount is paid in respect of a Fund or any other adjustment is made to the number of shares or units representing an interest in a Fund, then each Account that is credited with a Notional Investment in such Fund shall be credited, or adjusted in accordance with procedures established by the Plan Administrator from time to time to reflect such cash dividend, income distribution or other amount or adjustment.
 - 4.7 Accounts Confer No Interest in Assets.
- (a) Notwithstanding any other provision of the Plan that may be interpreted to the contrary, the Funds and Notional Shares are to be used only for the

{00835594-7} **-**XX**i-**

purposes of determining the amount of notional earnings to be credited and/or notional losses to be debited to a Participant's Account, and a Participant's designation of any such Fund or Notional Share, the crediting of Notional Investments and Notional Shares to an Account, the calculation of notional earnings and/or losses and the crediting or debiting of such notional earnings or notional losses to an Account shall not be considered or construed in any manner as an actual investment of an Account in any actual investment fund (including the Investment Funds under the Basic Plan) or in MMC Stock. In the event the Company or any Participating Company, in its sole discretion, decides to invest any amount in any or all of the actual investment funds (including the Investment Funds under the Basic Plan), no Participant shall have any rights in or to such investments themselves. Without limiting the foregoing, an Account shall at all times be a bookkeeping entry only and shall not represent an investment made on behalf of any Participant by the Company or any Participating Company.

- (b) Each Participant shall bear the full responsibility for all results associated with his or her designation of any Fund and/or Notional Shares under this Article 4 for notional investment, and neither the Company, Participating Company, Plan Administrator nor the Investment Committee shall have any responsibility or liability with respect to any Participant's designation of any Fund and/or Notional Shares.
- 4.8 <u>Valuation of Accounts and Account Statements</u>. The Plan Administrator may determine the notional value of all Accounts at such times as designated under procedures established by the Plan Administrator from time to time. The Plan Administrator shall furnish or make available, at any time designated under

{00835594-7} -XXII-

procedures established by the Plan Administrator from time to time, electronic or written statements to each Participant setting forth the number of Notional Shares and their value credited to his or her Account, the Notional Investments and their respective Notional Investment Values credited to his or her Account, and, if applicable, his or her Pre-2005 Sub-Account, as of the date designated in the statement.

4.9 <u>Changing Notional Investments</u>.

- (a) Account Balances. A Participant may change the allocation of Notional Investments credited to his or her Account among the Funds (including, for this purpose, Notional Shares) or transfer Notional Investments from a specific Fund into one or more replacement Funds in accordance with procedures and in increments established by the Plan Administrator from time to time. A Participant may not reallocate or transfer Notional Shares credited to his or her Account and/or Pre-2005 Sub-Account to any Fund.
- (b) <u>Compensation Reduction</u>. A Participant may change the notional allocation of future reductions in his or her Compensation to be credited to his or her Account pursuant to Section 4.2 among the Funds and Notional Shares in accordance with procedures established by the Plan Administrator from time to time.
- (c) <u>Matching Credits and MMC Fixed Company Credits</u>. A Participant may change the notional allocation of Matching Credits made pursuant to Section 4.3(a) and MMC Fixed Company Credits made pursuant to Section 4.3A among the Funds and Notional Shares in accordance with procedures established by the Plan Administrator from time to time; provided, however, that the notional investment election

{00835594-7} -XXIII-

in effect at any given time with respect to a Participant's Matching Credits (if any) shall also apply to such Participant's MMC Fixed Company Credits..

- 4.10 <u>Default Investment Direction</u>. If an Eligible Employee, selected in accordance with Section 2.1 to participate in the Plan, fails to timely complete and submit an Investment Direction to the Plan Administrator at the time of his or her initial enrollment in the Plan under Section 2.2, the Account of such Participant shall be deemed notionally invested in the default Investment Fund designated for the Plan by the Investment Committee until he or she submits a completed Investment Direction.
- 4.11 <u>Delayed Crediting of Notional Shares or Notional Investments</u>. If the Plan Administrator determines that the crediting of Notional Shares or Notional Investments to any Account at the time specified in this Article 4 would or could be a violation of applicable law, then such crediting shall be delayed until the Plan Administrator, in its sole discretion, determines it to be permitted. In the event of any such delay, the number of Notional Shares and Notional Investments that are eventually credited to the Account shall be determined based on the Fair Market Value of a share of MMC Stock and the Notional Investment Values determined on the date such Notional Shares and/or Notional Investments are credited to such Account.

ARTICLE 5

{00835594-7} **-XXİV-**

BENEFITS

- 5.1 Benefits. For purposes of determining the amount to which a Participant is entitled under the Plan, a Participant's Benefit payable at any time shall be equal to (i) the number of Notional Shares (including fractional Notional Shares) that have been credited to his or her Account at such time and (ii) the Notional Investment Value of the Notional Investments that have been credited to his or her Account at such time. Except as provided in Section 5.5 hereof or by procedures established from time to time by the Plan Administrator, the portion of the Benefit in respect of Notional Shares shall be paid only in shares of MMC Stock and the portion of the Benefit in respect of Notional Investments shall be paid only in cash. If a Participant has elected payment of his or her Benefit, or a portion of such Benefit, in installments pursuant to Section 5.2(b) and/or Section 5.3(c) or (d), the number of Notional Shares and value of Notional Investments credited to his or her Account shall be ratably reduced in accordance with the procedures established by the Plan Administrator from time to time.
 - 5.2 <u>Time and Method of Payment Elections of Pre-2005 Benefits</u>.
- (a) <u>In General</u>. Subject to the provisions of paragraph (b) of this Section 5.2, a Participant may, under uniformly applicable rules, elect the time and/or form of payment of his or her Pre-2005 Benefit. Any such election shall remain in effect until such time as a new election shall become effective hereunder.
- (b) <u>Time and Payment Election</u>. Under procedures established by the Plan Administrator from time to time, a Participant may elect that, in the event of his or her Retirement or death, his or her Pre-2005 Benefit be paid to him or her (or his or her Beneficiary) in (i) a single lump sum distribution, or (ii) annual installments over a

period of two (2) to fifteen (15) years as he or she may designate. The Participant may also elect that his or her Pre-2005 Benefit be paid, or begin to be paid if annual installments are elected, to him or her (or his or her Beneficiary) (x) as of the last business day of the calendar month (or, if it is not administratively practicable to make the payment in such month, the succeeding calendar month) following the date of such Retirement or death, (y) during the first calendar quarter of the Plan Year that immediately follows the Plan Year of such Retirement or death, or (z) during the first calendar quarter of the fifth (5th) Plan Year that follows the Plan Year of such Retirement or death. Any election under this paragraph (b) shall take effect twelve (12) months after the election is made and shall replace any distribution election made previously by the Participant with respect to his or her Pre-2005 Benefit. If a Participant separates from service before the twelve (12) month period above has elapsed, his or her previous distribution election which has been in effect for at least twelve (12) months shall control or, if none, distributions shall be made in accordance with Section 5.2(d). In the event a Participant elects that his or her Pre-2005 Benefit be distributed to him or her in the form of annual installments, the Participant shall also direct at the same time that in the event of his or her death whether the balance of his or her Pre-2005 Benefit shall be paid to his or her Beneficiary in the form of a single lump sum distribution or shall be paid in annual installments for the remainder of the installment period designated by the Participant. Notwithstanding the foregoing provisions of this paragraph (b), in the event a Participant separates from service for any reason other than Retirement or death, his or her Pre-2005 Benefit shall be paid in a single lump sum as soon as administratively practicable following such Participant's separation from service.

{00835594-7} -XXVi-

- (c) <u>Death Benefit</u>. In the event that a Participant dies before payment of his or her Pre-2005 Benefit commenced or was completed, the balance of his or her Pre-2005 Benefit shall be paid to his or her Beneficiary as soon as administratively practicable following the Participant's death in accordance with his or her last effective distribution election made in accordance with Section 5.2(b). If no election is in effect, the unpaid balance of such Pre-2005 Benefit shall be paid in a single lump sum distribution as soon as administratively practicable after the Participant's death.
- (d) <u>Default Method of Distribution</u>. In the event that a Participant has not made a timely election in accordance with Section 5.2(b), upon the Participant's Retirement or death, the Participant's Pre-2005 Benefit shall be paid to him or her (or his or her Beneficiary) in a single lump sum as soon as administratively practicable following such separation from service.
- (e) <u>Separation from Service</u>. For purposes of this Section 5.2, a Participant's separation from service for any reason other than Retirement or death shall be determined by the Plan Administrator under the rules and administrative practices of the Plan that were in effect prior to January 1, 2005 and in accordance with the employment practices of the Company at such time, as distinct from the rules that apply to a Participant's Termination of Employment for purposes of Section 5.3 of the Plan.
 - 5.3 <u>Time and Method of Payment Elections of Post-2004 Benefits</u>.
- (a) <u>In General</u>. A Participant may elect the form of payment of his or her Post-2004 Benefit by timely completing and submitting a Distribution Election to the Plan Administrator in accordance with procedures established the Plan Administrator from time to time. Any such Distribution Election shall remain in effect

{00835594-7} -XXVII-

until such time as a subsequent Distribution Election made in accordance with Section 5.3(d) shall become effective.

- (b) <u>Payment Date</u>. For purposes of this Section 5.3, the Payment Date of a Participant's Post-2004 Benefit shall be the date of the first to occur of: (i) the Participant's Retirement; (ii) the Participant's Disability; (iii) the Participant's death; (iv) a Change in Control; or (v) the Participant's Termination of Employment for any reason other than Retirement, Disability or death.
- (c) Initial Distribution Elections. In accordance with Section 2.2 and subject to Section 5.3(e), a

 Participant may direct in the Distribution Election that, in the event of his or her Retirement, Disability or death, his or her Post2004 Benefit be paid to him or her (or his or her Beneficiary) in (i) a single lump sum distribution, or (ii) annual installments over a period of two (2) to fifteen (15) years as he or she may designate. A Participant may also direct in the Distribution Election that his or her Post-2004 Benefit be paid or commence: (x) as soon as administratively practicable following his or her Payment Date but no later than ninety (90) days after such Payment Date; (y) during the first calendar quarter of the Plan Year that immediately follows the Plan Year of his or her Payment Date for Retirement, Disability or death; or (z) during the first calendar quarter of the fifth (5th) Plan Year that follows the Plan Year during which his or her Payment Date for Retirement, Disability or death occurs. In the event a Participant elects that his or her Post-2004 Benefit be distributed to him or her in the form of annual installments, the Participant shall also direct, in the event of his or her death, whether the balance of his or her Post-2004 Benefit shall be paid to his or her Beneficiary in the form of a single lump sum distribution or shall be paid in

{00835594-7} -XXVIII-

annual installments for the remainder of the installment period designated by the Participant.

- (d) <u>Default Distribution Election on Account of Termination of Employment</u>. Subject to Section 5.3(e), if a Participant fails to timely submit a completed initial Distribution Election in accordance with Section 2.2 and Section 5.3(c), his or her Post-2004 Benefit shall be paid in a single lump sum as soon as administratively practicable following his or her Payment Date for Termination from Employment for any reason other than Disability or death but no later than ninety (90) days after such Payment Date.
- (e) Subsequent Distribution Elections. Notwithstanding a Participant's initial Distribution Election under Section 5.3(c) or his or her default election under Section 5.3(d), a Participant may complete and submit to the Plan Administrator, in accordance with procedures established by the Plan Administrator from time to time, a subsequent Distribution Election in which he or she designates the time or form of the payment of his or her Post-2004 Benefit. Notwithstanding the foregoing, if a Participant designates annual installments as the form of payment in his or her initial Distribution Election, he or she may not reduce the number of annual installments previously elected; provided, however, a Participant may still designate, in accordance with the rules of this paragraph (e), the single lump sum distribution as the form of payment. Except in the cases of Disability or death, if a Participant submits a subsequent Distribution Election under this paragraph (e), such election must provide that payment shall be made or commence no sooner than five (5) calendar years from the time such payment otherwise would have been made or would have commenced. For purposes of

{00835594-7} -XXIX-

this paragraph (e), a subsequent Distribution Election shall be effective only if the Distribution Election is made at least twelve (12) months prior to the time of payment specified in the immediately preceding Distribution Election. The annual installment form of payment under Section 5.3 shall be treated as a single payment for purposes of this paragraph (e).

- (f) <u>Default Distribution Election on Account of Disability</u>. In the event a Participant fails to timely submit a completed initial Distribution Election in accordance with Section 2.2 and Section 5.3(c) and becomes Disabled, his or her Post-2004 Benefit shall be paid in a single lump sum in the calendar month, or immediately succeeding calendar month, that a determination has been made that the Participant is Disabled.
- (g) Death Benefit. In the event that a Participant dies before payment of his or her Post-2004 Benefit was made in a single lump sum distribution or commenced to be paid in annual installments, his or her Post-2004 Benefit shall be paid to his or her Beneficiary at the time and in the form designated in the Participant's Distribution Election in effect at the time of death. If no Distribution Election is in effect at the time of the Participant's death, the Post-2004 Benefit shall be paid to his or her Beneficiary in a single lump sum distribution as soon as administratively practicable following the Participant's date of death but no later than ninety (90) days after such Payment Date. If a Participant dies after annual installments have commenced, payment of the balance of his or her Post-2004 Benefit shall be made to his or her Beneficiary in accordance with his or her Distribution Election; provided, however, if the Distribution Election provided for the continuation of installment payments, then the remaining

{00835594-7} **-**XXX**-**

annual installments shall be paid to the Participant's Beneficiary on the annual installment payment date that is established by the Plan Administrator from time to time that applies to all annual installment payments made under the Plan next following the Participant's date of death.

- (h) <u>Termination of Employment for any Other Reason</u>. Notwithstanding a Participant's Distribution Election made under Section 5.3(c) or (e), in the event that a Participant Terminates Employment for any reason other than Retirement, Disability or death, his or her Post-2004 Benefit shall be paid in a single lump sum distribution as soon as administratively practicable following the date of the Participant's Termination from Employment for any reason other than Retirement, Disability or death, but no later than the 90-day period following such Termination from Employment.
- (i) <u>Determination of Payment Date</u>. Except as may otherwise be provided in Section 5.3(e) with respect to subsequent distribution elections, no Participant or Beneficiary shall have the right to designate the date or taxable year of payment of any distributions to be made under Sections 5.3(c), (d), (g) or (h) of the Plan.
- (j) <u>Specified Employee Rule</u>. Notwithstanding a Specified Employee's election under Section 5.3(c) or (e) that is effect on his or her Payment Date, the application of the default election rules under Section 5.3(d) or the distribution rules of Section 5.3(h), the payment or commencement of payment of a Specified Employee's Post-2004 Benefit following his or her Termination from Employment shall not be made or commence before the earlier of (i) the date that is six (6) months following his or her Termination of Employment other than for Disability, or (ii) the date of his or her death.

{00835594-7} -XXXi-

5.4 Source of Payment. The Benefit of each Participant shall be the obligation of the Participating Company or Companies that employed such Participant at the time reductions in the Participant's Compensation were made pursuant to Section 3.1, and shall be the general liability of such Participating Company or Companies. The claim of a Participant or Beneficiary to a Benefit shall at all times be merely the claim of an unsecured creditor of the Participating Company or Companies responsible therefor. No trust, security, escrow, or similar account need be established for the purpose of paying Benefits. However, the Company may in its discretion establish a custodial account or "rabbi trust" (or other arrangement having equivalent taxation characteristics under the Code and applicable regulations or rulings) to hold assets of the Participating Companies, subject to the claims of such Participating Companies' creditors in the event of insolvency, for the purpose of paying Benefits. If the Company establishes such an account or trust, amounts paid therefrom shall discharge the obligations hereunder to the extent of the payments so made.

5.5 Change in Control.

- (a) For purposes of Section 5.3 and this Section 5.5, a "Change in Control" shall have occurred if:
- (i) any "person", as such term is used in Sections 13(d) and 14(d) of the Securities Exchange Act (the "Exchange Act") (other than the Company, any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of MMC Stock), or more than one person

{00835594-7} -XXXII-

acting as a group is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company representing more than fifty percent (50%) of the combined voting power of the Company's then outstanding voting securities;

- (ii) during any 12-month period, individuals who at the beginning of such period constitute the Board, and any new director whose election by the Board of Directors or nomination for election by the Company's stockholders was approved by a vote of at least a majority of the directors then still in office who either were directors at the beginning of the period or whose election or nomination for election was previously so approved, cease for any reason to constitute at least a majority thereof;
- (iii) there is consummated a merger or consolidation of the Company with any other corporation, other than (A) a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving or parent entity) fifty percent (50%) or more of the combined voting power of the voting securities of the Company or such surviving entity (or any parent of the Company or such surviving entity) outstanding immediately after such merger or consolidation or (B) a merger or consolidation effected to implement a recapitalization of the Company (or similar transaction) in which no "person" (as herein above defined) or more than one person acting as a group acquired fifty percent (50%) or

{00835594-7} -XXXIII-

more of the combined voting power of the Company's then outstanding securities; or

- (iv) during any 12-month period, any person or more than one person acting as a group acquires all or substantially all of the Company's assets (or any transaction having a similar effect); provided that such assets have a total gross fair market value equal to or more than forty percent (40%) of the total gross fair market value of all of the assets of the Company and its subsidiaries.
- (b) Notwithstanding any contrary provision of the Plan, upon the occurrence of a Change in Control, the Company shall pay to each Participant (including a Participant who is a Specified Employee), his or her Benefit (i) in a single distribution of shares of MMC Stock in respect of the number of Notional Shares credited to his or her Account and a single lump sum distribution of cash equal to the Notional Investment Value of his or her Notional Investments credited to his or her Account or (ii) to the extent all of the shares of MMC Stock have been changed, exchanged or converted into cash, property or other securities of the Company in connection with such Change in Control, in such cash, property or other securities to which such Participant would have been entitled if his or her Benefit had been paid to him or her in the manner as set forth in clause (i) hereof immediately prior to the Change in Control. The Company shall make payments on or after the date of the Change in Control, but in no event shall payment be made later than the latest of (i) the last day of the calendar year in which such Change in Control occurs or (ii) the fifteenth (15th) day of the third (3th) calendar month following

{00835594-7} **-**XXX**i**V-

the date of the Change in Control, provided that the Participant shall not designate, directly or indirectly, the taxable year of payment.

- 5.6 <u>Payment on Account of Income Inclusion</u>. Notwithstanding any provision in the Plan to the contrary, in the event it is determined at any time that the Plan fails to comply with the requirements of Section 409A of the Code and/or Treasury regulations, a single lump sum distribution shall be paid to an affected Participant within thirty (30) days of such determination. Such payment may not exceed the amount required to be included in the income of such Participant as a result of the failure to comply.
- 5.7 Withholding All deferrals and payments under the Plan shall be subject to any applicable withholding requirements imposed by any tax or other law. The Participating Company or Companies responsible for payment of a Benefit shall have the right to (i) require as a condition of deferral and payment that the payee remit to such Participating Company or Companies an amount sufficient in its or their opinion to satisfy all applicable withholding requirements, or (ii) accelerate the time of a payment, or make a payment from the Plan, in order to pay employment taxes under Sections 3101, 3121(a) and 3121(v)(2) of the Code, wage withholding under Section 3401 of the Code and wage withholding under applicable state, local and foreign tax law. If the event giving rise to the withholding obligation is the payment of shares of MMC Stock, then the withholding obligation may be satisfied by having the Participating Company or Companies withhold shares of MMC Stock having a Fair Market Value equal to the amount of tax to be withheld. For this purpose, Fair Market

{00835594-7} **-**XXXV**-**

Value shall be determined by the Plan Administrator as of the date on which the amount of tax to be withheld is determined.

5.8 <u>Delays in Payment</u>. Notwithstanding any other provision of this Article 5, the Plan Administrator may delay the payment of a Benefit in accordance with Treas. Reg. §1.409A-2(b)(7)(i) if the Plan Administrator reasonably anticipates that making the payment would result in the loss of a Participating Company's right to a tax deduction for such payment due to the application of Section 162(m) of the Code. The Plan Administrator may also delay payment of a Benefit in accordance with Treas. Reg. §1.409A-2(b)(7)(ii) if the Plan Administrator reasonably anticipates that the making of the payment will violate Federal securities laws or other applicable law.

{00835594-7} **-**XXXV**i**-

ARTICLE 6 BENEFICIARIES

6.1 <u>Beneficiary Designation</u>.

- (a) A Participant shall designate a Beneficiary or Beneficiaries of his or her Benefit. The designated Beneficiary or Beneficiaries may or may not be such Participant's designated beneficiary or beneficiaries under the Basic Plan.
- (b) A Participant shall designate his or her primary Beneficiary or Beneficiaries and contingent

 Beneficiary or Beneficiaries by completing and submitting to the Plan Administrator a written designation on a form and in a

 manner specified by the Plan Administrator (which may include electronic means) from time to time.
- (c) In the event that there is no properly designated Beneficiary or contingent Beneficiary living at the time of a Participant's death, the Participant's unpaid Benefit shall be paid to his or her surviving Spouse, or, if there is no surviving Spouse, to the executors or administrators of his or her estate. The person or persons to whom such amount is paid shall be deemed to be the deceased Participant's Beneficiary for purposes of Article 5 of this Plan.
- 6.2 <u>Payment to Incompetent</u>. If any person entitled to benefits under this Plan shall be a minor or shall be physically or mentally incompetent in the judgment of the Plan Administrator, such benefits may be paid to the person to whom the corresponding benefits under the Basic Plan are paid pursuant to Section 11.3 thereof.

{00835594-7} -XXXVii-

ARTICLE 7 ADMINISTRATION

- 7.1 <u>Appointment of Plan Administrator</u>. The Plan shall be administered by the Plan Administrator. Without limiting the generality of the foregoing, the Plan Administrator shall have the power and discretion to:
- (a) make and enforce rules and regulations and prescribe the use of forms it deems appropriate for the administration of the Plan (including the discretion to prescribe the form or other method of communication, consistent with applicable law, for any particular purpose specified in the Plan, whether or not the Plan specifies that such communication be written);
- (b) construe all terms, provisions, conditions and limitations of the Plan and resolve ambiguities, inconsistencies and omissions;
- (c) determine all questions arising out of or in connection with the provisions of the Plan or its administration in any and all cases in which it deems such a determination advisable, such determinations to be final and conclusive on all persons;
- (d) delegate authority to agents and other persons to act on its behalf in carrying out the provisions and administration of the Plan, and to take or direct any action required or advisable with respect to the administration of the Plan.
- 7.2 <u>Claims Procedure</u>. If the Plan Administrator, or an individual delegated with the authority to make initial claim determinations, denies any Participant's or Beneficiary's claim for benefits under the Plan:
- (a) the Plan Administrator or individual delegated with the authority to make initial claim determinations shall notify such Participant or Beneficiary

{00835594-7} -XXXVIII-

of such denial by written notice which shall set forth the specific reasons for such denial; and

- (b) the Participant or Beneficiary shall be afforded a reasonable opportunity for a full and fair review by the Plan Administrator of the decision to deny his or her claim for Plan benefits, generally in the same manner for claims made under the Basic Plan.
- 7.3 <u>Service of Process</u>. The Company or such other person as may from time to time be designated by the Company shall be the agent for service of process under the Plan.
- 7.4 <u>No Bond Required</u>. No bond or other security shall be required of the Plan Administrator or any individual to whom the Plan Administrator delegates authority except as may be required by law.
- 7.5 Limitation of Liability; Indemnity. Except to the extent otherwise provided by law, if any duty or responsibility of the Plan Administrator or the Committee has been allocated or delegated to any other individual in accordance with any provision of this Plan, then neither the Plan Administrator nor Committee (as the case may be) shall be liable for any act or omission of such individual in carrying out such duty or responsibility. The Company shall indemnify and save the Plan Administrator and each person who is a member of the Committee, and each employee or director of a Participating Company harmless against any and all loss, liability, claim, damage, cost and expense which may arise by reason of, or be based upon, any matter connected with or related to the Plan or the administration of the Plan (including, but not limited to, any and all expenses reasonably incurred in investigating, preparing or

{00835594-7} **-**XXX**i**X**-**

defending against any litigation, commenced or threatened, or in settlement of any such claim) to the fullest extent permitted under applicable law, except when the same is judicially determined to be due to the gross negligence or willful misconduct of the Plan Administrator or such Committee member, employee or director.

7.6 <u>Payment of Expenses</u>. The Plan Administrator and its members shall serve without special compensation. Expenses of plan administration shall be paid by the Company.

{00835594-7} -xl-

ARTICLE 8 AMENDMENT AND TERMINATION

- 8.1 <u>Right Reserved.</u> (a) Subject to Section 8.2, the Board of Directors may at any time amend or terminate the Plan, retroactively or otherwise, provided that such amendment shall not cause the Plan to violate Section 409A of the Code or Treasury regulations thereunder. However, no such amendment or termination shall reduce any Participant's Benefit determined as though the date of such amendment or termination were the date of his Termination of Employment. The Chief Executive Officer or any officer designated by him also may amend the Plan to the extent permitted under the Employee Benefit Plan Guidelines adopted by the Board of Directors as of September 18, 2003, as from time to time amended.
- (b) In its discretion, the Company may upon termination of the Plan or at any time thereafter pay to every Participant (or Beneficiary) in a single distribution a number of shares of MMC Stock equal to the number of Notional Shares credited to his or her Account and a lump sum payment equal to the Notional Investment Value of his or her Notional Investments, whereupon all Accounts shall be terminated; provided, however, no payment shall be made under this paragraph (b) if the termination and liquidation of the Plan does not satisfy any of the permissible circumstances set forth in Treas. Reg. \$1.409A-3(j)(4)(ix).
- (c) If, upon termination of the Plan, the conditions and circumstances for liquidation of the Accounts under Treas. Reg. §1.409A-3(j)(4)(ix) are not satisfied, no more Eligible Employees shall become Participants, and reductions in Compensation under Section 3.1, the associated credits under Section 4.2 and 4.3, and the credits under Section 4.3A shall cease. Notwithstanding termination of the Plan, a

{00835594-7} -xli-

Participant may continue to provide Investment Directions to the Plan Administrator until his or her Benefit is distributed. For the purposes of this paragraph (c), Benefits shall be distributed in accordance with Article 5.

- (d) Notwithstanding a termination of the Plan, additional Notional Shares and Notional Investments shall continue to be credited to each Account as dividend reinvestments pursuant to Section 4.6 until such time as such Account is liquidated.
- 8.2 <u>Action to Bind Participating Company</u>. Upon the execution of the Plan by the Company, each Participating Company designates the Company as its agent to administer the Plan. Any amendment or termination of the Plan by the Company shall be binding upon each Participating Company.
- 8.3 <u>Cessation of Credits of Notional Shares</u>. Reductions in Compensation under Section 3.1 and the associated credits under Section 4.2 and 4.3 shall cease to be credited as Notional Shares effective as of May 14, 2013 or, if earlier, at such time as (a) the aggregate number of shares of MMC Stock paid pursuant to Article 5 on or after May 15, 2003, plus (b) the aggregate number of Notional Shares credited to Participants' Accounts on or after May 15, 2003, equals the share authorization set forth in Section 10.2.

{00835594-7} -xlii-

ARTICLE 9 MISCELLANEOUS

- 9.1 <u>Doubt as to Right to Payment</u>. If any doubt exists as to the right of any person to any benefits under this Plan or the amount or time of payment of such benefits (including, without limitation, any case of doubt as to identity, or any case in which any notice has been received from any other person claiming any interest in amounts payable hereunder, or any case in which a claim from other persons may exist by reason of community property or similar laws), the Plan Administrator may, in its discretion, direct that payment of such benefits be deferred until such right or amount or time is determined, or until a court of competent jurisdiction orders that such benefits be paid into court in accordance with appropriate rules of law, or the Plan Administrator may direct that payment be made only upon receipt of a bond or similar indemnification (in such amount and in such form as is satisfactory to the Plan Administrator).
- 9.2 <u>Spendthrift Clause</u>. No benefit, distribution or payment under the Plan may be anticipated, assigned (either at law or in equity), alienated or subject to attachment, garnishment, levy, execution or other legal or equitable process whether pursuant to a "domestic relations order" as defined in Section 414(p) of the Code or otherwise.
- 9.3 <u>Usage</u>. Whenever applicable, the masculine gender, when used in the Plan, includes the feminine gender, and the singular includes the plural.
- 9.4 <u>Data</u>. Any Participant or Beneficiary claiming a Benefit under the Plan shall furnish to the Plan Administrator such documents, evidence or information as the Plan Administrator shall consider necessary or desirable for the purpose of administering the Plan, or to protect the Plan Administrator. It is a condition

{00835594-7} -xliii-

of the Plan that each such Participant or Beneficiary shall furnish promptly true and complete data, evidence or information and sign such documents as the Plan Administrator may require before any benefits become payable under the Plan.

- 9.5 <u>Separability</u>. If any provision of the Plan is determined to be invalid, unenforceable or inconsistent with any applicable law, including Section 409A of the Code, its invalidity, unenforceability or inconsistency with any such applicable law shall not affect any other provisions of the Plan, and the Plan shall be construed and enforced as if such provision had not been included therein. Without limiting the application of the preceding sentence, a provision shall be considered invalid if its operation would cause the Basic Plan to fail to qualify under Section 401(a) of the Code.
- 9.6 <u>Captions</u>. The captions in this document and in the table of contents prefixed hereto are inserted only as a matter of convenience and for reference and in no way define, limit, enlarge or describe the scope or intent of the Plan and shall in no way affect the Plan or the construction of any provision thereof.
- 9.7 <u>Right of Discharge Reserved</u>. The establishment of the Plan shall not be construed to confer upon any employee any legal right to be retained in the employ of a Participating Company or give any employee or any other person any right to benefits, except to the extent expressly provided for hereunder. All employees shall remain subject to discharge to the same extent as if the Plan had never been adopted, and may be treated without regard to the effect such treatment may have upon them under the Plan.
- 9.8 <u>Limitations on Liability</u>. Notwithstanding any other provision of the Plan, no Participating Company nor any employee or agent of a Participating

{00835594-7} -xliv-

Company shall be liable to any Participant, Beneficiary or other person for any claim, loss, liability or expense incurred in connection with the Plan.

- 9.9 <u>Section 409A Compliance</u>. Notwithstanding any other provision of the Plan, the terms of each Plan shall in all instances be interpreted in a manner so as to comply with the requirements of Section 409A of the Code and Treasury Regulations issued thereunder.
- 9.10 Governing Law and Limitations on Actions. The Plan is intended to constitute an arrangement that is unfunded and maintained primarily for the purpose of providing deferred compensation for a select group of management or highly compensated employees, all within the meaning of the ERISA. All rights under this Plan shall be governed by and construed in accordance with rules of Federal law applicable to such plans. No action (whether at law, in equity or otherwise) shall be brought by or on behalf of any Participant or Beneficiary for or with respect to benefits due under this Plan unless the person bringing such action has timely exhausted the Plan's claim review procedure. Any action (whether at law, in equity or otherwise) must be commenced within three (3) years. This three (3) year period shall be computed from the earlier of (a) the date a final determination denying such benefit, in whole or in part, is issued under the Plan's claim review procedure and (b) the date such individual's cause of action first accrued (as determined under the laws of the State of New York without regard to principles of choice of laws).

{00835594-7} -xlv-

ARTICLE 10 EFFECTIVE DATE; SHARE AUTHORIZATION

10.1 Effective Date.

- (a) This amended and restated Plan shall be effective on January 1, 2022. The Plan was previously restated effective January 1, 2012. This January 1, 2022 amended and restated Plan shall supersede the prior restatement of the Plan. The provisions of this amended and restated Plan shall apply to the interests in this Plan and the prior Plan of Participants or Beneficiaries as of or subsequent to January 1, 2022, provided that Participants who are not Employees at any time on or after January 1, 2022 shall be subject to the provisions of the Plan as in effect at their Termination of Employment, except as otherwise provided herein.
- (b) For the period from January 1, 2005 through December 31, 2008 (i.e., the period prior to the January 1, 2009 restatement of the Plan), it was intended that the Plan be administered and operated in good faith compliance with Section 409A of the Code and in accordance with the transition rules provided in IRS Notice 2005-1, IRS Notice 2006-79 and IRS Notice 2007-86 including, but not limited to, the rules governing Participants' elections as to the time and method of payment.
- 10.2 <u>Shares Authorized</u>. Pursuant to the resolution adopted by the Board of Directors on May 15, 2003, 5,000,000 (five million) shares of MMC Stock (as may be adjusted for stock splits, stock dividends, reorganizations and the like) are authorized for issuance pursuant to this Plan on and after May 15, 2003.

{00835594-7} -xlvi-

IN WITNESS WHEREOF, MARSH & MCLENNAN COMPANIES, INC. has caused this amended and restated Plan to be executed by its duly authorized officer this 23 day of December, 2022.

MARSH & MCLENNAN COMPANIES, INC.

By: <u>/s/ Carmen Fernandez</u> Carmen Fernandez Chief People Officer

{00835594-7} -xlvii-

Marsh & McLennan Companies, Inc. List of Subsidiaries as of February 13, 2023

1 SWORKS LTD Unupuy 2 A. Consaminul & CIA. S.C. Unupuy 3 Access Equity Enhanced Fund CP, LLC United States 4 AFCO Premium Acceptance, Inc. United States 5 AFCO Premium Credit LLC United States 6 Apnew Higgins Pickering & Co. (Bermuda) Ltd Bermuda 8 Ald Float Mushrikach Lestisharal Alamal Jordian 8 Aldigate Trusieses Ltd United Kingdom 9 Alpha Consultants Limited New Zealand 10 Alia SA Chile 11 Alius Associates GP Ltd Guernsey 12 Alius Sacoiates Special Partners GP Limited Guernsey 12 Alius Celini (CSP) H LLC United States 14 Alius Celini (CSP) LLC United States 15 Alius Celini JP Limited Cayman Islands 16 Alius Real Assets Kelp LLC United States 18 Anal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Anison 18 (RF) Proprietary Limited South Arica		Company Name	Country
2 A. Constantinufial & Cla. S.C. Access Equity Enhanced Fund CP, LLC ARCO Premium Acceptance, Inc. Dinted States ARCO Premium Acceptance, Inc. ARCO Premium Acceptance, Inc. ARCO Premium Acceptance, Inc. Dinted States ARCO Premium Acceptance, Inc. Dinted Kingdom ARCO ARCO ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE ACCEPTANCE	1	• •	•
3. ACCOS Equity Enhanced Fund GP, LLC United States 4. AFCO Premium Acceptance, Inc. United States 5. AFCO Premium Credit LLC United States 6. Agnew Higgins Pickering & Co. (Bermuda) Ltd Bermada 8. Aldgate Trustees Ltd United Kingdom 9. Alpha Consultants Lesisharat Alamal Jordan 10. Ald SA Chile 11. Alfus Associates SP Ltd Guernsey 12. Alfus Associates SP Ltd Guernsey 12. Alfus Associates SP Ltd Guernsey 13. Alfus Client (GP) ILLC United States 14. Alfus Client (GP) LLC United States 15. Alfus Client (GP) LLC United States 16. Alfus Client (GP) LLC United States 17. Alfus Real Assests (GP) LLC United States 18. Alfus Client (GP) LLC United States 19. Alfus Real Assests (GP) LLC United States 10. Alfus Real Assests (GP) LLC United States 19. Alfus Real Assests (GP) LLC United States 20. Assure Consels Marsh S.A Sengal 21. Assurance Struces Limited (injudation) Sauth Alfus 22. Assure Consels Marsh S			<u> </u>
4 AFCO Premium Acceptance, Inc. United States 5 AFCO Premium Acceptance, Inc. United States 6 Agnew Higgins Pickering & Co. (Bermuda) Ltid Bermuda 7 Al Holoul Al Mushrikah Lestisharat Alamal Jordan 8 Aldgade Trustees Ltd United Kingdorm 9 Ajha Consultants Limited New Zealand 10 Alta SA Chile 11 Altius Associates Special Partners GP Limited Guernsey 12 Altius Client (GP) ILC United States 13 Altius Client (GP) ILC United States 14 Altius Client (GP) ILC United States 15 Altius Client 3P Limited Cayman Islands 16 Altius Real Assests (GP) LLC United States 17 Altius Real Assests (GP) LLC United States 18 Amal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Consells Marsh S.A. Senegal 21 Assurance Services Corporation United States 22 Assurance Services Corporation United States 23 BenPool Re IC United States 24 Best Insurance, Inc. (Fukouka Branch) United States			
5 AFCO Premium Credit LLC 6 Agnew Higgins Pickening & Co. (Bermuda) Ltd 7 Al Heloul Al Mushrikah Lestisharat Alamal 8 Aldigate Trustees Ltd 9 Ajhar Consultants Limited 10 Alta SA 10 Chile 11 Altius Associates Special Partners GP Limited 10 Alta SA 11 Altius Associates Special Partners GP Limited 12 Altius Associates Special Partners GP Limited 13 Altius Client (GP) ILC 14 Altius Client (GP) ILC 15 Altius Client (GP) ILC 16 Altius Real Assest Special Partners GP Limited 16 Altius Client (GP) ILC 17 Altius Client (GP) LLC 18 Altius Client JP Limited 18 Altius Client JP Limited 19 Altius Real Assests (GP) LLC 19 Altius Real Assests Management S.a.r.I (GP) (Luxembourg) 19 Annison 18 (RP) Proprietary Limited 19 Annison 18 (RP) Proprietary Limited 20 Assur Consells Marsh S.A 21 Assurance Services Corporation 21 Assurance Services Corporation 22 BenPool Re IC 23 BenPool Re IC 24 Assurance Services Corporation 25 Best Insurance, Inc. (Fukuyama Branch) 26 Best Insurance, Inc. (Fukuyama Branch) 27 Best Insurance, Inc. (Fukuyama Branch) 28 Best Insurance, Inc. (Fukuyama Branch) 29 Best Insurance, Inc. (Fukuyama Branch) 29 Best Insurance, Inc. (Fukuyama Branch) 20 Best Insurance, Inc. (Fukuyama Branch) 21 Best Insurance, Inc. (Fukuyama Branch) 22 Best Insurance, Inc. (Fukuyama Branch) 23 Biue Marbie Micro Limited 24 Dinted Kingdom 25 Best Insurance, Inc. (Fukuyama Branch) 26 Best Insurance, Inc. (Fukuyama Branch) 27 Best Insurance, Inc. (Fukuyama Branch) 28 Best Insurance, Inc. (Fukuyama Branch) 29 Best Insurance, Inc. (Fukuyama Branch) 30 Biuefin Insurance Group Limited 31 Dinted Kingdom 32 Biuefin Insurance Group Limited 41 Carpenter Marsh Fac Agentina Corredores de Reaseguros SA 41 Agentina 42 Carpenter Marsh Fac Argentina Corredores de Reaseguros SA 42 Carpenter Marsh Fac Colombia Corredores de Reaseguros SA 43 Carpenter Marsh Fac Colombia Corredores de Reaseguros SA 44 Carpenter Marsh Fac Colombia Corredores de Reaseguros SA 45 Carpenter Tumer SA, 45 Colombia 46 Carpenter Tumer SA, 45 Colombia Corredores de Reaseguros			
6 Agnew Higgins Pickering & Co. (Bernuda) Ltd Bernuda 7 Al Heloul Al Mushrikah Lestisharat Alamal Jordan 8 Aldpate Trustees Ltd United Kingdom 9 Alpha Consultaris Limited New Zealand 11 Alfus Consultaris Limited Guernsey 21 Alfus Associates Special Partners GP Limited Guernsey 12 Alfus Associates Special Partners GP Limited United States 14 Alfus Client (GP) I Lt.C United States 15 Alfus Client (GP) Lt.C United States 16 Alfus Client (GP) Lt.C United States 16 Alfus Real Assets (GP) Lt.C United States 16 Alfus Real Assets (GP) Lt.C United States 16 Alfus Real Assets (GP) Lt.C United States 17 Alfus Real Assets (GP) Lt.C United States 18 Anal Insurance Brokers Limited (In liquidation) Saudi Arabia 19 Annison 18 (RP) Proprietary Limited Sundi Arabia 20 Assurance Services Corporation United States 21 Assurance Services Corporation United States 22			
7 Al Heloul Al Mushrikah Lestisharat Alamal Jordan 8 Aldgate Trustees Ltd United Kingdom 9 Apha Consultants Limited New Zealand 10 Alta SA Chile 11 Altius Associates Special Partners GP Limited Guernsey 12 Altius Client (GP) LLC United States 14 Altius Client (GP) LLC United States 15 Altius Client (JP) LLT Limited Cayman Islands 16 Altius Real Assets (GP) LLC United States 17 Altius Real Assets (GP) LLC United States 18 Ambina Insurance Brosers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Consells Marsh S.A. Senegal 21 Assur Consells Marsh S.A. Senegal 22 Avascent Canada ULC Canada 23 Bero Re IC United States 24 Best Insurance, Inc. (Flukyama Branch) Verest States 27 Best Insurance, Inc. (Flukyama Branch) Verest States 28 Bert Insurance, Inc. (Flukyama Branch) Verest			
Aldgate Trustees Lid Alpha Consultants Limited New Zealand Alpha Consultants Limited New Zealand Alpha Consultants Limited College Alpha Consultants Limited College C			
9 Alpha Consultants Limited New Zealand 10 Altas Associates GP Ltd Chile 11 Altus Associates Special Partners GP Limited Guernsey 12 Altus Associates Special Partners GP Limited Guernsey 13 Altus Cilent (GP) LLC United States 14 Altius Cilent (GP) LLC United States 15 Altius Cilent (GP) LLC United States 16 Altius Real Assets (GP) LLC United States 17 Altius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 18 Amal Insurance Brokes Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Consells Marsh S.A. Senegal 21 Assurance Services Corporation United States 22 Avascent Canada ULC Canada 23 Bend Insurance, Inc. (Fukouka Branch) 24 Best Insurance, Inc. (Fukouka Branch) 27 Best Insurance, Inc. (Fukouka Branch) Verestrance, Inc. (Fukouka Branch) 28 Bert			
10 Altau SA SO Chile 11 Altius Associates GP Ltd Guernsey 12 Altius Associates Special Partners GP Limited Guernsey 13 Altius Client (GP) ILLC United States 14 Altius Client (GP) LLC United States 15 Altius Client JP Limited Cayman Islands 16 Altius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 17 Altius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 18 Amal Insurance Brokers Limited (in liquidation) Saulf Arabia 20 Assur Consells Marsh S.A. Senegal 21 Assur Consells Marsh S.A. Senegal 21 Assur Consells Marsh S.A. Senegal 21 Assur Consells Marsh S.A. Senegal 22 Assect Canada ULC Canada 23 Berolies C. United States 24 Best Insurance, Inc. (Fukoyama Branch) Ferential States 25 Best Insurance, Inc. (Fukoyama Branch) Ferential States 26 Best Insurance, Inc. (Nara Branch)		•	-
11 Altius Associates GP Ltd Guernsey 12 Altius Cilent (GP) ILC United States 14 Altius Cilent (GP) LLC United States 15 Altius Cilent (GP) LLC United States 16 Altius Real Assets (GP) LLC United States 17 Altius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 18 Amal Insurance Brokes Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Consells Marsh S.A. Sengal 21 Assurance Services Corporation United States 22 Avascent Canada U.C Canada 23 BenPool Re IC United States 24 Best Insurance, Inc. (Fukuyama Branch) Fee 25 Best Insurance, Inc. (Fukuyama Branch) Fee 27 Best Insurance, Inc. (Fukuyama Branch) Fee 28 Best Insurance, Inc. (Fukuyama Branch) Fee 29 Best Insurance, Inc. (Fukuyama Branch) Fee 20 Best Insurance, Inc. (Fukuyama Branch) Fee 21 Best Insurance, Inc. (Gokayama Branch) Fee 22 Bub Marble Micro Limited United Kingdom 30 Biue Marble Micro Limited <		•	
12 Altius Associates Special Partners GP Limited Guernsey 13 Altius Client (GP) ILLC United States 14 Altius Client (GP) LLC United States 15 Altius Client JP Limited Cayman Islands 16 Altius Real Assets (GP) LLC United States 17 Altius Real Assets (GP) LLC Luxembourg 18 Amal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Conseils Marsh S.A. Senegal 21 Assur Conseils Marsh S.A. Senegal 22 Avascent Canada ULC Canada 23 BenPool Re IC United States 24 Best Insurance, Inc. Japan 25 Best Insurance, Inc. (Fukuyama Branch) Tester States 26 Best Insurance, Inc. (Fukuyama Branch) Tester States 27 Best Insurance, Inc. (Nara Branch) Tester States 28 Best Insurance, Inc. (Nara Branch) Tester States 29 Best Insurance, Inc. (Nara Branch) Tester States 20 Best Insurance, Inc. (Nara Branch) Tester States 21 Bust Marshe Micro Limited United Kingdom 21 Blue Marshe Micro L			
13 Altius Client (GP) ILLC United States 14 Altius Client (GP) LLC United States 15 Altius Real Assets (GP) LLC United States 16 Altius Real Assets Management S.a.r.l (GP) (Luxembourg) Luxembourg 18 Anal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assurance Services Corporation United States 21 Assurance Services Corporation United States 22 Avascent Canada ULC Canada 23 Berpol Re IC United States 24 Best Insurance, Inc. Japan 25 Best Insurance, Inc. (Fukouka Branch) Ferman 26 Best Insurance, Inc. (Fukoyama Branch) Ferman 27 Best Insurance, Inc. (Fukoyama Branch) Ferman 28 Best Insurance, Inc. (Novagama Branch) Ferman 29 Best Insurance, Inc. (Novagama Branch) Ferman 30 Bilw Marbie Micro Limited United Kingdom 31 Bluefin Insurance Services			•
14 Altius Client (AP) LLC Cayman Islands 15 Altius Client AP Limited Cayman Islands 16 Altius Real Assets (CP) LLC United States 17 Allius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 18 Amal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RP) Proprietary Limited South Africa 20 Assur Conselis Marsh S.A. Senegal 21 Assur Conselis Marsh S.A. Senegal 22 Avascent Canada U.C Canada 23 BenPool Re IC United States 24 Best Insurance, Inc. (Fluxusma Branch) United States 25 Best Insurance, Inc. (Fluxusma Branch) Ference, Inc. (Fluxusma Branch) 26 Best Insurance, Inc. (Nayama Branch) Ference, Inc. (Okayama Branch) 27 Best Insurance, Inc. (Nayama Branch) United Kingdom 31 Blue Marble Microinsurance, Inc. Okara Branch 32 Blue Marble Microinsurance, Inc. Okara Branch 33 Bluefin Insurance Services Limited United Kingdom <td></td> <td></td> <td>•</td>			•
15 Altius Client JP Limited Cayman Islands 16 Altius Real Assets (CP) LLC United States 17 Altius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 18 Amal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assurance Services Corporation United States 21 Assurance Services Corporation United States 23 BenPool Re IC Canada 24 Avascent Canada ULC Canada 25 Best Insurance, Inc. (Fukouka Branch) Test Insurance, Inc. (Fukouka Branch) 26 Best Insurance, Inc. (Fukouka Branch) Test Insurance, Inc. (Riva Branch) 27 Best Insurance, Inc. (Nara Branch) Test Insurance, Inc. (Okayama Branch) 29 Best Insurance, Inc. (Okayama Branch) Test Insurance, Inc. (Okayama Branch) 29 Best Insurance, Inc. (Okayama Branch) United Kingdom 31 Blue Marble Microinsurance, Inc. United Kingdom 32 Bluefin Insurance Group Limited United Kingdom 33 Bluefin Insurance Services Limited United Kingdom		· ,	
16 Altius Real Assets (GP) LLC United States 17 Altius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 18 Amal Insurance Brokers Limited (in liquidation) Sould Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Conselis Marsh S.A. Senegal 21 Assurace Services Corporation United States 22 Avascent Canada ULC Canada 23 BenPool Re IC United States 4 Best Insurance, Inc. Japan 5 Best Insurance, Inc. (Fukouka Branch) Ferein Surance, Inc. (Fukouka Branch) 26 Best Insurance, Inc. (Fukouka Branch) Ferein Surance, Inc. (Hiroshima Branch) 27 Best Insurance, Inc. (Hiroshima Branch) Ferein Surance, Inc. (Nara Branch) 28 Best Insurance, Inc. (Nara Branch) Ferein Surance, Inc. (Nara Branch) 31 Blue Marble Micro Limited United Kingdom 31 Blue Marble Micro Limited United States 32 Bluefin Insurance Group Limited United States 33 Bluefin Insurance Services Limit		,	
17 Altius Real Assets Management S.a.r.I (GP) (Luxembourg) Luxembourg 18 Amal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Conseils Marsh S.A. Senegal 21 Assurance Services Corporation United States 22 Avascent Canada ULC Canada 23 BenPool Re IC United States 24 Best Insurance, Inc. Japan 25 Best Insurance, Inc. (Fickouka Branch) Frequence, Inc. (Fickouka Branch) 27 Best Insurance, Inc. (Hiroshima Branch) Frequence, Inc. (Kara Branch) 28 Best Insurance, Inc. (Koyama Branch) Frequence, Inc. (Koyama Branch) 29 Best Insurance, Inc. (Koyama Branch) Frequence, Inc. (Koyama Branch) 30 BHM Consultores S.A. Chile 31 Blue Marble Micro Limited United States 32 Blue Marble Micro Limited United Kingdom 33 Blue Marble Micro Limited United Kingdom 34 Blue Marble Micro Limited United Kingdom			
18 Amal Insurance Brokers Limited (in liquidation) Saudi Arabia 19 Annison 18 (RF) Proprietary Limited South Africa 20 Assur Conseils Marsh S.A. Senegal 21 Assurance Services Corporation United States 22 Avascent Canada ULC Canada 23 Berbool Re IC United States 24 Best Insurance, Inc. (Fukuwa Branch) Japan 25 Best Insurance, Inc. (Fukuyana Branch) Ferbilance, Inc. (Fukuyana Branch) 27 Best Insurance, Inc. (Hiroshima Branch) Ferbilance, Inc. (Nara Branch) 28 Best Insurance, Inc. (Nara Branch) Ferbilance, Inc. (Okayama Branch) 30 BHM Consultores S.A. Chile 31 Blue Marble Micro Limited United Kingdom 31 Blue Marble Micro Isurance, Inc. United Kingdom 34 Bluefin Insurance Group Limited United Kingdom 34 Bluefin Insurance Services Limited United Kingdom 35 Bowring (Bermuda) Investments Ltd. Bermuda 36 Bowring (Bermuda) Investments Ltd. Bermuda 37 Bowring Marsh (Hong Kong) Limited United Ki		· /	
19Annison 18 (RF) Proprietary LimitedSouth Africa20Assur Consells Marsh S.A.Senegal1Assurance Services CorporationUnited States22Avascent Canada ULCCanada23BenPool Re ICUnited States24Best Insurance, Inc.Japan25Best Insurance, Inc. (Fukouka Branch)Fest Insurance, Inc. (Fukouyama Branch)26Best Insurance, Inc. (Fluknyama Branch)Fest Insurance, Inc. (Mara Branch)27Best Insurance, Inc. (Nara Branch)Fest Insurance, Inc. (Okayama Branch)29Best Insurance, Inc. (Okayama Branch)Chile30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Micro LimitedUnited Kingdom33Blue Marble Micro LimitedUnited Kingdom34Bluein Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited Kingdom36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina41Carpenter Marsh Fac Brasil Corretora de Reaseguros LtdaBrazil42Carpenter Marsh Fac Chille Corredores de Reaseguros SA.Colombia45Carpenter Marsh Fac Chille Corredores de Reaseguros SA.Colombia46Carpenter M			ŭ .
20Assur Conseils Marsh S.A.Senegal21Assurance Services CorporationUnited States22Avascent Canada ULCCanada23BenPool Re ICUnited States24Best Insurance, Inc. (Fukouka Branch)Japan25Best Insurance, Inc. (Fukouka Branch)		, , ,	
21Assurance Services CorporationUnited States22Avascent Canada ULCCanada23BenPool Re ICUnited States24Best Insurance, Inc.Japan25Best Insurance, Inc. (Fukouka Branch)Japan26Best Insurance, Inc. (Fukouka Branch)Ference Inc. (Fukouka Branch)28Best Insurance, Inc. (Idiroshima Branch)Ference Inc. (Okayama Branch)29Best Insurance, Inc. (Okayama Branch)Ference Inc. (Okayama Branch)30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Micro LimitedUnited States33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited Kingdom36Bowring (Bermuda) Investments Ltd.Bermuda38Bowring Marsh (Bermuda) Ltd.Bermuda39Calm Treasury Holdings LimitedUnited Kingdom30Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Collebia Corredores de Reaseguros SAArgentina43Carpenter Marsh Fac Chile Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.Colombia46Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.Cyprus47Carpenter Marsh Fac Re LCCygrus48Carpe		, , , ,	
22Avascent Canada ULCCanada23BenPool Re ICUnited States24Best Insurance, Inc.Japan25Best Insurance, Inc. (Fukouka Branch)			
23BenPool Re ICUnited States24Best Insurance, Inc. (Fukouka Branch)Japan25Best Insurance, Inc. (Fukuyama Branch)		·	
24Best Insurance, Inc. (Fukouka Branch)25Best Insurance, Inc. (Fukouka Branch)26Best Insurance, Inc. (Fukuyama Branch)27Best Insurance, Inc. (Nara Branch)28Best Insurance, Inc. (Okayama Branch)29Best Insurance, Inc. (Okayama Branch)30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Micro LimitedUnited Kingdom33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Bourling (Bermuda) Investments Ltd.Bermuda36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring (Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedUnited Kingdom40Calm Treasury Foldings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Reaseguros LimitadaBrazil43Carpenter Marsh Fac Colombia Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Rea El LCUnited States46Carpenter Marsh Fac Rea El LCUnited States47Carpenter Turner Cyprus LtdCarpenter Turner Cyprus LtdCarpenter Turner Cyprus Ltd			
25Best Insurance, Inc. (Fukuyama Branch)26Best Insurance, Inc. (Hiroshima Branch)27Best Insurance, Inc. (Hiroshima Branch)28Best Insurance, Inc. (Nara Branch)29Best Insurance, Inc. (Okayama Branch)30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Micro LimitedUnited Kingdom33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited Kingdom36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina43Carpenter Marsh Fac Chile Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.Colombia46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner Cyprus LtdCyprus			
26Best Insurance, Inc. (Fukuyama Branch)27Best Insurance, Inc. (Hiroshima Branch)28Best Insurance, Inc. (Nara Branch)29Best Insurance, Inc. (Okayama Branch)30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Microinsurance, Inc.United States33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited Kingdom36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina43Carpenter Marsh Fac Chile Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.Colombia46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner Cyprus LtdCyprus		· · · · · · · · · · · · · · · · · · ·	Јара п
27Best Insurance, Inc. (Hiroshima Branch)28Best Insurance, Inc. (Okayama Branch)29Best Insurance, Inc. (Okayama Branch)30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Microinsurance, Inc.United States33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited States36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LitdaBrazil43Carpenter Marsh Fac Chile Corredores de Reaseguros S.A.Colombia44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.Colombia46Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner Cyprus LtdCyprus		. ,	
28Best Insurance, Inc. (Nara Branch)29Best Insurance, Inc. (Okayama Branch)30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Microinsurance, Inc.United Kingdom33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited States36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LimitadaChile44Carpenter Marsh Fac Chile Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner Cyprus LtdCyprus			
29Best Insurance, Inc. (Okayama Branch)30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Micro Insurance, Inc.United States33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited States36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LtdaBrazil43Carpenter Marsh Fac Chile Corredores de Reaseguros S.A.Colombia44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Turner Cyprus LtdCyprus48Carpenter Turner Cyprus LtdCyprus		, , ,	
30BHM Consultores S.A.Chile31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Microinsurance, Inc.United States33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited States36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LtdaBrazil43Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Re LLCUnited States46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece			
31Blue Marble Micro LimitedUnited Kingdom32Blue Marble Microinsurance, Inc.United States33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited States36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corredores de Reaseguros LtdaBrazil43Carpenter Marsh Fac Collombia Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Collombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece			Oleile
32Blue Marble Microinsurance, Inc.United States33Bluefin Insurance Group LimitedUnited Kingdom34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited States36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LtdaBrazil43Carpenter Marsh Fac Chile Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece			
Bluefin Insurance Group Limited 34 Bluefin Insurance Services Limited 35 Boulder Claims, LLC 36 Bowring (Bermuda) Investments Ltd. 37 Bowring Marsh (Bermuda) Ltd. 38 Bowring Marsh (Hong Kong) Limited 39 Calm Treasury Holdings Limited 40 Calm Treasury Services (Barbados) SRL 41 Carpenter Marsh Fac Argentina Corredores de Reaseguros SA 42 Carpenter Marsh Fac Brasil Corretora de Reaseguros Ltda 43 Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A. 44 Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A. 45 Carpenter Marsh Fac Peru Corredores de Reaseguros S.A. 46 Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C. 47 Carpenter Turner Cyprus Ltd 48 Carpenter Turner S.A. 49 United Kingdom 40 United Kingdom 40 United Kingdom 41 United Kingdom 42 Carpenter Marsh Fac Argentina Corredores de Reaseguros SA 43 Argentina 44 Carpenter Marsh Fac Brasil Corredores de Reaseguros Limitada 45 Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.C. 46 Carpenter Turner Cyprus Ltd 47 Carpenter Turner Cyprus Ltd 48 Carpenter Turner S.A. 48 Carpenter Turner S.A. 49 Greece			
34Bluefin Insurance Services LimitedUnited Kingdom35Boulder Claims, LLCUnited States36Bowring (Bermuda) Investments Ltd.Bermuda37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LtdaBrazil43Carpenter Marsh Fac Chile Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece		,	
Boulder Claims, LLC Bermuda Hong Kong United Kingdom Calm Treasury Holdings Limited United Kingdom Barbados Carpenter Marsh Fac Argentina Corredores de Reaseguros SA Argentina Carpenter Marsh Fac Brasil Corredores de Reaseguros Ltda Brazil Carpenter Marsh Fac Chile Corredores de Reaseguros Limitada Chile Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A. Colombia Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C. Peru Carpenter Marsh Fac Re LLC United States Carpenter Turner Cyprus Ltd Carpenter Turner Cyprus Ltd Carpenter Turner S.A. Greece		·	
Bermuda Bermud			-
37Bowring Marsh (Bermuda) Ltd.Bermuda38Bowring Marsh (Hong Kong) LimitedHong Kong39Calm Treasury Holdings LimitedUnited Kingdom40Calm Treasury Services (Barbados) SRLBarbados41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LtdaBrazil43Carpenter Marsh Fac Chile Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece			
Bowring Marsh (Hong Kong) Limited United Kingdom Calm Treasury Holdings Limited United Kingdom Calm Treasury Services (Barbados) SRL Barbados Carpenter Marsh Fac Argentina Corredores de Reaseguros SA Argentina Carpenter Marsh Fac Brasil Corretora de Resseguros Ltda Brazil Carpenter Marsh Fac Chile Corredores de Reaseguros Limitada Chile Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A. Colombia Carpenter Marsh Fac Peru Corredores de Reaseguros S.A. Colombia Carpenter Marsh Fac Re LLC United States Carpenter Turner Cyprus Ltd Cyprus Carpenter Turner S.A. Greece			
39 Calm Treasury Holdings Limited 40 Calm Treasury Services (Barbados) SRL 41 Carpenter Marsh Fac Argentina Corredores de Reaseguros SA 42 Carpenter Marsh Fac Brasil Corretora de Resseguros Ltda 43 Carpenter Marsh Fac Chile Corredores de Reaseguros Limitada 44 Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A. 45 Carpenter Marsh Fac Peru Corredores de Reaseguros S.A. 46 Carpenter Marsh Fac Re LLC 47 Carpenter Turner Cyprus Ltd 48 Carpenter Turner S.A. 48 Carpenter Turner S.A. 49 United States 40 Carpenter Turner S.A. 40 Greece			
40 Calm Treasury Services (Barbados) SRL 41 Carpenter Marsh Fac Argentina Corredores de Reaseguros SA 42 Carpenter Marsh Fac Brasil Corretora de Resseguros Ltda 43 Carpenter Marsh Fac Chile Corredores de Reaseguros Limitada 44 Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A. 45 Carpenter Marsh Fac Peru Corredores de Reaseguros S.A. 46 Carpenter Marsh Fac Re LLC 47 Carpenter Turner Cyprus Ltd 48 Carpenter Turner S.A. 49 Barbados Argentina Barbados Argentina Brazil Chile Colombia Chile United Colombia Colombia Carpenter Marsh Fac Re LLC United States Cyprus Greece		- ,,	· •
41Carpenter Marsh Fac Argentina Corredores de Reaseguros SAArgentina42Carpenter Marsh Fac Brasil Corretora de Resseguros LtdaBrazil43Carpenter Marsh Fac Chile Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece		<u> </u>	•
42Carpenter Marsh Fac Brasil Corretora de Resseguros LtdaBrazil43Carpenter Marsh Fac Chile Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece			
43Carpenter Marsh Fac Chile Corredores de Reaseguros LimitadaChile44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece			·
44Carpenter Marsh Fac Colombia Corredores de Reaseguros S.A.Colombia45Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C.Peru46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece			
45 Carpenter Marsh Fac Peru Corredores de Reaseguros S.A.C. 46 Carpenter Marsh Fac Re LLC 47 Carpenter Turner Cyprus Ltd 48 Carpenter Turner S.A. 48 Carpenter Turner S.A. 49 Greece			
46Carpenter Marsh Fac Re LLCUnited States47Carpenter Turner Cyprus LtdCyprus48Carpenter Turner S.A.Greece		· · · · · · · · · · · · · · · · · · ·	
47 Carpenter Turner Cyprus Ltd Cyprus 48 Carpenter Turner S.A. Greece		· · · · · · · · · · · · · · · · · · ·	
48 Carpenter Turner S.A. Greece		·	
·		·	* '
40 Concede Degional Holdings Limited		•	
49 Cascade Regional Holdings Limited United Kingdom	49	Cascade Regional Holdings Limited	United Kingdom

		11.5.116.1
50 E1	Central Insurance Services Limited	United Kingdom
51	Chartwell Healthcare Limited	United Kingdom
52	Clark Thomson Insurance Brokers Limited	United Kingdom
53	Client Provide Limited	New Zealand
54	Consultores 2020 C.A.	Venezuela, Bolivarian Republic of
55	Continental Owner Operators Insurance Services, Inc.	United States
56	CPRM Limited	United Kingdom
57	Darwin Technologies Holdings Limited	United Kingdom
58	Darwin Technologies Limited	United Kingdom
59	Darwin Technologies S.R.L.	Romania
60	Deasterra Partners, S.L.	Spain
61	Deasterra Services, S.L.	Spain
62	DeLima Marsh S.A Los Corredores de Seguros S.A.	Colombia
63	DVA - Deutsche Verkehrs-Assekuranz-Vermittlungs GmbH	Germany
64	Echelon Australia Pty Limited	Australia
65	Echelon Claims Consultants Sdn Bhd	Malaysia
66	EnBW Versicherungs Vermittlung GmbH	Germany
67	English Pension Trustees Limited	United Kingdom
68	Epsilon (US) Insurance Company	United States
69	Epsilon Insurance Company, Ltd.	Cayman Islands
70	Evolve Intelligence Pty Ltd	Australia
71	Evolve Intelligence Unit Trust	
72	Exmoor Management Company Limited	Bermuda
73	Freedom Trust Services Limited	Ireland
74	Gard Fund General Partner S.a.r.l.	Luxembourg
75	GC Genesis LLC	United States
76	Gem Insurance Company Limited	Bermuda
77	Global Premium Finance Company	United States
78	Group Promoters Pty Limited	Australia
79	Guy Carpenter & Cia., S.A.	Spain
80	Guy Carpenter & Co. Labuan Ltd.	Malaysia
31	Guy Carpenter & Company AB	Sweden
82	Guy Carpenter & Company Corredores de Reaseguros SpA	Chile
33	Guy Carpenter & Company Corretora de Resseguros Ltda.	Brazil
84	Guy Carpenter & Company GmbH	Germany
35	Guy Carpenter & Company Gmbh (Austria Branch)	Commany
36 36	Guy Carpenter & Company GmbH (Czech Branch)	
87	Guy Carpenter & Company GmbH (Czech Branch)	
88	Guy Carpenter & Company GmbH (Switzerland Branch)	
39	Guy Carpenter & Company GmbH (UK Branch) Guy Carpenter & Company Limited	United Kingdom
90	• • • • • • • • • • • • • • • • • • • •	United Kingdom
91	Guy Carpenter & Company Limited	New Zealand
92	Guy Carpenter & Company Peru Corredores de Reaseguros S.A.C.	Peru
93	Guy Carpenter & Company Private Limited	Singapore
94	Guy Carpenter & Company Proprietary Limited	South Africa
95	Guy Carpenter & Company Pty. Ltd.	Australia
96	Guy Carpenter & Company S.A. (Netherlands Branch)	
97	Guy Carpenter & Company S.r.l.	Italy
98	Guy Carpenter & Company, Limited	Hong Kong
99	Guy Carpenter & Company, LLC	United States
100	Guy Carpenter & Company, LLC (Taiwan Branch)	
	Guy Carpenter & Company, LtC (Taiwan Branch) Guy Carpenter & Company, Ltd./Guy Carpenter & Compagnie, Ltee	Canada
100 101 102		Canada Belgium

104	Guy Carpenter & Company, S.A.S.	France
105	Guy Carpenter (Middle East) Limited	United Arab Emirates
106	Guy Carpenter Bermuda Ltd.	Bermuda
107	Guy Carpenter Broking, Inc.	United States
108	Guy Carpenter Colombia Corredores de Reaseguros Ltda.	Colombia
109	Guy Carpenter Insurance Brokers (Beijing) Co. Ltd.	China
110	Guy Carpenter Insurance Brokers (Beijing) Co., Ltd (Shanghai Branch)	
111	Guy Carpenter Japan, Inc.	Japan
112	Guy Carpenter Mexico Intermediario de Reaseguro, S.A. de C.V.	Mexico
113	Guy Carpenter Reasurans Brokerligi Anonim Sirketi	Turkey
114	Guy Carpenter Reinsurance Broker Philippines, Inc.	Philippines
115	Hamilton Bond Limited	United Kingdom
116	HAPIP GP 2009, LLC	United States
117	HAPIP GP, LLC	United States
118	Hayward Aviation Limited	United Kingdom
119	HSBC Insurance Brokers International (Abu Dhabi) LLC (in liquidation)	United Arab Emirates
120	Induslynk Training Services Private Limited	India
121	Insbrokers Ltda.	Uruguay
122	InSolutions Limited	United Kingdom
123	Insure Direct - Jardine Lloyd Thompson Limited	United Arab Emirates
124	Insure Direct (Brokers) LLC	United Arab Emirates
125	Insure Direct (Brokers) LLC [BAHRAIN BRANCH]	
126	InsurTech Alliance, LLC	United States
127	International Catastrophe Insurance Managers, LLC	United States
128	International Loss Control Services Limited (In Liquidation)	Ireland
129	Irish Pensions Trust Limited	Ireland
130	Isosceles Insurance (Barbados) Limited	Barbados
131	Isosceles Insurance Ltd	Bermuda
132	Isosceles PCC Limited	Guernsey
133	J&H Marsh & McLennan Limited	Hong Kong
134	J.W. Terrill Benefit Administrators, Inc.	United States
135	Japan Affinity Marketing, Inc.	Japan
136	Jardine Lloyd Thompson Limited	Thailand
137	Jelf Commercial Finance Limited	United Kingdom
138	Jelf Financial Planning Limited	United Kingdom
139	Jelf Insurance Brokers Limited	United Kingdom
140	Jelf Limited	United Kingdom
141	Jelf Risk Management Limited	United Kingdom
142	JI Holdings Limited	Mauritius
143	JIB Group Holdings Limited	United Kingdom
144	JIB Group Limited	United Kingdom
145	JIB Holdings (Pacific) Limited	Virgin Islands, British
146	JIB Overseas Holdings Limited	United Kingdom
147	JIB UK Holdings Limited	United Kingdom
148	JLT Advisory Limited	United Kingdom
149	JLT Advisory Services Limited	India
150	JLT Asesorias Ltda	Chile
151	JLT Asia Holdings BV	Netherlands
152	JLT Chile Holdings SpA	Chile
153	JLT Colombia Retail Limited	United Kingdom
154	JLT Colombia Wholesale Limited	United Kingdom
155	JLT Consultants & Actuaries Limited	United Kingdom
156	JLT EB Services Limited	United Kingdom
157	JLT Employee Benefits Holding Company (PTY) LTD	South Africa

158	JLT Financial Planning Limited (In Liquidation)	Ireland
159	JLT Group Holdings Limited	United Kingdom
160	JLT Group Services Pty Limited	Australia
161	JLT Holdings (Barbados) Ltd	Barbados
162	JLT Holdings (Bermuda) Ltd.	Bermuda
163	JLT Independent Consultancy Services Private Limited	India
164	JLT Insurance Brokers Ireland Limited (In Liquidation)	Ireland
165	JLT Insurance Brokers Limited	Hong Kong
166	JLT Insurance Brokers Limited (Macao Branch)	Macao
167	JLT Insurance Brokers SA	Argentina
168	JLT Insurance Group Holdings Ltd	United Kingdom
169	JLT Intellectual Property [UK Branch]	United Kingdom
170	JLT Intellectual Property Limited	Ireland
171	JLT Investment Management Limited	United Kingdom
172	JLT LATAM (Southern Cone) Wholesale Limited	United Kingdom
173	JLT Latin American Holdings Limited	United Kingdom
174	JLT Life Assurance Brokers Limited	Thailand
175	JLT Management Pte. Ltd.	Singapore
176	JLT Management Services Limited	United Kingdom
177	JLT Mexico Holdings Limited	United Kingdom
178	JLT Mexico, Intermediario de Reaseguro, S.A. de C.V.	Mexico
179	JLT Pension Trustees Ireland Limited (In Liquidation)	Ireland
180	JLT Pensions Administration Limited	United Kingdom
181	JLT Peru Reinsurance Solutions Limited	United Kingdom
182	JLT Peru Retail Limited	United Kingdom
183	JLT Peru Wholesale Limited	United Kingdom
184	JLT QFM Services Limited	Ireland
185	JLT Re Argentina Corredores de Reaseguros S.A.U.	Argentina
186	JLT Re Limited	United Kingdom
187	JLT Re Pty Ltd	Australia
188	JLT Reinsurance Brokers Limited	United Kingdom
189	JLT Risk Solutions Pty Ltd	Australia
190	JLT SA IB Holdings Company Proprietary Limited	South Africa
191	JLT Singapore Holdings Pte. Ltd.	Singapore
192	JLT Singapore Holdings Pte. Ltd. JLT Specialty (SWISS Branch)	Switzerland
193	JLT Specialty Limited	United Kingdom
194	JLT Specialty Pte. Ltd (Hong Kong branch)	Cinconore
195	JLT Specialty Pte. Ltd.	Singapore
196	JLT Towner Insurance Management (Anguilla) Limited	Anguilla
197	JLT Trust Services (Barbados) Ltd	Barbados
198	JLT UK Investment Holdings Limited	United Kingdom
199	JLT Vantage Risk and Benefit Consulting Private Limited	India
200	JLT Wealth Management Limited	United Kingdom
201	JMIB Holdings BV	Netherlands
202	Johnson & Higgins (Bermuda) Limited	Bermuda
203	JSC "Marsh - consulting services"	Russian Federation
204	Kepler Associates Limited	United Kingdom
205	Kessler & Co AG	Switzerland
206	Kessler & Co Inc.	Liechtenstein
207	Kessler Consulting Inc.	Switzerland
208	Kessler Prevoyance Inc.	Switzerland
209	KFAS GP S.a.r.l.	Luxembourg
210	Lambert Brothers Brokers (Employee Benefits) Limited	Hong Kong
211	Lambert Brothers Holdings Limited	Hong Kong

212	Lambert Brothers Insurance Brokers (Hong Kong) Ltd	Hong Kong
213	Lavaretus Underwriting AB	Sweden
214	Lavaretus Underwriting AB (BRANCH - Finland)	Finland
215	MACC Asistencias SpA	Chile
216	MAG SpA	Italy
217	Malcolm Investment Holdings Limited	Barbados
218	Mangrove Cell 5 IC	United States
219	Mangrove Cell 6 IC	United States
220	Mangrove Delaware Insurance Solutions Inc	United States
221	Mangrove Insurance Europe PCC Limited	Malta
222	Mangrove Insurance Guernsey PCC Limited	Guernsey
223	Mangrove Insurance Solutions PCC Limited	Isle of Man
224	Mangrove Insurance Solutions, PCC	United States
225	Marine, Aviation & General (London) Limited	United Kingdom
226	Marley Eternit Fund General Partner S.a.r.l.	Luxembourg
227	Marsh & McLennan (PNG) Limited	Papua New Guinea
228	Marsh & McLennan Agencies AS	Norway
229	Marsh & McLennan Agencies Limited	Hong Kong
230	Marsh & McLennan Agency Limited	New Zealand
231	Marsh & McLennan Agency LLC	United States
232	Marsh & McLennan Agency Pty Ltd.	Australia
233	Marsh & McLennan Asia Business Services Sdn. Bhd.	Malaysia
234	Marsh & McLennan Colombia S.A.S	Colombia
235	Marsh & McLennan Companies Asia Pacific Treasury Center Limited	Hong Kong
236	Marsh & McLennan Companies Finance Center (Luxembourg) S.a.r.l.	Luxembourg
237	Marsh & McLennan Companies Finance Center (Luxembourg) S.a.r.l. (Barbados Branch)	
238	Marsh & McLennan Companies France S.A.S.	France
239	Marsh & McLennan Companies Holdings (Luxembourg) S.a.r.l.	Luxembourg
240	Marsh & McLennan Companies UK Limited	United Kingdom
241	Marsh & McLennan Companies, Inc.	United States
242	Marsh & McLennan Companies, Inc. (UK Establishment)	Simou States
243	Marsh & McLennan Deutschland GmbH	Germany
244	Marsh & McLennan Europe S.a.r.l.	Luxembourg
245	Marsh & McLennan Europe S.a.r.I. (Barbados Branch)	Laxembourg
246	Marsh & McLennan Finance Unlimited Company	Ireland
247	Marsh & McLennan Global Broking (Bermuda) Ltd.	Bermuda
248	Marsh & McLennan Global Services India Pvt. Ltd.	India
249	Marsh & McLennan Holding GmbH	Germany
250	Marsh & McLennan Holdings (Canada) ULC	Canada
251	Marsh & McLennan Innovation Centre Limited	Ireland
252	Marsh & McLennan Insurance Services Limited	Hong Kong
253	Marsh & McLennan Ireland Limited	Ireland
254	Marsh & McLennan Management Services (Bermuda) Limited	Bermuda
255	Marsh & McLennan Risk Capital Holdings, Ltd.	United States
256	Marsh & McLennan Servicios, S.A. De C.V.	Mexico
257	Marsh & McLennan Shared Services Canada Limited	Canada
258	Marsh & McLennan Shared Services, LLC	United States
259	Marsh (Bahrain) Company WLL	Bahrain
260	Marsh (Beijing) Risk Management Consulting Co., Ltd.	China
201	Marsh (China) Insurance Brokers Co., Ltd.	China
261		
262	Marsh (China) Insurance Brokers Co., Ltd. (Chengdu Branch)	
262 263	Marsh (China) Insurance Brokers Co., Ltd. (Guangdong Branch)	
262		

266	Marsh (China) Insurance Brokers Co., Ltd. (Jiangsu Branch)	
267	Marsh (China) Insurance Brokers Co., Ltd. (Liaoning Branch)	
268	Marsh (China) Insurance Brokers Co., Ltd. (Shandong Branch)	
269	Marsh (China) Insurance Brokers Co., Ltd. (Shanghai Branch)	
270	Marsh (China) insurance Brokers Co., Ltd. (Shenzhen Branch)	
271	Marsh (China) Insurance Brokers Co., Ltd. (Tianjin Branch)	
272	Marsh (China) Insurance Brokers Co., Ltd. (Xiamen Branch)	
273	Marsh (Hong Kong) Limited	Hong Kong
274	Marsh (Insurance Brokers) LLP	Kazakhstan
275	Marsh (Malawi) Limited	Malawi
276	Marsh (Middle East) Limited	United Kingdom
277	Marsh (Middle East) Limited (Abu Dhabi Branch)	
278	Marsh (Middle East) Limited (Egypt Rep Office)	
279	Marsh (Middle East) Limited (Iraq Branch)	
280	Marsh (Namibia) Proprietary Limited	Namibia
281	Marsh (Singapore) Pte. Ltd.	Singapore
282	Marsh (Singapore) Pte. Ltd. (Myanmar Rep Office)	3.4
283	Marsh A/S	Denmark
284	Marsh AB	Sweden
285	Marsh Advantage Insurance Pty Ltd.	Australia
286	Marsh Advisory S.A.C.	Peru
287	Marsh Advisory S.r.I.	Italy
288	Marsh Advisory Services S.R.L.	Romania
289	Marsh Africa (Pty) Ltd	South Africa
290	Marsh AG	Switzerland
291	Marsh Argentina S.R.L.	Argentina
292	Marsh AS	Norway
293	Marsh Associates Proprietary Limited	South Africa
294	Marsh Austria G.m.b.H.	
295	Marsh B.V.	Austria Netherlands
296	Marsh Broskman (Proprietary) Limited	Botswana
297	Marsh Brockman y Schuh, Agente de Seguros y de Fianzas, S.A. de C.V.	Mexico
298	Marsh Broker de Asigurare-Reasigurare S.R.L.	Romania
299	Marsh Broker Japan, Inc.	Japan Kana
300	Marsh Brokers (Hong Kong) Limited	Hong Kong
301	Marsh Canada Limited/Marsh Canada Limitee	Canada
302	Marsh Company Management Services Cayman Ltd.	Cayman Islands
303	Marsh Corporate Services (Barbados) Limited	Barbados
304	Marsh Corporate Services Isle of Man Ltd	Isle of Man
305	Marsh Corporate Services Limited	United Kingdom
306	Marsh Corporate Services Malta Limited	Malta
307	Marsh Corretora de Seguros Ltda.	Brazil
308	Marsh d.o.o. Beograd	Serbia
309	Marsh d.o.o. za posredovanje u osiguranju	Croatia
310	Marsh Emirates Consultancy LLC	United Arab Emirates
311	Marsh Emirates Insurance Brokerage LLC	United Arab Emirates
312	Marsh Emirates insurance Brokerage LLC (Abu Dhabi Branch)	
313	Marsh Employee Benefits Limited	Ireland
314	Marsh Engineering Consulting (Shanghai) Co., Ltd.	China
315	Marsh EOOD	Bulgaria
316	Marsh Eurofinance B.V.	Netherlands
317	Marsh Europe S.A.	Belgium
318	Marsh Europe S.A. (Romania Permanent Establishment)	
319	Marsh Europe S.A. (Slovakia Branch)	

320	Marsh Europe S.A. (Slovenia Branch)	
321	Marsh Europe S.A. (Ukraine Branch)	
322	Marsh FJC International Insurance Brokers Limited	Nigeria
323	Marsh for Insurance Brokerage S.A.E.	Egypt
324	Marsh for Insurance Consulting	Egypt
325	Marsh for Insurance Services - Jordan	Jordan
326	Marsh Franco Acra, S.A.	Dominican Republic
327	Marsh GmbH	Germany
328	Marsh GSC Servicos e Administracao de Seguros Ltda.	Brazil
329	Marsh Guy Carpenter Reinsurance Brokers Saudi Arabia	Saudi Arabia
330	Marsh Holding AB	Sweden
331	Marsh India Insurance Brokers Private Limited	India
332	Marsh Insurance & Investments LLC	United States
333	Marsh Insurance Agencies Limited	Hong Kong
334	Marsh Insurance and Reinsurance Brokers LLC	Azerbaijan
335	Marsh Insurance Brokers (Macao) Limited	Macao
336	Marsh Insurance Brokers (Malaysia) Sdn Bhd	Malaysia
337	Marsh Insurance Brokers AO	Russian Federation
338	Marsh Insurance Brokers Limited	Cyprus
339	Marsh Insurance Brokers Limited	Uganda
340	Marsh Insurance Consulting Saudi Arabia	Saudi Arabia
341	Marsh International Broking Holdings Limited	United Kingdom
342	Marsh International Holdings, LLC	United States
343	Marsh Ireland Brokers Limited	Ireland
344	Marsh Ireland Brokers Limited (UK Branch)	notation .
345	Marsh Israel (1999) Ltd.	Israel
346	Marsh Israel (Holdings) Ltd.	Israel
347	Marsh Israel Consultants Ltd.	Israel
348	Marsh Israel Insurance Agency Ltd.	Israel
349	Marsh Israel International Brokers Ltd. (in liquidation)	Israel
350	Marsh Japan, Inc.	Japan
351	Marsh JLT Ireland Holdings Limited	Ireland
352	Marsh JLT Proprietary Limited	South Africa
353	Marsh Kft.	Hungary
354	Marsh Kindlustusmaakler AS	Estonia
355		
	Marsh Korea, Inc. Marsh Limited	Korea, Republic of
356		United Kingdom
357	Marsh Limited [Fiji]	Fiji
358	Marsh Limited [New Zealand]	New Zealand
359	Marsh Limited [PNG]	Papua New Guinea
360	Marsh LLC	United States
361	Marsh LLC [Ukraine]	Ukraine
362	Marsh LLC Insurance Brokers	Greece
363	Marsh Lorant, Agente de Seguros y de Fianzas, S.A. de C.V.	Mexico
364	Marsh Ltd. [Taiwan Branch]	
365	Marsh Ltd. [Wisconsin]	United States
366	Marsh Management Services (Barbados) Limited	Barbados
367	Marsh Management Services (Bermuda) Ltd.	Bermuda
368	Marsh Management Services (Dublin) Limited	Ireland
369	Marsh Management Services (MENA) Limited	United Arab Emirates
370	Marsh Management Services Cayman Ltd.	Cayman Islands
371	Marsh Management Services Guernsey Limited	Guernsey
372	Marsh Management Services Inc.	United States
373	Marsh Management Services Isle of Man Limited	Isle of Man

374	Marsh Management Services Labuan Limited	Malaysia
375	Marsh Management Services Luxembourg S.a.r.l.	Luxembourg
376	Marsh Management Services Luxembourg S.a.r.l. (Zurich/Switzerland Branch)	
377	Marsh Management Services Luxembourg, S.a.r.l. (Vaduz/Liechtenstein Branch)	
378	Marsh Management Services Malta Limited	Malta
379	Marsh Management Services Singapore Pte. Ltd.	Singapore
380	Marsh Management Services Sweden AB	Sweden
381	Marsh Marine & Energy AB	Sweden
382	Marsh Marine (Pty) Ltd	South Africa
383	Marsh McLennan (Australia) Pty	Australia
384	Marsh McLennan Agency A/S	Denmark
385	Marsh McLennan Alpha Limited	United Kingdom
386	Marsh Mclennan India Holdings Limited	United Kingdom
387	Marsh McLennan India Limited	United Kingdom
388	Marsh McLennan International Treasury Services Pte. Ltd.	Singapore
389	Marsh McLennan Investment B.V.	Netherlands
390	Marsh McLennan Morocco	Morocco
391	Marsh McLennan Regional Colombia Holdings, Ltd.	Bermuda
392	Marsh McLennan Regional Holdings Limited	United Kingdom
393	Marsh MEA Ltd	United Arab Emirates
394	Marsh Medical Consulting GmbH	Germany
395	Marsh Mercer Holdings (Australia) Pty Ltd	Australia
396	Marsh Morocco	Morocco
397	Marsh Oman LLC	Oman
398	Marsh Oy	Finland
399	Marsh PB Co., Ltd.	Thailand
400	Marsh Peru S.A.C. Corredores de Seguros	Peru
401	Marsh Philippines, Inc.	Philippines
402	Marsh Proprietary Limited	South Africa
403	Marsh Pty Ltd	Australia
404	Marsh Qatar LLC	Qatar
405	Marsh Resolutions Pty Limited	Australia
406	Marsh Risk and Consulting Services (Pty) Ltd	Namibia
407	Marsh Risk Consulting B.V.	Netherlands
408	Marsh Risk Consulting Limitada	Chile
409	Marsh Risk Consulting Ltda.	Colombia
410	Marsh Risk Consulting, S.L.	Spain
411	Marsh S.A. Corredores De Seguros	Chile
412	Marsh S.A.S.	France
413	Marsh S.p.A.	Italy
414	Marsh s.r.o.	Czech Republic
415	Marsh s.r.o.	Slovakia
416	Marsh SA (Luxembourg Branch)	
417	Marsh SA (UK Branch)	
418	Marsh SA [Argentina]	Argentina
419	Marsh SA [Belgium]	Belgium
420	Marsh SA [Uruguay]	Uruguay
421	Marsh Saldana Inc.	Puerto Rico
422	Marsh Saudi Arabia Insurance & Reinsurance Brokers	Saudi Arabia
423	Marsh Secretarial Services Limited	United Kingdom
424	Marsh Semusa, S.A.	Panama
425	Marsh Services Limited	United Kingdom
426	Marsh Services Spolka z.o.o.	Poland
	a.c Co. vioco oponia 2.0.0.	i olaila

427	Marsh SIA	Latvia
428	Marsh Sigorta ve Reasurans Brokerligi Anonim Sirketi	Turkey
429	Marsh Soken, Inc	Japan
430	Marsh Spolka z.o.o.	Poland
431	Marsh Takaful Brokers (Malaysia) Sdn Bhd	Malaysia
432	Marsh Treasury Services Limited	United Kingdom
433	Marsh Tunisia S.a.r.l.	Tunisia
434	Marsh USA Borrower LLC	United States
435	Marsh USA LLC	United States
436	Marsh Venezuela C.A. Sociedad de Corretaje de Seguros	Venezuela, Bolivarian Republic of
437	Marsh Vietnam Insurance Broking Company Ltd	Vietnam
438	Marsh Zambia Limited	Zambia
439	Marsh, Lda.	Portugal
440	Marsh, S.A. Mediadores de Seguros	Spain
441	Matthiessen Assurans AB	Sweden
442	MedPool Re IC	United States
443	Mercer (Argentina) S.A.U.	Argentina
444	Mercer (Australia) Pty Ltd	Australia
445	Mercer (Austria) GmbH	Austria
446	Mercer (Belgium) SA-NV	Belgium
447	Mercer (Canada) Limited Mercer (Canada) limitee	Canada
448	Mercer (China) Limited	China
449	Mercer (China) Limited (Beijing Branch)	
450	Mercer (China) Limited (Guangzhou Branch)	
451	Mercer (China) Limited (Shanghai Branch)	
452	Mercer (China) Limited (Shenzhen Branch)	
453	Mercer (China) Limited (Zhejiang Branch)	
454	Mercer (Colombia) Ltda (Sucursal Peru/Peru Branch)	
455	Mercer (Colombia) Ltda.	Colombia
456	Mercer (Denmark) A/S	Denmark
457	Mercer (Finland) Oy	Finland
458	Mercer (France) SAS	France
459	Mercer (Hong Kong) Limited	Hong Kong
460	Mercer (Hong Kong) Limited (Macao Branch)	
461	Mercer (Ireland) Limited	Ireland
462	Mercer (Malaysia) Sdn. Bhd.	Malaysia
463	Mercer (N.Z.) Limited	New Zealand
464	Mercer (Nederland) B.V.	Netherlands
465	Mercer (Norge) AS	Norway
466	Mercer (Polska) Sp.z o.o.	Poland
467	Mercer (Portugal) - Recursos Humanos, Lda	Portugal
468	Mercer (Singapore) Pte. Ltd.	Singapore
469	Mercer (Sweden) AB	Sweden
470	Mercer (Taiwan) Ltd.	Taiwan (Province of China)
471	Mercer (Thailand) Ltd.	Thailand
472	Mercer (US) LLC	United States
473	Mercer Administration Services (Australia) Pty Limited	Australia
474	Mercer Africa Limited	United Kingdom
475	Mercer Alternatives (Luxembourg) S.a.r.l.	Luxembourg
476	Mercer Alternatives AG	Switzerland
477	MERCER ALTERNATIVES LIMITED	United Kingdom
478	Mercer Asesores de Seguros S.A.	Argentina
479	Mercer Asesores es Inversion Independientes S.A. de C.V.	Mexico
480	Mercer Broking Ltd.	Taiwan (Province of China)
-500	more broking Eta.	raiwan (i rovince oi Chilla)

481	Mercer Consulting (Australia) Pty Ltd	Australia
482	Mercer Consulting (Chile) Limitada	Chile
483	Mercer Consulting (France) SAS	France
484	Mercer Consulting (India) Private Limited	India
485	Mercer Consulting Limited	United Kingdom
486	Mercer Consulting Limited (Saudi Arabia Branch)	
487	Mercer Consulting Venezuela, C.A.	Venezuela, Bolivarian Republic of
488	Mercer Consulting, S.L.U.	Spain
489	Mercer Corredores de Seguros Limitada	Chile
490	Mercer Danismanlik Anonim Sirketi	Turkey
491	Mercer Deutschland GmbH	Germany
492	Mercer Employee Benefits - Mediacao de Seguros Unipessoal Lda.	Portugal
493	Mercer Employee Benefits Limited	United Kingdom
494	Mercer Financial Advice (Australia) Pty Ltd	Australia
495	Mercer Financial Services Limited Liability Company	Morocco
496	Mercer Financial Services Middle East Limited	United Arab Emirates
497	Mercer Global Investments Canada Limited Mercer Gestion mondiale d'investissements Canada limitee	Canada
498	Mercer Global Investments Europe Limited	Ireland
499	Mercer Global Investments Europe Limited (Germany Branch)	
500	Mercer Global Investments Europe Limited (Sweden Branch)	
501	Mercer Global Investments Europe Limited (UK Branch)	
502	Mercer Global Investments Management Limited	Ireland
503	Mercer Health & Benefits (Singapore) Pte. Ltd.	Singapore
504	Mercer Health & Benefits Administration LLC	United States
505	Mercer Health & Benefits LLC	United States
506	Mercer Holdings, Inc. [Philippines]	Philippines
507	Mercer Holdings, LLC	United States
508	Mercer HR Consulting Borrower LLC	United States
509	Mercer HR Services, LLC	United States
510	Mercer Human Resource Consulting Ltda	Brazil
511	Mercer Human Resource Consulting S.A. de C.V.	Mexico
512	Mercer ICC Limited	Guernsey
513	Mercer Infrastructure General Partner S.a.r.l.	Luxembourg
514	Mercer Investment Management (Shanghai) Co., Ltd	China
515	Mercer Investment Solutions (Singapore) Pte. Ltd.	Singapore
516	Mercer Investments (Australia) Limited	Australia
517	Mercer Investments (HK) Limited	Hong Kong
518	Mercer Investments (TR) Elimieu Mercer Investments (Japan), Ltd	Japan Japan
519	Mercer Investments (Japan), Eta	United States
520	Mercer Irrestitients LLC Mercer Ireland Holdings Limited	Ireland
521	Mercer Italia Societa d'Intermediazione Mobiliare SpA	
	'	Italy
522 523	Mercer Italia Srl Socio Unico Mercer Japan Ltd.	Italy
	·	Japan Karaa Basublia of
524	Mercer Korea Co., Ltd.	Korea, Republic of
525	Mercer Limited	United Kingdom
526	Mercer Master Trust No. 1 Designated Activity Company	Ireland
527	Mercer Master Trust No. 2 Designated Activity Company	Ireland
528	Mercer Master Trust No. 3 Designated Activity Company	Ireland
529	Mercer Mauritius Ltd.	Mauritius
530	Mercer MC Consulting Borrower LLC	United States
531	Mercer Outsourcing (Australia) Pty Ltd	Australia
532	Mercer Outsourcing, S.L.U.	Spain
533	MERCER PE FUND II SCOTLAND LP	United Kingdom

534	Mercer PE II GP Ltd	Guernsey
535	Mercer Pensionsfonds AG	Germany
536	Mercer Pensionsraadgivning A/S	Denmark
537	Mercer Philippines, Inc.	Philippines
538	Mercer Private Investment Partner IV General Partner S.a.r.l.	Luxembourg
539	Mercer Private Investment Partner V General Partner S.a.r.l.	Luxembourg
540	Mercer Private Investment Partners VI General Partner S.a.r.l.	Luxembourg
541	Mercer Private Investment Partners VII General Partner S.a.r.l.	Luxembourg
542	Mercer Private Markets Advisers (US) AG	Switzerland
543	Mercer Schweiz AG	Switzerland
544	Mercer Services Australia Pty Ltd	Australia
545	Mercer Services Poland Sp. z.o.o.	Poland
546	Mercer Sigorta Brokerligi Anonim Sirketi	Turkey
547	Mercer South Africa Proprietary Limited	South Africa
548	Mercer Superannuation (Australia) Limited	Australia
549	Mercer System Services LLC	United States
550	Mercer Treuhand GmbH	Germany
551	Mercer Trust Company LLC	United States
552	Mercer Trustees Limited	United Kingdom
553	Mercer Wealth (India) Private Limited	India
554	Mercury Insurance Services Pty Ltd	Australia
555	Mesos Asistencia SpA	Chile
556	Mesos Corredores de Seguros Ltda	Chile
557	<u> </u>	Chile
558	Mesos Gestion y Servicios S.A. MIPP S.a.r.l.	
559		Luxembourg
	MIPP VI S.a.r.l.	Luxembourg
560	MM Asistencias S.A.	Argentina Ctates
561	MMA Asset Management LLC	United States
562	MMA Securities LLC	United States
563	MMB Consultores S.A.	Argentina
564	MMC (Singapore) Holdings Pte. Ltd.	Singapore
565	MMC Borrower LLC	United States
566	MMC Brazilian Holdings B.V.	Netherlands
567	MMC CAPITAL SOLUTIONS LP	United Kingdom
568	MMC Capital Solutions UK Limited	United Kingdom
569	MMC Capital Solutions US LLC	United States
570	MMC Finance (Australia) Limited	United Kingdom
571	MMC FINANCE (EUROPE) LIMITED	United Kingdom
572	MMC Finance (Singapore) Limited	United Kingdom
573	MMC Finance (US) Limited	United Kingdom
574	MMC Finance Holdings (US) Limited	United Kingdom
575	MMC FINANCE HOLDINGS LTD	United Kingdom
576	MMC FINANCE UK LIMITED	United Kingdom
577	MMC Funding (US) Limited	United Kingdom
578	MMC Group Services sp. z o.o.	Poland
579	MMC Holdings (Australia) Pty Ltd	Australia
580	MMC Holdings (New Zealand) ULC	New Zealand
581	MMC Holdings (UK) Limited	United Kingdom
582	MMC International Finance (Barbados) SRL	Barbados
583	MMC International Holdings LLC	United States
584	MMC International Limited	United Kingdom
585	MMC International Treasury Centre Limited	United Kingdom
586	MMC Management Services Proprietary Limited	South Africa
587	MMC Middle East Holdings Limited	United Kingdom
551		

588	MMC Oliver Wyman Bermuda Ltd.	Bermuda
589	MMC Poland Holdings B.V.	Netherlands
590	MMC Regional LATAM Holdings B.V.	Netherlands
591	MMC Securities (Ireland) Limited	Ireland
592	MMC Securities Limited	United Kingdom
593	MMC Securities LLC	United States
594	MMC ShunTak Insurance Brokers Limited	Hong Kong
595	MMC ShunTak Insurance Brokers Limited (Macao Branch)	Macao
596	MMC UK Group Limited	United Kingdom
597	MMC UK Pension Fund Trustee Limited	United Kingdom
598	MMOW Limited	United Kingdom
599	Moola Systems Limited	United Kingdom
600	Mountlodge Limited	United Kingdom
601	MPIP III GP LLC	United States
602	MPIP IV GP LLC	United States
603	MPIP V GP, LLC	United States
604	MPIP VI GP, LLC	United States
605	Muir Beddal (Zimbabwe) Limited	Zimbabwe
606	National Economic Research Associates, Inc.	United States
607	National Economic Research Associates, Inc.	United States
608	National Economic Research Associates, Inc Sucursal en Espana (Spain branch)	
609	NERA Australia Pty. Ltd.	Australia
610	NERA Economic Consulting GmbH	Germany
611	NERA Economic Consulting Limited	New Zealand
612	NERA S.R.L.	Italy
613	NERA SAS	France
614	NERA UK Limited	United Kingdom
615	NERA UK Limited (Belgium Branch)	3
616	Neuburger Noble Lowndes GmbH	Germany
617	Normandy Reinsurance Company Limited	Bermuda
618	Oliver Wyman (Hong Kong) Limited	Hong Kong
619	Oliver Wyman (Perth) Holdings Pty Ltd	Australia
620	Oliver Wyman (Perth) Pty Ltd	Australia
621	Oliver Wyman AB	Sweden
622	Oliver Wyman Actuarial Consulting, Inc.	United States
623	Oliver Wyman AG	Switzerland
624	Oliver Wyman Austria GmbH	Austria
625	Oliver Wyman B.V.	Netherlands
626	Oliver Wyman Co., Ltd.	Thailand
627	Oliver Wyman Consulting (Shanghai) Ltd	China
628	Oliver Wyman Consulting (Shanghai) Ltd (Beijing Branch)	China
629	Oliver Wyman Consultoria em Estrategia de Negocios Ltda.	Brazil
630	Oliver Wyman Energy Consulting Limited	United Kingdom
631	Oliver Wyman FZ-LLC	United Arab Emirates
632	Oliver Wyman GmbH	
	•	Germany
633	Oliver Wyman Group KK	Japan Linited Kingdom
634	Oliver Wyman Limited	United Kingdom
635	Oliver Wyman Limited (Abu Dhabi branch)	
636	Oliver Wyman Limited (India Branch)	
637	Oliver Wyman Limited (Saudi Arabia Branch)	Crassa
638	Oliver Wyman Limited Liability Company	Greece
639	Oliver Wyman LLC	Qatar
640	Oliver Wyman LLC	Russian Federation
641	Oliver Wyman Ltd.	Korea, Republic of

642	Oliver Wyman Proprietary Limited	South Africa
643	Oliver Wyman Pte. Ltd.	Singapore
644	Oliver Wyman Pty. Ltd.	Australia
645	Oliver Wyman S.A.S.	Colombia
646	Oliver Wyman S.L.	Spain
647	Oliver Wyman S.r.l.	Italy
648	Oliver Wyman Sdn. Bhd.	Malaysia
649	Oliver Wyman Services Limited	United Kingdom
650	Oliver Wyman Servicios, S. de R.L. de C.V.	Mexico
651	Oliver Wyman SNC	France
652	Oliver Wyman sp. z o.o.	Poland
653	Oliver Wyman SRL	Belgium
654	Oliver Wyman Thailand Holdings, LLC	United States
655	Oliver Wyman, LLC	United States
656	Oliver Wyman, LLC (Hong Kong Establishment)	
657	Oliver Wyman, S. de R.L. de C.V.	Mexico
658	Oliver, Wyman Limited/Oliver, Wyman limitee	Canada
659	Oliver, Wyman Limited/Oliver, Wyman limitee (Barbados Branch)	
660	Omega Indemnity (Bermuda) Limited	Bermuda
661	Organizacion Brockman y Schuh S.A. de C.V.	Mexico
662	Orizon Underwriters SL	Spain
663	OWL Marine Insurance-Brokers GmbH & Co.KG	Germany
664	OWL Marine Verwaltungs GmbH	Germany
665	Pallas Marsh Servicos Ltda.	Brazil
666	Pavilion Advisory ULC	Canada
667	Pavilion Financial Corporation Holdings UK Limited	United Kingdom
668	Pavilion Financial Corporation Holdings US III, LLC	United States
669	Pavilion U.S. Investments Holdco LLC	United States
670	Pension Trustees Limited	United Kingdom
671	Perils AG	Switzerland
672	PFT Limited	United Kingdom
673	PI Indemnity Company, Designated Activity Company	Ireland
674	Private Client Services by Mercer China Limited	China
675	Private Client Services by Mercer China Limited (Beijing Branch)	
676	Private Client Services by Mercer China Limited (Guangzhou Branch)	
677	Private Client Services by Mercer Holdings Pte. Ltd.	Singapore
678	Private Client Services by Mercer Limited	Hong Kong
679	Private Client Services by Mercer Pte. Ltd.	Singapore
680	Private Client Services by Mercer SA	Switzerland
681	Profund Solutions Limited	United Kingdom
682	PT Marsh Indonesia	Indonesia
683	PT Marsh Reinsurance Brokers Indonesia	Indonesia
684	PT Mercer Indonesia	Indonesia
685	PT Nexus Asia Pacific	Indonesia
686	PT Oliver Wyman Indonesia	Indonesia
687	PT Quantum Computing Services	Indonesia
688	PT Quantum Investments	Indonesia
689	PT Quantum Support Services	Indonesia
690	Pymetrics, Inc.	United States
691	Rightpath Reinsurance SPC, Ltd.	Cayman Islands
692	Rivers Group Limited	United Kingdom
693	Sail Insurance Company Limited	Bermuda
694	SAM	France
695	SCIB (Bermuda) Limited	Bermuda
093	JOID (DOMINGO) LIMITED	Demidua

696	SCM Global Real Estate Select GP LLC	United States
697	SCM International Private Equity Select III GP LLC	United States
698	SCM LT General Partner S.a.r.l.	Luxembourg
699	SCM PE General Partner S.a.r.l.	Luxembourg
700	Seabury & Smith Borrower LLC	United States
701	Seabury & Smith LLC	United States
702	Sedgwick Financial Services Limited	United Kingdom
703	Sedgwick Forbes Middle East Limited	Jersey
704	Sedgwick Group (Zimbabwe) Limited	Zimbabwe
705	Sedgwick Group Limited	United Kingdom
706	Sedgwick Internationaal B.V.	Netherlands
707	Sedgwick Management Services (Barbados) Limited	Barbados
708	Sedgwick Noble Lowndes (UK) Limited	United Kingdom
709	Sedgwick Noble Lowndes Limited	United Kingdom
710	Sedgwick Overseas Investments Limited	United Kingdom
711	Sedgwick Private Limited	Singapore
712	Sedgwick Trustees Limited	United Kingdom
713	Sedgwick Ulster Pension Trustees Limited	United Kingdom
714	Settlement Trustees Limited	United Kingdom
715	SF Private Debt General Partner S.a.r.l.	Luxembourg
716	Shanghai Mercer Insurance Brokers Company Ltd (Beijing Branch)	
717	Shanghai Mercer Insurance Brokers Company Ltd (Guangzhou Branch)	
718	Shanghai Mercer Insurance Brokers Company Ltd (Shenzhen Branch)	
719	Shanghai Mercer Insurance Brokers Company Ltd.	China
720	Shorewest Insurance Associates, LLC	United States
721	Sirota Consulting UK Limited	United Kingdom
722	SME Insurance Services Limited	United Kingdom
723	Smith Long Term Disability Management Group, Inc.	United States
724	Societe d'Assurances et de Participation Guian SA	France
725	Southern Marine & Aviation Underwriters, Inc.	United States
726	Suedzucker Versicherungs-Vermittlungs GmbH	Germany
727	SureBuilt Insurance Company IC	United States
728	The Benefit Express Holdings Limited	United Kingdom
729	The Carpenter Management Corporation	United States
730	The Positive Ageing Company Limited	United Kingdom
731	The Recovre Group Pty Ltd	Australia
732	Torrent Technologies, Inc.	United States
733	Tower Hill Limited	United Kingdom
734	Tower Place Developments (West) Limited	United Kingdom
735	Tower Place Developments Limited	United Kingdom
736	U.T.E. AMG	Spain
737	UAD BB Marsh Lietuva	Lithuania
738	Vezina Assurances Inc.	Canada
739	Victor Deutschland GmbH	Germany
740	Victor Insurance Australia Pty Ltd	Australia
741	Victor Insurance Europe B.V.	Netherlands
743	Victor Insurance Italia S.r.I.	Italy
744	Victor Insurance Managers Inc./Gestionnaires d'assurance Victor inc.	Canada
745	Victor Insurance Managers LLC	United States
746	Victor Insurance Pty Ltd	Australia
747	Victor O. Schinnerer & Company Limited	United Kingdom
748	Wellnz Limited	New Zealand

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in Registration Statements on Form S-8 (Registration File Nos. 2-58660, 33-48807, 33-59603, 333-176084, 333-146400, 333-229776, 333-209684, 333-226429, 333-226431, 333-240223), Registration Statement on Form S-3 (Registration File No. 333-258194) and Registration Statement on Form S-4 (Registration File No. 333-163405) of our reports dated February 13, 2023, relating to the consolidated financial statements of Marsh & McLennan Companies, Inc. and subsidiaries (the "Company"), and the effectiveness of the Company's internal control over financial reporting, appearing in this Annual Report on Form 10-K for the year ended December 31, 2022.

/s/ Deloitte & Touche LLP

New York, New York February 13, 2023

CERTIFICATIONS

- I, John Q. Doyle, certify that:
 - 1. I have reviewed this Annual Report on Form 10-K of Marsh & McLennan Companies, Inc. (the "registrant");
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 13, 2023 /s/ John Q. Doyle

John Q. Doyle

President and Chief Executive Officer

CERTIFICATIONS

- I, Mark C. McGivney, certify that:
 - 1. I have reviewed this Annual Report on Form 10-K of Marsh & McLennan Companies, Inc. (the "registrant");
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 13, 2023 /s/ Mark C. McGivney

Mark C. McGivney

Chief Financial Officer

Certification of Chief Executive Officer and Chief Financial Officer

The certification set forth below is being submitted in connection with the Annual Report on Form 10-K for the year ended December 31, 2022 of Marsh & McLennan Companies, Inc. (the "Report") for the purpose of complying with Rule 13a-14(b) or Rule 15d-14(b) under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and Section 1350 of Chapter 63 of Title 18 of the United States Code.

John Q. Doyle, the President and Chief Executive Officer, and Mark C. McGivney, the Chief Financial Officer, of Marsh & McLennan Companies, Inc. each certifies that, to the best of his knowledge:

- 1. the Report fully complies with the requirements of Section 13(a) or 15(d) of the Exchange Act; and
- 2. the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Marsh & McLennan Companies, Inc.

Date: February 13, 2023 /s/ John Q. Doyle

John Q. Doyle

President and Chief Executive Officer

Date: February 13, 2023 /s/ Mark C. McGivney

Mark C. McGivney Chief Financial Officer